# THE UNIVERSITY OF HULL

# Living in Flux: Risk, Protective Factors, and Meaning-Making in Transmasculine Experiences of Substance Use

being a thesis submitted for the Degree of

Doctor of Philosophy in Psychology

at the University of Hull

by

Tyler James Mills

March 2022

# Acknowledgements

I would like to express my heartfelt thanks to following people for their support throughout this project.

Firstly, I would like to thank my supervisors, Dr David Smith, and Dr Liz Price. Since working with Dr Smith on my undergraduate dissertation, he has provided unwavering support throughout my academic career. Without the invaluable supervision I have received from Dr Smith during my PhD, I would not be the researcher that I am today. Dr Price has also been instrumental in the completion of this thesis, and I could not have done it without her guidance, encouragement, and insightful feedback.

I would also like to thank all the staff and my fellow PhD students in the Psychology Department at the University of Hull who offered a supportive and inclusive academic environment. I am particularly appreciative of Dr Kim Dent-Brown and Professor Marie Reid, for their insights and advice.

I am deeply grateful to my family, who continue to astound me with their kindness, support, and belief in me. I would especially like to thank my parents for everything they have done for me. Andy, who has always stood by me, and my Mam, who has always inspired and encouraged me, have both influenced the man I am today, and I would not have made it this far without them.

I also would like to thank Kirstie Riddell, for being the most encouraging and accepting friend I could ask for. As well as providing valuable contributions and suggestions throughout this project, she has supported me through thick and thin, and I can't emphasise enough how grateful I am. Thanks also to Leo, Reggie, Millie, and Penny, who bring me happiness every day.

Finally, I would like to thank everyone who participated in this study for contributing their ideas, experiences, and time. This thesis is dedicated to the transgender community, whose persistence, resilience, and strength inspires me every day.

#### Abstract

Transgender individuals are particularly vulnerable to issues such as poor mental health, social isolation, and gender-related discrimination and violence which may contribute to developing and sustaining substance use disorders (SUD). The problems faced by transmasculine individuals are compounded by a poor understanding of factors which contribute to SUD in this population. This thesis is a mixed-methods, longitudinal study which examines factors associated with high-risk substance use in the transmasculine community and explores their lived experiences of SUD.

The main study adopted a sequential, longitudinal, mixed-methods design. 105 transmasculine individuals participated and 13 individuals were invited for interviews. Correlational and multiple regression analyses were performed on variables of SUD, quality of life, gender minority stress, gender role conflict, psychopathological symptoms, and personality traits. ANOVAs were performed to explore change over time.

Quantitative analyses revealed that over half of the sample engaged in high-risk substance use, and this increased over time. Gender minority stress, psychopathological symptoms, and personality traits were all major contributors to high-risk substance use. Qualitative analyses indicated that difficulties accessing transition-related healthcare, transphobia, distress (in)tolerance, and navigating masculinities and social expectations were all associated with increased distress that led to increased substance use.

The impact of COVID-19 on transmasculine individuals was substantial, and substance use significantly increased during the pandemic. Participants felt anxious, isolated, and distressed by poor access to healthcare and reported using substances to cope with these issues.

Overall, the results imply that SUD is a critical issue in the transmasculine community. Factors which protect against SUD, psychological interventions to improve mental health and reduce substance use, and limitations of this study are discussed, and topics of future study are suggested.

# **Table of Contents**

1. Introduction	1
1.1. Overview and Background	1
1.2. Aims and Research Questions	2
1.3. Layout of the Thesis	4
2. Literature Review	6
2.1. Introduction	6
2.2. Substance Use and Substance Use Disorders	6
2.3. Gender Minority Stress and Resilience	
2.4. Gender Role Conflict and Masculinities	
2.5. Quality of Life	
2.6. Mental Health and Psychopathological Symptoms	
2.7. Personality	
2.8. Summary and Conclusion	
3. Method	
3.1. Research Design	47
3.2. Researcher position and Reflexivity	
3.3. Pilot study - Method	
3.4. Transmasculine Experiences of Substance Use Project	
4. Pilot Study	
4.1. Literature Review	
4.2. Results	74
4.3. Discussion	
5. TESUP-1	
5.1. Descriptive Statistics	
5.2. Inferential Statistics	
5.3. TESUP-1 IPA Results	
5.4. TESUP-1 Discussion	
5.5. Conclusions	
6. TESUP-2	
6.1. Descriptive Statistics	
6.2. Inferential Statistics	
6.3. Longitudinal Statistics	
6.4. TESUP-2 TA Results	
6.5. TESUP-2 IPA Results	
6.6. TESUP-2 Discussion	

7. Conclusions	
7.1. Summary of Findings	253
7.2. Addressing the Research Questions	254
7.3. Limitations	
7.4. Concluding Thoughts	
References	
Appendix A: Pilot Study SSIS for Interviews	
Appendix B: Pilot Study SSIS for Focus Group	
Appendix C: TESUP-1 SSIS for Interviews	
Appendix D: TESUP-2 SSIS for Interviews	
Appendix E: TESUP-1 Stage of Transition Comments	
Appendix F: SCL-90 Comparison Table	
Appendix G: WHOQOL-BREF Comparison Table	
Appendix H: TESUP-1 Correlations Additional Information	
Appendix I: TESUP-1 AUDIT Multiple Regression	
Appendix J: TESUP-1 DUDIT Multiple Regression	
Appendix K: TESUP-2 Correlations Additional Information	
Appendix L: TESUP-2 AUDIT Multiple Regression	
Appendix M: TESUP-2 DUDIT Multiple Regression	

#### List of Tables, Boxes, & Figures

### 2. Literature Review

Figure 2.1: Testa et al.'s (2015) GMS framework Figure 2.2: O'Neil's GRC framework

## 3. Method

Figure 3.1: Model of research design Table 3.1: Pilot study (interview) demographics Table 3.2: Pilot study (focus group) demographics Table 3.3: TESUP 1+2 demographics Table 3.4: TESUP 1+2 AUDIT/DUDIT means Table 3.5: TESUP 1 (interview) demographics Table 3.6: TESUP 2 (interview) demographics Table 3.7: TESUP 1+2 drug use frequencies

## 4. Pilot Study

Box 4.1: Quotes: Conceptualising the Process of Transition Box 4.2: Quotes: NHS Communication and Support Box 4.3: Quotes: Medicalisation, Power, and Non-disclosure

#### 5. TESUP-1

Table 5.1: Demographic Correlations Table 5.2: QOL Correlations Table 5.3: GRCS Correlations Table 5.4: GMSR Correlations Table 5.5: GMSR SD Table 5.6: SCL Correlations Table 5.7: SCL SD

- Table 5.8: TCI Correlations
- Table 5.9: Binary Logistic Regression Tobacco Use
- Table 5.10: Multiple Regression AUDIT
- Table 5.11: Multiple Regression DUDIT
- Table 5.12: TESUP-1 IPA themes
- Box 5.1: Conceptualising the Transgender Self
- Box 5.2: Navigating Transition-related Healthcare
- Box 5.3: LGBTQ Community Connections
- Box 5.4: Substance Use to Facilitate Socialising
- Box 5.5: Impact of Masculinities and Social Expectations
- Box 5.6: Coping with Frustration and Uncertainty
- Box 5.7: Unhealthy Strategies to Regulate Negative Emotions
- Box 5.8: The Symbiotic Relationship of Mental Health and Substance Use
- Box 5.9: Impulsivity and Novelty Seeking
- Box 5.10: Personal and family history of SUD
- Box 5.11: Attitudes and Behaviours of peers
- Box 5.12: Harm avoidance
- Table 5.14: t-test TCI

#### 6. TESUP-2

- Table 6.1: Demographic Correlations
- Table 6.2: QOL Correlations
- Table 6.3: GRCS Correlations
- Table 6.4: GMSR Correlations
- Table 6.5: GMSR SD
- Table 6.6: SCL Correlations
- Table 6.7: SCL SD
- Table 6.8: TCI Correlations
- Table 6.9: Binary Logistic Regression Tobacco Use
- Table 6.10: Multiple Regression AUDIT

- Table 6.11: Multiple Regression DUDIT
- Table 6.12: ANOVA SUD & Demographics
- Table 6.13: ANOVA QOL
- Table 6.14: ANOVA GRCS
- Table 6.15: ANOVA GMSR
- Table 6.16: ANOVA SCL
- Table 6.17: ANOVA TCI
- Box 6.1: TA Impact on Substance Use
- Box 6.2: TA Impact on Transition
- Box 6.3: TA Impact on GMS
- Box 6.4: TA Impact on Mental Health
- Box 6.5: TA Impact on QOL
- Box 6.6: TA Impact on GRC
- Box 6.7: TA Impact on Personality
- Table 6.18: TESUP-2 IPA Themes
- Box 6.8: Barriers to Transition-Related Healthcare
- Box 6.9: Maintaining Health and Wellbeing
- Box 6.10: Personal Impact of Pervasive Transphobia
- Box 6.11: Negotiating Identity
- Box 6.12: COVID-19: Isolation, Mental Health, and Substance Use
- Box 6.13: Distress Intolerance and Maladaptive Coping Mechanisms
- Box 6.14: Strategies to Reduce Substance Use
- Box 6.15: Novelty Seeking and Impulsivity
- Box 6.16: Harm Avoidance
- Table 6.19: ANOVA SUD of participants who completed both vs. only 1st questionnaire

#### Acronyms

AFAB – assigned female at birth

**ANX** - anxiety

ASD – autism spectrum disorder

AUDIT - alcohol use disorders identification test

BIPOC – Black, Indigenous, and people of colour

CBWFR - conflict between work and family relations

CC – community connectedness

**CO** – cooperativeness

**D** – gender-related discrimination

**DEP** - depression

**DUDIT** – drug use disorders identification test

**GIS** – gender identity services

GMS – gender minority stress

GMSR – gender minority stress and resilience measure

**GRC** – gender role conflict

- GRCS gender role conflict scale
- **GSI** global severity index
- HA harm avoidance

**HOS** – hostility

**HRT** – hormone replacement therapy

**INS** – interpersonal sensitivity

**IPA** – interpretative phenomenological analysis

**IT** – internalised transphobia

LGBTQ – lesbian, gay, bisexual, trans, queer

NA – non-affirmation

ND - non-disclosure

**NE** – negative expectations for future events

NICE - national institute for health and care excellence

- NS novelty seeking
- NSSI non-suicidal self-injury
- **OBC** obsessive-compulsive symptoms
- **P** identity pride
- PAR paranoid ideation
- **PHO** phobic anxiety
- **PS** persistence
- PSDI positive symptom distress index
- **PST** positive symptom total
- PSY psychoticism
- QOL quality of life
- **R** gender-related rejection
- RABBM restrictive affectionate behaviour between men
- RD reward dependence
- **RE** restricted emotionality
- SCL symptom checklist
- SD self-directedness
- **SOM** somatic symptoms
- **SPC** success, power, and competition
- ST self-transcendence
- SUD substance use disorder

#### **TA** – thematic analysis

- **TCI** temperament and character inventory
- **TERF** trans exclusionary radical feminist/feminism
- **TESUP** transmasculine experiences of substance use project
- TRIM transgender resilience intervention model
- V gender-related victimisation
- WHO world health organisation

#### **1. Introduction**

#### 1.1. Overview and Background

This thesis is designed to address major deficits in knowledge regarding substance use and substance use disorders (SUD) in the transmasculine population. Current research regarding substance use and SUD in this vulnerable population is severely lacking. Lived experiences of substance use in transmasculine individuals are rarely examined and there is little longitudinal research into the factors which may be related to how SUD are developed and sustained. The research literature shows that the transgender community is subject to a number of health, economic, and social disparities which leave its members particularly vulnerable to negative outcomes. These include poor physical and mental health (Newfield et al., 2006), social isolation and familial rejection (Budge et al., 2013a), and gender-related discrimination and violence (Miller & Grollman, 2015). Additionally, the daily problems faced by transmasculine people are exacerbated by difficulties accessing adequate healthcare; the poor understanding of the healthcare services regarding the development and maintenance of SUD within the transgender community; in addition to a distinct gap in the knowledge surrounding the transmasculine population's specific treatment needs in regard to SUD (Keuroghlian et al., 2015).

Individual responses to adversity and stress are influenced by the personality traits and related neuropsychological mechanisms of the individual (George & Koob, 2017). Yet research is lacking regarding the personality style and traits of transmasculine individuals who experience SUD. An individual's personal response to challenging and novel situations can be informative regarding formulation of interventions to address mental health and substance use difficulties (Vassileva & Conrod, 2019). LGBTQ experiences of substance use are different from that of cisgender and heterosexual individuals, and there is a consistently reported disparity in prevalence of SUD and related negative outcomes (Hunt, 2012). Furthermore,

sexual minority identities and transgender identities have typically been conflated in previous literature, despite research which demonstrates that the transgender community differs from LGB individuals in their experiences of discrimination, stigmatisation, and the uniquely gendered challenges they face on a daily basis (Shelton, 2017). Therefore, it is crucial that efforts to understand the issues facing the transgender community are focused on their lived experiences and take a gender-affirmative and intersectional approach.

There exists a multitude of biopsychosocial factors which affect if and how an individual may develop a SUD (Connolly & Gilchrist, 2020) – all these factors take place in the context of transmasculine individuals' attempts to navigate the complexities and bureaucracy of the medico-legal system in the UK. This is combined with unique stress-inducing daily life events which transmasculine individuals must tackle such as: the disclosure or non-disclosure of one's transgender identity (each of which presents unique challenges to one's safety and sense of self); gender minority stress (such as discrimination, rejection, and negative expectations for future events); and gender dysphoria. It is vital that the personal and socio-cultural contexts that an individual is situated within during their transition is taken into consideration, in order to fully understand the complicated interplay of factors which may contribute to the development and maintenance of a SUD.

#### **1.2.** Aims and Research Questions

This thesis is an investigation of the correlates and experiences of substance use in transmasculine individuals. The aim of this research is to explore gaps in knowledge related to how gender minority stress, masculinities, quality of life, psychopathological symptoms, and personality traits may contribute to the development and maintenance of SUD in this population. By using a longitudinal mixed-methods approach this research aims to identify

factors associated with high-risk substance use and explore protective factors associated with low-risk substance use. By relating quantitative findings to the lived experiences of transmasculine individuals, this thesis will provide a holistic and in-depth exploration of substance use in the context of change over time. The central research question is:

What are the factors that contribute to the development and maintenance of substance use in transmasculine individuals?

The quantitative aspects of this study aim to address the following research questions:

- 1. Is stage of transition related to substance use?
- 2. Is low quality of life associated with high-risk substance use?
- 3. Does increased gender role conflict correlate with higher substance use?
- 4. Does increased gender minority stress correlate with higher substance use?
- 5. Is resilience a protective factor against issues with substance use?
- 6. Are psychopathological symptoms associated with substance use?
- 7. Is there a specific personality profile for individuals who engage in high-risk substance use?

Additionally, the qualitative aspects of this study aim to explore the following research questions:

- 8. What is the impact of transitioning and masculinities on substance use?
- 9. What contexts and situations are associated with substance use and what do these mean to the participants?
- 10. What factors contribute to a cessation or reduction of substance use?

Finally, this project originally aimed to explore factors which may contribute to changes to substance use *over time*. However, due to the COVID-19 pandemic occurring between the phase one and phase two time points, the longitudinal analysis was inherently

affected by this event. Therefore, the longitudinal analyses changed focus in order to explore the impact of the pandemic and aimed to answer the following question:

11. What was the impact of the COVID-19 pandemic on substance use within the transmasculine community?

#### **1.3.** Layout of the Thesis

Chapter one begins with a broad overview of substance use within the transgender community. It introduces the particular areas of concern which this thesis will investigate in relation to their impact on substance use and provides a rationale for their inclusion within the research. This chapter highlights the aims of this thesis and the research questions that this study will explore and concludes by outlining the layout of this thesis.

Chapter two is an in-depth review of the literature that relates to transmasculine substance use. It begins with an overview of substance use and substance use disorders and explores how these issues affect the transmasculine community. This is followed by a full discussion of relevant literature regarding gender minority stress and resilience, gender role conflict and masculinities, quality of life, mental health and psychopathological symptoms, and personality traits. This chapter concludes with a summary of salient literature and its relevance to the present thesis.

Chapter three provides an overview of the methodology and methods used to collect data in this study. The chapter begins by outlining the research design and rationale for choosing a longitudinal, mixed-methods approach. This is followed by a consideration of my own positionality in relation to the research topic and a discussion of reflexivity. The following sub-sections of the chapter outline the specific steps taken to collect and analyse data in the pilot study, and in both phases of the central research project (named the Transmasculine Experiences of Substance Use Project; TESUP).

Chapter four presents the main findings of the pilot study. This chapter begins with a consideration of the relevant literature and offers a rationale for conducting a pilot study. The key themes are discussed and the contributions that the pilot study made to the TESUP are highlighted.

Chapter five presents the quantitative and qualitative findings of the first phase of research (TESUP-1) and offers an in-depth discussion which integrates all of the results and connects them to the research questions.

Chapter six presents the quantitative and qualitative findings of the second phase of research (TESUP-2) and discusses these results in a longitudinal context, with particular attention paid to the influence of COVID-19 in this phase of study.

The thesis concludes with chapter seven. This chapter summarises the findings of the thesis in the context of the initial research questions, draws together recommendations for future study, identifies the key contributions to theory and practice, and ends with a discussion of the limitations of this research.

#### 2. Literature Review

#### **2.1. Introduction**

This study is designed to address major deficits in knowledge regarding substance use and substance use disorders (SUD) in the transmasculine population. In this study, transmasculine is used to refer to individuals who are transgender, were assigned female at birth (AFAB), and align with a masculine gender role. This may include identifying as male, masculine, non-binary, genderfluid, or a variant of this nature. The term transgender is used as an umbrella term to refer to all individuals who do not identify with the gender they were assigned at birth. Cisgender, therefore, is anyone who continues to identify with the gender they were assigned at birth.

This chapter will illustrate why investigating substance use and SUD in transmasculine individuals is of critical importance, particularly in the current socio-political context of the UK. This chapter includes a review of all the factors that this study proposes are integral to understanding the experiences of transmasculine individuals in the context of transition and substance use/SUD. These factors include experiences of gender minority stress and resilience; masculinities and gender-role conflict; quality of life and social support; mental health and psychopathological symptoms; and personality traits.

#### 2.2. Substance Use and Substance Use Disorders

#### 2.2.1. Defining Substance Use and Substance Use Disorders.

To discuss substance use, substance misuse, and substance use disorders (SUD) it is first necessary to delineate the meaning of these terms. In this study, substance misuse is defined as the use of tobacco, alcohol, illegal drugs, and legal drugs taken in an illicit manner (e.g., taking too many prescribed drugs or taking prescribed drugs that were not prescribed to you). I will define substance use disorder (SUD) based on recommendations from DSM-5 (American Psychiatric Association, 2013). The DSM-5 reports that the "essential feature of a substance use disorder is a cluster of cognitive, behavioural, and physiological symptoms indicating that the individual continues using the substance despite significant substancerelated problems" (American Psychiatric Association, 2013, p.483). SUD can be classified as mild (2-3 symptom criteria), moderate (4-5 symptom criteria), or severe (6 or more symptom criteria). There are 11 symptom criteria which can be categorised within the themes of (A) impaired control (4 symptom criteria); (B) social impairment (3 symptom criteria); (C) risky use (2 symptom criteria); and (D) pharmacological symptoms (2 symptom criteria). The 11 symptoms criteria are: (1) increasing amount or frequency of substance used (impaired control); (2) unsuccessful efforts to cut down or stop substance use (impaired control); (3) spending a lot of time obtaining, using, or recovering from substance use (*impaired control*); (4) strong urges or cravings for the substance (impaired control); (5) failure to fulfil responsibilities at home, work, or school due to substance use (social impairment); (6) continued use despite social and interpersonal issues due to substance use (*social impairment*); (7) giving up or reducing recreational, occupational, and social activities due to substance use (social impairment); (8) repeated substance use even when it is physically hazardous (risky use); (9) continued substance use despite knowledge of a physical or psychological problem which is made worse by substance use (*risky use*); (10) tolerance – needing increasingly more amounts of the substance to reach the desired effect (pharmacological); and (11) withdrawal symptoms which are relieved by taking more of the substance (*pharmacological*).

The substances discussed in this study may be legal (e.g., alcohol and tobacco), illegal (e.g., cannabis, ecstasy), or controlled substances (e.g., pain-controlling medication such as tramadol, prescribed by medical professionals) and all can be misused or develop into a SUD. These types of substances can be further categorised into seven sections according to their

behavioural, psychological, and pharmacological effects: Nicotine (e.g., chewing tobacco, vaping, cigarettes); Alcohol (e.g. beer, wine, spirits); Cannabinoids (e.g. marijuana, hashish); Opioids (e.g. heroin, Tramadol, Oxycodone); Depressants (e.g. benzodiazepines, barbiturates); Stimulants (e.g. cocaine, methamphetamine, amphetamines); and Hallucinogens (e.g. LSD, MDMA) (McLellan, 2017).

#### 2.2.2. LGBTQ Substance Use and SUD

The UK Trans Mental Health Study (McNeil et al., 2012) explored areas of transgender individual's lives, for example employment, housing, medical care, mental health, and substance use. McNeil et al. (2012) state that just over half of the sample of transgender individuals had smoked tobacco at some point, with 19% identifying as current smokers (N=583). This statistic is somewhat higher than that of the general population in the UK, with 14.7% being current smokers (ONS, July 2019). The ONS (2019) publication also examined figures relating to LGB individuals; this demonstrated that (in 2017) smoking prevalence in the lesbian and gay (23.1%) and bi (23.3%) population were significantly higher than that of straight people (15.9%). Though there is a clear lack of up-to-date information regarding smoking behaviours of transgender individuals in the UK, one can likely assume from the figures regarding LGB individuals that rates may be similarly increased, however the mechanisms by which substance abuse is developed and maintained is highly variable between L, G, B, and T groups (Shelton, 2017).

Early studies into LGBTQ substance use and SUD began in the 1970s and tended only to focus on lesbian and gay identities while neglecting to examine the experiences of bisexual and transgender individuals (Shelton, 2017, p.12). This can be seen as reflecting the attitudes of the time; while gay and lesbian liberation movements gained momentum throughout the 70s and 80s, knowledge of bisexual identities was only just beginning to expand (Taylor, 2018) and the Harry Benjamin International Gender Dysphoria Association (now known as the World Professional Association for Transgender Health; WPATH) had just published the first iteration of the Standards of Care (Walker et al., 1979) setting the stage for increasing visibility and acceptance of transgender identities. In 2011, the Institute of Medicine (IOM) released a report which examined the disparities in physical and psychological health between LGBTQ communities and cisgender, heterosexual populations. This report revealed the lack of research which focused on bisexual and transgender identities, highlighting that much of the limited existing research was based on convenience sampling and could not be reliably generalised to the larger population, and was also lacking insights into intersectional identities, such as people of colour, people with disabilities, and transgender people who identify as LGBQ.

There has been a recent increase in more methodologically rigorous studies into LGBTQ substance use; internet-based methods have improved the sampling issues of the past, and meta-analyses and longitudinal studies offer a better understanding of the development of SUD and which interventions are most useful in this population (Shelton, 2017, p.14). However, there still exists a tendency to treat gender identity and sexual orientation as one and the same, and the conflation of these two identities often leads to a lack of exploration of the intersection between the two or the differences between them (Galupo et al., 2014). Research into transgender individuals' experiences of sexual orientation often indicate that it is conceptualised as dynamic and fluid, and that shifts in gender identity can prompt changes in sexual orientation (Auer et al., 2014). This type of research emphasises that transgender individuals experience uniquely gendered forms of discrimination (transphobia) and often experience more stigmatisation than LGBQ individuals (Nadal et al., 2012). This indicates that to fully understand transgender experiences they should be discussed separately from the experiences of LGBQ individuals.

Research consistently indicates the increased prevalence of SUD in the transgender community. A large-scale cross-sectional study comparing prevalence rates of transgender SUD in US adults demonstrated that, compared to cisgender people, transgender adults had significantly higher prevalence of nicotine SUD (16.6% vs 5.4%), alcohol SUD (2.6% vs 0.9%), drug SUD (4.3% vs 1.2%), and polysubstance SUD (2% vs 0.5%) (Hughto et al., 2021). A systemic review of literature relating to substance use in the transgender community indicated that correlates of substance use support the minority stress theory; transphobia discrimination or violence, visible gender non-conformity, and intersectional minority status (e.g., being transgender and Black) were all connected to substance use in this population (Connolly & Gilchrist, 2020). The authors also state that additional research into transmasculine and non-binary experiences of substance use and SUD are necessary due to the over-representation of transfeminine identities in the literature.

Where studies have focused on exploring the experiences of transmasculine individuals in relation to substance use they have often demonstrated differences in substances used and rates of use between transmasculine people and other sub-groups of the transgender community. Barger et al. (2021) reported that transmasculine participants and queer participants were more likely to have reported marijuana use in the past year, compared to cisgender women and lesbian women, respectively. Another study reported that 17% of trans women and 33% of trans men self-reported past problems with alcohol use (Testa et al., 2012). Schweizer and Mowen's (2020) research provided similar results, with trans men having significantly higher levels of substance use than trans women – a potential explanation offered by Schweizer and Mowen is that in an effort to enact a masculine identity, trans men more heavily endorse stereotypical masculine behaviours such as greater substance use.

LGBTQ people experience the same risk factors for engaging in substance use as cisgender/heterosexual people. These risk factors can include efforts to feel good or feel better (Sinha, 2008), peer pressure or to fit in with one's peer group (Keyzers et al., 2020); or to enhance performance at work, school, or in sports (McDuff & Baron, 2005; Schelle et al., 2015). However, transgender people are faced with additional, compounding risk factors which may increase the likelihood of problematic substance use or the development of a SUD. Reisner et al. (2015) reported that enacted and anticipated discrimination within healthcare settings was significantly associated with substance use in transmasculine adults; substance use to cope with enacted discrimination (such as refusal of care) was frequently reported and participants also reported delaying necessary medical care when unwell or injured due to the anticipation of discrimination based on past experiences. Further research has indicated that the process of 'coming out' as LGBTQ is a particularly high-risk period for engaging in substance use, potentially because this is an anxiety inducing and socially isolating event which may cause individuals to self-medicate with substances (Shelton, 2017, p.44). Additionally, Hunt's (2012) report into why the LGBTQ community are facing disparities in SUD highlighted that because bars and clubs are traditionally safe spaces for this community to socialise and reduce stress, LGBTQ people who frequent bars and clubs are at higher risk of being exposed to risky alcohol and drug use, and for it to become a normal aspect of socialising, leading to increased rates of SUD in this community.

#### 2.2.3. SUD Interventions: Barriers and Best Practice

Interventions for SUD vary depending on the type of substance used and the personal, social, and medical characteristics of the individual (Shelton, 2017, pp.54). In the UK, the National Institute for Health and Care Excellence (NICE) has published guidelines on best

practices for addressing both alcohol use and drug use disorders. In their principles for all interventions (NICE, 2011) a comprehensive assessment is recommended to understand the severity of the SUD, co-occurring health and social problems (such as mental illness or homelessness), and cognitive function testing. Meaningful service user involvement is also recommended whereby the service user is involved in their own care and development of their treatment plan, in addition to becoming involved in delivering peer mentoring and developing service user-led services (NICE, 2011). The aim of service user involvement is to increase confidence and trust in the interventions that are offered, to increase commitment on behalf of the service users, and offer the ability to make meaningful and informed decisions about healthcare (Schulte et al., 2007).

Interventions for SUD can range from acute care (e.g., pharmaceutical interventions for the management of alcohol withdrawal) to self-help mutual aid groups (e.g., Self-Management and Recovery Training; SMART Recovery), to formal psychosocial interventions within the community or as an inpatient (NICE, 2007). The basic principles of SUD interventions are the same for all people, however assessment and interventions for transgender individuals involve additional considerations, in part because this community faces additional difficulties which may contribute to their SUD such as poor access to healthcare (Reisner et al., 2015), discrimination and stigma (Connolly & Gilchrist, 2020), family and social problems (Klein & Golub, 2016) and increased levels of psychopathological symptoms (Keuroghlian et al., 2015). Gaining an understanding of social, psychological, and cultural influences which may affect LGBTQ service users, and using this information to develop appropriate interventions for this diverse cultural group, is known as *cultural competency* (Boroughs et al., 2015).

The concept of cultural competency, as applied to the LGBTQ community, involves building awareness of one's own beliefs and biases and how these aspects may shape interactions with service users, and the ability to employ skills and tools to provide culturally sensitive assessment and interventions (Sue et al., 2009). Treatment facilities are typically illequipped to meet the unique needs of transgender service users; Nuttbrock (2012) reports that conventional substance use interventions display a lack of cultural competence at the outset with intake and referral procedures usually being organised according to binary and heteronormative assumptions. Questions about sexual and gender identity are not the norm and inpatient facilities are generally segregated based on gender (male or female) with no consideration for transgender identities or identities outside of this binary (Lombardi, 2007).

Considering the unique treatment needs of the transgender population, specialised SUD intervention programs are frequently recommended, in particular, multiple studies recommend that utilising a Minority Stress framework in SUD interventions for transgender people would be highly beneficial (Wolf & Dew, 2012; Connolly & Gilchrist, 2020). In Glynn and van den Berg's (2017) review of interventions aimed at reducing SUD among transgender individuals, they state that service-user involvement in SUD interventions is paramount in order to promote greater rapport and ease of communication between everyone involved.

One limitation to much of the research on the topic of transgender experiences of SUD is that the transmasculine community is under-represented and poorly understood, as previously mentioned by Connolly and Gilchrist (2020). It can be argued that due to the masculine identity of transmasculine individuals they may be at further risk of factors which would encourage engaging in substance use and developing a SUD. Research that has focused on cisgender men has consistently demonstrated that strong endorsement of traditional masculine ideals can be linked to poor health outcomes (Courtenay, 2009) and increased substance use and SUD (Vogel & Heath, 2016). Schrock and Schwalbe (2009) argue that the social and public performance of masculinity is key to solidification and validation of one's

masculine (and male) identity. They argue that conformity to masculine norms through performing traditional acts of manhood in social and interpersonal contexts is key to gaining recognition as masculine and therefore, as male.

In relation to substance use, Darcy's (2018) exploration of masculinities and men's illicit drug use illustrated how the use of illicit drugs can be a symbolic act, a vehicle to communicate the drug taker's gender identity and social role, in addition to being a socially approved manner to perform masculinities. Furthermore, in the context of transmasculine substance use, these individuals can be seen to be enacting a form of *compensatory masculinity*. In the pursuit of social recognition of their masculine or male identity, transmasculine individuals may adopt hyper-masculine traits to communicate their masculine identity to others and to make up for a perceived lack of masculinity or maleness (Vegter, 2013). This theory is supported by recent research into transgender substance use. As mentioned previously, Schweizer and Mowen (2020) theorised that the significantly higher levels of transmasculine substance use in their own research could be partially explained by the trans male participants in their study heavily endorsing conventional masculine behaviours (such as increased substance use) in order to communicate a masculine identity or to 'pass' as male in social settings.

Considerations of the role of masculinities in transmasculine lives is important when assessing for SUD and planning interventions. Cisgender men are less likely to both seek help and accept help for physical and mental healthcare problems (Courtenay, 2009); therefore, it is possible that transmasculine individuals face these same barriers. In addition to this, the transgender community are already hesitant to approach healthcare services due to negative experiences of discrimination and stigma, as previously mentioned (Reisner et al., 2015). Therefore, it is crucial to explore the potential for increased difficulties when these barriers to treatment co-occur within the transmasculine community, particularly when exploring experiences of SUD and potential barriers to receiving culturally competent assessment and interventions.

#### 2.3. Gender Minority Stress and Resilience

#### 2.3.1. Gender Minority Stress

The Gender Minority Stress (GMS) framework is a conceptualisation of the impact of multiple social stressors which are specific to transgender and gender diverse individuals and negatively impacts their mental health and wellbeing (Testa et al., 2015). The concept of GMS is built upon Meyer's (1995, 2003) minority stress theory, which states that individuals who belong to a stigmatised minority group face increased exposure to unique social stressors such as discrimination and prejudice which cause chronically high levels of stress in an individual and lead to poor physical and mental health. Meyer's theory was developed to provide explanations for the disparate mental health outcomes of lesbian, gay, and bisexual (LGB) people compared to heterosexual people, and multiple studies have confirmed the value of using minority stress theory to understand the health disparities faced by the LGB community (Pascoe & Smart Richman, 2009; Frost et al., 2015). However, the utility of Meyer's theory when applied to transgender and gender diverse individuals was inadequate; research has demonstrated that the stressors faced by transgender people, and the coping methods available to them, are different from those of LGB people (Su et al., 2016; Tan et al., 2019). Therefore, Testa et al. (2015) subsequently developed the GMS framework to better represent the range of stressors which are specific to the transgender community.

Similar to Meyer's (2003) minority stress theory, Testa et al.'s (2015) framework proposes that stressors are divided into two types: distal stressors and proximal stressors. Distal

stressors are those which originate from an external source and includes discrimination, rejection, victimisation, and non-affirmation of gender identity. Proximal stressors are internal processes and subjective experiences which are presumed to occur as a result of repeated exposure to distal stressors, this includes internalised transphobia, negative expectations for future events, and non-disclosure of transgender identity. Additionally, consistent with minority stress theory, the GMS framework suggests that coping strategies, or protective factors, exist to buffer the negative impact of GMS. These protective factors (called resilience in the GMS framework) can be divided into the community-level and individual-level. Within the GMS framework community is a community-level protective factor, and identity pride (i.e., acceptance of one's own transgender identity and embracing one's self-worth) is an individual-level protective factor. Figure 2.1 depicts Testa et al.'s (2015) proposed structure of the GMS framework.



**Figure 2.1.** Testa et al.'s (2015) proposed structure of the gender minority stress framework. Grey arrows indicate a negative effect. Distal stressors have a negative effect on outcomes and on proximal stressors. Resilience factors buffer the negative effect of distal and proximal stressors on mental and physical health outcomes.

#### 2.3.2. Gender Minority Stress, Health, and the Role of Resilience.

The GMS framework has been applied to explore various outcomes related to the physical and mental health of transgender people. Multiple studies have demonstrated how experiences of GMS can lead to negative health outcomes such as increased depression (Jaggi et al., 2018), increased suicidal ideation (Testa et al., 2017), and increased levels of bodily inflammation (McQuillan et al., 2021). However, although the effects of GMS on transgender people have been widely examined using Testa et al.'s theoretical framework, very few studies utilise the complete model (four distal stressors, three proximal stressors, two resilience factors) when exploring health outcomes. Instead, numerous studies choose only one distal stressor to measure the overall effect of distal stressors, or one proximal stressor to measure the overall effect of proximal stressors (e.g., Scandurra et al., 2017; Rood et al., 2016, respectively). Additionally, some studies (Timmins et al., 2017; Austin & Goodman, 2017) have adapted measures of LGB minority stress for use with transgender participants which lack the reliability or validity that have been established with Testa et al.'s (2015) gender minority stress and resilience measure (GMSR). This presents a challenge in drawing conclusions from the range of studies which use the GMS framework in differing ways, therefore the following findings are discussed with these limitations in mind.

The majority of studies which use the GMS framework have focused on exploring the impact of distal and/or proximal stressors on mental health and psychological wellbeing. One of the most comprehensive studies to date which utilises the complete GMS framework explored associations between GMS and depressive symptoms in 143 transitioned<sup>1</sup> Swiss

<sup>&</sup>lt;sup>1</sup> This study defines 'transitioned' individuals as 1) not currently attending trans-specific therapy; 2) not having taken hormone therapy for at least one year; 3) no plans for surgical interventions within the next year. The criterion of not currently taking hormone therapy is questionable because this is usually a life-long requirement, especially for those who have had hysterectomy or orchidectomy. Additionally, the concept of having completed medical transition is complicated, with many trans individuals feeling that their transition is never fully 'complete' and with others feeling that their transition is 'complete' without ever undertaking medical transition (see Tatum et al. [2020] for further discussion of this concept).

transgender people (Jaggi et al., 2018). The results of this study largely support the proposed GMS framework as the multivariate analyses identified that distal stressors had a significant direct effect on depressive symptoms and proximal stressors had a significant indirect effect on depressive symptoms. However, the study failed to support the assumed moderating effect of resilience factors between GMS stressors and depressive symptoms; the authors suggest a refinement of definitions of resilience factors, and in particular, improvements in the operationalisation of the concepts of 'identity pride' and 'community connectedness'. The lack of significance of resilience factors in the GMS framework has also been noted in other studies, which leads one to question the appropriateness of how resilience in the GMS framework is operationalised. Puckett et al. (2019) examined how different types of social support are associated with resilience and with symptoms of depression and anxiety. Their results suggest that good community connections are less important than strong familial support or support from friends, and the authors question whether community *support* may be more applicable to resilience than the individual simply having connections to the trans community. One overlooked aspect that may contribute to a better understanding of community connectedness is access to community-level resources and institutional support that may be gained from sexual health services and psychotherapy which is specifically aimed at transgender and gender diverse individuals (Meyer & Frost, 2013). Such gender-affirming resources may have the effect of not only establishing connections to other members of the transgender community but may also decrease GMS through addressing negative health outcomes and strengthening individual-level resilience, such as identity pride, and building positive coping strategies.

When exploring the nature of the relationship between resilience factors and GMS, the literature is conflicting. For example, Breslow et al.'s (2015) examination of individual and group level resilience factors (personal resilience and collective action, respectively) revealed that the relationship between psychological distress and community connections or collective

action is more complex than the GMS framework suggests. Breslow et al.'s study indicated that contrary to previous studies which found that engaging in collective action (i.e., community participation in social activism) is an effective protective factor against GMS (Velez & Moradi, 2016), high levels of collective action actually increased the association between the proximal stressor of internalised transphobia and psychological distress. Breslow et al. (2015) suggest that high levels of activism may expose individuals to increased levels of transphobia and consequently may be associated with fatigue and depression. It should be noted that previous studies into the buffering effect of collective action against minority stress and psychological distress have mainly focused on sexual minority groups. It could be argued that transgender and gender diverse groups have less control over disclosure of their gender identity or transgender status, making them more vulnerable to being exposed to direct and indirect transphobia. Whereas LGB individuals are often able to conceal their sexual orientation if they wish, potentially limiting their exposure to homophobia and turning the public demonstration of identity pride into an active process. In this way, community connections, in the form of collective action, has a unique interaction with identity pride; further studies which take the complex relation of the two into account are necessary to delineate more precise definitions and operationalisations of resilience factors.

As mentioned previously, various distal and proximal stressors have been found to have significant associations to psychological distress and poor mental wellbeing. Timmins et al. (2019) used Meyer's (2003) framework of minority stress to comprehensively examine, through structural equation modelling, the relationship between minority stress and psychological distress, including rumination. This study indicated that prejudice events, expectations of rejection, and self-stigma (regarding sexual orientation and gender identity) were all significantly associated with psychological distress and the full model could account for 54.5% of variance in psychological distress. While this study did not use the GMSR

framework and elected instead to adapt existing LGB minority stress measures to use with transgender participants, which calls into question the validity of the measures used, it nevertheless provides support for Testa et al.'s (2017) study which had similar results. Testa et al.'s (2017) research used path analysis to explore the relationship between GMSR and suicidal ideation; although the model fit was only adequate, GMSR factors could explain 20% of variance in current suicidal ideation, and the model supported the hypothesised mediation role of proximal stressors between distal stressors and suicidal ideation. Furthermore, trans men and transmasculine individuals scored significantly higher in measures of discrimination than trans women, trans feminine, and genderqueer groups, however there was no significant difference between groups in suicidal ideation. These studies provide good initial data on the effect that GMS has on mental wellbeing for transmasculine individuals and demonstrates support for the GMS framework.

#### 2.3.3. Gender Minority Stress and Substance Use

Emerging research is beginning to explore the relationship between GMS and substance use, particularly exploring substance use as a reaction to, or method of coping with, experiences of GMS. As stated previously in the subsection *'LGBTQ Substance Use and SUD'*, there are associations between GMS and substance use which support the GMS framework (Connolly & Gilchrist, 2020), and one study indicated that 28% of adult transgender men engage in substance use to cope with negative expectations and previous experiences of discrimination in healthcare settings (Reisner et al., 2015). While few studies which explore this topic elect to use the GMSR measure, those which situate their research within the GMS framework have described the relationship between specific distal or proximal stressors and substance use in detail. Rood et al. (2016) adopted a qualitative approach to understanding lived experiences of the expectation of rejection among 30 transgender adults. The results of their study indicated that substance use was a 'typical' coping strategy in response to managing the expectation of rejection ('typical' was defined as applying to 16 to 28 participants; this strategy was described by 17 participants).

One recent meta-analysis indicated that of 97 studies which focused on minority stress and substance use in youths, adolescents, and young adults, only nine articles (9.3%) included findings which were specific to transgender and gender diverse individuals (Kidd et al., 2018). In addition to the lack of focus on gender minority stress and substance use, there is also a lack of longitudinal research in this area. Those few studies that exist have demonstrated longitudinal connections between gender-related victimisation and subsequent increases in binge-drinking (Dermondy et al., 2016). Similarly, one experience sampling survey which monitored experiences of discrimination and substance use six times daily for 14 days, found that individuals were more likely to engage in substance use in the hours following an experience of discrimination (Livingstone et al., 2017). Dyar et al.'s (2020) longitudinal research into associations between GMS and substance use in transgender individuals assigned female at birth demonstrated that participants who reported more microaggressions and genderrelated victimisation were more likely to have concurrent alcohol and cannabis use, but there was no association between these experiences and future substance use. However, this study only utilised the distal stressor of victimisation, therefore it is difficult to draw conclusions about the relationship between the full model of GMS and substance use.

#### 2.3.4. (Trans)Masculinities and Gender Minority Stress

Transgender men and transmasculine individuals are uniquely situated in experiencing GMS (i.e., chronic stress from inhabiting a minority identity) and identifying as male and/or

masculine, which, in the GRC framework, can include a desire to conform to traditional or stereotypic conventions of masculinity (O'Neill, 2013). This leads one to question the relationship between inhabiting these two identities, and related needs and desires. To date there has been no investigations into the relationship between gender minority stress and masculinities in transgender people. Due to the lack of any studies on this topic, the following subsection will explore what is known about sexual minority cisgender men in relation to masculinities and minority stress and how these findings could be applied to transmasculine individuals.

Previous studies have highlighted that some gay men who identify as 'feminine' or 'androgynous' have less GRC than those who identify as 'masculine' (Choi et al., 2011), or those who desire public conformity to masculine ideals (Sánchez et al., 2010). In regard to this, Kimmel and Mahalik's (2005) study focused on the roles of minority stress and conformity to masculine norms in relation to the body image concerns of gay men. This research found that both a desire to conform to masculine norms and minority stress factors were associated with masculine body ideal distress (i.e., distress arising from the failure to meet the social expectations of a masculine body, such as being muscular). Additionally, Parker and Harringer's (2020) research indicates that a desire of conformity to masculine norms and high levels of gender role conflict can contribute to disordered eating in gay men. This is a finding which may also exist in transgender men and transmasculine individuals; McGuire et al.'s (2016) qualitative exploration of transgender young adult's body satisfaction revealed that bodily distress is related to gender dysphoria and that both reflect cultural expectations of what a male or masculine body should look and act like. This study suggests that for transmasculine individuals who desire to be perceived as male and/or masculine, the pursuit of the ideal male/masculine body can include restricted eating in order to reduce bodily dysphoria and suppress secondary sex characteristics (such as body fat distribution around the hips and breast tissue). Additionally, Watt et al. (2018) reported that transgender men whose voices sounded more congruent with their gender (i.e., voices that sounded more masculine) reported better psychological wellbeing and quality of life than those whose voices were less gender congruent. This finding has important implications for understanding the relationship between masculinity and gender minority stress in transmasculine individuals. It is possible that those who have a more gender congruent voice may have more control over disclosure of their trans identity, and therefore may have less exposure to, or experiences of, discrimination and victimisation due to being transgender; as previous studies have demonstrated that individuals who experience more GMS have worse psychological wellbeing (Testa et al., 2017; Jaggi et al., 2018). However, the relationship between masculinities or gender role conflict, and gender minority stress in transmasculine individuals is currently only theoretical, as no studies have yet explored these potential associations.

#### 2.4. Gender Role Conflict and Masculinities

#### 2.4.1. Conceptualising Gender Role Conflict

Gender role conflict (GRC) refers to "a psychological state in which socialized gender roles have negative consequences for the person or others. GRC occurs when rigid, sexist, or restrictive gender roles result in restriction, devaluation, or violation of others or self... The ultimate outcome of GRC is the restriction of a person's human potential or the restriction of another person's potential" (O'Neil, 2008, p. 362). GRC is a construct that has been put forward to help explain how and why people experience conflicts during what O'Neil refers to as the 'gender role socialization process'. The emphasis of almost all of the research in this area is the GRC of cisgender males. One of the early influential papers regarding the restrictiveness of the male gender role was O'Neil's (1981a) article, which described 40 patterns of gender role conflict which males are taught during gender role socialisation. According to O'Neil (1981a) GRC is best conceptualised in the context of early gender role socialization. This occurs between 18 months and 3 years when infants become aware of their own gender identity and the gendered behaviours of those around them (Newman & Newman, 2017). Children of this age also demonstrate a preference for gender-stereotyped toys (e.g., boys play with cars, girls play with dolls) (Zosuls et al., 2009) and gendered preferences for play-types (e.g., boys play rough, girls play cooperatively) (Miller, Trautner, & Ruble, 2006). However, research has also demonstrated that in preschool settings, the beliefs and perceptions of gender stereotypes of the educators' influence that of the children in their care, which contributes to the gendered separation of play preferences (Chapman, 2016). Additionally, Kollmayer et. al. (2018) demonstrated that parents preferred their child to play with same-gender-typed toys or gender-neutral toys as opposed to cross-gender toys, which in turn diminished their play repertoire and limited the development of varied cognitive skills.

Gender-role socialisation can be a dangerous concept for those who do not, or cannot, adhere to the gender binary. Conformity to gender roles is enforced throughout society in Western cultures, at all stages of life, which sometimes means that when children express gender non-conformity their parents and families attempt to force that child's gender role to align with their sex assigned at birth (Stieglitz, 2010). In one of the few studies regarding gender socialisation of transgender individuals Dietert and Dentice (2013) explored the early socialisation experiences of 37 transgender individuals and concluded that when participants failed to enact normative gender roles and expression they were confronted with numerous challenges by their family, peers, and schools. These challenges included demands for conformity, harassment and even violence.

GRC is operationally defined over four psychological domains. These are the *cognitive* domain which refers to thoughts and questions one has about gender roles. For example, simply

thinking that one does not meet society's strict standards of masculinity can cause GRC. Next, there is the *affective* domain, which pertains to the emotions felt about gender roles; in this case, negative emotions can result in GRC. There is also the *unconscious* aspect which covers how gender roles affect one's actions beyond one's own awareness. For example, O'Neil (2015) claims that early pioneers of psychoanalysis, such as Freud and Jung, were discussing GRC in the context of the individual's unconscious conflict with gender roles as manifested through their unconscious actions and what they have said. There is then the *behavioural* domain which refers to how one acts and interacts with others due to gender roles, specifically behaviours and interactions that produce negative intra- and inter-personal problems (O'Neil, 2008). GRC is also defined over numerous situational contexts, for example, one could experience GRC in the context of a gender-role transition, such as becoming a father, or when a man deviates from gender-role norms such as wearing make-up.



Figure 2.2. A visual depiction of the major concepts included in the GRC paradigm

To help illustrate the major concepts included within the GRC paradigm and how these concepts connect to the four empirically derived patterns within the Gender Role Conflict Scale (GRCS) refer to figure 2.2. O'Neil (1981a) listed 40 patterns of male GRC which are linked to male gender role socialization. These include limited emotional intimacy among men, difficulties in interpersonal communication, and experiencing work-related burn-out and fatigue which can result in serious threats to one's health. Similarly, Levant and Richmond's (2008) review of research on masculinity ideologies indicated that endorsing restrictive masculine norms was significantly correlated with interpersonal conflict and psychological problems such as depression and increased stress. The male socialisation and masculine norms which are core ideas of GRC are both conceptually related to a rejection of all that is perceived as feminine (e.g., homosexuality).

Figure 2.2 also demonstrates the four empirically derived patterns of GRC, as measured by the Gender Role Conflict Scale (O'Neil et al., 1986). These include restrictive emotionality (RE); restrictive affectionate behaviour between men (RABBM); success, power, and competition issues (SPC); and conflicts between work and family relations (CBWFR) (O'Neil, 2013). RE is defined as experiencing difficulties and fears regarding expressing emotion, as well as struggling to find the words to express basic emotions. RABBM is defined as having difficulties discussing one's own feelings and thoughts with other men. SPC represents personal beliefs, thoughts, and attitudes about achieving success through means of competition and gaining power. CBWFR describes the conflict of attempting to balance work and home life. Over 85 psychological problems have been correlated with men's GRC which include depression, anxiety, stress, and addiction/substance use (O'Neil, 2013).
Important critiques have been made by various authors regarding the GRC model and the GRCS. Numerous authors (Good et al., 2004; Tokar et al., 2000) have concluded that the GRC model is too simplistic, and a more complex and robust model is necessary to capture the lived experiences of GRC. In addition to this, the GRC research programme has been criticised for lacking empirical studies which assess GRC longitudinally, these would provide GRC researchers with much-needed information regarding the development and maintenance of GRC over time. Furthermore, multiple authors (Jome & Tokar, 1997; Tokar et al., 2000; Walker et al., 2000) have expressed concerns regarding the validity of the CBWFR domain within the GRC model, specifically stating that the low correlations between this factor and measures of masculinity indicate that it may not be theoretically connected to the male gender role. O'Neil (2008) addresses some of these concerns, stating that the CBWFR domain has been empirically linked to increased stress, shame, and anxiety in men; however, he also states that predictive and discriminate validity studies are recommended to assess whether the CBWFR sub-scale truly measures men's experiences of GRC.

# 2.4.2. GRC, Behaviour, and Psychological Distress

Male GRC has been repeatedly linked to the experience of psychological distress such as struggling with social interactions and personal relationships, mental illnesses such as depression or anxiety, aggression, and a reduced likelihood of seeking help for these issues. Research utilising various groups of men have indicated that all four domains of GRC correlate with increased depression (Good & Mintz, 1990; Yang, 2015); anxiety (Sharpe & Heppner, 1991); sexual and physical aggression (Schwartz et al., 2005; Cohn & Zeichner, 2006); and homophobia (Kassing et al., 2005; McDermott et al., 2014). Furthermore, all four domains of male GRC correlate with decreased help-seeking attitudes and behaviours (Good & Wood, 1995; Vogel et al., 2014); self-esteem (Mahalik, et al., 2001; Schartz et al., 2005); and marital satisfaction (Campbell & Snow, 1992).

Wester et al.'s (2007) study into male GRC, psychological distress, and social support demonstrated that social support acts as a mediator between both RE and RABBM. This result suggests that men who score highly in these domains have decreased levels of social support. Furthermore, social support acted as a moderator only between RABBM and psychological distress; this demonstrates that the nature and severity of psychological distress was dependent on the man's social support network. Overall, these findings suggest that social support plays an important role in reducing the negative effects of male GRC, particularly in the context of RE and RABBM. However, it should be noted that the sample used in the study was mainly white (89%) and only consisted of young university students. In a similar study, Osbourne (2004) found that perceived available social support partially mediated the effect between male GRC and psychological help-seeking; this indicated that men with high GRC and high perceived social support experienced more willingness to seek and accept psychological help. These studies illustrate that social support plays an important role in reducing the negative psychological effects of male GRC. However, it is clear that further studies are needed, with more diverse samples, to fully comprehend the function of social support in relation to male GRC.

According to O'Neil (2013) addiction and substance abuse in the context of male GRC is classified as one of the various negative outcomes that are associated with GRC. Eleven studies have investigated the connection between male GRC and substance abuse – seven of them found statistically significant associations (O'Neil, 2008). Peterson (1999) suggests that high GRC is a risk-factor for certain addictive behaviours, including smoking cigarettes; however, Sloan et al.'s (2015) study into masculinities and health behaviours found no significant link between masculinities and smoking. In Blazina and Watkin's (1996) study into

male university students' substance use, psychological well-being and help-seeking attitudes, a significant correlation between the SPC factor and increased alcohol intake was noted. Furthermore, a significant correlation between RE and negative attitudes towards help-seeking was also found. Additionally, Monk and Ricciardelli (2003) demonstrated that higher scores on the RE dimension predicted increased alcohol and cannabis use in adolescent boys. These findings suggested that male GRC is connected to increased addiction and substance use. However, the majority of these studies were based solely on university students, and it is therefore hard to apply these findings to the general population.

#### 2.4.3. GRC and Personality

Eleven studies have been conducted in the area of GRC and personality (O'Neil, 2008). Overall, these studies indicated that male GRC is significantly correlated to personality, though it has been linked to both positive and negative personality attributes and styles. Cortese's (2004) research into GRC, personality style and help-seeking revealed that high GRC was associated with an "introversive personality style" whereas low GRC was linked to a "sociable personality style" indicating that high GRC results in unfavourable intra-personal outcomes, and potentially, worse social skills. Furthermore, Chamberlin's (1993) study into GRC, leadership, and personality style, which fosters poor leadership. However, GRC has also been significantly correlated to personality styles of neuroticism, openness, and agreeableness (Fischer, 2007), which indicates that GRC is not only related to negative attributes of men's personality but also positive attributes.

Five of the eleven studies into GRC and personality have focused on the five-factor model of personality (Tokar et al., 2000; Kratzner, 2003; Serna, 2004; Sipes, 2005; Fischer,

2007). In two of the five studies, personality was indicated as a mediator variable between GRC and parental relationship quality (Fischer, 2007) and counselling-related variables (Tokar et al. 2000). The research into GRC and personality is limited and while the majority of studies indicate a link between GRC and personality, the nature of this relationship is still unclear.

### 2.4.4. GRC, LGB Health, and Trans Lives

GRC has been explored in numerous studies with gay and bisexual men and women; however, there has only been one paper concerning transgender women, and none regarding transgender men. One early study conducted by Simonsen et al. (2000) examined GRC and psychological distress in gay men and found that male GRC was significantly negatively correlated with psychological well-being. In particular, men who scored lower in the RABBM domain reported fewer symptoms of depression, anxiety, and anger. This study suggests that GRC is a concept that is not exclusive to heterosexual males. Findings of another paper exploring GRC in relation to gay men indicated that high GRC is most strongly associated with 'masculine' gay men rather than 'feminine' or 'androgynous' gay men (Choi et al., 2011). Further, gay men in this study who scored highly in the SPC domain also scored highly in the negative aspect of masculinity, indicating that the more 'traditional' or stereotypically masculine gay men encountered struggles due to being highly competitive and striving for power.

Wester et al. (2010) applied a GRC theoretical perspective to transgender women's lived experiences of transitioning, specifically in a counselling context. The authors claimed that counselling for trans women based on GRC theory is uniquely appropriate as it addresses the conflicts that arise between situational demands of transitioning and male gender-role socialisation. In this paper, GRC theory is applied to Lev's (2004) theory of the transition

process which includes non-linear developmental 'stages': awareness, seeking information, exploration, disclosure, and integration. Wester et al. (2010) hypothesized that different GRC domains will become problematic throughout the transition process. For example, they suggest that, for trans women in the awareness stage, the most common patterns of GRC will be in the domains of RE and RABBM. Whereas for trans women in the later stage of disclosure, SPC and CBWFR may be the more problematic domains, as the individual's public identity shifts from male to female, and successful career achievement in the context of family relations becomes salient. Research into GRC and the LGBT population is vast, and still growing, however, a glaring limitation is the lack of empirical studies including transgender individuals; to date there has been none.

#### 2.5. Quality of Life

# 2.5.1. The Concept of Quality of Life

The concept of quality of life (QoL) has historically been difficult to define and can be conceptualised in multiple ways. It is generally agreed upon that QoL is a multidimensional concept which is informed by cultural, social, and environmental contexts (Felce & Perry, 1995). For example, the subjective experience of QoL can vary within different patient groups (e.g., severe mental illness, chronic health conditions, cancer, etc.) in addition to differing between healthy populations and those with health-related conditions (Karimi & Brazier, 2016). The World Health Organisation (WHO) have developed one of the most popular and widely used measures of QoL (WHOQOL) which measures QoL over four domains: (1) physical health; (2) psychological health; (3) social relationships; and (4) environment. They define QoL as "an individual's perception of their position in life in the context of the culture and value systems in which they live and in relation to their goals, expectations, standards, and concerns" (WHOQOL Group, 1997, p.1). In this project, QoL is based on the WHO definition

of QoL and is defined as an individual's perception of their own health, wellbeing, resources, safety, and ability to participate in or enjoy life events, in the context of their own culture and value systems and in relation to their goals, expectations, standards, and concerns.

# 2.5.2. The Quality of Life of Transgender People

Numerous studies on the quality of life of transgender people have consistently found that transgender people have lower quality of life, independent of the domain investigated, when compared to the general population (Bockting et al., 2016). A recent systematic review and meta-analysis of the QoL of treatment seeking transgender adults (Nobili et al., 2018) revealed that transgender people reported significantly lower QoL than cisgender people, particularly in the domain of mental health. However, participants on hormone-replacement therapy (HRT) and who had received any type of gender-affirming surgery reported higher QoL scores than those who had not yet received any transition-related medical interventions. The possibility that transition-related treatment is associated with improvements in QoL is supported by longitudinal studies and further meta-analyses, which also suggest that transitionrelated treatment is associated with lower depression and anxiety among the transgender population (Manieri et al., 2014; Baker et al., 2021). It is difficult to understand the changes to an individual's QoL as they progress through transition-related treatment because very few of the studies included in the meta-analyses assessed individuals prior to the commencement of any transition-related treatment. Furthermore, all of these studies are focused on the QoL of treatment-seeking individuals and there is a lack of research regarding transgender people who cannot, or do not want to, seek transition-related treatment.

As previously stated in the 'Gender Minority Stress and Resilience' section, GMS is associated with psychological distress, which in turn impacts an individual's QoL. Surprisingly, QoL and GMS is an under-studied area of research in transgender health, however, while the use of the complete GMSR model in empirical research is lacking, some studies have explored various aspects of GMS in relation to QoL. Başar et al. (2016) investigated how perceived discrimination can impact the QoL of transgender individuals. This study revealed that perceived discrimination was significantly associated with worse QoL in the social and environmental domains of the WHOQOL. Additionally, trans men reported worse psychological QoL than trans women, though this was not significantly associated with perceived discrimination. Furthermore, a narrative review of discrimination and resilience in transgender people indicated that resilience, in the form of strong social support and enhanced coping skills, can improve the quality of life of transgender people (McCann & Brown, 2017).

# 2.5.3. Quality of Life, Health, and Substance Use

When considering aspects which influence quality of life, psychological and physical health are among the most frequently studied factors. Downing and Przedworski (2018) examined the health of transgender adults in the US and report that compared with cisgender men, trans men had higher odds of having a diagnosis of arthritis and multiple chronic conditions, and compared to cisgender women and cisgender men, they also had higher odds of having diagnosed depression. Additionally, in one study which examined health, disability, and QoL in Swedish transgender adults, over half of the participants in the study self-reported having a disability, and previous negative healthcare experiences were associated with poor self-rated health and lower QoL (Zeluf et al., 2016). Furthermore, the authors report that illicit drug use was also associated with worse self-rated health and lower QoL.

There is a lack of research into QoL in transgender adults who have a SUD. However, studies which have explored the relationship between QoL and substance use in cisgender

adults have reported that unhealthy alcohol use has a negative impact on health-related QoL (Tiffany et al., 2012). Despite this, further research has revealed that other correlates of poor QoL such as mental wellbeing and social support have more influence on QoL than SUD-specific factors, such as type of substance used or frequency of use (Muller et al., 2016). These contradictory findings reveal that the impact of substance use on QoL is complex, and it is difficult to extend these results to apply to transgender adults. Previous research has demonstrated that factors such as negative experiences of healthcare and perceived discrimination are associated with worse QoL, and QoL often increases as transgender people access more transition-related healthcare. So, taking into consideration Muller et al.'s (2016) research, it is likely that factors such as these are more influential than SUD-specific factors in how one rates their QoL. Therefore, further research is necessary to delineate the contributions of each of these factors in the overall QoL of transgender adults.

#### 2.6. Mental Health and Psychopathological Symptoms

# 2.6.1. Mental Health in the Trans Community

It is well documented that the transgender community often report increased psychological distress and it is generally accepted that the higher rates of mental health issues in this community are largely due to gender minority stress (Valentine & Shipherd, 2018). This disparity in mental health and psychological wellbeing between transgender and cisgender populations is further exacerbated by barriers to healthcare which are typically experienced by the transgender community (Wright et al., 2021). Furthermore, transgender people also experience inequality in other areas of life which are established as contributing to poorer mental and physical health. For example, the LGBT in Britain: Trans Report (Stonewall, 2018) found that 51% of trans employees hide their transgender identity at work out of fear of

discrimination, 41% of transgender people had experienced a hate crime, and 25% had experienced homelessness.

Some of the most frequently researched health outcomes in the transgender community are depression, anxiety, and suicidality. One study (Witcomb et al., 2018) which examined depression in transgender people found that individuals who desired transition-related medical interventions, but had not yet received any, were four times more likely to have depression in comparison to a cisgender control group. However, the prevalence of depression decreased in transgender individuals who had accessed medical interventions such as HRT. This is a consistent finding in the literature; however, there is a lack of research which includes transgender people who do not wish to have any medical interventions, or which focuses on non-binary individuals. Suicidality in the transgender population has been described as a "public health crisis" (Dickey & Budge, 2020); among marginalised groups, transgender people report one of the highest rates of suicide attempts and it has been estimated to be between 30% and 81% (Narang et al., 2018). One study (Lefevor et al., 2019) found significant differences between transgender and cisgender participants in current suicidal ideation (57% trans, 36% cis women, 37% cis men) and lifetime suicide attempts (45% trans, 21% cis women, 13% cis men). This study drew from a sample of university students; therefore, it is not clear how suicidal behaviour might affect older transgender people, or those who cannot access higher education.

An important factor which relates to increased psychopathological symptoms is the greater prevalence of non-suicidal self-injury (NSSI) in transgender people compared to cisgender people. NSSI is defined as direct and intentional injury to one's own body, in the absence of suicidal intent (Nock & Prinstein, 2004). According to a recent systematic review (Cipriano et al., 2019), some common functions of NSSI are to regulate negative emotions, or to inflict self-punishment. In relation to transgender individuals, Claes et al. (2015) explored

the prevalence of NSSI and its associations with psychopathological symptoms. This study revealed that although psychopathological symptoms were increased in trans women when compared to trans men, NSSI was still more frequent among trans men, with 57.7% of the trans men in this study reporting a history of NSSI compared to 26.2% of trans women. Furthermore, individuals who reported engaging in NSSI had significantly higher scores in all psychopathological symptoms than participants without NSSI.

It is important to understand the prevalence and characteristics of psychiatric and neurodevelopmental diagnoses in the transgender population. One large cross-sectional study combined samples of five datasets of non-clinical cohorts (overall, this included 641,860 cisgender and transgender individuals) to investigate rates of neurodevelopmental and psychiatric diagnoses, by comparing the transgender sample to the cisgender sample (Warrier et al., 2020). This study revealed that across all datasets, transgender individuals were 3.03 to 6.36 times more likely to have a diagnosis of autism spectrum disorder (ASD). In addition to this, the transgender sample scored significantly higher than the cisgender sample on traits of ASD, such as sensory sensitivity and lower empathy, and the transgender sample were more likely to report that they suspected they had undiagnosed ASD and other psychiatric diagnoses, compared to the cisgender sample.

It should be noted that the measurement of empathy used in Warrier et al.'s paper rests on shaky theoretical foundations; the Empathy Quotient (EQ; Lawrence et al., 2004) was developed from Baron-Cohen's controversial 'extreme male brain' theory. Briefly, this theory argues that because males score lower on measures of empathy and higher on measures of systemising (i.e., a tendency to be analytical and to easily recognise patterns), similar to the cognitive features of ASD, that individuals with ASD have an 'extreme male brain' regardless of the gender or sex of the individual (Baron-Cohen & Hammer, 1997). Not only does this theory uphold a biologically essentialist view of gender and behaviour, but it also ignores the evidence that historically, women have been systematically underdiagnosed with ASD, and often present with a different clinical symptomatology in comparison to men (Perrykkad & Hohwy, 2019). Therefore, measures of ASD traits which are frequently used (such as the EQ) have been designed and validated to fit with stereotypically male presentations of ASD, and as such, inherently support the 'extreme male brain' theory. It is clear how this biological essentialism in the measurement of autistic traits could be complicated for transgender individuals and calls into question the validity of these measures when used in such samples.

Cross-sectional studies of existing databases have revealed the increased frequency of psychiatric diagnoses in transgender individuals. Wanta et al. (2019) explored the prevalence of psychiatric diagnoses in a sample of 10,270 transgender patients in the US healthcare system. Compared to cisgender patients (n = 53,499,400), the transgender sample had increased prevalence of all psychiatric diagnoses which were investigated (including mood disorders, anxiety disorders, eating disorders, personality disorders, and substance use disorders, among others). Approximately 58% of all transgender patients had at least one DSM-5 diagnosis, compared to 13.6% of cisgender patients. Furthermore, 10% of transgender patients had a SUD, compared to just 2.6% of the cisgender sample. The authors note that prevalence of diagnoses in the sample of transgender patients may be inflated, due to the necessity of a psychiatric assessment before starting any transition-related medical interventions. These studies demonstrate the importance of understanding transgender mental health in order to improve mental health outcomes and support transgender people throughout their lives.

Studies into characteristics of people diagnosed with a SUD have demonstrated that comorbid mental health difficulties are frequent (Alsuhaibani et al., 2021); when substance use and psychiatric disorders present together this is commonly termed 'multimorbidity' (Bhalla & Rosenheck, 2018). There is a lack of research into multimorbidity in transgender individuals -- what does exist typically focuses on the impact of human immunodeficiency virus (HIV) in this population. One study (Rich, 2021) which does include both transgender people with and without HIV, highlights that transgender people overall have higher prevalence of chronic physical and mental health conditions, including SUD, and this rate is increased for transgender people living with HIV. The co-existence of a SUD and a psychiatric diagnosis is an important factor when assessing an individual and planning interventions; these disorders often mutually exacerbate one another and the interaction between the two dictates treatment options. This can be further complicated when an individual is transgender because healthcare services often lack cultural competency. This can include ignorance of transgender-specific risk factors and treatment facilities being segregated by gender and being inherently cis/heteronormative.

#### 2.7. Personality

# 2.7.1. Personality – An Overview

Personality psychology aims to investigate individual differences and similarities between people, which are thought to be patterns of cognition, behaviours, and motivations which can influence how an individual relates to others (Corr & Matthews, 2009). Allport (1937) described two major approaches to investigating personality: the nomothetic (general principles and traits which can be applied to most people and can be measured or categorised), and the idiographic (a focus on the unique aspects of an individual as a whole which cannot be suitably measured or compared). Historically these approaches have been conceptualised as an oppositional dyad, however more recently they are beginning to be thought of as complementary and interdependent approaches which can be used together to better understand both individual differences and similarities between people (Salvatore & Valsiner, 2010).

#### 2.7.2. Understanding the Temperament and Character Model of Personality

Currently, the five-factor model dominates the field of research into personality (Aluja & Blanch, 2011). This can be measured by the Big Five Inventory (BFI; John et al., 1991), among others. The five-factor model consists of: Neuroticism, Extraversion, Conscientiousness, Agreeableness, and Openness (John et al., 1991). However, this approach is criticised due to the ambiguous nature of the descriptions of traits and because there may exist social bias in the description of words such as 'neuroticism' (negative bias) and 'extraversion' (positive bias) (Trofimova, 2014).

Differing from the five-factor model, which offers no underlying biological or causative aspects to personality traits, Cloninger (1986) proposed a unified psychobiological model of personality based on two components which represent traits associated with biology (Temperament) and traits related to learning and culture (Character). Cloninger proposes that personality is a complex and dynamic expression of neurobiological mechanisms within an individual, which change and develop over time as adaptive processes and situational interactions inform components of personality (for more detail on this see Cloninger, 2015). Within this Temperament and Character model, Cloninger et al. (1993) proposed various dimensions which constitute Temperament and Character in order to measure personality traits. Within the third iteration of the Temperament and Character Inventory (TCI-R; Cloninger et al., 1999), there are four multi-faceted dimensions of Temperament and three multi-faceted dimensions of Character which I will now cover briefly<sup>2</sup>.

The Temperament dimensions include Novelty Seeking (NS), which is defined as impulsive exploratory activity in response to novel stimulation and efforts to avoid

 $<sup>^{2}</sup>$  A discussion of the full complexities of the genetic and psychobiological foundations of Temperament and its relationship to aspects of Character is beyond the scope of the current literature review, for more information please see Cloninger et al., 2019.

frustration. It is thought to be associated with dopaminergic activity, specifically the D4 dopamine receptor gene (D4DR; Ebstein et al., 1996). Harm Avoidance (HA) is characterised by excessive worry and fear, and the tendency to avoid potentially aversive stimuli. It is thought to be associated with high serotonergic activity (Koller et al., 2008). Reward Dependence (RD) is characterised by a strong tendency to sustain behaviour in response to signals of reward, such as social approval. It is thought to be related to noradrenergic activity, specifically, low norepinephrine (Ham et al., 2005). Persistence (PS) is defined as the tendency to persevere despite fatigue and frustration. It has been associated with increased blood flow in the ventral striatum, anterior cingulate cortex, and orbitofrontal cortex, which relate to behavioural persistence despite frustrative non-reward (Gusnard et al., 2003).

The Character dimensions are proposed to be rooted in one's culture and informed by social and personal experiences (Cloninger et al., 1993). The Character dimensions are: Self-Directedness (SD), which refers to an individual's ability to adapt behaviour and regulate oneself according to the needs of a situation. Cooperativeness (CO), which is defined as one's ability to identify with and accept other people. Finally, Self-Transcendence (ST) is defined as conceptualising oneself, the world and everything in it as interrelated parts of a unitive whole.

# 2.7.3. Personality and Transmasculinity

The TCI is a well-suited instrument to examine personality traits in both nonclinical and clinical populations (Cloninger, 2015) and has been utilised to explore differences between genders (typically men vs. women) and cross-culturally (Brändström et al., 2001). Furthermore, the TCI has consistently been used to investigate the relationship between personality traits and various mental health conditions (Joyce et al., 2003; Pelissolo et al., 2015). Therefore, it seems uniquely situated to examine personality traits in transgender people.

One of the first studies to examine personality traits in transgender people using the TCI was conducted by Gomez-Gil et al. (2013) and focused on profiling and comparing the personality traits of transgender people to cisgender people. This study found that the personality profile of transgender individuals was more similar to that of others who shared their gender identity, rather than that of others who shared their sex assigned at birth (i.e., trans men were more similar to cis men than cis women). Additionally, this study reported that transgender participants scored lower than cisgender participants in RD and CO, suggesting that the transgender individuals in this sample were more independent and more self-concerned than the cisgender sample. The authors speculate that these differences might be indicative of the impact of the disproportionate personal and social difficulties that the transgender community faces. This implies a connection between GMS and personality traits, though GMS is not directly discussed by the authors.

A similar study which aimed to compare personality profiles of Japanese trans men and trans women to Japanese cis men and cis women (Miyajima et al., 2014) has reported differing results from that of Gomez-Gil et al.'s (2013) research. Miyajima et al.'s research found that trans men had higher RD than cis men, which is the opposite finding to that of Gomez-Gil et al., perhaps indicating different personality profiles in Eastern and Western populations. Miyajima et al.'s research also indicated that that CO was higher, and ST was lower in the trans male sample, the authors argue that these results together potentially indicate better social functioning and lower psychopathological symptoms for transgender men, in comparison to transgender women.

These studies demonstrate differences in personality traits between trans individuals and cis individuals, in addition to between trans men and trans women. However, as these studies have employed cross-sectional designs it is unclear whether these differences exist prior to medical transition or develop during/after medical transition. One study which explored the effect of testosterone therapy on the personality traits of trans men did use a longitudinal design to assess these differences (Metzger & Boettger, 2019). This research used the five-factor model of personality, as opposed to the psychobiological model which was used in the previous studies presented in this section. However, a comparison study demonstrated that all of the TCI dimensions are significantly related to dimensions within the five-factor model. For example, HA is positively associated with neuroticism, CO is positively associated with agreeableness, and NS is positively associated with extraversion and openness to experience (De Fruyt et al., 2000). Metzger & Boettger (2019) demonstrated that during the first three months of testosterone therapy, trans men showed a decrease in neuroticism and an increase in extraversion and agreeableness in comparison to controls, and this difference persisted after six months of treatment. The authors suggest that testosterone therapy in trans men impacts personality traits, and also increases psychological wellbeing and social functioning.

#### 2.7.4. Personality and Substance Use

It is generally agreed that one aspect that might influence the development and maintenance of a SUD is personality, and in particular, the psychobiological predisposition to develop a problematic relationship with substances (Kotov et al., 2010). It is for this reason that the TCI has often been used to explore the relationship between personality traits and different types of SUD. One study examined the predictive utility of the TCI for SUD diagnoses in a cross-sectional and longitudinal design (Sher et al., 2000). The results of this study indicated that the NS dimension was the most significant element of the TCI in predicting tobacco use and drug use disorders over a seven-year period, however it did not reliably predict alcohol use disorders. That the NS dimension is substantially related to substance use, particularly illicit drug use, is a finding that has been replicated over numerous studies and is suggested to be a causal factor in the development of SUD (Foulds et al., 2017; Hashemi et al., 2019).

Sher at al.'s research raises some interesting questions about potential differences between personality profiles of individuals who use different substances, for example, why might some people use alcohol rather than illicit substances, and *vice versa*, and can this be explained by personality differences? One study which explored the differences in personality traits between people with heroin dependence, alcohol dependence, and a control group (Le Bon et al., 2004). In this study, the heroin-use group had significantly higher NS, especially exploratory excitability, in comparison to the alcohol-use group and the control group. Additionally, HA was higher in both substance use groups compared to the controls, but the alcohol-use group showed significantly more fear of uncertainty and anticipatory worry than the heroin-use group. However, this study is limited in its conclusions as the alcohol-use group took the TCI test up to 3 weeks after detoxification, whereas the heroin-use group were still actively using heroin. Therefore, there may be differences between active alcohol dependent people and people like those in Le Bon et al.'s study who had gone through a detoxification programme. Furthermore, there is a lack of literature which explores differences between treatment-seeking individuals with SUD and those who are not seeking treatment or presenting to detoxification/rehabilitation services.

# 2.7.5. Personality and Mental Health

Cloninger (2006) has argued that overall psychological wellbeing is largely dependent on dimensions of Character which are essential for enhancing overall wellbeing and perceiving oneself as 'happy'. Some studies have supported this hypothesis; Cloninger and Zohar (2011) report that SD was strongly associated with all aspects of wellbeing (such as good health, social support, and happiness). Additionally, they reported that personality traits could account for almost half of the variance in self-reported levels of happiness in this sample.

The psychobiological model has often been used to explore impulsivity-related disorders such as sex addiction and gambling disorders (Farre et al., 2015), bipolar disorder (Harley et al., 2011) and ADHD (Perroud et al., 2016). One study (del Pino-Gutierrez et al., 2017) aimed to explore the personality profile of three groups of diagnoses (SUD, gambling disorder, and bulimia nervosa) and two groups of dual diagnosis (gambling disorder + SUD and bulimia nervosa + SUD). The results of this study indicate that NS was significantly associated with all diagnosis groups and was most strongly related to both of the dual diagnosis groups. Additionally, high scores on ST were most strongly related to a diagnosis of SUD, followed by associations with both dual diagnosis groups. This research supports previous studies which consistently indicate that NS and ST are strongly associated with SUD (Kotov et al., 2010; Foulds et al., 2017; Hashemi et al., 2019).

One particular strength of the psychobiological model of the TCI is that it demonstrates good predictive validity of the diagnosis of personality disorders (Dell'Orco et al., 2018), and has often been used to explore the personality profiles of individuals with various personality disorders. However, although the TCI has often been used to identify potential personality disorders, it frequently is unable to differentiate between disorders. A recent study (Paolini, 2016) found that personality disorders were most consistently identified by SD (described in this study as 'impairment of the self') and CO (described in this study as 'impairment of interpersonal functioning'). In both of these studies, however, they were unable to differentiate between personality disorders, lending credibility to the earlier suggestion that the TCI is unable to specify a particular personality disorder even though it can correctly identify the existence of one. Though it should be noted that the categorical, descriptive approach to personality disorder diagnosis (and the practice of medical diagnosis a whole within psychology) has often been criticised for not being sufficiently empirically grounded (Herpetz et al., 2017) and for pathologizing normal variations of human behaviour, emotion, and functioning (Ratnayake, 2022).

Though there is a scarcity of information about identification of psychiatric symptoms and personality disorders using the TCI in transgender people, there are some studies which indicate that Cluster B personality disorders (antisocial, emotionally unstable/borderline, histrionic, and narcissistic) are the most frequently diagnosed in this population (Madeddu et al., 2009). The consideration of personality disorders in the transgender community is of high importance because SUD are prevalent in both personality disorder populations and transgender populations (Grant et al., 2011). There is a potentially increased risk that transgender people may struggle with a SUD and might present to medical services with dual diagnosis of personality disorder and SUD. Therefore, it is of utmost importance that a gender-affirming framework considers the impact of gender minority stress when assessing transgender individuals for personality disorders (Goldhammer et al., 2019).

# 2.8. Summary and Conclusion

This chapter has provided an overview of the most important literature that relates to SUD in the transgender community. In particular, this chapter has highlighted the lack of research into transgender substance use. A review of the literature within this chapter has emphasised the importance of using the gender minority stress framework when considering issues which affect the transgender community, particularly in relation to psychopathological symptoms.

It has been demonstrated that transmasculine individuals face unique stressors in their lives, such as experiences of gender minority stress and issues surrounding masculinities, and in particular, how this community faces unique struggles with SUD and faces many barriers when attempting to access healthcare for SUD, mental health, or physical health.

This chapter has explored how various factors such as quality of life, psychopathological symptoms, gender minority stress, gender role conflict, and personality traits may all relate to substance use in transmasculine individuals. However, to date no study has included all of these factors in one research project, and no studies at all have focused on GRC in transmasculine individuals.

#### 3. Method

#### **3.1. Research Design**

This study uses a mixed-methods approach; this design incorporates quantitative and qualitative data collection, analysis, and reporting of results. A mixed-methods design was chosen to capture the intricacies and complexities of understanding the nature of substance use in the transmasculine community. Qualitative analysis is focused on understanding phenomena from the perspective of the participants; where quantitative analysis focuses on measurable facets of the phenomenon studied, qualitative approaches focus on lived experiences and meaning-making of the participants as it relates to the phenomenon (Hayes, 2013). Within a mixed-methods design, quantitative and qualitative approaches can be considered complementary, and each can build on the overall understanding of the studied phenomenon and allow for a more complete and holistic analysis.

This study adopts an explanatory, sequential mixed-methods, and longitudinal design. Within an explanatory, sequential design quantitative measures are employed first to gather initial broad information about the sample. This design is well suited for this study as it enables qualitative analysis to explain significant or surprising results and allows a more indepth analysis of substance use in this community. Additionally, by first exploring the quantitative characteristics of the sample, it was possible to use the data and information gathered to guide purposive sampling for the qualitative data collection and explore the experiences of those at low-risk and high-risk for substance use. In this study, equal emphasis was placed on the quantitative and qualitative components. The inherent strength of quantitative analysis is to be able to provide measurable change over time within the longitudinal analysis in this study and has an increased likelihood of generating generalisable results (Carr, 1994). On the other hand, qualitative research enables a deeper, idiographic understanding of the phenomenon from the perspective of the participant, as well as allowing for participants to raise issues which might not have been included in a more rigid, quantitative design (Choy, 2014). Quantitative and qualitative data were integrated through an overarching narrative which involved writing-up and interpreting both quantitative and qualitative findings on a concept-by-concept basis.

# 3.2. Researcher position and Reflexivity

As a methodology, IPA seeks to interpret what participants say in order to make sense of lived experiences, and as such, researchers using IPA consequently adopt an 'insider perspective' to interpret how individuals make sense of their experiences (Shaw, 2010). By engaging reflexively with one's own preconceptions, biases, and beliefs throughout an experiential qualitative inquiry, the researcher is able to enter a double hermeneutic with rigour and clarity, conscious of their own preconceptions and perspectives as they arise and understanding how this may affect the interpretation and analysis of their results (Clancy, 2013).

On commencing this research project, I felt confident about my ability to understand the lives of transmasculine individuals. I occupy both an 'insider' and 'outsider' perspective; being a member of the transmasculine community I share an identity with my participants, however being a researcher, I knew I must take an observational, neutral position while exploring the very real and often distressing lives of my participants. My 'insider' position was, at first, extremely beneficial; I shared knowledge and language with participants which made communication easy, and I was often quick to build rapport. However, in collecting data I began to realise where my biases lay. Participants would sometimes say, in response to a question, something like "obviously the waiting times are long, it's difficult, you know how it is", and without question I would agree ('*yes, I know how it is, it's very frustrating and*  *difficult')*, and move on from this exchange in the belief that I had understood them and correctly interpreted what they had told me.

Through reflexivity, I began to identify areas of bias and assumptions that I was making. One of which was that I often presumed that I understood my participants' experiences easier because of my insider status. I had to adjust this assumption; I reflected on how each participant I had interviewed had different life experiences which had led them to the point of being interviewed by myself. They had different motivations not just for using/not using substances but also for wanting to take part in this project. By attempting to honestly acknowledge how my past experiences and identity as a trans man were influencing my assumptions about what my participants told me, I was able to address this in the data collection, and subsequent analysis. In the instances where I identified with what a participant disclosed, I took extra care to explore their entire interview and build up 'evidence' that supported my interpretation of a statement that previously I would have presumed to understand.

While working on building my reflexivity skills and acknowledging the dual position I inhabited of 'insider/outsider', I began to notice that where I disagreed with a participant's opinion, I was able to put my personal beliefs to one side and approach those statements with empathy and academic curiosity. In particular the interactions and analysis of one participant, Darren, highlighted my own biases, but also demonstrated my growing abilities to employ reflexivity during and after textual interpretation. Darren and myself built rapport while discussing our own histories of studying Psychology at university, and I think this initial development of rapport and mutual understanding aided in the conversation which happened soon after. Darren began a discussion about perceived rifts in the transmasculine community; namely the difference between those who believe you *must* have gender dysphoria to be trans, and those who think that gender dysphoria is *unnecessary* to identify as trans. Darren was in

the former group and his comments forced me to examine my own position as it related to this issue. One extract which was particularly impactful was his discussion of his own experiences of rejection from the trans community.

"D: trying to find a group where people are **actually** trans ... trying to find that is really hard and it makes me feel really angry because I can't find a community that supports me being in it (...) they all get annoyed with me and are like "why can't you just believe in this?" and it's like because I just don't T: yeah you can't force yourself to change your beliefs D: yeah ... like I'm born with this condition and that doesn't mean I have to have certain beliefs about things ... and that's a big problem in this community is that you **have** to have a certain perspective otherwise you get very badly rejected and that makes everything to do with being trans even worse"

This exchange with Darren forced me to face some uncomfortable truths that were becoming apparent in my research and challenged my perspective as an insider. This conversation made me reflect on how I, personally, had contributed to people feeling the way that Darren felt – rejected from his own community. At this point in the research the importance of trans community connections was becoming clear, and rejection from your own community, from the people who should understand you the most, was very obviously having a negative effect on Darren and people like him. My interview with Darren forced me to address biases I took for facts such as 'the trans community is accepting and inclusive' – that is not Darren's experience and responding with empathy and working to truly understand Darren's perspective in context of his lifeworld, I could see how the trans community at large had failed him. This experience was integral to my development as a reflexive researcher; I brought forward the enhanced skills of reflection, empathy, and interpretation to my future interviews and was able to adopt a more gentle, self-interrogative approach to interpreting participants' experiences. It allowed me to see subtle hints that other participants had not found the acceptance they desired within the trans community, and I was able to respond and interpret these from my dual position of insider/outsider. I was able to bracket my assumptions while learning about the ways I may still hold biases and how to mitigate these in the analysis. I took extra care to centre the voice of *all* participants, no matter if I personally agreed or disagreed with their sentiment and was careful not to over-represent participants that I felt more connected with or with whom I shared commonalities.

My initial focus when writing this reflexive statement was about my shared identity with the participants of being transgender. However, another shared experience, which has been neglected in this section, is that of substance use. When I ask myself "why leave out my own experiences of substance use when that is the main focus of this research?", the answer is largely my own desire for professionalism. I wanted to appear to the participants as competent and trustworthy, but by avoiding discussions of my own experiences with substance use, the implication is that any experience with substance use and being competent or trustworthy are mutually exclusive, which is not my belief. Despite my hesitance to discuss my own experiences in this thesis, when my participants asked about my own history of substance use, I discussed it honestly, and it brought a level of trust and openness that enhanced the interviews. One conversation with a participant, Simon, demonstrates the impact on participants that researcher authenticity can have.

"S: Can I ask you, have you ever done anything like this [substance use] before? T: Yeah ... when I was younger, yeah 51

# S: But you stopped?

*T*: Yeah, I did ... this was years ago like I said, but I did take steps to stop S: ah ... it gives me hope you've said that ... because then it makes me think that **I** can stop too"

This exchange taught me that participants aren't necessarily looking for professionalism, they're often looking for honesty, understanding, and connection on a human level. It brought to the forefront of my mind the conflicting emotions that surround substance use, for both me and the participants, and offered me a deeper appreciation of the double hermeneutic inherent in IPA methodology.

I hope this personal reflexive journey has produced a more considered and empathetic interpretation of *all* of the collected data, along with an analysis with increased reliability and credibility due to the steps I took to actively engage with reflexivity and put into practice what I learned from my own reflexive accounts, and the experiences of participants.

#### 3.3. Pilot study - Method

The aim of conducting a pilot study was first to understand the appropriate terminology regarding transgender and transmasculine individuals. Though I am myself transgender, it is crucial to use the language of the population sample and to not project prior assumptions of preferred language onto the community. The second aim was to investigate how the transgender community conceptualises the process of transition, to best operationalise this concept quantitatively within the main project (Transmasculine Experiences of Substance Use Project; TESUP). Ethical approval was granted for both the pilot study and the main study (TESUP) by the Faculty of Health Sciences Research Ethics Committee at the University of Hull (REF: FHS128). The pilot study was conducted in two phases. The first phase involved three interviews with transgender men; this was to explore the area of study initially and to narrow down questions for the more limited context of a focus group. The second phase consisted of a focus group attended by nine transmasculine individuals.

# 3.3.1. Participants

Participants had to be 18 years or older, live in the UK, and be able to give full informed consent. They had to be assigned female at birth and no longer identify this way; most of the participants identified their gender as male, non-binary, or transmasculine. There were no exclusionary criteria relating to identification within a binary gender system, length, route, or destination of transition, or sexual orientation. The participants for the interviews were recruited from my personal Facebook account. Participants were made aware of the participation opportunity through a public Facebook post in a transmasculine-specific support group and were not contacted directly. The participants for the focus group were recruited from a trans-led voluntary organisation which coordinated a weekend retreat for transgender individuals. Participants were made aware of the participation opportunity through a leaflet that was given to all attendees of the retreat and were not contacted directly. All attendees were invited to attend the focus group with a limit of nine participants on a first-come firstserved basis.

The total number of participants in the pilot study was 12. This consisted of three in the interview phase and nine in the focus group phase. All three of the participants in the pilot interviews identified as male and as white British. The ages of participants ranged from 24 to 27 (see Table 3.1). The individuals in the focus group identified their gender in a variety of ways, their ages ranged from 23 to 44 (see Table 3.2).

Name	Age	Gender	Ethnicity	Sexuality	Annual income (£)	Education	Stage of transition
Romeo	25	Male	White	Heterosexual	10,000	GCSE	Waiting for hormones
			British				
Charlie	24	Male	White	Mostly	16,000	A-Level	On hormones, waiting for
			British	straight			top surgery
Mike	27	Male	White	Bisexual	10,000	A-level	On hormones, waiting for

Table 3.1. Pilot Study - Demographics of Interview Participants.

Table 3.2. Pilot Study - Focus Group Participant Demographics

British

Name	Age	Gender	Ethnicity	Sexuality	Annual income (£)	Education	Stage of transition
Elliot	26	Non-binary	White British	Bisexual	N/A	BSc	Finished top surgery
Francis	44	Genderqueer	White	Straight	N/A	Post-grad	Testosterone
Greg	36	Non-binary	White British	Bisexual	N/A	MSc	Finished phalloplasty
Raj	23	Transmasculine	British	Queer	20,000	PGDip	Waiting for hormones
Paul	27	Non-binary, demi-boy	White British	Queer	0	N/A	Late-ish - hormones and top
Martin	N/A	N/A	N/A	N/A	N/A	Post-grad	I don't know how to answer this
Amit	24	Non-binary	Indian	N/A	10,000	University	Mid-transition
Terry	33	Male	Slavic	Queer	N/A	University	Phalloplasty
Alex	28	Non-binary	White British	Queer	20,000	A-Level	Testosterone, top surgery

On hormones, waiting for

top surgery

# 3.3.2. Materials

A semi-structured interview schedule (SSIS) was created to align with the research aims of the pilot study. A SSIS was chosen in line with recommendations from Smith, Flowers, & Larkin (2009); it was designed to elicit detailed responses with open-ended, nonleading questions relating to the phenomenon being investigated. The focus of the interviews was on the process of transition in the UK for transmasculine individuals, in addition to language preferences in the community. The SSIS for the interviews (see Appendix A) consisted of eight discussion prompts which included questions such as "How would you describe your method of transition?" A separate SSIS was created for the focus group (see Appendix B) having been narrowed in focus and informed by the initial interviews. The SSIS for the focus group consisted of questions such as "What role has medical care played in your transition?".

#### 3.3.3. Procedure

**Data Collection.** For the interviews, a public post on my personal Facebook page briefly summarised the research and people were invited to send me a private message indicating their interest in taking part. Individuals who contacted me were provided with a copy of a Participant Information Sheet which outlined the aims of the research and what it would entail, in addition to possible risks, confidentiality agreements, and ethical considerations which were in place.

All three participants were interviewed in a private room in a library. The interviews lasted between 56 and 94 minutes and were audio recorded. I transcribed the interviews verbatim, and ensured they were fully anonymised. Participants were reminded that they were able to withdraw from the study at any point, without having to provide a reason. If they agreed to continue, they were asked to sign a consent form and fill out a demographic information sheet. Immediately following the interview, the participants were debriefed and were provided with further information regarding the study and contact details for the research team and external support agencies should they feel distressed or have any concerns. They were also given the opportunity to ask any questions they may have had.

For the focus group, I contacted a trans-led transgender support group and requested permission to attend a private retreat aimed at transgender and transmasculine individuals in the UK. The focus group was then advertised to attendees of the weekend retreat in the form of a leaflet, and they were requested to sign up as spaces were limited to nine. Individuals who were interested in taking part were provided with information sheets on their arrival which outlined confidentiality agreements, ethical considerations, and the aims and process of the research.

The focus group was held in a private room that had been booked for the retreat. Participant information sheets were given out and each participant who wished to continue was asked to sign a consent form and fill out a demographic information sheet. The focus group lasted for 118 minutes. Following the focus group, the participants were given further information about the project in addition to being provided with contact information for the research team and for external support agencies. The participants were provided with time at the end of the focus group to ask any questions or raise any concerns about the project.

**Data Analysis.** Interpretative Phenomenological Analysis (IPA) was used to analyse the data. The analytic process was largely informed by published IPA guidelines (Smith et al., 2009) and peer-reviewed papers related to IPA in a focus-group context (Palmer, Larkin, de Visser, & Fadden, 2010; Tomkins & Eatough, 2010). IPA was chosen as an appropriate analytic framework for both the pilot interviews and focus group due to its suitability to conceptualise phenomenological experiences, particularly from a social constructionist background (Shinebourne, 2011).

In line with the idiographic nature of IPA, each interview was analysed in-depth and was re-read at least twice, then initial impressions were noted. Exploratory comments, indicating analytic commentary regarding descriptive content, language use, and more abstract, conceptual concepts were noted in a separate margin (Pietkiewicz and Smith, 2014). The next stage involved using post-it notes to indicate themes which emerged from both the transcript and the commentary. A list of emergent themes was then constructed in chronological order; this list was printed, and each emergent theme was cut out. After this, the emergent themes were moved around to generate clusters of related themes, named super-

ordinate themes. The final stage of analysis involved investigating connections across the three cases. This stage consisted of clustering super-ordinate themes to create overarching themes, according to their shared higher-order qualities, ultimately representing the most salient and interesting themes within the data set.

The focus group analysis necessitated a somewhat different approach; given the complex interactional context of focus groups they have not often been used within the IPA approach. While focus groups are inherently less phenomenological, as the experiential claims and concerns are embedded in the social and environmental interactions within the group setting, they have also been demonstrated to bring additional elements which add to the richness of the data (Palmer et al., 2010). In line with the IPA focus group framework outlined in Palmer, Larkin, de Visser, and Fadden's (2010) article the first round of analysis consisted of identifying experiential claims and concerns central to the transcript. I then went through the transcript once again for each stage of analysis in which I examined positionality (how both myself and the participants were approaching the questions posed); roles and relationships within the group, organisations and systems present in the lives of participants, use of stories and the language used by participants. I completed the analysis of the focus group data by following the framework for emergent and super-ordinate theme identification as laid out by Smith, Flower, and Larkin (2009).

## 3.4. Transmasculine Experiences of Substance Use Project

The main study of this thesis, named The Transmasculine Experiences of Substance Use Project (TESUP), aims to investigate the various factors which contribute to the development and maintenance of substance use (tobacco, alcohol, drugs) and substance use disorders in the transmasculine population. The factors investigated included stage of transition, quality of life, male gender role conflict, gender minority stress and resilience, psychopathological symptoms, and personality traits. This is a mixed-methods study which adopted an explanatory, sequential design; quantitative data was first collected and analysed, this then informed the qualitative data collection. It is also a longitudinal study; the same measures were repeated by the same participants one year after completion of the original questionnaire and interviews (See Figure 3.1). Henceforth, the first stage of quantitative and qualitative analysis shall be called TESUP-1 and the second stage of quantitative and qualitative analysis shall be called TESUP-2.

Phase 1 (TESUP-1)



Figure 3.1. Explanatory, sequential mixed-methods, and longitudinal design model

## 3.4.1. Participants

In the first quantitative stage (Figure 3.1), 105 transgender individuals who identified as male, masculine, or non-binary were recruited to take part in an online questionnaire in November 2019. The ages of participants ranged from 18 to 52 years (M = 25, SD = 5.84) (see Table 3.3 for TESUP-1 demographic information; see Table 3.4 for TESUP-1 ages and substance use scores). Participants had to be living in the UK to take part. All participants received a £10 Amazon voucher on completion of the questionnaire. In the first qualitative stage (Figure 3.1), 13 participants were invited to take part from those who had completed the quantitative stage, these participants comprised 7 of the highest scoring and 6 of the lowest scoring on the measures of alcohol and drug use. For reference, the highest possible score on AUDIT is 40 points and on DUDIT is 44 points. The ages of participants ranged from 19 to 34 years (M = 25.62, SD = 5.347) (see Table 3.5 for qualitative TESUP-1 demographic information with substance use scores). Participants who took part in the interviews each received a £20 Amazon voucher.

	<b>TESUP-1</b> (N = 105)		TESUP.	-2(N = 82)
	N	%	0	%
Gender Identity		,,,		,.
Binary (Male)	74	70.5	54	65.9
Non-binary	31	29.5	28	34.1
Race/Ethnicity				
White	98	93	75	91.5
Mixed Race	4	4	4	4.9
Black	3	3	2	2.4
Indian	1	1	1	1.2
Sexual Identity				
Bisexual	29	28	27	32.9
Queer	20	19	16	19.5
Heterosexual	17	16	6	7.3
Gay	16	15	19	23.2
Pansexual	9	9	4	4.9
Asexual	7	7	9	11
Other	6	6	1	2.1
Disability Status				
No disability	-	-	61	74.4
Disability	-	-	21	25.6
Employment				
Employed	66	62.9	56	68.3
Student	26	24.7	15	18.3
Unemployed	13	12.4	11	13.4
Education (Qualification Levels)				
No qualifications	1	1	1	1.2
Level 2 (GCSE)	13	12.4	6	7.3
Level 3 (A level)	20	19.1	15	18.3
Level 4 (CertHE/NVQ)	12	11.4	5	6.1
Level 5 (Foundation degree)	7	6.7	4	4.9
Level 6 (Degree with honours)	41	39	41	50
Level 7 (Postgraduate/MA/MSc)	11	10.4	10	12.2
Stage of Transition				
Pre-transition	4	3.8	2	2.4
Early transition	20	19	10	12.2
Mid-transition	41	39	32	39
Late transition	28	26.7	29	35.5
Post-transition	12	11.5	7	8.5
Current Tobacco Use				
Daily	12	11.4	10	12.2
Less than daily	33	31.4	33	40.2
None	60	57.1	39	47.6

**Table 3.3.** Questionnaire demographic information in both TESUP-1 and TESUP-2 of thequestionnaire. Count totals and percentages.

	TE	SUP-1	TESUP-2		
	MEAN	SD	MEAN	SD	
Age	25	2.84	26	6.21	
AUDIT	7.49	5.9	11.96	8.81	
DUDIT	5.7	6.57	10.98	11.55	

 Table 3.4. Age and AUDIT and DUDIT scores in TESUP-1 and TESUP-2. Means and standard deviations.

Table 3.5. Interview Demographic Information in TESUP-1. Including tobacco, AUDIT, and DUDIT scores. Group split by low/high substance use.

Pseudonyms	Age	Gender	Ethnicity	Sexuality	Group	Current	AUDIT	DUDIT
					ID	Tobacco	score	score
Aaron	26	Male/trans/genderqueer	White (Slavic)	Bisexual	Low	Not at all	2	0
Darren	21	Male	White (British)	Unspecified	Low	Not at all	4	0
Jeremy	31	Male	White (British)	Uncertain	Low	Not at all	2	0
Cody	19	Transmasculine	White (British)	Bisexual	Low	Not at all	6	0
Bailey	34	Transmasculine	White	Pansexual	Low	Not at all	2	0
Ezra	28	Transmasc	White	Bisexual	Low	Not at all	2	2
Mean [SD]	26.5[5.75]	-	-	-	-	-	3[1.67]	.33[.82]
Owen	21	Male	White (European)	Bisexual	High	Not at all	13	3
Rowan	20	Male (transgender)	Mixed race:	Pansexual	High	Not at all	8	1
			Black and Indian					
Parker	28	Non-binary/transmasc	White (British)	Pansexual	High	Not at all	3	10
Zack	19	Trans male	White (British)	Bisexual	High	Daily	8	9
Simon	26	Male	White (British)	Straight	High	Daily	13	0
Gale	34	None	White	Queer	High	Less than	12	7
						daily		
Logan	26	Male	White (Scottish)	Gay	High	Daily	10	17
Mean [SD]	24.86[5.31]						9.57[3.59]	6.71[5.96]

In the second quantitative stage (Figure 3.1), all 105 of the original participants were invited to take the same questionnaire again one year later in November 2020. Of those invited, 82 completed the questionnaire. The ages of participants ranged from 18 to 53 (M = 26, SD = 3.28) (see Table 3.3 for quantitative TESUP-2 demographic information; see Table 3.4 for TESUP-2 ages and substance use scores). All participants received a £10 Amazon voucher on completion of the questionnaire. In the second qualitative stage (Figure 3.1), all of the original 13 participants were invited to be interviewed. Of those invited, nine took part in the follow-up interview. The ages of participants ranged from 20 to 35 years (M = 25.56, SD = 5.725) (see Table 3.6 for qualitative TESUP-2 demographic information with substance use scores). Participants who took part in the interviews each received a £20 Amazon voucher.

Pseudonym	Age	Gender	Ethnicity	Sexuality	Group	Current	AUDIT	DUDIT
					ID	Tobacco	score	score
					(TEP1)			
Darren	22	Male	White (British)	Bisexual	Low	Not at all	8	0
Jeremy	32	Male	White (British)	Queer	Low	Not at all	4	0
Cody	20	Transmasculine	White (British)	Bisexual Ace	Low	Less than	15	6
						daily		
Bailey	35	Transmasculine	White	Bisexual	Low	Not at all	4	1
				Demisexual				
Ezra	29	Transmasc	White	Bisexual	Low	Not at all	3	0
				(male pref)				
Mean [SD]	27.6[6.43]	-	-	-	-	-	6.8[4.97]	1.4[2.61]
Owen	22	Male	White (European)	Bisexual	High	Less than	24	27
						daily		
Rowan	21	Male (transgender)	Mixed race:	Bisexual	High	Less than	12	6
			Black and Indian			daily		
Parker	29	Non-binary/transmasc	White (British)	Bi/Pan	High	Not at all	15	14
Zack	20	Trans male	White (British)	Bisexual	High	Daily	24	14
Mean [SD]	23[4.08]						18.75[6.185]	15.25[8.694

### 3.4.2. Materials

**Interviews.** A semi-structured interview schedule (SSIS) was designed for use in the TESUP-1 interviews. The SSIS consisted of 12 open-ended questions, such as "In what situations do you typically use substances?" and "What do you do to cope if you have the urge to use substances but are unable?" (See Appendix C). A separate SSIS was created for the TESUP-2 follow-up interviews. This SSIS consisted of 10 open-ended questions such as
"Can you tell me about what may have changed in your life since we spoke one year ago?" and "Can you describe to me a time in the past year where you have taken drugs or drank alcohol?" (See Appendix D). This SSIS also included time for topics specific to each individual to be raised, based on issues they had emphasised in their previous TESUP-1 interview.

**Demographics**. In TESUP-1, questions regarding age, sexuality, gender identity, ethnicity, and SES were included. To measure 'stage of transition' participants were asked to place themselves on a 1-5 scale of self-defined transition stages (1 = pre-transition, 2 = early transition, 3 = mid-transition, 4 = late transition, 5 = post-transition). In TESUP-2 these same questions were retained and additional questions regarding disability status were added.

Substance Use. Tobacco use was measured by three questions relating to quantity and frequency of current and past use. Alcohol use was assessed using the 10-item Alcohol Use Disorders Identification Test (AUDIT; Babor et al., 1992). Scores range from 0 - 40 and a cut-off value of 8 or more points indicates hazardous alcohol use. The internal consistency of the AUDIT is reported as acceptable (Cronbach alpha = .86) and test-retest reliability is .90 (Rubio et al., 1998). Drug use was assessed using the 11-item Drug Use Disorders Identification Test (DUDIT; Berman et al., 2005). Scores range from 0 - 44 and a cut-off value of 6 or more points indicates harmful drug use. Internal consistency of the DUDIT is reported by the authors as .80. In TESUP-2 an open question was added to assess the participants perceptions of change in substance use since TESUP-1.

**Quality of Life.** Quality of life was measured using the WHOQOL-BREF scale (World Health Organisation, 1996). The WHOQOL-BREF is a 26-item assessment that evaluates quality of life over four domains: physical heath, psychological health, social relationships, and environment. Higher scores indicate increased endorsement of items within

that domain. The internal consistencies range from .66 (social relationships) to .84 (physical health) (WHOQOL group, 1998). In TESUP-2 an open question was added to assess the participants perceptions of change in quality of life since TESUP-1.

Gender Minority Stress and Resilience. Gender Minority Stress and related protective factors were measured using the Gender Minority Stress and Resilience measure (GMSR; Testa et al., 2015). The GMSR includes 58-items and nine subscales: Gender-related Discrimination (D), Gender-related Rejection (R), Gender-related Victimisation (V), Nonaffirmation of Gender Identity (NA), Internalised Transphobia (IT), Pride (P), Negative Expectations for the Future (NE), Non-disclosure (ND), and Community Connectedness (CC). Higher scores indicate increased experiences and attitudes specific to that subscale. All subscales had adequate internal consistencies, and good divergent and convergent validity has been demonstrated among transgender adults (Testa et al., 2015). In TESUP-2 an open question was added to assess the participants perceptions of change in gender minority stress (e.g., experiences of discrimination, rejection, victimisation, etc.) since TESUP-1.

**Gender Role Conflict**. Gender Role Conflict was measured using the Gender Role Conflict Scale (GRCS; O'Neil, et al., 1986). The GRCS is a 37-item scale which includes four factors: Success, Power, and Control (SPC), Restricted Emotionality (RE), Restricted Affectionate Behaviour Between Men (RABBM), and Conflict Between Work and Family Relations (CBWFR). Higher scores indicate increased endorsement of items in that factor. The authors report that all factors had adequate internal consistencies (SPC = .85, RE = .82, RABBM = .83, CBFWR = .75). In TESUP-2 an open question was added to assess the participants perceptions of change in their own masculinity (e.g., thoughts, feelings about themselves or other men, restricted emotionality, etc.) since TESUP-1. **Psychopathological Symptoms**. Psychopathological symptoms were measured using the Symptom Checklist 90 – Revised (SCL-90-R Derogatis et al., 1994). The SCL-90-R is a 90-item assessment which contains nine symptom dimensions: Somatisation (SOM), Obsessive-compulsive (OBC), Interpersonal Sensitivity (INS), Depression (DEP), Anxiety (ANX), Hostility (HOS), Phobic Anxiety (PHO), Paranoid Ideation (PAR), and Psychoticism (PSY). It also includes three global measures of distress: Global Severity Index (GSI; severity of distress and number of symptoms endorsed), Positive Symptom Total (PST; number of symptoms endorsed), and Positive Symptom Distress Index (PSDI; average severity of distress reported). Higher scores indicate increased experiences of symptoms within that dimension. Internal consistencies have been reported to range from .79 (PAR) to .90 (DEP) (Horowitz et al., 1988). In TESUP-2 an open question was added to assess the participants perceptions of change in mental health symptoms since TESUP-1.

**Personality.** Personality traits were measured using the Temperament and Character Inventory – 140 (TCI-140; Cloninger et al., 1994). The TCI-140 is a 140-item inventory which includes 7 personality dimensions. There are four temperament dimensions: Novelty Seeking (NS), Harm Avoidance (HA), Reward Dependence (RD), and Persistence (PS), and there are three character dimensions: Self-Directedness (SD), Cooperativeness (CO), and Self-Transcendence (ST). Higher scores indicate increased agreement of statements that are specific to that domain. Internal consistencies are reported as ranging from .78 (NS) to .89 (HA, PS, ST) (Farmer & Goldberg, 2008). In TESUP-2 an open question was added to assess the participants' perceptions of change in personality traits since TESUP-1.

**COVID-19.** Due to the COVID-19 pandemic beginning between TESUP-1 and TESUP-2, open-ended questions relating to whether the participants perceived the pandemic to have had an impact on any of the variables measured were added. This included questions

such as "Has COVID-19 affected your use of substances such as tobacco, alcohol, and drugs?" and "Has COVID-19 affected your transition?".

#### 3.4.3. Data Collection

Participants for the TESUP-1 questionnaire were recruited via posting on private Facebook groups for transgender people in the UK and a public advertisement on the Gender Identity Research & Education Society (GIRES) website. Interested individuals could click on a link which directed them to a Participant Information Sheet which outlined the purpose of the study, ethical considerations, and confidentiality. If they wished to proceed with the questionnaire, they then gave informed consent before taking part. At the end of the questionnaire, participants were provided with further information about the study and contact information for any questions and support. Participants also had the opportunity to indicate if they would like to be contacted to participate in an interview at a future date. The questionnaire was conducted and analysed prior to qualitative data collection.

A nested sample of 13 participants were invited to take part in the semi-structured interview for TESUP-1. Individuals who indicated they would like to participate in further research were contacted via email and were provided with a Participant Information Sheet which outlined the purpose of the interviews, ethical considerations, and confidentiality. If they chose to participate, they gave informed consent prior to the interviews. The interviews lasted between 46 and 87 minutes and on completion the participants were provided with further information about the study and the contact details of organisations if they required any support. The interviews were audio recorded, transcribed verbatim, and anonymised.

One year after initially participating in TESUP-1 questionnaire, all participants who indicated they would like to take part in future research (N = 101) were sent an email inviting them to take part in the TESUP-2 questionnaire. Interested individuals could click on a

personal link which had been generated for them to anonymously take part. This allowed me to match responses to their personal ID number from TESUP-1 without their contact details being connected to the response. On following their personal link, participants were directed to a Participant Information Sheet which outlined the purpose and aims of the study, ethical considerations, and confidentiality. If the participants continued, they had the opportunity to give their informed consent before starting the questionnaire. On completion of the questionnaire, participants were provided with more information about the aims and possible findings of the study, in addition to being given contact details for support organisations should they require additional support. Overall, 82 participants took part in TESUP-2, with an attrition rate of 18.6%. The TESUP-2 questionnaire was conducted and analysed prior to TESUP-2 interview data collection.

One year after initially participating in TESUP-1 interviews, all 13 participants were contacted via email inviting them to take part in another interview. Nine participants agreed to another interview, one participant declined, and three did not respond. The email invitation included a participant information sheet which outlined the purpose of the interview, potential risks and benefits of participation and confidentiality. Individuals who took part gave informed consent prior to the interview. The interviews lasted between 52 and 76 minutes and on completion the participants were provided with contact details of relevant support organisations and further information about the study. Interviews were audio recorded, transcribed verbatim, and anonymised.

#### 3.4.4. Data Analysis

**Quantitative Analysis.** Descriptive statistics and inferential statistics, such as correlations, ANOVAs, binary logistic regression, and multiple regression analysis were

performed using SPSS version 27. All statistical tests were conducted at the 95% confidence level.

**Qualitative Analysis.** Thematic Analysis (TA; Braun & Clarke, 2006) was used to analyse the textual data from the questionnaire. The responses were collated and from these, emergent themes were generated which reflected the important issues and concerns reported by participants. This method allows for the emergence of topics that were not anticipated or directly prompted by the survey questions.

Interpretative Phenomenological Analysis (IPA) was used to analyse the interview data (Smith et al., 2009). All data analysis was performed using NVIVO version 12. After reading each transcript multiple times, initial observations were noted regarding the descriptive content, linguistic style, and conceptual aspects of the data. From these notes, a chronological list of emergent themes was created for each case, before exploring connections across multiple cases. The low-risk substance use group and high-risk substance use group were first analysed separately to allow for a more nuanced analysis of responses, followed by an overall analysis of the super-ordinate themes in both groups and bringing them together. By clustering important super-ordinate themes across all cases, overarching themes were created; these represent the most relevant and interesting themes throughout all cases.

### 4. Pilot Study

This pilot study arose from the need to understand current best practice of language use within the transmasculine community in the UK, in addition to attempting to understand how the process of transitioning is conceptualised within a UK context. This research aimed to understand the currently accepted nomenclature regarding trans experiences. This study also explored how transmasculine individuals conceptualise the process of transition, with a focus on access to healthcare.

## 4.1. Literature Review

In the UK the term transgender is most often used as an umbrella term to encompass all individuals whose personal experience and conceptualisation of their own gender differs from the sex they were assigned at birth (usually assigned along binary lines of male or female) (Ellis et al., 2015). The term transmasculine, in this paper, is used to refer to individuals who are transgender, were assigned female at birth, and align with a masculine gender role, they may identify themselves as male, masculine, non-binary, genderfluid, or a variant of this nature. Not all transmasculine people desire medical interventions and many will not engage with any gender-related services (Austin & Goodman, 2018); this is one reason that it is difficult to determine the true population of the trans community. Some transmasculine people will transition socially by changing their name, pronouns, and appearance, whereas others will also decide to engage with private or NHS GIS. Similarly, the extent to which medical interventions are desired differs between individuals. For example, one study indicated that transmasculine frequently "top surgery" chest people more seek (i.e.,

masculinisation/mastectomy) than "bottom surgery" (i.e., surgically constructing a penis) (Yerke & Mitchell, 2011).

For adults wishing to seek medical interventions or support for transitioning there are seven NHS funded GIS in the UK, and an additional one which caters to under 18s (GIRES, 2015). According to new figures, over 13,500 individuals are on a waiting list for their first appointment with an NHS GIS in the UK (BBC, 2020). The current NHS guidelines state that the maximum wait between GP referral and first appointment should be 18 weeks; however, the reality is that some people face waits of up to 8 times as long, meaning that the wait for their first appointment can be nearly 3 years (NHS, 2019). For adults approaching the NHS for diagnosis and treatment of gender dysphoria, the World Professional Association for Transgender Health (WPATH) Standards of Care (SOC), version 6, recommended a triadic therapy approach which consists of 12-24 months of 'real life experience', hormone therapy, and sex reassignment surgery (WPATH, 2001). The updated SOC -7 (WPATH, 2012) now recommends "continuous months of living in a gender role that is congruent with a person's gender identity"; here WPATH does not refer to this requirement as 'real life experience' but effectively creates the same stipulation without a specific period of time. The vagueness in guidance allows considerable space for subjective interpretation of clinicians for how and when a person has met the aforementioned SOC-7 criteria. In spite of the removal of the term 'real life experience' this remains a dominant idea within clinical practice, as is demonstrated in multiple publications released after the SOC-7 (Collazo, et al., 2013; Austin & Craig, 2015).

The interpretation of WPATH guidelines in the UK is variable, as there are no NICE guidelines for transition-related interventions. Though the WPATH Standards of Care (SOC-7) state that "treatment is individualised" (WPATH, 2012, p.5), the NHS service specifications for transition-related treatment pathways imply a linear model of social transition, hormonal interventions, then surgical interventions (NHS England, 2019). Extant literature regarding

71

transgender service-user attitudes to this approach depict dissatisfaction and demonstrates how it is perceived as a barrier to autonomy and individualised care (Ellis et al., 2015; Harrison et al., 2020).

Additionally, to access NHS GIS, individuals must first meet the criteria for a diagnosis of Gender Dysphoria (GD), as outlined in the DSM-5 (APA, 2013). However, many trans activists and researchers have disputed the necessity of a GD diagnosis, claiming that this diagnostic model pathologizes the lived experiences of trans individuals and overlooks the possibility that the distress reported is not arising from an underlying psychological impairment, but is instead as a result of society's negative response to gender nonconformity (Ashley, 2019; Schulz, 2018). Transgender individuals face discrimination and stigma in many forms, from increased rates of homelessness and unemployment (Spicer, 2010) and to being the victims of abuse, assaults, and violence (Whittle et al., 2007). Despite this, it has been argued that the diagnostic/medical model overlooks these societal sources of distress and instead situates it as arising *only* from internal emotional distress and severe bodily dysphoria (Schulz, 2018).

The diagnostic/medical model of transgender identities, as advocated by the DSM and WPATH, promotes a view of the transgender experience based on distress and dysphoria, which has led some transgender activists and academics to view this model as a dehumanising barrier to transition-related healthcare (Ashley, 2019). This is due in part to disparate definitions between service-users and clinicians regarding the route and goals of transition; those seeking transition often view clinicians as 'gatekeepers' and, in anticipation of encountering barriers, can feel forced into adopting the 'distress narrative' to access diagnosis and treatment (Waszkiewicz, 2006). Likewise, the diagnostic/medical model, as is utilised by the NHS, is perceived by transgender service-users to place additional barriers to transition-related treatment for those who do not fit into the discrete binary categories of 'male' or

'female', which may lead to non-binary or agender people to feel forced into hiding their identity in order to access necessary medical services (Green, 2004). This is in spite of NHS publications which emphasise their commitment to meet the healthcare needs of non-binary service-users (NHS England, 2019). Research which focuses on understanding the lived experiences of non-binary individuals to best provide healthcare treatment emphasises an affirmative and intersectional approach (Taylor, et al., 2018). This approach encourages and supports individuals to explore and articulate their gender identity, while considering the cultural and social context the individual exists within. This would necessitate careful consideration of aspects of intersecting identities, such as religion, sexuality, class, and ethnicity, and may also challenge the assumptions and beliefs of the service provider (Eyssel et al., 2017).

Building on this approach, and in response to the barriers experienced by transgender individuals, further research suggests abandoning the diagnostic/medical model in favour of the 'Informed Consent Model' (Informed Consent for Access to Trans Health, n.d., Cavanaugh, Hopwood, & Lambert, 2016). This model attempts to ameliorate the 'gatekeeping' aspect of accessing transgender healthcare services by removing the necessity of a GD diagnosis. Instead of being assessed by a mental health professional who then grants access to transition services, transgender service users are given the power to decide if they are ready to access transitionrelated healthcare through the routes and methods they would prefer (Schulz, 2018). Here, the role of the health practitioner is to support and collaborate with the service user to come to an informed decision by providing information regarding the social, psychological, physical, and financial aspects of receiving transition-related healthcare. In this way, the Informed Consent Model works to de-pathologize and destigmatise transgender experiences. The Informed Consent Model has been increasingly practiced in trans healthcare in various countries worldwide but has not yet been used within NHS services. A review of the clinical practices of USA non-profit, private, and academic clinics which adopted the Informed Consent Model demonstrated that within 12 clinics representing a total of 1,944 patients there were only three cases of 'regret' (0.1%) resulting in reversal of transition-related treatment (Deutsch, 2012). In comparison to this, research into WPATH surgeons' experiences of patient regret of transition-related treatment (Danker et al., 2018) found that within approximately 22,725 patients 62 (0.28%) expressed a desire for 'reversal' of their transition. This result provides support that the Informed Consent Model does not increase cases of transition-related 'regret'.

At the same time as seeking and navigating medical healthcare services, a transgender individual will also be undergoing a process of identity development as they come to understand themselves and their gender identity. Various theories of transgender identity development have been put forward, the majority of which are 'stage' models, which often emphasise the role of distress and medical interventions (Kuper et al., 2018). These stage models (Devor, 2004; Hiestand & Levitt, 2005) follow similar themes: early stages are often characterised by confusion and distress; after the initial discovery of their transgender identity the individual often seeks out relationships with LGBTQ people and conducts their own research; with this new information, the individual begins to identify with a transgender identity, now exploring labels and gender enactments more intensely, in relation to their own sense of authenticity; having accepted their status as transgender, the individual must decide if they want to transition; following transitioning, the individual learns to accept and develop pride in their transgender identity, manage stigma and discrimination, and ultimately engage in advocacy. Despite the popularity of stage models of transgender identity development, there are concerns that such models overlook generational, cultural, and geographical differences (Bilodeau & Renn, 2005). Furthermore, they indicate that identity develops along a linear developmental trajectory and do not allow for diversity and flexibility of gender expression, gender identities, goals, and methods that the transgender community expresses and celebrates.

When the participants in this study were given a space to speak openly and honestly about their experiences, they took this opportunity to detail the difficulties of accessing transspecific healthcare and the struggles they face when consolidating their needs with the reality of what is available to them.

## 4.2. Results

This study used IPA methodology to analyse three interviews and a focus group (for a full discussion of methods and participants please see Chapter 3; 3.3). Three super-ordinate themes arose throughout the interviews and focus group: (1) conceptualising the process of transition; (2) NHS communication and support; and (3) medicalisation, power, and non-disclosure. Several participant extracts are provided to illustrate each theme following Smith's (2004) recommendations.

## 4.2.1. Conceptualising the Process of Transition

While many of the individuals conceptualised their transition as a physical and personal "journey" toward the authentic and true self, it was simultaneously conceptualised in terms of what was made accessible to them through GIS. It was made clear by the participants that while this was a personal and individualised odyssey, it was only truly accessible within the strict confines of NHS funding and GIS protocol. The participants perceived considerable inflexibility in the chronology of the process of transition as recommended by GIS. When discussing his transition "route", one of the interviewees, Romeo, described how he had "always been told, it's hormones, then top surgery, then bottom surgery" and while he agreed this was the route for him, he also acknowledged that this is not the case for all those seeking GIS intervention (see Box 4.1; 1.1). The way that GIS dictate and control the process of

transition was noted as being "strict" and was given as a motivation for opting for private care (1.2).

Most participants conceptualised their transition as marked by various "processes" or "stages". Within the focus group there was a consensus that transition begins with a necessary period of introspection. It was emphasised by multiple members of the group that this process of identity discovery is highly individualised, yet they did not believe that GIS see it this way and reported instead that GIS promote a narrow and stereotypical trans narrative (1.3). The order of the entire transition process was debated amongst the focus group who concluded that while transitioning is non-linear and unique to each individual, it can be described as an overarching process marked by various stages, and that undertaking one's own external research and searching for community was an integral aspect of initial identity discovery (1.4).

The entire transition 'journey' was conceptualised in multiple ways, but all had threads of commonality. The major theme among the descriptions was a sense of having to convince or persuade healthcare professionals at GIS of the authenticity of their transgender identity (1.5). It was often framed as a series of arbitrary and intrusive challenges set by GIS which were to be defeated; all who mentioned this process alluded to the huge amount of time and the element of luck required (1.6). **Box 4.1**. Representative quotes for "Conceptualising the Process of Transition". Quotes are from Focus Group (FG) or interviews if not FG

- 1.1. "I know for some people this route just isn't for them because maybe they want chest surgery first or something, but I know the gender clinic told me that this is the best route" (Romeo)
- **1.2.** "They [GIS] tell you that you'll get a better outcome if you're taking testosterone [before top surgery], I dunno if that's medically verified or just their opinions" (Alex, FG)
- **1.3.** "It's important to note that this [identity discovery] can happen at any age, and it's the gender clinics that are pushing the narrative that we were all tomboys, we all knew as a little kid and that we hated dresses" (Raj, FG)
- **1.4.** "For me, and a lot of men and women, there was a prolonged searching phase where you search for answers, and proof that you're not alone" (Amit, FG)
- **1.5.** "I'd say the first stage is the convincing stage, then admin, then testosterone, then more admin, then top surgery" (Charlie)
- **1.6.** "Once you've made it past the waiting lists you're ready to face the first boss, the gender clinic (...) they're gonna make you jump through hoops and will ask a million intrusive questions about your body and your sex life, and if you're lucky, they're gonna prescribe you hormones and maybe two years after that you level up and **bang** you hit the second boss, it's top surgery" (Greg, FG)

## 4.2.2. NHS Communication and Support

Whether seeking support and treatment for gender dysphoria, mental health, or sexual and reproductive health, navigating NHS services was an emotionally draining and convoluted process. All participants believed their transgender status negatively affected their access to NHS services. When approaching their general practitioner (GP) for information or seeking a referral, participants anticipated that the GP would lack the knowledge to help or would be unsupportive. This apprehension to approach their GP was compounded by the belief that a smooth transition is dependent on a supportive and well-informed GP (Box 4.2; 2.1). Participants in both individual interviews and the focus group discussed the necessity of being well prepared and the impact of GP expectations of a well-informed patient; one man described himself as "lucky" because he had done prior research and his GP was "ignorant of the process" (2.2). Participants also discussed how a lack of GP support was a barrier to accessing mental health services; the focus group in particular voiced concerns that accessing mental health services would risk stopping their referral to GIS (2.3). Participants who did seek help from their GP for mental health reasons reported feeling dismissed and suspicious of GP motives (2.4); GPs were viewed as not taking trans mental health seriously and placing the onus of responsibility on GIS for the participants' mental health care. Previous negative experiences with a GP served as reinforcement for participant's avoidance of seeking help for any issue with their GP, not just mental health (2.5). Additionally, participants who had accessed NHS mental health services expressed the belief that their transgender identity negatively impacted the care that they received (2.6).

Only two participants reflected positively on their referral process for their initial GIS appointment. One participant, Alex, reported that his experience was "actually okay" though he felt he had been let down by his GP. The other participant, Charlie, sought private medical care when he found that NHS waiting times far exceeded NHS guidelines. The remaining ten participants reported various complaints about the GIS referral process.

Multiple participants expressed significant discomfort and anxiety around the lack of communication from GIS once they had been referred (2.7), which led many to feel isolated and ignored by the healthcare system. This feeling was compounded by the excessively long waiting times the participants were subject to, which led some participants to the conclusion that GIS are ignorant of the true impact that waiting lists have on the lives of trans individuals (2.8).

Box 4.2. Representative quotes for "NHS Communication and Support"

<b>2.1.</b> "[I	had] a difficult GP and it's so important when you're beginning your transition that your GP is on your side, if they're not you're gonna have a hard time of it" (Mike)
<b>2.2.</b> "Y	ou could just see that they didn't really want to be having this conversation () they were really unprepared and ignorant of the process" (Charlie)
<b>2.3.</b> "I'	d just been referred [to GIS] my mental health wasn't great but I didn't wanna go and access my GP for medication for mental health because I'd risk the implications of having that referral [not] put through (noises of agreement) and so I thought okay I'll deal with my mental health problems, but first this referral needs to be put through, and that was the priority (Raj, FG)
<b>2.4.</b> "W	Then you're trying to access a GIC they say they don't know how to do it but when you're not trying to access trans-related healthcare in regards to mental health they just wanna send you off there" (Paul, FG)
<b>2.5.</b> "T	his entire experience has just made me even more apprehensive than I already was in relation to accessing any type of medical service" (Raj, FG)
<b>2.6.</b> "T	he second I mentioned I was trans, everything, just the whole conversation came back to me being trans and then it's like I was treated like a 2-D cardboard cut-out with no more aspects to my personality other than having to change my body and my gender history" (Alex, FG)
<b>2.7.</b> "Y	ou wait alone for your appointment, there's no support or information () you're just left in the dark with no idea" (Charlie)

# 4.2.3. Medicalisation, Power, and Non-disclosure

Participants felt that GIS, and the NHS, acted as adversaries. Rather than providing support and facilitating treatment, GIS were viewed as gatekeepers of medical diagnosis and subsequent treatment. Often referring to their experiences with GIS early in their transition as the 'convincing' stages of transition (Box 4.1; 1.5), participants described a reluctance to be wholly authentic in their interactions with medical professionals due to the perceived risk (Box 4.3; 3.1). Individuals across interviews and the focus group described how they believed the halting or delaying of treatment to be a punitive response by GIS in reaction to perceived doubt (3.2). This rigid conceptualisation of the transmasculine experience is incongruent with the flexibility and individuality of gender identity, as described by the participants in this study.

Thus, participants attempted to avoid potential barriers by conforming to the expectations of GIS and not asking questions (3.3).

Once having finally attained an initial appointment, participants described how they felt let down as their lived experiences were ignored and that they felt pressure to 'perform' and 'jump through hoops' for GIS (3.4). Non-binary individuals deliberated over whether to disclose their non-binary identity (3.5), while others discussed feeling forced to 'assimilate' by GIS (3.6). These negative experiences of having their identities repeatedly invalidated, whilst feeling unable to discuss matters of mental health or gender nonconformity, intensified existing distress for these individuals.

The participants in this study were aware of the power exerted over them by GIS and of the negative effects that are felt due to this power imbalance. One participant, Amit, attributed this power imbalance as the reason why they could not be truly authentic with GIS (3.7). Furthermore, some participants voiced the opinion that they felt powerless in their own transition and were 'stuck in the system' (3.8). This lack of control over their own situation was felt so profoundly, particularly within the focus group, that it provoked a passionate discussion of the necessity for, and the costs of fighting for your rights, and the rights of those in your community (3.9). This discussion culminated in a call to 'take back that power', for the empowerment of transgender individuals, and for the right to live your authentic life.

Box 4.3. Representative quotes for "Medicalisation, Power, and Non-disclosure"

- 3.1. "They put you in the position where I'm gonna hide stuff because of the risk" (Raj, FG)
- **3.2.** "Even if I said the slightest thing, like I feel a little bit depressed, then they'd just jump on it and stop the whole process, it's almost as if they're looking for a reason to stop you, or a reason to say you're doubting it, or they doubt you" (Romeo)
- **3.3.** "They take it as doubt whenever you ask a question, so it's best not to I've learned (Charlie)
- **3.4.** "[GIS] just want me to jump through hoops for them and perform for them" (Romeo)
- **3.5.** "They might withhold treatment, it's a very real fear ... and so when I went to my first assessment I was like, do I disclose that I'm non-binary?" (Amit, FG)
- **3.6.** "They only think about the medical side of it and they're pushing and pushing for you to assimilate, and they don't take into account the everyday processes we go through on the social side of it" (Mike)
- **3.7.** "[GIS are] considered experts so they do hold that power, so it means you can't tell them certain things" (Amit, FG)
- **3.8.** "I'm stuck in the system, it's like addiction now because they can't take away treatment, not for any fucking reason" (Francis, FG)
- **3.9.** "Elliot: When I had my first appointment, I told them that I'm **male**, when I'm not, I'm non-binary. That ended up hurting my health, but I knew I wouldn't get treatment otherwise"
  - Amit: You're right Elliot, and I think it's tricky because there are so many different narratives. We need to **take** back that power and we need to fight, but these things have a cost, we have to make choices like you [gestures to Elliot] had to make a choice. So, it's like you can either choose to be complicit and get your treatment on time, or you can fight, and be true and authentic and fight for your community, and make it so that non-binary people, we can keep fighting together. It's tricky deciding what do we sacrifice? We've already sacrificed so much" (FG)

## 4.3. Discussion

This study adds to existing literature regarding transgender identity development and offers a deeper understanding of how the NHS can be better equipped to provide the appropriate healthcare to the transgender community. This study sought to clarify and explore the idea of 'stages' of transition. This was with the purpose of creating an ethical and reliable way of documenting the chronological point at which a participant sits in their transition. Understanding how the transgender community conceptualises stages of transition would allow

future research to examine experiences that take place within them and provides parameters for understanding and comparing experiences across the transition timeline. The topic of discussion, however, evoked more than a grass-roots definition of transition and understanding of the transgender experience. The group seized the opportunity to share their views on the externally imposed transition pathways and the intricacies of navigating the GIS.

The stages of transition outlined by the GIS as an appropriate guide to a successful transition were rejected by the group. The idea of 'transition stages' was viewed as a way of controlling and gatekeeping access to medical treatment. The focus group argued that one's process of transition should be guided by introspection, information-seeking, and flexibility. While many 'stages' described by the participants are mirrored in the stage models of transgender identity, such as the work of Devor (2004), the predominant view was that such linear models do not accurately account for the fluidity, complexity, or diversity of lived experiences of transmasculine individuals. In light of this, it was suggested that broader descriptive categories (such as pre-transition, mid-transition, post-transition etc.) would be best placed to measure stage of transition, in order to capture the flexibility and uniqueness of transitioning that the participants described.

While everyone held their own personal concept of transition and the goals and route they prefer, they also expressed that they felt as if their transition was subject to the whims of the NHS. The NHS GIS pathway was criticised by participants for its paradoxical, one-sizefits-all approach. Rather than being guided through their transition, participants described lacking support and feeling "stuck in the system". These perspectives reveal the nature of the group's relationship with the GIS as necessary but also with the potential to be a harmful and difficult process. These findings build on the work of Ellis et al. (2015) which reported that 46% of respondents in their study reported that they had experienced difficulties obtaining treatment or assistance at their GIS. Participants in the current study reported that their transgender status negatively impacts their access to all NHS services. It was a major source of anxiety that their GP would be ill-informed about transgender healthcare which presented a significant barrier to accessing services. This is mirrored in previous research such as one systematic review (Heng et al., 2018) which found that a lack of healthcare provider knowledge was a frequent experience for transgender people. Furthermore, Taylor's (2013) exploration of transmen's experiences with healthcare reflects on the necessity of self-advocacy and health literacy, something which the participants in the current study also reported.

The vital information-seeking that the participants stated defined many of the initial 'stages' of transgender identity development was avoided in healthcare contexts. Asking questions and seeking information was perceived to cause the GIS to doubt their transgender status or their readiness for treatment. As a result, participants reported feelings of being made to 'perform' for healthcare professionals. To stave off any possible doubt from medical professionals, participants were hesitant to disclose sensitive information for fear that there would be a punitive response through the delaying or halting of treatment. This sentiment is reflected throughout the limited previous work which examines transgender experiences of GIS, for example, a recent UK study (Harrison et al., 2020) reported similar findings whereby many of their participants reported feeling pressured to 'prove' their gender identity to GIS. Participants in the present study described how the pressure to quickly convince healthcare professionals of their readiness to medically transition was intensified by the prospect of lengthy waiting lists with little to no communication from GIS. These findings provide support for prior studies which focus on transgender access to healthcare, these studies report that participants often express feeling a lack of support during their transition, particularly during the wait for their initial assessment and between appointments (Harrison et al., 2020; Linander et al., 2017).

The rigid diagnostic framework of NHS GIS and power inequalities inherent in this framework were apparent in the reports from participants; feeling powerless and lacking in autonomy was frequently described. The participants in this study expressed frustration with the "gatekeeping" role of healthcare providers, which has been described as "unethical" and "dehumanising" by leading transgender academics (Ashley, 2019). In response to various literature which clearly communicates the transgender community's growing dissatisfaction with the current diagnostic model (Ellis et al., 2015; Harrison et al., 2020; Waszkiewicz, 2006), and the rallying cries to "take back that power" voiced in the current study, a transition to the Informed Consent Model (Schulz, 2018) is suggested. Such a change in protocol would mitigate many of the barriers faced by this community when accessing transition-related healthcare, it would empower patients to make informed choices in collaboration with healthcare professionals and would place the UK and the NHS at the forefront of transgender healthcare. Within this model, trans individuals would not have to choose between identity and interventions, nor would they have to "sacrifice" or "fight", as was often mentioned by participants in this study. Previous studies indicate that healthcare providers who offer the Informed Consent diagnostic model are currently growing in popularity for both service providers and patients (Reisner et al., 2015).

In advance of a fundamental shift in the diagnostic model, some immediate practical recommendations are suggested, based on the experiences of the current study's participants and of extant literature in this field. Increased support for service users during waiting periods is necessary; participants reported feeling increased anxiety while waiting, therefore service users should be offered appropriate support. This support may include transgender-specific counselling or psychotherapy as was recommended by Harrison et al. (2020) or might consist of more frequent contact with service users. Furthermore, a lack of trans-specific training and knowledge was frequently mentioned by participants; an NHS-wide commitment to LGBTQ-

specific training is recommended to improve access to all services for the transgender population. Additionally, measures to facilitate multi-disciplinary working for GIS serviceusers is suggested because participants frequently discussed the difficulty of accessing mentalhealth services and GIS services simultaneously.

Criticisms of the inner working of the NHS and GIS which were mentioned frequently, are here described through the lens of the patient, and accounts of these processes may differ from that of the care providers themselves. It is hoped that perceived conflicts between service users and providers might be better understood by conducting future research into the perceptions of care providers in these settings. The participants were likely in receipt of care under different local authorities and health services, though this information was not gathered during this study. I am therefore unable to delineate regional discrepancies of care. However, there was a strong consensus in the type of criticisms made of healthcare provision from those that took part in this research.

In the context of the larger TESUP study, this pilot study sought to clarify the most accurate and appropriate method of operationalising 'stage of transition'. In addition to this, currently accepted language conventions were explored within the focus group. The results revealed that the concept of 'transition stages', particularly in a medical context, were rejected by the participants as being too strict and inflexible. Participants suggested, that for the purposes of the current research, broad descriptive categories that participants could self-define would allow for sufficient fluidity and diversity of stages of transitioning. These categories were decided within the focus group as being pre-, early, mid-, late, and post-transition, with no 'start' or 'end' point being defined by the researcher. Furthermore, the participants in the focus group agreed that the term 'transmasculine' was appropriate for the group that TESUP is aimed at. These suggestions informed the project design and language used within the TESUP study.

#### **5. TESUP-1**

#### 5.1. Descriptive Statistics

### 5.1.1. Demographics

The age of the 105 participants ranged between 18–52 years (M = 24.98, SD = 5.84). 29.5% (N = 31) of the sample identified their gender identity to be non-binary which included identities such as transmasculine, genderqueer, agender, non-binary, and genderless. The remaining 70.5% (N = 74) identified as male. The largest proportion of participants considered themselves to be mid transition (39%, N = 41), however, other participants (N =12) said they felt these options were limiting. Additionally, some participants (N = 38) took the opportunity to go into detail about their transition journey; these comments demonstrate the different paths taken during their transition (Appendix E). The most common sexual orientation was bisexual (N = 29), and the majority of the sample identified as White British (N = 98, 93%).

Completed level of education was assessed using the established levels of educational attainment in the UK (Gov.uk, 2022) which ranges from 0 (no completed education) to 8 (doctoral level). Levels in this sample ranged from 0 to level 7 (postgraduate master's level). Almost half of the sample (49.5%, N = 52) reported completing either an undergraduate or postgraduate degree or certificate. The majority of participants were in paid employment (62.9%, N = 66). Please see Table 3.3 in Chapter 3 for quantitative demographics variables.

**Tobacco Use.** Use of tobacco was determined through questions relating to currentand past-tobacco use. The majority (60.9%, N = 64) of the sample indicated that they had used tobacco in the past. Fewer participants (42.8%, N = 45) reported current use of tobacco. 39% (N = 41) of the sample reported never having used tobacco. Please see Table 3.3 for quantitative tobacco use data.

Of those who reported using tobacco daily currently (N = 45), the most frequently used method of using tobacco was smoking manufactured (68.8%, N = 31) or hand-rolled (40%, N = 18) cigarettes. The frequency with which tobacco was used on a daily basis ranged between 1–75 times per day, the most common frequency of use was between 1–10 times per day, with only five participants reporting a frequency of use greater than 10 times per day.

Alcohol Use. Use of alcohol was determined through administering the AUDIT (Babor et al., 1992). A cut-off score of 8 or over is suggestive of harmful and/or hazardous drinking habits and a cut-off score of 20 or over indicates that the individual may be at risk of, or are experiencing, alcohol dependence (Babor et al., 1992). For this study's sample, the mean score on the AUDIT was 7.49 (SD = 5.9), with scores ranging from 0–31 points. 49.5% (N = 52) of the sample met or exceeded the cut-off score of 8, and a further 6% (N = 7) of the sample met or exceeded the cut-off score of 8, and a further 6% (N = 7) of the sample met or exceeded the cut-off score of 8. And a further 6% (N = 7) of the sample met or exceeded the cut-off score of 8. And a further 6% (N = 7) of the sample met or exceeded the cut-off score of 20. Please see Table 3.4 for means and standard deviations of AUDIT.

**Drug Use.** Use of drugs was determined through administering the DUDIT (Berman et al., 2005). A cut-off score of 6 or more points is recommended to identify those with drug-related problems (Berman et al., 2005). For this sample, the mean score on the DUDIT was 5.7 (SD = 6.57), with a range of 0–22 points. 43.8% (N = 46) of the sample met or exceeded the cut-off score of 6. Please see Table 3.4 for means and standard deviations of DUDIT.

Participants were asked to indicate which drugs they usually take and had the option to select multiple drugs. A total of 59 participants indicated at least one drug that they typically used. The most commonly used drug was cannabis (23.8%, N = 25), followed by hallucinogens (17.1%, N = 18). Drugs also indicated in free-text comments but not included

in the questionnaire are DMT, NOS, Valium, and Ayahuasca. Heroin was the only substance included in the questionnaire which was not indicated by any participant. Please see Table 3.7 for frequencies of drug type used.

## 5.1.2. Correlation Results

Please refer back to each scales' description in the Methods chapter (*section 3.4.2. TESUP Materials*) for definitions of the abbreviations in each scale. Please see Appendix H for additional information and scattergrams of all correlations.

**Demographics.** Table 5.1 shows the correlations between demographic information (age, gender, education level), transition stage, and substance use (tobacco, alcohol, and drugs) on data collected from 105 transmasculine individuals. Not surprisingly, there were significant positive correlations between age and transition stage (r(103) = .230, p = .018), age and education ( $\rho(103) = .427, p < .001$ ), and transition stage and education (r(103) = .244, p = .013). There was also a significant positive correlation between age and gender (r(103) = .236, p = .016), indicating that younger participants tended to identify as binary male whereas the older participants more often identified as non-binary. Factors such as completed level of education, stage of transition, and gender identity (such as male, genderqueer, agender, etc) showed no significant correlations with the use of tobacco, alcohol, or drugs.

Interestingly, while the use of drugs had a moderate positive correlation to the use of alcohol (r(103) = .416, p < .001), 95% CI [.210, .618], the use of drugs was moderately negatively correlated with the use of tobacco (r(103) = ..443, p < .001), 95% CI [-.613, - .234].

Variable	М	SD	1	2	3	4	5	6	7
1. Age ( <i>r</i> )	24.98	5.841							
2. Gender binary $(r_{pb})$			.233* [.028, .449]						
3. Transition Scale ( <i>r</i> )	3.23	1.012	.236* [006, .422]	188 [391, .026]					
4. Education (ρ)	4.70	1.723	.427** [.228, .622]	009 [218, .189]	.244* [.058, .419]				
5. Tobacco score ( <i>r</i> )	4.57	1.358	191 [398, 019]	083 [254, .114]	.028 [194, .242]	100 [278, .157]			
6. AUDIT score ( <i>r</i> )	7.49	5.900	011 [181, .103]	001 [210, .148]	.028 [041, .328]	.047 [143, .280]	132 [361, .073]		
7. DUDIT score ( <i>r</i> )	5.70	6.571	.123 [125, .235]	.048 [154, .216]	.149 [083, .216]	.190 [140, .390]	443** [613, 234]	.416** [.210, .618]	

**Table 5.1.** Demographics, stage of transition, and substance use correlations. Means, standard deviations, and correlations with confidence intervals

*Note. M* and *SD* represent the mean and standard deviation, respectively. *R* and  $\rho$  indicate Pearson's correlation and Spearman's Rank correlation, respectively.  $r_{pb}$  indicates point-biserial correlation. Values in square brackets are the 95% confidence interval for each correlation.

\* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

**Quality of Life.** Table 5.2 shows correlations between the four domains of the WHOQOL-BREF and substance use (tobacco, alcohol, drugs). There were no significant correlations between quality of life and tobacco or alcohol use. Results demonstrated a weak negative relationship between physical health QoL and drug use, r(103) = -.212, p = .03, 95% CI [-.387, .011]. There was also a weak negative relationship between environment and drug use, r(103) = -.195, p = .04, 95% CI [-.372, .003].

Variable	М	SD	1	2	3	4	5	6	7
1. Physical Health QoL (r)	13.62	2.924							
2. Psychological Health QoL (r)	11.63	2.993	.520** [.364, .647]						
3. Social and relationships QoL ( <i>r</i> )	13.40	3.476	.280** [.093, .447]	.436** [.266, .579]					
4. Environment QoL ( <i>r</i> )	13.35	2.655	.570** [.424, .686]	.725** [.619, .804]	.471** [.307, .607]				
5. Tobacco score ( <i>r</i> )	4.57	1.358	088 [275, .105]	.069 [124, .257]	.082 [111, .269]	.128 [065, .312]			
6. AUDIT score ( <i>r</i> )	7.49	5.900	075 [262, .118]	.120 [073, .304]	039 [228, .153]	.130 [063, .313]	132 [361, .073]		
7. DUDIT score $(r)$	5.70	6.571	212* [387, .011]	.021 [171, .211]	114 [299, .079]	195* [372, - .003]	443** [613, - .234]	.416** [.210, .618]	

Table 5.2. Quality of Life and substance use correlations. Means, standard deviations, and correlations with confidence intervals

*Note. M* and *SD* represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation. \* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

**Gender Role Conflict.** Table 5.3 shows correlations between the four factors of the GRCS and substance use (tobacco, alcohol, drugs). The factors SPC, RE, and CBFWR showed no significant correlations to substance use. Results demonstrated a weak positive relationship between RABBM and alcohol use (r(103) = .215, p = .028, 95% CI [-.037, .484]) and a moderate positive relationship between RABBM and drug use (r(103) = .331, p = .001, 95% CI [.156, .504]).

Variable	М	SD	1	2	3	4	5	6	7
1. SPC ( <i>r</i> )	43.30	11.418							
2. RE ( <i>r</i> )	37.33	10.231	.213* [.048, .466]						
3. RABBM ( <i>r</i> )	25.38	8.817	.406** [.260, .612]	.670** [.530, .773]					
4. CBWFR ( <i>r</i> )	19.70	5.429	.423** [.209, .597]	.241* [.073, .500]	.227* [.060, .463]				
5. Tobacco score ( <i>r</i> )	4.57	1.358	050 [259, .161]	133 [355, .069]	094 [343, .044]	153 [345, .056]			
6. AUDIT score ( <i>r</i> )	7.49	5.900	.177 [032, .383]	.067 [153, .320]	.215* [037, .484]	.048 [146, .221]	132 [361, .073]		
7. DUDIT score ( <i>r</i> )	5.70	6.571	.114 [128, .333]	.181 [001, .374]	.331** [.156, .504]	.148 [080, .351]	443** [613, - .234]	.416** [.210, .618]	

Table 5.3. Gender Role Conflict Scale and substance use correlations. Means, standard deviations, and correlations with confidence intervals

*Note. M* and *SD* represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation. \* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

**Gender Minority Stress and Resilience.** For reasons of available space Table 5.4 shows correlations between the nine sub-scales of the GMSR and substance use (tobacco, alcohol, drugs) and Table 5.5 shows respective confidence intervals. The sub-scales D, NA, IT, NE, ND, and CC were found to have no significant associations with the use of any substance. The sub-scale V was found to have a weak negative relationship with tobacco use, r(103) = -.215, p = .026, 95% CI [-.433, -.069]. Results show that this sub-scale also had a moderate positive relationship with the use of drugs, r(103) = .364, p < .001, 95% CI [.131, .588]. The sub-scale R had a weak positive relationship with the use of drugs, r(103) = .194, p = .047, 95% CI [-.031, .404]. The sub-scale P was found to have a weak positive relationship with drug use (r(103) = .253, p = .018, 95% CI [-.287, .120].

Variable	М	SD	1	2	3	4	5	6	7	8	9	10	11	12
1. Discrimination ( <i>r</i> )	4.20	2.128												
2. Rejection ( <i>r</i> )	4.64	2.624	.631**											
3. Victimisation ( <i>r</i> )	3.96	3.060	.481**	.597**										
4. Non-affirmation ( <i>r</i> )	12.75	6.906	.006	.184	.189									
5. Internalised Transphobia (r)	15.87	8.072	.096	.196*	.261**	.322**								
6. Pride ( <i>r</i> )	12.63	6.405	.128	.201*	.080	.356**	165							
7. Negative Expectations ( <i>r</i> )	18.36	8.033	.140	.231*	.260**	.534**	.309**	.210*						
8. Non-disclosure ( <i>r</i> )	13.11	4.772	.033	003	005	003	.119	410**	.230*					
9. Community Connectedness (r)	11.49	4.788	090	154	038	.041	277**	.191	187	088				
10. Tobacco score ( <i>r</i> )	4.57	1.358	113	039	215*	.040	.073	028	016	001	015			
11. AUDIT score ( <i>r</i> )	7.49	5.900	.102	.072	.135	014	.005	.128	.033	177	.005	132		
12. DUDIT score ( <i>r</i> )	5.70	6.571	.187	.194*	.364**	.006	056	.253**	.107	162	048	443**	.416**	

Table 5.4. Gender Minority Stress and Resilience measure and substance use correlations including means and standard deviations

*Note. M* and *SD* represent the mean and standard deviation, respectively. See table 7 for confidence intervals for each correlation value. \* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

Variable	М	SD	1	2	3	4	5	6	7	8	9	10	11	12
1. Discrimination $(r)$	4.20	2.128												
			[.507.											
2. Rejection $(r)$	4.64	2.624	.7481											
Jan ()			[.289,	[.469,										
3. Victimisation ( <i>r</i> )	3.96	3.060	.672]	.711]										
ζ,			[219,	[073,	[.014,									
4. Non-affirmation ( <i>r</i> )	12.75	6.906	.262]	.385]	.361]									
			[120,	[020,	[.069,	[.154,								
5. Internalised Transphobia (r)	15.87	8.072	.299]	.376]	.435]	.513]								
			[119,	[.110,	[145,	[.156,	[395,							
6. Pride ( <i>r</i> )	12.63	6.405	.381]	032]	.331]	.521]	.019]							
			[104,	[.002,	[.054,	[.367,	[.070,	[035,						
7. Negative Expectations (r)	18.36	8.033	.377]	.402]	.433]	.659]	.442]	.415]						
			[195,	[190,	226,	[222,	[122,	[585,	[.015,					
8. Non-disclosure $(r)$	13.11	4.772	.236]	.195]	.225]	.213]	.325]	220]	.450]					
			[324,	[417,	[221,	166,	[467,	[013,	[376,	[325,				
9. Community Connectedness (r)	11.49	4.788	.117]	.083]	.172]	.332]	057]	.418]	.049]	.153]				
			[315,	[265,	[433,	[226,	[205,	[245,	[291,	[236,	[202,			
10. Tobacco score $(r)$	4.57	1.358	.097]	.156]	069]	.228]	.247]	.146]	.157]	.242]	.226]			
			[067,	[118,	[072,	[207,	[230,	[096,	[199,	[378,	[173,	[361,		
11. AUDIT score $(r)$	7.49	5.900	.281]	.249]	.339]	.149]	.233]	.364]	.248]	.040]	.184]	.073]		
			[041,	[031,	[.131,	[198,	[287,	[287,	[104,	[369,	[264,	[613,	[.210,	
12. DUDIT score $(r)$	5.70	6.571	.403]	.404]	.588]	.190]	.120]	.120]	.291]	.071]	.175]	234]	.618]	

Table 5.5. Gender Minority Stress and Resilience measure and substance use correlation confidence intervals, with means and standard deviations

*Note. M* and *SD* represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation (see table 6 for correlation values.

**Psychopathological Symptoms.** For reasons of available space Table 5.6 shows correlations between all 12 dimensions of the SCL-90-R and substance use (tobacco, alcohol, drugs) and Table 5.7 shows respective confidence intervals. Typically, GSI (the average score of all 90 items of the measure) is thought to be the best indicator for the current level of the symptoms (Derogatis, 1994). Results from this sample demonstrate that GSI was moderately positively correlated with the use of drugs, r(103) = .464, p < .001, 95% CI [.205, .600]. PST (the number of symptoms that are reported) was found to have a moderate positive correlation with both alcohol use (r(103) = .273, p = .021, 95% CI [-.020, .406]) and a moderate positive correlation with drug use (r(103) = .502, p < .001, 95% CI [.340, .647]). Additionally, PSDI had a weak positive correlation to alcohol use, r(103) = .261, p < .001, 95% CI [.340, .647].

The use of tobacco had no correlations to any symptom dimensions and the use of alcohol had a weak positive correlation to three symptom dimensions: HOS (r(103) = .240, p = .028, 95% CI [.005, .427]), PAR (r(103) = .251, p = .010, 95% CI [-.035, .378]), and PSY (r(103) = .326, p < .001, 95% CI [.067, .408]).

The use of drugs was significantly correlated with all nine symptom dimensions. A moderate positive correlation was found between the use of drugs and DEP (r(103) = .311, p < .001, 95% CI [.106, .481]), INS (r(103) = .368, p < .001, 95% CI [.106, .491]), OBC (r(103) = .393, p < .001), 95% CI [.114, 513]). Additionally, a moderate positive relationship was demonstrated between the use of drugs and PAR (r(103) = .407, p = <.001, 95% CI [.123, .520]), ANX (r(103) = .444, p < .001, 95% CI [.164, .602]), SOM (r(103) = .440, p < .001, 95% CI [.195, .559]), PHO (r(103) = .452, p < .001, 95% CI [.181, .697]), HOS (r(103) = .442, p < .001, 95% CI [.270, .585]), and PSY (r(103) = .586, p < .001, 95% CI [.357, .693]).

Variable	М	SD	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
1. SOM ( <i>r</i> )	1.2659	.82383															
2. OBC ( <i>r</i> )	1.8371	.91707	.716**														
3. INS ( <i>r</i> )	1.6506	.91761	.712**	.799**													
4. DEP ( <i>r</i> )	1.7778	.88898	.679**	.842**	.793**												
5. ANX ( <i>r</i> )	1.4505	.93327	.777**	.834**	.816**	.863**											
6. HOS ( <i>r</i> )	1.2184	.98878	.712*	.614**	.729**	.637**	.752**										
7. PHO ( <i>r</i> )	1.2184	1.03379	.757**	.717**	.764**	.712**	.845**	.780**									
8. PAR ( <i>r</i> )	1.1843	.86460	.727**	.639**	.770**	.581**	.711**	.761**	.676**								
9. PSY ( <i>r</i> )	1.1733	.85229	.816**	.723**	.763**	.701**	.854**	.801**	.820**	.789**							
10. GSI ( <i>r</i> )	1.4540	.80762	.869**	.877**	.897**	.885**	.949**	.834**	.882**	.829**	.912**						
11. PST ( <i>r</i> )	59.10	25.239	.868**	.801**	.806**	.752**	.840**	.779**	.831**	.772**	.887**	.923**					
12. PSDI ( <i>r</i> )	2.0736	.54337	.569**	.760**	.766**	.837**	.785**	.638**	.617**	593**	.623**	.790**	.045				
13. Tobacco score $(r)$	4.57	1.358	058	135	086	160	175	082	123	064	165	142	171	152			
14. AUDIT score $(r)$	7.49	5.900	.225*	.038	.105	042	.121	.240*	.152	.251*	.326**	.149	.273**	011	132		
15. DUDIT score $(r)$	5.70	6.571	.440**	.393**	.368**	.311**	.444**	.442**	.452**	.407**	.586**	.464**	.502**	.261**	443**	.416**	

Table 5.6. Symptom Checklist - 90 - Revised and substance use correlations with means and standard deviations.

Note. M and SD represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation.

\* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

Variable	М	SD	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
1. SOM ( <i>r</i> )	1.2659	.82383															
2. OBC ( <i>r</i> )	1.8371	.91707	[.495, .797]														
3. INS ( <i>r</i> )	1.6506	.91761	[.524, .847]	[.626, .832]													
4. DEP ( <i>r</i> )	1.7778	.88898	[.510, .729]	[.775, .886]	[.671, .837]												
5. ANX ( <i>r</i> )	1.4505	.93327	[.598, .835]	[.699, .960]	[.699, .860]	[.760, .933]											
6. HOS ( <i>r</i> )	1.2184	.98878	[.517, .849]	[.403, .772]	[.563, .812]	[.478, .715]	[.620, .848]										
7. PHO ( <i>r</i> )	1.2184	1.03379	[.528, .819]	[.533, .874]	[.612, .822]	[.584, .784]	[.781, .910]	[.575, .840]									
8. PAR ( <i>r</i> )	1.1843	.86460	[.573, .796]	[.466, .733]	[.621, .815]	[.458, .710]	[.579, .810]	[.595, .815]	[.529, .771]								
9. PSY ( <i>r</i> )	1.1733	.85229	[.667, .859]	[.599, .779]	[.618, .814]	[.618, .814]	[.771, .907]	[.658, .854]	[.696, .924]	[.715, .859]							
10. GSI ( <i>r</i> )	1.4540	.80762	[.743, .958]	[.789, .907]	[.814, .959]	[.829, .919]	[.917, .961]	[.708, .876]	[.809, .908]	[.722, .876]	[.849, .950]						
11. PST ( <i>r</i> )	59.10	25.239	[.767, .950]	[.653, .822]	[.711, .895]	[.649, .792]	[.775, .895]	[.702, .872]	[.749, .871]	[.699, .847]	[.849 .939]	[.899, .954]					
12. PSDI ( <i>r</i> )	2.0736	.54337	[.432, .738]	[.474, .839]	[.133, .564]	[.571, .913]	[.517, .906]	[408, .744]	[009, .469]	[010, .418]	[.392, .770]	[.413, .840]	[151, .245]				
13. Tobacco score (r)	4.57	1.358	[280, .105]	[405, 005]	[360, .037]	[411, .038]	[394, .016]	326, .092]	[356, .053]	[296, .073]	[392, 069]	[387, 002]	[340, .003]	[317, .113]			
14. AUDIT score ( <i>r</i> )	7.49	5.900	[031, .395]	[156, .165]	[097, .216]	[298, .110]	[146, .220]	[.005, .427]	[098, .265]	[035, .378]	[.067, .408]	[131, .284]	[020, .406]	[377, 012]	[361, .073]		
15. DUDIT score ( <i>r</i> )	5.70	6.571	[.195, .559]	[.114, .513]	[.106, .491]	[.106, .481]	[.164, .602]	[.270, .585]	[.181, .697]	[.123, .520]	[.357, .693]	[.205, .600]	[.340, .647]	[.128, .381]	[613, 234]	[.210, .618]	

Table 5.7. Symptom Checklist - 90 - Revised and substance use correlation confidence intervals with means and standard deviations.

Note. M and SD represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation (see table 6 for correlation values.

**Personality.** Table 5.8 shows the correlations between the seven dimensions of the TCI-140-R and substance use (tobacco, alcohol, drugs). Tobacco use was only correlated with one dimension; results demonstrate a moderate negative relationship between tobacco use and NS (r(103) = -.382, p < .001, 95%CI [-.576, -.228]).

The personality profiles of alcohol users and drug users in this sample had very similar results. There was a moderate positive relationship between both NS and alcohol use (r(103) = .244, p = .012, 95% CI [.076, .447]) and NS and drug use (r(103) = .360, p < .001, 95% CI [.187, .544]). There was also a moderate positive relationship between both ST and alcohol use (r(103) = .374, p = .002, 95% CI [.184, .496]) and ST and drug use (r(103) = .293, p = .004, 95% CI [.049, .459]). There was a moderate negative relationship between both HA and alcohol use (r(103) = -.333, p < .001, 95% CI [-.519, -.198]) and HA and drug use (r(103) = -.257, p = .009, 95% CI [-.418, -.028]). Additionally, there was a moderate negative relationship between both CO and alcohol use (r(103) = -.386, p < .001, 95% CI [-.526, -.195]) and CO and drug use (r(103) = -.293, p = .003, 95% CI [-.500, -.036]).

Table 5.8. Temperament and Character Inventory – 14	- Revised and substance use correlations. Means, standard deviations, and correlations with confidence intervals
· · · · · · · · · · · · · · · · · · ·	

I I I I I I I I I I I I I I I I I I I		,					·					
Variable	М	SD	1	2	3	4	5	6	7	8	9	10
1. Novelty Seeking ( <i>r</i> )	59.91	9.395										
2. Harm Avoidance ( <i>r</i> )	70.56	14.373	278** [496, - .098]									
3. Reward Dependence (r)	64.04	10.225	.230* [.008, .439]	.140 [151, .303]								
4. Persistence ( <i>r</i> )	61.34	12.073	.036 [163, .334]	152 [436, .028]	.091 [182, .282]							
5. Self-directedness ( <i>r</i> )	57.79	11.341	233* [448, .040]	236* [437, 051]	.132 [079, .430]	.362** [.094, .561]						
6. Cooperativeness (r)	70.37	11.673	167 [357, .102]	.443** [.251, .634]	.454** [.319, .677]	.115 [120, .300]	.303** [.051, .491]					
7. Self-transcendence ( <i>r</i> )	44.07	9.829	.320** [.127, .453]	263* [511, 056]	.104 [095, .323]	.184 [012, .429]	251* [466, - .025]	225* [395, .020]				
8. Tobacco score ( <i>r</i> )	4.57	1.358	382** [576, 228]	046 [240, .201]	046 [290, .121]	122 [364, .069]	.073 [113, .262]	069 [210, .210]	114 [339, .106]			
9. AUDIT score ( <i>r</i> )	7.49	5.900	.244* [.076, .447]	333** [519, 198]	157 [313, .035]	.090 [061, .300]	070 [263, .115]	386** [526, 195]	.374** [.184, .496]	132 [361, .073]		
10. DUDIT score (r)	5.70	6.571	.360** [.187, .544]	257* [418, - .028]	099 [291, .163]	057 [238, .183]	170 [356, .036]	293* [500, 036]	.293** [.049, .459]	443** [613, - .234]	.416** [.210, .618]	

*Note. M* and *SD* represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation. \* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

## **5.2. Inferential Statistics**

Separate multiple regression analyses using the forward method were carried out for the outcome variables of alcohol and drug use to investigate whether the predictor variables which most highly correlated with those outcomes could significantly predict participant's AUDIT and DUDIT scores, respectively. A binary logistic regression was conducted between the variables which most highly correlated with smoking to investigate if they could predict whether an individual was a current smoker or non-smoker. For a predictor variable to be included in the regression analyses the inclusion criteria was a correlation coefficient of  $r \ge .35$  and a p-value of  $p \le .01$ 

## 5.2.1. Binary Logistic Regression – Smoking Status

Based on the inclusion criteria stated above, a single predictor variable was included which was Novelty Seeking (NS). Please refer to Table 5.8 for the correlations and p-values. A binary logistic analysis was conducted to explore if NS could significantly predict whether an individual was a current smoker or non-smoker.

105 cases were included in analysis, with three cases excluded for missing NS data. In total 102 cases were analysed, and the full model significantly predicted smoking status (omnibus  $\chi^2 = 13.07$ , df = 1, p < .001). The model accounted for between 12% and 23.3% of variance in smoking status, with 100% of current smoking status successfully predicted and 25% of current non-smoking status successfully predicted. Please see Table 5.9 for coefficients, Wald statistics, probability values and standardised beta (with 95% confidence intervals). Overall, 91.2% of predictions were accurate. NS significantly contributed to predicting smoking status (p = .002); for each point increase in NS, participants were 1.142 times less likely to be a current smoker.

Table 5.9. Binary Logistic Regression predicting current smoking status (smoker/non-smoker).

Variable	df	Sig.	В	SE	Exp(β) [95%CI]
Constant	1	.000	10.481	2.799	.000
Novelty Seeking (NS)	1	.002**	133	.042	1.142 [1.052, 1.240]

*Note.* Omnibus  $\chi^2 = 13.074$ , df = 1, p <.001. R<sup>2</sup> = .233 (Nagelkerke). R<sup>2</sup> = .120 (Cox & Snell). \*Significant at the .01 level.
### 5.2.2. Multiple Regression – Alcohol Use

Based on the inclusion criteria previously stated, five predictor variables were included, these were Novelty Seeking (NS), Harm Avoidance (HA), Cooperativeness (CO), Self-Transcendence (ST), and Global Severity Index (GSI). Please see Appendix I for the correlations and p-values. A multiple regression analysis using the forward method was conducted to explore if the five predictor variables could significantly predict alcohol use.

105 cases were included in analysis, with 13 cases excluded for missing data in any of the five predictor variables. In total 92 cases were analysed, and the results indicated that the model which was the most significant predictor of AUDIT scores was model 2 (see Table I.1), which included the predictor variables CO (p = .002) and NS (p = .041). This model could account for 16.3% of variance and was a significant predictor of alcohol use status, F(2, 91) = 8.669, p < .001, R<sup>2</sup> = .163 (see Table 5.10). The final most successful predictive model was:

AUDIT score = 
$$11.08 + (-.155 \times CO) + (.121 \times NS)$$

AUDIT score decreased by .155 for each point increase in CO and increased by .121 for each point increase in NS.

Variable	В	SE B	β	t	р
(Constant)	11.082	5.292		2.094	.039*
CO	155	.048	316	-3.211	.002**
NS	.121	.058	.204	2.077	.041*

**Table 5.10.** Multiple regression of AUDIT using forward method, Model 2: CO andNS

*Note:* B = unstandardized beta, SEB = standard error for unstandardized beta,

 $\beta$  = standardized beta.

\* Significant at the 0.05 level. \*\* Significant at the 0.01 level.

## 5.2.3. Multiple Regression – Drug Use

Based on the inclusion criteria previously stated, four predictor variables were included, these were Novelty Seeking (NS), Global Severity Index (GSI), Positive Symptom Total (PST), and Gender-related Victimisation (V). Please see Appendix J for the correlations and p-values. A multiple regression analysis using the forward method was conducted to explore if the four predictor variables could significantly predict drug use.

105 cases were included in analysis, with 8 cases excluded for missing data in any of the three predictor variables. In total 97 cases were analysed, and the results indicated that the model which was the most significant predictor of DUDIT scores was model 2 (see Table J.1), which included the predictor variables NS (p = .001) and GSI (p < .001). This model could account for 35% of variance and was a significant predictor of drug use status, F(2, 96) =25.259, p < .001, R<sup>2</sup> = .35 (see Table 5.11). The final predictive model was:

DUDIT score = 
$$-11.38 + (3.233 \text{*GSI}) + (.206 \text{*NS})$$

DUDIT score increased by 3.233 for each point increase in GSI and increased by .206 for each point increase in NS.

**Table 5.11**. Multiple regression of DUDIT using forward method, Model 2: GSI and NS

Variable	В	SE B	β	t	р
(Constant)	-11.382	3.573		-3.185	.002*
GSI	3.233	.710	.400	4.554	.000**
NS	.206	.059	.305	3.466	.001**

*Note*: B = unstandardized beta, SEB = standard error for unstandardized beta,  $\beta$  = standardized beta.

\* Significant at the 0.05 level. \*\* Significant at the 0.01 level.

### **5.3. TESUP-1 IPA Results**

Participants in the qualitative TEP1 study consisted of six of the lowest-scoring and seven of the highest-scoring on measures of alcohol and drug use from the quantitative TEP1 study. The following results were identified across both the low- and high-risk substance use groups. There was a degree of shared experiences across the two groups as well as differences. Overall, six themes (See Table 5.12) were identified across both groups, with the exception of theme 5 which was only relevant to those in the high-risk group. Each theme includes up to three sub-themes. The themes are: (1) Conceptualising the Transgender Self; (2) Navigating Transition-related Healthcare; (3) Social Contexts for Substance Use; (4) Distress (In)tolerance; (5) Impulsivity and Novelty Seeking; and (6) Protective and Prohibitive Factors in Substance Use. Participant extracts from each group are provided to illustrate each theme.

Sub-Themes			
1.1. Journey to the Authentic Self			
1.2. Loss of the Self			
2.1. Lack of Knowledgeable and Inclusive			
Healthcare Services			
2.2. Navigating a Complex Healthcare System			
3.1. LGBTQ Community Connections			
3.2. Substance Use to Facilitate Socialising			
3.3. Impact of Masculinities and Social			
Expectations			
4.1. Coping with Frustration and Uncertainty			
4.2. Unhealthy Strategies to Regulate Negative			
Emotions			
4.3. The Symbiotic Relationship of Mental Health			
and Substance Use			
5.1. Pursuit of Novel Stimulation			
5.2. Impulsivity: Risks and Consequences			
6.1. Personal and Family History of Substance Use			
Issues			
6.2. Attitudes and Behaviours of Peers			
6.3. Harm Avoidance			

 Table 5.12. TESUP-1 IPA results: themes and related sub-themes

# 5.3.1. Conceptualising the Transgender Self

**Journey to the Authentic Self.** The majority of participants across both the low-risk (L) and high-risk (H) substance use groups discussed the process of discovering and identifying with a transgender identity as one fraught with difficulties. Participants emphasised the complexities of navigating a trans identity but ultimately all participants except for one reported an improvement in their mental health and QoL after beginning to transition.

For the participants in this study the initial experience of recognizing their own transgender identity was characterised by difficulty interpreting one's own experiences and a lack of language to communicate their feelings to other people or themselves. This type of difficulty could be described as hermeneutical injustice (Fricker, 2006), which occurs when people are unable to understand or articulate their experiences due to social inequality and historic exclusion from contributing to the collective understandings that people use to make sense of their experiences. In this study, hermeneutic injustice often arose as a result of conservative socio-cultural norms and being raised within a Christian religion which has historically excluded trans people (see Box 5.1, extract 1.1). Exposure to positive depictions of transgender people was frequently a key turning point in the lives of the participants and remain as important life events which they can vividly recall (1.2).

The process of transitioning (both medically and socially) was most often reported as being a positive and validating experience. For Parker (H; 1.3), even negative experiences of receiving homophobic slurs were interpreted in a positive light because it indicated that they had not been perceived as female. However, despite the process of transitioning generally increasing the quality of life of the participants, there existed multiple barriers and difficulties in becoming their authentic selves. To cope with these difficulties, multiple participants from both L and H groups reported using alcohol or drugs. Jeremy (L) was experiencing unemployment and struggling with complications from phalloplasty surgery, which left him concerned about his ability to manage his alcohol use (1.4). On the other hand, some participants reported that their motivation for drinking alcohol had changed since they began to medically transition, and their relationship to alcohol and themselves had improved (1.5)

**Box 5.1.** Representative extracts for "Conceptualising the Transgender Self". Quotes are from both low substance use group (L) and high substance use group (H).

- **1.1.** "I grew up in Austria which is a fairly Christian and conservative country and so being trans wasn't really a thing there [...] I did an exchange program to England when I was 16 and I met a trans man ... it was a very lightbulb-y moment of oh god that's a thing they can do here ... and so I came out to some of my friends but not my family ... I was very scared" (**Owen, H**)
- **1.2.** "I never actively identified as female ever in my life but looking at my body I could sort of go 'oh I can see why people describe me in that way' but I didn't have the knowledge or any sense of anything else ... I suppose I never behaved or identified with the way everyone around me expected me to be ... the big point for me that I recall was they had the documentary ... on Channel 4 in 2011 ... it was My Transsexual Summer ... and that was the first time I really saw proper transmasculine experiences ... and it was accessible ... and I saw those experiences and went 'OH! Oh **that** describes how I feel' ... and that really opened the door for me to have the language and knowledge to look at things on the internet and say 'this really fits with how I feel about things' (**Jeremy, L**)
- **1.3.** "I had someone yell 'faggot' at me in the street once and I was ... obviously it's not nice but at the same time I was like ... it was quite validating in a way because they read me as a guy [laughs]" (**Parker, H**)
- **1.4.** "I don't know when I'll be employed again and I'm struggling ... coupled with that surgical issue it's gonna be hard in the future [...] I think these two factors are making me question my alcohol use ... and that is probably my biggest fear in terms of slipping ... slipping back into problematic behaviours" (Jeremy, L)
- **1.5.** "a year or two ago I was spending five days out of the week drinking ... this was before starting T and I was in a difficult place and I just didn't want to be sober [...] after I started T it's like I'm a lot more confident and happier and now I go [drinking] to have a good time ... it's completely reversed reasons" (**Logan**, **H**)
- **1.6.** "I sometimes see glimpses of how I could have been and I find that frustrating when I think about how my life could have been different ... it makes me depressed a lot really ... thinking of what I've lost before I've had the chance to do anything ... all the things that have been taken away from me and the opportunities I haven't had ... like if I wasn't trans I wouldn't be depressed ... it makes me quite bitter I suppose" (**Darren, L**)
- 1.7. "a lot of the time I get angry with myself ... like why do I feel this way and why am I like this and how can I stop being like this ... I get angry and wish I could just be a girl because it would be easier" (**Darren, L**)

Loss of the Self. Almost all of the participants described how their transition had helped them feel "at peace with myself" (Logan, H) and "happier and more content" (Aaron, L). However, one participant did not feel this way: Darren (L) frequently reported feeling "quite angry" and "isolated". He expressed intense bodily dysphoria ("there's something massively wrong with my body ... I look disgusting and wrong") and conceptualised being transgender as a medical condition with no cure ("I'm stuck with this condition and no one understands ... and there's no treatment for it"). Where other participants described transitioning as a journey to their authentic self, Darren described it as a loss of the self, the self that could have been (1.6), and was the only participant who vocalised his unhappiness with identifying as male (1.7). While no participants other than Darren felt similarly, Darren's negative experiences with his transgender identity were so frequently discussed throughout his interview and had such a substantial impact on the entirety of his life, the author felt it important that his views were centred within a single sub-theme.

## 5.3.2. Navigating Transition-Related Healthcare

Lack of Knowledgeable and Inclusive Healthcare Services. Many of the participants who were seeking transition-related medical interventions discussed the value of trans-inclusive healthcare. The few participants who received appropriate care from their GP typically described themselves as "very lucky" (Parker, H; Box 5.2; 2.1), or emphasised their appreciation for GPs who tried to provide appropriate healthcare, despite lacking any experience in it previously (2.2).

A more often reported experience was struggling to receive appropriate healthcare from primary care to gender identity services (GIS). One participant (Owen, H) was assumed to be struggling with depression and anxiety, due to his request for a GIS referral (2.3). Another participant (Bailey, L) experienced a lot of distress in their attempts to access a

referral to GIS (2.4). In contrast to Parker's inexperienced GP who tried to increase their

knowledge and provide the correct care, Bailey perceived their GP as having "no idea what

she's doing" and preventing Bailey from accessing timely and low-cost healthcare.

**Box 5.2.** Representative extracts for "Navigating Transition-related Healthcare". Quotes are from both low substance use group (L) and high substance use group (H).

- 2.1 "I am very lucky that my GP is amazing and she does shared care ... she's been really good even though GenderGP haven't sent much information she's been trying to fight for me ... we've actually not moved [house] so I can stay with this GP ... I've never had a GP this good before and it makes a huge difference" (**Parker, H**)
- 2.2 "they [GP] have openly said that at the time they didn't have any trans patients and were sort of quite new and naïve to knowing how to deal with it and if anything ... they took my journey ... and it was a learning experience for them" (Jeremy, L)
- 2.3 "I went to my GP and said I want a referral to the gender clinic [...] and she just gave me a test for depression and anxiety ... I mean I did get the diagnosis but it was weird to immediately assume 'oh you're trans you must also be depressed" (Owen, H)
- 2.4 "my GP has never had a trans patients and so ... it's really annoying because she wants to help but has no idea what she's doing ... and the NICE guidelines are incredibly vague ... so she doesn't know what she can or cannot do and ends up basically doing nothing ... she referred me to the GIC but everything else has hit a road block of pedantic nonsense" (**Bailey, L**)
- **2.5** "As a non-UK national the hardest thing for me ... and it's still ongoing ... it's that I can't change my documents ... and I do think it affects my drinking habits ... if I'm really stressed out like if there's a visa deadline or something like that then I find it just makes me want to have a drink and some cigarettes (**Aaron, L**)
- 2.6 "Interviewer: have you been referred to a GIC? Cody: Somewhat ... kind of ... I think so Interviewer: so you're assuming you're on the waiting list? Cody: yeah I mean ... I can't afford privately because I'm a student and get jack shit ... I had a private consultation a while ago and I didn't have the money for it at the time ... so I guess at the time it was like my way of solving that was just to have a drink" (Cody, L)
- 2.7 "A lot of it comes down to what the manager of the GP surgery says ... and they don't wanna spend money unnecessarily ... so basically they dictate who gets treatment and who doesn't" (**Bailey, L**)
- **2.8** "It's frustrating because you can't get on with what you **know** that you need ... you **know** you need this and they hold you back so it takes a long time ... sometimes I think they do it on purpose ... hold you back" (**Simon, H**)
- 2.9 "I feel sort of ... [sigh] not ungrateful ... I don't know the word ... sort of like ... I've let them down ... like they gave me testosterone and I've let them down because I'm taking a horrible drug *Interviewer:* why do you feel like you've let them down? *Simon:* because they trusted me to look after my own body and they gave me this testosterone to feel better and it's like I'm ... I'm **purposefully** putting blocks in the way ... not really purposefully but subconsciously ... like trying to make problems for myself ... like there's no problem because now I'm on testosterone so maybe it's like I need a problem in my life to dwell on ... like subconsciously ... I definitely feel like I've let them down ... and especially my mum as well ... that's a massive thing" (Simon, H)

**Navigating a Complex Healthcare System.** Where participants had managed to secure a referral to GIS, this process was far from simple. Some participants described how they were "put through so many hoops" before a referral was successful (Ezra, L). For participants who were non-UK nationals there was additional difficulties regarding navigating transition; changing names and gender markers on legal documents were referenced as additional challenges that might trigger the urge to drink alcohol and smoke cigarettes (2.5).

The waiting time between referral and first appointment was mentioned by all of the participants, they described this waiting time as "**extremely** just ridiculously long" (Logan, H) and like being "stuck in a nightmare" (Parker, H). The length of waiting time, the uncertainty of whether their referral would be accepted or when their appointment might be, and the financial stress of attempts to access private healthcare were often depicted as experiences which triggered increased frequency and quantity of substance use (2.6). It is interesting to note that four of the six L participants had accessed private transition-related healthcare, whereas only one of the seven H participants had been able to access private healthcare. Additionally, it was more often the H group who used emotional language to describe their difficulties accessing healthcare, describing waiting as a "nightmare" (Parker, H), and depicted delayed appointments as "a kick in the face". The L group had similar experiences but were more reserved in their description of events ("I suppose it was upsetting" Jeremy, L) and tended to minimise the emotional impact ("my appointment was moved back which was admittedly disappointing" Ezra, L).

Throughout interviews with all of the participants, there was a sense that GIS could not be trusted and there was a lack of understanding of GIS motivations behind decisionmaking processes regarding HRT and referrals for surgery. For example, Bailey (L) suspected that they had been denied shared care simply so that their GP surgery could save money (2.7), and Simon (H) wondered if GIS were purposefully preventing him from accessing testosterone (2.8). Simon's conflicted feelings about his relationship with GIS was particularly apparent when discussing his use of cocaine and his feelings of shame because he thought that he had "let them down" when they had "trusted [him] to look after [his] own body" (2.9).

# 5.3.3. Social Contexts for Substance Use

LGBTQ Community Connections. Many of the participants discussed their relationships with other LGBTQ people and the LGBTQ community as a whole. The L group tended to feel disconnected from other trans people and sometimes felt as though they did not really belong to the LGBTQ community. On the other hand, the H group often felt strong connections with other LGBTQ people and would attend trans support groups and LGBTQspecific nightclubs and bars. Many individuals in the H group defined themselves as activists for trans rights and their social circles largely consisted of other LGBTQ people.

Some participants discussed how socialising with other trans people when they were younger or when they had just come out as trans was a strong source of validation and support (Box 5.3; 3.1). Many participants in the H group described how a lot of their social experiences with other LGBTQ people took place in the context of substance use. Logan (H) felt as though the community that he had found in "queer bars" helped him gain confidence and he felt a sense of safety in these contexts, compared to non-queer bars (3.2). However, the benefits of seeking community in bars and clubs was not felt by all participants. For example, Parker (H) described how many events for LGBTQ people centred around alcohol, and they felt that this limited their opportunities for socialising when they did not want to drink (3.3). It is notable that participants who had positive experiences of socialising with other trans people had all done so 'in real life' as opposed to participating online. Other participants who had tried to connect with other trans people but had done so online had strikingly negative experiences, and many of them who had looked for support in online trans support groups described how they found it "difficult to get along" with other trans people (3.4; Bailey, L), and felt like they had "nothing in common with them" (3.5; Darren, L).

Most of the L group had less LGBTQ community connections in comparison to the H group, however they tended to report that they did not have much interest in pursuing this, and that they had other friendships that were based on shared hobbies or having grown up together as children. The exception to this was Darren (L) who consistently described being socially isolated and claimed that he didn't "have any friends or anyone to talk to". He felt disconnected from the trans community because he perceived being trans as having been "born with a condition" and he reported that this led to him getting into arguments online with other trans people. These arguments were typically based on a division within the trans community over the necessity of dysphoria to identify as trans, with Darren stating, "you **do** need dysphoria to be trans … it's quite literally part of the diagnosis". This had led to him feeling rejected from the community, increasing his feelings of shame and self-hatred, and led to an increase in drinking alcohol (3.6).

The idea that engaging with other trans people is related to substance use was also explored by another participant, Zack (H). However, where Darren (L) conceptualised disconnection from the trans community as contributing to his substance use, Zack (H) described how his connections with other trans people increased his substance use. He felt that in his friendship group, being trans was connected to heavy substance use, and they would encourage these behaviours in each other (3.7). He directly related identifying as trans, and being around other trans people, as a precipitant to increased substance use ("when you're in a group and you're all queer ... well you're not exactly sitting down for bible study"). In his opinion having an identity which made him "a social outcast" made it easier to participate in other behaviours which were not socially acceptable (3.8). It is interesting that both Darren (L) and Zack (H) related increased substance use with feelings of rejection from their peers. However, for Darren it was rejection from the "mainstream" trans community that influenced his substance use, whereas for Zack it was rejection from "mainstream" society as a whole (mainstream for Zack largely meant cis/het society).

**Box 5.3.** Representative extracts for "LGBTQ Community Connections" in "Social Contexts for Substance Use". Quotes are from both low substance use group (L) and high substance use group (H).

- **3.1** "I'm from a really small town and I had a couple of gay friends but it wasn't ... the same type of queer ... and coming to university and being able to meet people ... like having a group of trans friends that are a bit older and are like ... big brothers ... that was really nice" (**Cody**, **L**)
- **3.2** "You tend to get that sense of community more in queer bars definitely ... I went to one that had gender neutral toilets and people were dressed up in drag and it was such a ... nice community ... it felt really safe [...] places like that just help me to feel more confident ... help me socialise more and create a sense of community" (Logan, H)
- **3.3** "I think it's a thing where a lot of queer events and spaces centre around alcohol ... which I think is an issue in the queer community for sure ... and so I just try not to go out because sometimes it's hard not to drink" (**Parker, H**)
- **3.4** "I've been on a couple of forums but ... it sounds awful ... but a lot of them like ... they went to all girl's schools and their experiences and their ... attitudes ... just didn't mesh with me ... I just found them quite difficult to get on with [laughs] I dunno if it's a culture thing or what" (**Bailey, L**)
- **3.5** "I've tried joining LGBT and trans groups online ... and I constantly have issues ... I just don't seem to have anything in common [...] I've noticed increasingly in trans groups online that there's lots of things about new genders and ... like ... there's a kind of Tumblr style of trans person who identifies as 30 genders and as a fox or whatever and they use weird pronouns ... it's frustrating that I can't find spaces for trans people who are like me ... like who don't agree with that stuff and who are **actually** trans" (**Darren, L**)
- **3.6** "it feels like you **have** to have certain beliefs otherwise you get badly rejected and it makes everything to do with being trans worse ... I do think it has pushed me to drinking sometimes ... because I end up hating myself because of the attitudes in the community [...] so I'm part of one of these communities of rejected trans people and it's an awful place where everyone is really depressed" (**Darren, L**)
- **3.7** "A lot of the trans friends I've made actively partook in some sort of substance use ... we created the atmosphere where ... not only were we the queer group in college but we were also the smokers and the people who did drugs and partied and stuff ... we were the alternative kinda punk crowd ... we created an atmosphere where we'd let ourselves do all those things and everyone was trying to win the bet of who can get the most screwed up" (**Zack, H**)
- **3.8** "almost every queer person I know likes drugs ... and I'm not too sure why that is ... maybe if you're already going against the grain it opens up ... like if you're already a social outcast ... it's a lot easier to get into things like that ... if you take this lifestyle where a lot of people are like 'oh that's bad' then I think it's very easy to do other things that people think are bad as well' (**Zack, H**)

**Substance Use to Facilitate Socialising.** Most of the participants in both the L and H groups conceptualised substance use as something which enhances social experiences and increases their confidence. Participants in the L group reported that they did not use substances when alone, and that they mostly drank in social situations to enhance an already good time (Box 5.4; 4.1). This sentiment was mirrored in the H group, but members of the H group also reported that they occasionally would drink alcohol or take drugs on their own, although this was not always depicted as a positive experience (4.2).

Participants in the H group reported more instances of experiencing peer pressure to drink more alcohol or take more drugs than they wanted to. Some participants reported that they would use more substances to stop their friends from mocking them and "taking the piss out of [them]" (4.3). Simon (H; 4.4) felt it was better to give in to peer pressure because "they don't know I'm trans so I can't single myself out for attention like that". For Simon, his need to conceal his trans identity from his friendship group led to his struggles with excessive use of alcohol and cocaine.

In both the L group and H group, participants discussed how getting drunk would encourage themselves and others to talk about topics which they would not have done while sober; some participants would take the opportunity while drunk to come out as trans to their friends (4.5). Some participants described how they had purposefully got drunk so that they could have difficult conversations, particularly if they want to talk about their mental health or experiences of dysphoria (4.6). These participants also all described that when they were sober the next day, they would have feelings of regret about talking about something they usually would not talk about and feeling that it was "embarrassing" (4.7). **Box 5.4.** Representative extracts for "Substance Use to Facilitate Socialising" in "Social Contexts for Substance Use". Quotes are from both low substance use group (L) and high substance use group (H).

- **4.1** "I only drink in social situations ... never on my own ... and in social situations I'm already feeling pretty good ... that fun feeling is more attributed to the people I'm with rather than alcohol ... the alcohol is like ... it's like the cherry on top and the people I'm with is the whole ice cream sundae [laughs]" (**Ezra**, **L**)
- **4.2** "When you're doing [drugs] with your friends you can justify it as having a good time but also like now I'm having an even better time .. but when you're on your own ... that's when you're really like oof [laughs] y'know ... you do feel ... more reflective the morning after" (**Zack**, **H**)
- **4.3** "Some of my friends will take the piss out of me and say 'come on you've not been out out for ages' and by out out they mean taking drugs all night and not coming home ... and it's nice to be appreciated and feel that other people want to encourage me to get high and hang out with them ... but at the same time they will pester me and just take the piss if I say no" (Gale, H)
- **4.4** "If I didn't take [cocaine] then my mates would take the piss or something ... and they don't know that I'm trans so I can't single myself out for attention like that ... it's better to fit in" (**Simon, H**)
- **4.5** "I came out to people at university when I was really drunk at a party ... because it's a relaxing ... a loosening of inhibitions thing ... and most of my comings out have actually happened while tipsy or drunk ... it just makes it easier to talk about" (**Owen**, **H**)
- **4.6** "I do sometimes have a drink to speak more openly with people ... like about mental health ... or I know that among my group of trans friends ... people will tend to talk about their experiences of dysphoria when they're drunk ... I think people are a lot more open to talk about it ... and there's a level of trust among queer friends that you ... you can't replicate that elsewhere" (Cody, L)
- **4.7** "Alcohol does lower my anxiety levels ... to a lower level where I can say more things and have those important discussions that I wouldn't really have while sober [...] but then there's the negative consequences the next day when you've said something embarrassing or something that you end up regretting" (**Owen, H**)

**Impact of Masculinities and Social Expectations.** Most of the participants in the H group discussed how the decision to use substances, or the type of substances used, was often dependent on accessibility of the substance and the social context (Box 5.5; 5.1). Participants would compare their own drug use to others in an attempt to normalise their own use of drugs. For example, Gale (H) rationalised their own use of drugs by comparing themselves to others who take more drugs in less acceptable circumstances (5.2). Drinking with friends was seen as more acceptable than drinking alone. For example, Rowan (H; 5.3) would purposefully avoid drinking alone to prevent "addiction" but would take advantage of being with his friends to engage in heavy drinking. He also conceptualised non-drinkers as

"boring", and so he was actively navigating the boundaries of what he found acceptable and

what society thinks of as acceptable drinking behaviours.

**Box 5.5.** Representative extracts for "Impact of Masculinities and Social Expectations" in "Social Contexts for Substance Use". Quotes are from both low substance use group (L) and high substance use group (H).

- 5.1 "If it's not the right settings I probably wouldn't have coke or ket or whatever ... we do that stuff at parties and if there's no party then there's no point to take drugs" (Zack, H)
- **5.2** "I can take or leave the harder stuff and I'd quite happily go three to six months without being inspired ... I wanna keep it special or you can disappear into a hole where obviously other people might take everything known to man in my opinion too often and they get used to it and end up casually having some ... having some coke after their tea ... that doesn't offer any appeal to me" (Gale, H)
- **5.3** "I go out **a** lot to drink ... I'm not one to drink alone because I have this thing in my head like ... if I drink alone I'll fall into this addictive personality ... this addiction ... and so I consciously did that purposefully so I wouldn't drink alone ... and I was pretty much not drinking unless I go out but going out ... I will go out a lot and I would just drink ... and drink and drink and drink and drink *Interviewer:* Would you feel comfortable going out when others are drinking but not having a drink? *Rowan:* um kinda ... it depends ... I have some preconceptions about people who don't drink and it's like ... they just seem boring or something ... so in my opinion having drinks with your mates is normal and fine and whatever" (**Rowan, H**)
- **5.4** "there's such a different culture with your geeky guys than there is with more ... jock guys ... that sort of lad culture [...] I think that within the geek subculture there's less of a necessity to drink ... I dunno ... I've always found that my role-play buddies drink significantly less than the more laddish buddies I have" (**Bailey, L**)
- **5.5** "*Interviewer:* Do you think that your substance use might actually help in your job [as a youth worker for at-risk kids] *Gale:* Well yeah because I think it means I've not been a geek for my entire life which means I can identify with some of the difficult kids I work with" (**Gale, H**)
- 5.6 "I do not get into lad's culture or any of that ... I'm a fan of cocktails and I like tasty ciders and wine ... I don't have their attitude to drinking and I really don't like binge-drinking *Interviewer:* do you consider yourself to be masculine? *Aaron:* yeah I do ... I mean ... it's one of those 'in the eye of the beholder' things ... but yeah ... I think I do" (Aaron, L)
- **5.7** "I'm quite conscious about what I drink and that I should be having a pint ... it's like I would choose my drink so that people will perceive my gender right ... and sometimes it's like all the guys are having pints so maybe I should have a pint and not a gin and tonic" (**Parker, H**)
- **5.8** "What you drink is inherently masculine or feminine ... I mean it's completely social and means nothing ... but men drink pints and women drink cocktails ... so that stuff affects what you drink because it's that stupid social dysphoria isn't it ... it's stupid but I think if I drink that beer then people won't misgender me" (**Cody**, **L**)
- **5.9** "I think there's a masculine or feminine way of drinking alcohol and I try to do it in the masculine way ... and do that with drugs too ... since transitioning I'm more aware of my gendered and I purposefully do gendered things ... to make sure it's all done in a masculine way ... like with alcohol I act really funny like saying 'lads lads let's get messed up' ... and smoking cigarettes to be that *cool guy who smokes cigarettes* ... and drugs y'know it's fun to be that cool mysterious dude who's done loads of psychedelics and trying to be more masculine definitely drove all that ... just that atmosphere ... of trying to be as cool and as masculine as possible" (**Zack, H**)

A cultural norm related to substance use which was frequently discussed by participants in both groups was something that they described as "lad culture" (Bailey, L; Aaron, L; Jeremy, L; Cody, L; Rowan, H; Zack, H; Parker, H; Logan, H). Participants in the L group often made a distinction between themselves and 'lad culture', which they associated with drinking large quantities of alcohol, particularly beer, in the pub while watching football. This 'lad culture' was typically framed in contrast to the subculture that the participants in the L group described themselves as belonging to: "nerdy" or "geek culture". They reported that this subculture included less of an expectation to drink alcohol and instead they focused their socialising on shared hobbies such as role-playing games (Bailey, 5.4). Although it was typically the L group which made this distinction, one participant in the H group (Gale) implied that because they do drink alcohol and use drugs, they are not a 'geek' (5.5). Many of the participants in the L group made this distinction between "geck" guys and "lad" guys which includes a suggestion of the existence of multiple masculinities. This was made clear by Aaron (L; 5.6) who stated that masculinity is subjective and implied that there are multiple ways to inhabit masculinity.

Although participants from both groups discussed 'lad culture' in the context of substance use, it was participants from the H group who more frequently reported that they might participate in 'lad culture' in order to pursue masculinity and 'pass' as male through gendered substance use (5.7). Participants in the H group often expressed concern that if they did not consume substances in a 'masculine' way then they would not be perceived as male by others, Rowan (H) described this behaviour as "it's all about **how** you drink and **when** you drink and the **things** you drink ... and if you don't do that then you're purposefully leading yourself to get misgendered". Cody (L) was one of the few participants in the L group who discussed how they make efforts to conform to gendered expectations of alcohol use, and they reported that they engage in this due to "social dysphoria" (5.8).

In a similar sentiment to Cody (L), Zack (H) also discussed conforming to gendered expectations of substance use, however where Cody conceptualised this as avoiding dysphoria, Zack framed his behaviours as seeking gender euphoria through masculine use of substances (5.9). His description of motivations for engaging in substance use were framed in the context of pursuing masculinity; "I smoke cigarettes so I can be that *cool guy who smokes cigarettes*", "I feel proper manly when I hold a pint", "I just wanted to be that cool mysterious dude who's done loads of psychedelics ... just trying to be as cool and masculine as possible".

While on the face of it, avoiding gender dysphoria and seeking gender euphoria through substance use may seem to be two sides of the same coin, it is worth noting the differences in the underlying intent and mechanisms through which they were attained. For those participants whose motivation for substance use was avoidance of dysphoria, this can also be framed as attempts to avoid distress, or avoidance of having to psychologically cope with distress. As is seen in Rowan's description of why they engage in dysphoria avoidance behaviours, when this is not done 'successfully', the individual may blame themselves, potentially leading to worse gender dysphoria and self-esteem. Whereas the desire to seek gender euphoria, as told by Zack (5.9), can be framed as a performative act, which not only serves to communicate his gender identity to others, but also is a source of self-esteem and validation (and likely lessens the negative impact of gender dysphoria).

# 5.3.4. Distress (In)Tolerance

**Coping with Frustration and Uncertainty.** For some of the participants in this study, their attempts to cope with uncertain or frustrating situations sometimes led to an increase in substance use in order to regulate those negative emotions. Some participants

reported that conflict in personal (most often, romantic) relationships could trigger the urge to drink alcohol (Box 5.6; 6.1). One participant (Parker, H; 6.2) reported that even when they were teetotal, if they were enduring significant emotional turmoil related to arguing with their partner, then they would usually drink alcohol to deal with their emotions.

Another area of the participants' lives which consistently caused frustration and feelings of uncertainty was their medical transition. The financial burden of getting private healthcare and trying to change one's name and gender on legal documents was cited by some participants as difficult situations which might trigger the urge to use alcohol to excess (see Box 5.2; 2.7 and 2.8). Some participants found the official NHS and government guidance on medical transition to be vague and confusing which caused feelings of frustration and confusion, and participants also reported drinking alcohol to cope with that (6.3). Some participants reported that they used alcohol "as a coping mechanism for dysphoria" (Cody, L), particularly when they were uncertain about when they would receive GIS appointments or transition-related treatment (See Box. 5.1; 1.4).

**Box 5.6.** Representative extracts for "Coping with Frustration and Uncertainty" in "Distress (In)Tolerance". Quotes are from both low substance use group (L) and high substance use group (H).

<sup>6.1 &</sup>quot;If there's more stress and there's more emotional conflict in my relationships that I'm unable to express for whatever reason then I'd be quite tempted to have a drink" (Aaron, L)

<sup>6.2 &</sup>quot;I'd definitely drink more if I'm stressed or if I've had an argument with my partner ... even when I wasn't drinking at all if we had an argument ... especially if she had stormed out ... then a lot of the time then I'd drink ... or I'd really want a drink ... but yeah I do drink more then ... like if I'm really down or frustrated or really stressed" (Parker, H)

**<sup>6.3</sup>** "All the information that you find is so vague ... and name change and all that is an absolute headache ... all you need is a deed poll on fancy paper but so many places say that you need more ... because they don't really know ... and again the guidance is so vague ... it's all a nightmare and I don't want to do it ... I find it incredibly stressful and it all just makes me want to have a drink" (**Bailey, L**)

**Unhealthy Strategies to Regulate Negative Emotions.** One of the reasons for engaging in substance use frequently discussed by participants in both groups was to reduce stress, and to distract themselves from feeling negative emotions. Many of the participants felt they lacked healthy coping mechanisms and did not know how to deal with stressful situations or negative emotions, so they would drink to cope, even though for some it might cause worse outcomes such as suicidal ideation (Box 5.7; 7.1). Some participants would turn to alcohol to deal with stress related to their job (7.2) and their housing situation (7.3). Another participant reflected on a traumatic event which he believed was the catalyst for his difficulties with substance use. As a teenager, Simon (H) witnessed his father take an overdose of heroin and, looking back on it, he identified it as the experience which caused his excessive alcohol use, and more recently, his cocaine use as well (7.4).

One participant in the L group (Bailey) who had previously suffered from chronic pain from migraines, reported how in an effort to reduce their pain they would combine opiate painkillers and alcohol so they could "pass out for a few hours" (7.5). Although Bailey reported that they no longer suffered so severely from migraines, they still crave alcohol if they are experiencing any negative emotions because "if I'm feeling down and really bad then I want a drink ... I suppose I still associate it with feeling better". **Box 5.7.** Representative extracts for "Unhealthy Strategies to Regulate Negative Emotions" in "Distress (In)Tolerance". Quotes are from both low substance use group (L) and high substance use group (H).

- 7.1 "I really haven't got any coping mechanisms for negative emotions ... other than just wanting to drink ... and when I want something I usually let myself have it otherwise I'll just think about it and think about it until I do have it ... so yeah just I drink ... but I have had to learn not to drink too much when I'm sad and alone ... because it leads to situations like when you tried to throw yourself into the river at 3am ... and that's not very good" (**Rowan, H**)
- 7.2 "When I have an upcoming project deadline and there's a lot of stress it makes me want a drink and some cigarettes ... this happened in my old job too ... I would just be so exhausted on all levels that I'd regularly have a drink" (Aaron, L)
- **7.3** "When I was really unhappy in my job and I lived in a really shitty place where there was nothing to do me and my partner would just sit inside and drink wine every night ... it got to the point where we were drinking and sharing a bottle every night and then it progressed to a bottle of wine **each** a night" (**Parker, H**)
- 7.4 "When I was 18 I went through a lot of stress with my dad and I mean massive stress ... and that's what made me first start drinking so much *Interviewer:* what was the stress that you were going through? *Simon:* I was with my dad and he took a heroin overdose ... and I think it massively affected me ... well it must have done ... I don't feel like it's affecting me but ... it's obvious because I did start having alcohol problems just after that ... and now I have a coke habit too" (Simon, H)
- 7.5 "I used to drink a lot ... as a way of dealing with the fact I was in almost constant pain ... I was the only student I knew who had a freakin' liquor cabinet (!) ... and so if I had a headache or a migraine then I would have a drink to help the painkillers ... and that was a habit I got into of taking my very strong opiate painkillers and washing them down with some straight whatever I had ... and then pass out for a few hours ... I know it's not a good thing but when you're in that much pain and you're that upset about it all then you're like 'I'm desperate just give me the booze" (Bailey, L)
- 7.6 "before starting T I was having a really difficult time and I just ... didn't want to be sober ... it started out like partying with friends but then it turned into me going out to drink and get really smashed and try to hook up with someone ... like for validation ... because I'd just come out and I was feeling like I wasn't passing very well ... I just wanted people to find me attractive ... sometimes it worked but not always ... I just wanted that validation and affection from someone else" (Logan, H)
- 7.7 "one of the worst parts [about being trans] is certainly the dysphoria … sometimes it's very debilitating … it's difficult to do anything … it's still very difficult to have a shower and to go outside … and it's not easy for me to sleep with my girlfriend but when I've had a few drinks it's much easier to loosen up … so yeah usually I'll have some drinks before we sleep together because it's harder for me when I'm not drunk or at least tipsy … it does help to lessen the dysphoria" (Owen, H)

Alcohol use to cope with dysphoria was sometimes reported, though only from participants in the H group. One participant, Logan (H) recounted a story about how prior to being on testosterone he would spend most of the week drinking because he felt so much dysphoria that he "didn't want to be sober". During this period, he would go out drinking in clubs alone in order to seek validation of his gender identity by "hook[ing] up with someone" (7.6). Additionally, another participant in the H group, Owen, disclosed that to reduce dysphoria he would typically have some alcohol to be intimate and have sex with his girlfriend (7.7).

The Symbiotic Relationship of Mental Health and Substance Use. Many of the participants described how their mental health and their substance use were often interrelated and interdependent. Participants in the L group, many of whom had previously struggled with substance use but no longer engaged in it, reflected on how they had become reliant on alcohol as a method of avoidance related to mental health and gender dysphoria. Bailey (L) described substance use as "a crutch" to cope with worsening mental health (Box 5.8; 8.1) and similarly, Jeremy (L) conceptualised his previous alcohol use as a maladaptive coping mechanism to deal with depression, anxiety, and dysphoria (8.2). Both Bailey and Jeremy alluded to the relationship between their gender dysphoria and struggles with their mental health; it's notable that both their reliance on alcohol and their symptoms of mental illness reduced as they progressed through their transitions. Another participant, Rowan (H), identified his substance use as stemming from his struggles to cope with ADHD when he was younger and his feeling that he was unable to do anything other than "be the life of the party" (8.3).

Conversely, some participants discussed the negative impact that substance use would have on their mental health. Parker (H) discussed how increased alcohol use would lead to a deterioration in their mental health, which they linked to weight gain that often led to increased bodily dysphoria (8.4). The negative self-image that resulted from poor mental health and weight gain caused a cycle of increased drinking in response to mental health struggles, and worse mental health as a result of increased substance use (8.5). Simon (H) also described how using cocaine would cause intense feelings of shame and guilt (8.6).

Substance use was discussed by some participants in relation to self-harm, though there was significant divergence of these experiences and the contexts from which they emerged. Early in his interview, Zack (H) reported that he tends to drink heavily on his own when his mental health is declining (8.7). Later in the interview, Zack recounted the first time that he smoked a cigarette as a twelve-year-old and discloses that he justified it by wanting to "look cool in front of [his] older brother" but that secretly it was "just another form of selfharm" (8.8). At the end of his interview, when asked if he would be willing to seek help for substance use-related issues he denied this, explaining that "I deserve to feel to like crap and to be on drugs [...] I don't wanna bother people with my stupid alcoholism" (8.9). As the interview progressed, the relationship between Zack's mental health and substance use became demonstrated more explicitly, and the contradictory nature of Zack's beliefs about substance use (a method of self-harm vs. a method of seeking gender euphoria) shed light on his complicated relationship with substances and with himself. Darren (L) also discussed how substance use and self-harm interact in his life, but his experiences and the personal meanings are contrasting to that of Zack. For Darren, alcohol was used as a method of preventing himself from self-harming because he perceived it to "calm me down a bit" (8.10), while Darren acknowledged that excessive alcohol use is harmful and potentially addictive, he viewed his alcohol use as a less damaging coping mechanism than self-harm (8.11). The damaging nature of alcoholism, for Zack, however, was part of his motivation for excessive alcohol use.

**Box 5.8.** Representative extracts for "The Symbiotic Relationship of Mental Health and Substance Use" in "Distress (In)Tolerance". Quotes are from both low substance use group (L) and high substance use group (H).

- **8.1** "I mean at the time ... it was before I started transitioning or anything ... and [alcohol] was very much a crutch ... and as my mental health has improved an awful lot the need for those crutches has gone away" (**Bailey, L**)
- 8.2 "I can remember how sort of … how reliant I got in terms of … when I wasn't busy and occupied I was using [alcohol] as a device to not have to deal with that stuff but these days the anxiety and depression has got a lot less since I've got through the process of transitioning … and so I don't need those coping mechanisms so much *Interviewer:* do you think that drinking was connected to the struggles with your gender that you were having at the time? *Jeremy:* absolutely yeah … definitely it was easier [to drink] than to deal with it because at that point I didn't … have the necessary language and knowledge of anything LGBT […] so turning to alcohol to take the edge off it was definitely my way of coping but not really coping at the time … just to not have to think about things or feel things and just … not have to deal with any of it" (Jeremy, L)
- **8.3** "A lot of things associated with ADHD are things which have been copping up as issues in my life **a lot** [...] when I was 14 I would get sensory overloads and changing moos and whatever ... and I was unable to do anything ... like in general but also about the way I was and it led to me going out drinking several times a week and doing the only thing I knew I could do ... just be the life of the party" (**Rowan, H**)
- **8.4** "If I have a drink I know I'll feel down for a few days afterward ... and then when I drink regularly my mental and physical health gets a lot worse" (**Parker, H**)
- 8.5 "I've been drinking a lot and I've put on a lot of weight and that's really upsetting because it all sits around my stomach and hips ... right now I hate my body ... and I usually just end up more miserable and eating and drinking more ... it just ends up as this cycle" (Parker, H)
- **8.6** "I always feel really guilty afterwards ... I mean **really** guilty ... because I've let them down ... especially my mum ... I mean it's disgusting and I feel ashamed of myself everyday" (**Simon, H**)
- 8.7 "When my mental health is on a decline I kinda ... drink on my own a lot ... I get those situations a lot where I just am on my own and feel ... very much like I wanna be off my face right now ... it doesn't happen all the time but it is kinda often that I have gotten really messed up on my own just because I felt like I needed to" (Zack, H)
- 8.8 "When I was about 12 er ... my brother at that point he was 17 and he'd like started smoking because he was a cool 17-year-old who has like a bottle of whiskey in his room and a packet of cigarettes ... and ... my twin brother he goes to me like 'oh hey y'know he let me take a puff of his cigarette and I coughed up like coughed my lungs out' and I was like 'what?! I want a puff' y'know 'what's this about' ... so I sneaked downstairs ages later ... and rolled myself the most god-awful- because I'm a 12-year-old the worst cigarette of my life and then I tried to smoke it out of my window and half of me was justifying it by saying okay I'm preparing myself so when he asks us if we want a toke I'm not gonna cough ... I'm gonna look cool in front of our older brother ... and then like the other half was-by that point I'd also started self-harming and it was kind of just a form of that to be honest" (Zack, H)
- 8.9 "I wouldn't seek help and definitely not if I'm in the peak of a depressive episode *Interviewer:* Why do you think you wouldn't seek help? *Zack:* Basically I'd feel like I deserve it ... like I deserve to feel like crap and to be on drugs and it's very much a ... self-eating snake of feeling like I deserve it and also wanting at the same time to feel this sadness ... like I wanna feel this bad because I deserve it and so ... y'know I'd feel like I wasn't ... worth ... getting help and I don't wanna bother people with my stupid alcoholism" (Zack, H)
- **8.10**"I've got quite a lot of mental health issues and I'll just try to block it a bit with alcohol [...] I have the urge [to self-harm] daily and alcohol can reduce the chances of self-harming ... it just calms me down a bit and it's like taking a different measure to change how I'm feeling inside" (**Darren, L**)
- **8.11**"[alcohol] is not something I want to rely on much for a coping mechanism ... so I try not to treat it in that way because I don't want to get dependent on it" (**Darren**, **L**)

# 5.3.5. Impulsivity and Novelty Seeking

**Pursuit of Novel Stimulation.** For participants in the H group, the experiences that they recounted indicated that substance use was not always conceptualised as a coping method for negative emotions or to facilitate socialising, as it was for members of the L group. Instead, some participants in the H group discussed how substance use was an enjoyable activity that they sought out in the pursuit of novel experiences. Logan (H; Box 5.9; 9.1) described how he had a "really positive outlook on drinking" because it enabled him to have new experiences and enhanced his enjoyment of activities such as festivals and parties with friends. Similarly, Gale (H; 9.2) would engage in substance use to enhance their enjoyment of music festivals. They described how their substance use would be carefully planned in order to get maximum effects from the drugs they were using, and also so that the type of drug would complement the activity they were doing (i.e., marijuana for low energy activities and cocaine for high energy activities).

Where Gale would plan to use specific substances at different times, other participants described how they would combine substances to enhance their effects. For example, Zack reported that he enjoyed experimenting with various drugs in different combinations (9.3), and Simon (9.4) enthusiastically described how he typically uses alcohol and cocaine together because it is "the best feeling in the world".

**Box 5.9.** Representative extracts for "Impulsivity and Novelty Seeking" Quotes are from the high substance use group (H).

- **9.1** "I guess I [drink alcohol] a lot just to have a new experience ... or a different experience ... like even getting really sweaty and disgusting is all part of it ... especially at a gig or a festival ... and if you go to a club that is **packed** and the music is loud and you're literally sardined in ... it's like you enjoy the whole experience and the drinking takes it up a notch ... the whole thing is so fun and ... yeah I definitely have overall a really positive outlook on drinking (Logan, H)
- 9.2 "I like MD because it lasts longer than coke … but I'll very much time it if like I want to come up at this time because I want to see these bands under the influence … for certain bands when I'm looking at a festival line-up I'll be like okay so Thursday is gonna be my MD heavy day … and then Friday I'm just gonna drink because there's not much I'm arsed about er … Saturday I wanna be … I'm gonna watch some acoustic stuff so I'm just gonna have a couple of joints and then Sunday I wanna be sort of … er y'know I wanna get stuck in for a couple of bands between 6 and 8 and then after that there's nothing so I'd think well I don't want to have MD because I'll be awake pointlessly dancing to shit I don't like [laughs] so that's sort of how I can organise a 4 day festival in what I'm having to make sure that I can peak because there's no point trying to peak on MD twice in a weekend it's just not gonna happen" (Gale, H)
- **9.3** "To give a broad list of everything I've done ... weed Xanax Valium LSD 2CB DMT ... coke ket ... there's 1 or 2 more ... I've forgot its actual name but it's some weird synthetic heroin type thing ... and I used to smoke **a lot** of weed ... I tried LSD and I was like yeah this sounds really cool ... also I've done MDMA a few times and sometimes would take it with acid because that's what's called a Candy Flip and it's a whole lot of fun" (Zack, H)
- **9.4** "[alcohol and cocaine] gives you the best feeling in the world and you get a rush of euphoria and you're having a drink and you're like 'oh I feel a bit drunk' so you have a sniff or a line and you're just like 'wow' and you get the best feeling ever you just get a warm feeling and it's like your heart slows down and you're like wow that's just amazing honestly it's the best feeling ever ... best feeling ... like the feeling of alcohol **makes** me want the coke and the coke makes me wanna drink ... they go together easy" (**Simon, H**)
- **9.5** "I can't buy a bottle of gin and just keep it in the cupboard because I wouldn't stop drinking it ... I would just drink it every single night so I need to buy cans to limit how much I drink ... I just really think I'm someone who just shouldn't drink at all ... I just ... I don't think I have the control to drink in moderation" (**Parker, H**)
- **9.6** "when my sort of reasons to behave ... aren't there anymore then it's like well if everyone else is doing that and they're having a great time why wouldn't I sort of be in the same headspace as everyone that I'm with ... so it's normally like if I haven't got the game the next day er ... sort of my self-control goes out the window 'cos there's nothing to control myself for ... and using alcohol I think I'm more likely to vomit when I've had alcohol because I don't know when to stop" (**Gale, H**)
- 9.7 "Alcohol is the worst trigger for (cocaine use) it's like my number one trigger ... it lowers my inhibitions so you don't think about the consequences or the future ... you just think 'ah fuck it' like act now and think later *Interviewer:* do you consider yourself to be an impulsive person? *Simon:* yeah definitely ... if I was at a friend's house and we were having a drink and they said let's go out then I'd just be like 'yeah let's go' ... I'm that impulsive ... and it's not a good thing in circumstances when we're drinking" (Simon, H)
- **9.8** "If I'm drinking then I act more impulsive and weird ... like especially with dancing and my favourite thing about going out is the dancing ... and I look like an idiot dancing on the floor so I need to be quite drunk" (**Rowan, H**)
- **9.9** "when I'm not doing well mentally then I won't particularly care ... and sometimes I have a depressive episode or I go very ... manic ... and then I'm quite impulsive ... one reason I took acid on my own was ... it was 10pm and I had college the next day so I wanted to see what a micro-dose felt like ... but I ended up taking the entire tab anyway and went to college having not slept and still tripping" (**Zack, H**)

**Impulsivity: Risks and Consequences.** The participants in the H group also discussed how impulsivity would lead to increased substance use and that substance use, in turn, would make them feel more impulsive. This is in contrast to the L group, none of whom mentioned impulsive behaviours related to substance use. Some participants in the H group, like Parker (9.5) reported that they had difficulties stopping drinking alcohol once they had started and perceived themselves to lack self-control. Gale (9.6) also talked about lacking self-control, but only when they thought they had no reason to try to moderate their behaviour, such as work commitments. Simon conceptualised himself as an impulsive person, and that his impulsivity was heightened when he was under the influence of alcohol (9.7).

Other participants related impulsive substance use to their experiences of neurodivergence and mental illness. Rowan (9.8), who reported having ADHD, explained that alcohol makes them act "more impulsive and weird", and Zack (9.9) described how his impulsive behaviours would increase when he was experiencing symptoms of depression and mania.

## 5.3.6. Protective Factors in Substance Use

**Personal and Family History of Substance Use Issues.** Participants in both the L and H groups described having struggled with substance use in their own lives. Participants in the H group discussed their current struggles with their relationship to substance use, whereas the L group tended to have past issues with excessive substance use which they now managed, but some reported that they currently struggled with the urge to engage in it again.

Past negative experiences of substance use were often cited as a motivating factor to reduce use for participants in both groups. The consequences of a particular instance of excessive drinking prevented Ezra (L; Box 5.10; 10.1) from drinking so heavily again. The

physical side-effects of drinking also prompted Bailey (L, 10.2) to reduce their alcohol intake after they began a medication which made them unwell when mixed with alcohol. For some participants in the H group, hangovers or feeling unwell after a session of heavy drinking were not always motivation enough to reduce their use. For example, Simon (H; 10.3) reported how he will drink heavily and feel unwell the next day but will repeat the same behaviours a week later. However, Simon also described how he decided to seek help from an alcohol and drug recovery service for cocaine use after getting an infection in his nose (10.4).

Additionally, some participants in both L and H groups had a family history of substance use which acted as a deterrent in their relationship with substances. Aaron (L; 10.5) described how because his dad is a "chain smoker" and "an alcoholic" he acted as "a good example of why not to [use substances]". Similarly, Simon's dad (H; 10.6) had an addiction to heroin and had overdosed on it in the presence of Simon, which made him avoid drugs and drug users when he was growing up because he was "proper against drugs".

**Box 10.** Representative extracts for "Personal and Family History of Substance Use" in "Protective and Prohibitive Factors in Substance Use". Quotes are from both low substance use group (L) and high substance use group (H).

- 10.1 "Two conventions ago I drank a lot ... and ended up being so sick and having a 3-day hangover ... it was horrible I felt so ill and fragile and really really embarrassed as well ... because I didn't remember my behaviour or anything ... and I was so mortified that I'd gotten into that state ... so it put me off so much that I've never gotten into that state again" (Logan, L)
- 10.2"I was put on medication for hormonal issues which meant I couldn't drink or I would get very very sick ... so I was teetotal for a while ... and that was probably the best thing I could have done because it really curbed [my drinking] and now I barely drink (Bailey, L)
- 10.3"Normally I'd go out and drink drink until 9 o'clock in the morning and then go home ... wake up later feeling anxious and depressed ... chest pain ... feels like I'm gonna die for a full week ... and I say 'I'm never ever doing it again" but then a week later [throws hands in air and shrugs]" (Simon, H)
- 10.4"I got a nose infection from sharing the snooters ... I told my doctor I've been taking cocaine and he had a proper look and was gonna send me to the ENT ... to see how much damage I'd done ... and it had got to the point where I was so sick of it ... I messaged my tutor at college while I was still pissed up and asked for help and she referred me onto [local alcohol & drug recovery service] ... so my appointments there have just started" (Simon, H)
- 10.5"Part of the reason why I've never had strong substance use is my dad ... is a good example of why not to do it [laughs] I had a fear when I was younger that if I started then ... what if I turn into my dad ... that's my self-control ... he's a chain smoker and he's ... okay ... I do think he's an alcoholic ... but it's like the way it was entwined in his life and became a part of him ... I just didn't want that for myself ... it affected his judgement especially the drinking ... like whatever you're dealing with mate this is not the way to deal with it ... I mean I know it's a common thing in Russia ... like there's literally a saying ... like there's a woman in a marriage and she's unhappy people will say like hey at least he's not a smoker and at least he's not a drinker and he doesn't beat you ... like the bar is very low [laughs]" (Aaron, L)
- **10.6**"I used to be proper against drugs because of my dad ... yeah he messed me up about them especially after his heroin overdose right in front of me ... and so I wanted to avoid them and avoid people that took them ... I guess because they reminded me of my dad and I didn't want that" (**Simon, H**)

Attitudes and Behaviours of Peers. The behaviours and attitudes of the participants'

friends were one of the strongest influences on substance use. Where friends engaged in substance use this would influence the participant to do that also; where friends did not drink alcohol or use drugs, the participants reported less frequency and quantity of substance use. Five of the six participants in the L group stated that their peer group did not often drink or use drugs and that this meant that they also did not either (Box 5.11; 11.1). Aaron (11.2) stated that because his partner "really frowned upon smoking" and was "sensitive to alcohol mismanagement", this was a protective factor which prevented him from substance use. While all the participants in the H group spent time with friends who also frequently used alcohol or drugs, Simon (H; 11.3) had recently made new friends who did not use substances.

This was having a positive influence on his own attitude to alcohol and drug use because it

meant that he was exposed to "the normality of what life can be like without all the chaos".

**Box 5.11.** Representative extracts for "Attitudes and Behaviours of Peers" in "Protective and Prohibitive Factors in Substance Use". Quotes are from both low substance use group (L) and high substance use group (H).

11.1"If I'm at a social event then maybe I'll have one [drink] or two at a push ... it's really only a couple [of drinks] with friends these days and since my friends don't drink much then I don't feel the need to drink either" (Jeremy, L)

- 11.2"My partner really frowned upon smoking so I have multiple deterrents from getting back into bad habits ... both of us have parents who have **bad** relationships with substances ... or difficult or ... I'm not sure if bad is the right word to use but ... they do have a relationship with substances which affected their children badly yeah [laughs] so he's quite sensitive to alcohol mismanagement and ... I'm sensitive to become a person who manages alcohol insensitively" (Aaron, L)
- 11.3"I think meeting more people has helped ... it's good for me to see other people and the patterns in their lives ... like seeing how people my age are living their lives with no worries and I'm seeing that and thinking I could be doing that ... you know I get to see some normality of what life can be like without the chaos and it makes me think drinking a lot during the week isn't actually **that** normal" (Simon, H)

Harm Avoidance. Participants in both groups discussed methods of limiting potential

harm which may arise through substance use. In the L group, Jeremy (Box 5.12; 12.1)

described how reflecting on his past behaviours and restricting excessive alcohol

consumption is necessary for him to stay in control of his substance use, and similarly Cody

(12.2) reported that recognising unhealthy patterns of substance use in himself was key to

managing his mental health and limiting harm from substance use.

Participants in the H group, who all currently reported frequent use of alcohol and/or drugs, described how they engaged in various methods of harm avoidance in relation to substance use. Gale reported that they were more likely to drink less alcohol if they were also taking drugs (12.3) and made sure to look after their friends when they were all taking drugs (12.4). Rowan (12.5) would make sure his alcohol consumption on nights out was limited by being conscious of how much he was spending, and Owen (12.6) would limit his alcohol

consumption because he would begin to feel tired (as a precursor to feeling sick) and made

sure to stop drinking at that point.

**Box 5.12.** Representative extracts for "Harm Avoidance" in "Protective and Prohibitive Factors in Substance Use". Quotes are from both low substance use group (L) and high substance use group (H).

- 12.1"I'm quite aware of how I've been in the past and I try not to [drink] for reasons of intoxication and only doing it for something that I want to enjoy for the taste ... I've definitely reached the point where I'm self-aware and in control ... it's definitely manageable ... although I have thought about it a lot and so I'm aware that I have a potentially addictive personality" (Jeremy, L)
- 12.2"At times I have been drinking in a way that was a bit unhealthy but it also doesn't take long to recognise that drinking is making [my mental health] worse ... so I do try to recognise these patterns and generally try to change it when it's really not making things any better" (Cody, L)
- **12.3**"Obviously if I have a Class A then I wind down how much alcohol I'm taking ... and sort of be aware of getting through a 500ml bottle of water every hour by just sipping it and looking at the clock and being careful not to overdo it or underdo it ... I'm much more likely to be sensible with drink if I **have** had Class A's" (**Gale, H**)
- 12.4" we all order off the same dude so we all are on the same stuff ... like pick one person's dodgy geezer and stick with him for that night out ... we're all eating out the same bag ... and at least you know sort of how much everyone has had and who came up first and what to expect or ... if someone is having a funny turn we can be like well they shouldn't have any more ... and we do try to look after each other" (Gale, H)
- 12.5" I've only been blackout drunk very few times ... I try not to get to that stage ... and I guess I don't because I'm not very financially sure ... I don't want to spend more than a certain amount and I put limits on that ... like it already costs to get into the club and I don't want to spend much so I'm very careful about that" (**Rowan, H**)
- 12.6"When I drink it can get to the point where it's not fun anymore and I just want to lie down and I feel sick really suddenly *Interviewer:* do you find you usually drink to that point? *Owen:* No I usually stop before then ... before I feel sick I start to feel really tired ... so it's sort of a natural barrier because I get really sleepy and just go to bed [laughs]" (Owen, H)

## 5.4. TESUP-1 Discussion

This study investigated correlates and experiences of substance use in transmasculine individuals. The aim of this research is to expand our understanding of how substance use is initiated and maintained in this population, explore gaps in knowledge related to how masculinities, gender minority stress, quality of life, personality traits, and psychopathological symptoms may contribute to substance use, and to better understand the lived experiences of transmasculine individuals in relation to substance use. This investigation will contribute to both a theoretical knowledge of substance use in transmasculine individuals and will also provide much needed information to healthcare professionals and service providers who work with this population in the UK.

In the first phase of research, the questionnaire responses of 105 transmasculine individuals were analysed, and from this sample, 13 participants were interviewed. From this sub-sample six participants engaged in low-risk substance use (denoted L in excerpts) and seven participants were categorised as engaging in high-risk substance use (denoted H in excerpts), as measured by the AUDIT and DUDIT. Participants were classified as high-risk if they scored 8 or over in the AUDIT or 6 or over in the DUDIT.

The central research question for this study was: What are the factors that contribute to the development and maintenance of substance use in transmasculine individuals?

There were several specifically quantitative research questions:

- 1. Is stage of transition related to substance use?
- 2. Is low QoL associated with higher substance use?
- 3. Does increased GRC correlate with higher substance use?
- 4. Does increased GMS correlate with higher substance use?
- 5. Is resilience a protective factor against issues with substance use?

- 6. Are psychopathological symptoms associated with substance use?
- 7. Is there a specific personality profile for individuals who engage in high-risk substance use?

Following on from the questionnaire, the interviews sought to explore the differences and similarities in experiences between the low-risk group and high-risk group. The associated research questions (below) reflect the aims of the interviews as attempting to provide a deeper exploration of GMS, masculinities, contexts of substance use, and protective factors to prevent high-risk substance use.

- 8. What is the impact of transitioning and masculinities on substance use?
- 9. What contexts and situations are associated with substance use and what do these mean to the participants?
- 10. What factors contribute to a cessation or reduction of substance use?

In this chapter an integrated discussion of quantitative and qualitative findings is presented. First a discussion of quantitative and qualitative results as it relates to each research question is provided in the context of four main themes which emerged throughout both quantitative and qualitative analysis. These four themes are: Transition, Masculinities, and the Function of Substance Use; Contexts of Substance Use and Meaning-Making; Risk Factors in Substance Use; and Protective Factors in Substance Use. The discussion of these core themes is then followed by a discussion of how each of these themes relate to the main research question. Finally, the chapter concludes with a discussion of practical implications, recommendations for future study, and limitations.

## 5.4.1. Transition, Masculinities, and the Function of Substance Use

**Stage of Transition.** The findings of this study indicate that the process of discovering and accepting one's transgender identity, often followed by socially and medically transitioning was, for some participants, related to increased substance use. Although there was no significant correlation between stage of transition and substance use in the quantitative data, the responses from interview participants indicated that they did perceive their transition to have had an impact on their substance use. For example, Jeremy (L) reported that complications from phalloplasty surgery left him feeling concerned about "slipping back into problematic behaviours". The qualitative results may suggest that while the more general 'stage of transitioning may represent risk factors for increased substance use. There have been very few studies which have explored stage of transition in relation to substance use. However, one study (Budge et al., 2013) which explored coping processes throughout gender transition, highlighted that substance use as a coping mechanism was highest for individuals in the 'pre-transition' stage, reduced slightly during transition, and was not mentioned by participants in the 'post-transition' stage.

Although the analysis of quantitative data in the current thesis did not indicate that substance use decreases as an individual reaches more self-defined goals in their transition as in the Budge et al. (2013) paper, the interview results often contradicted this. Participants in both groups consistently described how transitioning reduced their dysphoria and increased their confidence and happiness. These participants all conceptualised these changes as being related to a reduction in their substance use or to a 'reversal' of their motivation to use substances from maladaptive ones to healthy ones. This supports the findings from previous studies which have demonstrated that transition-related treatment improves the QoL of treatment-seeking transgender people, in addition to being associated with lower depression

130

and anxiety (Baker et al., 2021). The present results may suggest that it is not the precise 'stage of transition' which contributes to substance use, but instead that it is a feeling of security and stability in one's gender identity and transition which may be more important. Guzman-Parra et al. (2014) reported that for treatment-seeking transgender individuals cannabis use was associated with anxiety and suggested that cannabis may be used in this population to self-medicate anxiety. The current study provides support for this finding; anxiety symptoms were significantly associated with both alcohol and drug use, and of the 59 participants who indicated that they typically consume drugs, cannabis was the most frequently reported drug (N = 25).

The experiences of non-binary transmasculine people compared to binary trans men in relation to medical transition and dysphoria vary in multiple ways. For example, in this study although there was no significant correlation between binary/non-binary gender identity and substance use, in the interviews it emerged that all the participants who reported substance use to cope with body dysphoria or non-disclosure of their trans identity identified as binary, and none of the non-binary participants related their substance use to these struggles. This may be because binary individuals are more likely to seek out medical interventions; one study (James et al., 2016) found that 95% of binary transgender participants desired HRT whereas 49% of non-binary participants desired it. Other literature reports that non-binary people are less likely to report bodily gender dysphoria, but instead emphasise the social roots of dysphoria such as misgendering and micro-aggressions (Galupo et al., 2021). While both binary and non-binary participants in the current study experienced similarly high levels of substance use, the motivations for engaging in substance use differed between the two groups.

It is also important to note that the findings related to 'stage of transition' could be limited by the operationalisation of the concept. The current formulation allowed for individuals to place themselves at pre-, early, mid-, late, and post-transition; a structure which was suggested by participants in the pilot study focus group. Within the focus group it was noted that transgender identity discovery and related transition processes are flexible and highly unique to the individual. Therefore, 'stage of transition' was measured through participant self-identification with a 'stage' and no specific beginning or end point was suggested by the author. This means that participants who identified as 'mid-transition' for example, had experiences which ranged from waiting to begin HRT to having had top surgery and waiting for lower surgery. It is possible that with an improvement in the operationalisation of 'stage of transition' the validity of this concept would increase. If the measurement of this concept could allow for non-linear flexibility but was still accurate enough to capture discrete groups of individuals, then possible relationships between transition stage and substance use could be revealed.

**GRC, Gender Socialisation, and Masculinities.** Another area of exploration in this study which revealed seemingly contradictory results between expected findings in the quantitative and qualitative data was the impact of GRC and masculinities on substance use. To discuss the GRC scores of the participants in the present study, scores will be compared to the normative data of adult gay men (Ervin, 2004). This decision has been made because only 16.2% (N = 17) of participants in the present study identified as heterosexual. The remaining participants all expressed an attraction to men or masculine-identified people except for 6.6% (N = 7) who identified themselves as asexual. There are some immediate issues with the use of this normative data, as well as with the measure of GRC itself. The main limitation of the GRC and the related normative data is that it is all based on binary men; there is no data, and to my knowledge, no study other than the present research, which has included non-binary masculine-identified people. First, the language used within the measure is aimed at binary men (e.g., "expressing my emotions to other men is risky"); although the individuals in the

current study all identified with masculinity to some extent, they were not all males. One participant provided feedback: "I'm not a man and my experiences are not those of a man and so these questions do not relate to me at all". This raises questions about the validity of the GRCS as a whole if it does not measure what it purports in all masculine-identified samples. Additionally, some participants provided feedback which stated that some of the questions would only be applicable in a context where they are presenting as 'stealth' among cis men, but the answer would change in a situation where they are among other transmasculine people. This calls into question whether the GRCS is an appropriate measure for use in transgender populations or perhaps if GRC in transmasculine individuals could be better understood by having participants complete the questionnaire in different conditions (i.e., socialising with cisgender men vs. with transmasculine people).

The correlational analysis of the GRCS in relation to substance use revealed that the only factor out of the four factors to be associated with substance use was Restrictive Affectionate Behaviour Between Men (RABBM), which was significantly correlated to both alcohol and drug use. However, it is important to note that while most of the factors were not associated with substance use, the mean score in two factors was higher than that of the normative data based on adult gay cisgender men (Ervin, 2004). The factors which were higher in the current study were RE (37.33 vs. 28.30 in Ervin, 2004) and RABBM (25.38 vs. 18.80 in Ervin, 2004). Interestingly, when comparing the present study's sample with the normative data of adult cisgender men in the UK (Tate, 1998) it is demonstrated that RE was also higher in the present study (37.33 vs. 28.77 in Tate, 1998), but so was SPC (43.30 vs. 40.84 in Tate, 1998). When considering what these results might mean in relation to transmasculine individuals, the interview data can shed some light on the meaning-making surrounding experiences of masculinity and GRC.

When discussing GRC it is important to note that GRC refers to the "psychological state in which socialised gender roles have negative consequences for the person or others" (O'Neil, 2008, p. 362). Therefore, it is not the belief of stereotypic norms of masculine behaviour, nor the pursuit of a hegemonic masculine ideal, which results in the psychological state of GRC. Instead, it is the behavioural and affective restrictions inherent in the socialised male gender role which limit the individual's ability to adaptively respond to various situational demands (O'Neil et al., 2017). One aspect of O'Neil's (2008) definition which I believe is important when considering GRC in transmasculine individuals is "socialised gender roles".

As mentioned in chapter two, O'Neil (1981) states that GRC is best conceptualised in the context of early gender role socialisation as children. Wester et al. (2010) suggest that transgender youths face negative consequences for not conforming to the expected gender role according to their assigned sex at birth. For example, a young trans boy who does not confirm to a traditional feminine gender role will often likely face harassment, shame, and demands for conformity from others (Dietert & Dentice, 2013). It is important to understand what gender socialisation refers to, typically it is thought of as "a process by which individuals develop, refine, and learn to 'do' gender through internalising gender norms and roles as they interact with key agents of socialisation, such as their family, social networks, and social institutions" (John et al., 2017, p.6). In this definition it is clear that gender socialisation is a two-way street; it involves the matter of which specific cultural norms and expectations are placed on an individual, but crucially, it also consists of which of these are identified with, internalised, and enacted. Furthermore, gender roles are learned both explicitly and implicitly, so an AFAB child will be learning about the female gender role by being directed to perform female gender norms but will also be learning about gender roles implicitly by way of observation and comparisons to how others are learning and doing
gender. Therefore, while norms are being applied to that child, the child is also integrating or rejecting those norms into their own beliefs and behaviours.

Based on this definition of gender socialisation, I contend that claims that an AFAB transmasculine individual is always socialised female are inaccurate. Such a child may be being taught and encouraged into a female gender role, but if they reject or fail to meet these norms, they often then receive a socialisation punishment, shame, and demands for conformity to a gender with which they do not identify (Dietert & Dentice, 2013). This is exemplified in the participants' interview extracts: Jeremy (L) reflected on his childhood and stated that he "never behaved or identified with the way everyone around [him] expected [him] to be" and pointed out that he lacked the knowledge and language to describe his own feelings in relation to his transgender identity. This unique experience of gender socialisation, which can be framed in the context of hermeneutic injustice (Hänel, 2020), demonstrates first how transgender individuals have a distinct gender socialisation experience, and secondly how this unique experience of lacking other- and self-recognition can have detrimental effects on understanding oneself and on the ability to communicate this. This might then be translated into the experiences described in Dietert and Dentice's (2013) study, in that trans youths who will not, or cannot, conform to gendered expectations are socially punished through exclusion and harassment, and by the internalisation of shame.

Therefore, to understand GRC in this population, it is first necessary to explore how transmasculine individuals conceptualise and how they 'do' masculinity, particularly in the context of substance use. Jeremy believed that because he "fit other people's expectations of masculinity" it meant that he did not feel the need to "overcompensate or do things that aren't my natural inclination" in relation to masculinity. This overcompensation in order to fit other's expectations of masculinity relates to the concept of *precarious manhood*, the idea

that manhood must be consistently enacted appropriately, and that it is a social status that is difficult to achieve and is easy to lose through acts of femininity (Vandello & Bosson, 2013). Related to this concept is a theory which is especially relevant for transmasculine individuals termed *compensatory masculinity*. This is the idea that an individual must enact hypermasculine behaviours and attitudes in order to communicate their masculine identity and to compensate for a perceived lack of masculinity or maleness (Vegter, 2013). The concept of compensatory masculinity was often alluded to in the participant interviews as a major motivator to engage in substance use. A common concern among the participants in this study was that if they did not consume substances in a masculine way then they would not be perceived as male at all by others. This attitude was exemplified in an extract from Rowan (H) who described compensatory masculine behaviours as "it's all about **how** you drink and the **things** you drink … and if you don't do that then you're purposefully leading yourself to get misgendered" (emphasis in original statement).

The GMSR results may help us understand why masculine substance use was so important to many of the participants in this study. Correlational analysis demonstrated that previous experiences of gender-related victimisation were associated with tobacco use and drug use, and experiences of gender-related rejection were associated with drug use. It is possible that individuals engaged in compensatory masculinity not only to receive external validation of their gender identity, but also to avoid further victimisation or rejection. The language that Rowan (H) uses in the previous extract when explaining why he engages in compensatory masculinity implies that the onus of responsibility is on him to ensure that he is correctly perceived as male, and therefore, the negative consequences are also his fault. On one hand, conceptualising his behaviour in this way may give him a sense of control over his gender identity and how he is perceived by others. Conversely, the constant self-monitoring of behaviour and rumination on how one is being perceived could be a contributing factor to the high levels of anxiety and paranoid ideation which were found in this sample. Both anxiety and paranoid ideation were associated with high-risk drug use, additionally, paranoid ideation was associated with high-risk alcohol use.

Typically, it was members of the high-risk group who described engaging in substance use in the pursuit of masculinity. One of the few members of the low-risk group, Cody, who described pursuing masculine substance use stated that it was because of "social dysphoria". Similarly, Zack (H) framed all of his motivations for substance use in the context of pursuing masculinity and seeking gender euphoria. Gender euphoria is a frequently used term in the transgender community, which is not often recognised in published literature. Typically, it is used to refer to feelings of joy and validation that occur when one's gender identity is recognised and accepted by oneself or others. These results indicate that context is important when considering the impact of GRC on transmasculine substance use. This lends support to Vandello & Bosson's (2013) discussion of GRC and laboratory-induced threats to manhood; that GRC is highly dependent on context and is responsive to situation-specific cues, as opposed to being a chronic psychological state.

As previously mentioned, RABBM was the only GRCS factor which was significantly associated with alcohol use and drug use. When taken into consideration with the qualitative findings, some potential explanations emerge. First, the interview participants expressed concern over the risk of being misgendered or 'outed' as transgender, particularly in the context of substance use. The idea that anti-masculine behaviours were a risk to their masculine status has some commonalities with RABBM, which includes statements such as "I am hesitant to show affection to other men because of how others might perceive me". Anxiety about demonstrating affectionate behaviour to men may arise from previous experiences of gender-related victimisation and rejection, therefore it was not just precarious manhood which was a cause for concern but also the very real risk of danger. The avoidance of this may be a motivator for excessive alcohol and drug use in this sample.

## 5.4.2. Contexts of Substance Use and Meaning-Making

Navigating Transition-Related Healthcare. In this study, all but one of the interview participants were seeking transition-related medical interventions. All of the participants who discussed the process of seeking transition-related healthcare reported difficulties in accessing NHS services. Many of the participants described how healthcare professionals often lacked knowledge and experience in providing care to transgender patients. For some participants in this study, disclosing their trans identity and seeking referral to a GIS often resulted in barriers to accessing appropriate and timely interventions. For example, when Owen (H) asked for a referral to a GIS, he reported that his GP immediately assumed "oh you're trans you must also be depressed" which resulted in significant delays to his referral. Owen's experience is echoed in other recent research into trans experiences of accessing NHS healthcare. Wright et al. (2021) reported that many participants in their qualitative study of access to gender-affirming healthcare had been subject to unnecessary referrals to mental health services due to GPs' lack of knowledge about appropriate care pathways.

Two participants in the low-risk group, Jeremy and Bailey, described how their GPs had never worked with a trans patient and therefore lacked the experience necessary to provide appropriate healthcare. However, whether they perceived this as a positive or negative experience was dependent on their GP's approach. In the absence of straightforward referral guidelines, Jeremy's GP conceptualised the provision of his transition-related care as a learning opportunity which resulted in Jeremy having a positive appraisal of the experience. This is in contrast to Bailey, who reported their GP had "no idea what she's doing". As a result, Bailey was struggling with the high cost of private healthcare. Lack of healthcare provider knowledge is a concern that has been reported in various studies. For example, the Trans Lives Survey (TransActual, 2021) found that 45% of participants reported that their GP did not have a good understanding of their needs and 14% of respondents reported that their GP had refused to provide healthcare due to their transgender status. Furthermore, a recent study (Willis et al., 2020) which explored older transgender adult's experiences of accessing primary care found that they were positioned as 'reluctant educators' for GPs and that they perceived self-advocacy to be a necessary and often stressful aspect of accessing transition-related healthcare. The results of the current study support Willis et al. (2020) and demonstrate that even for younger transgender adults, who may have less experience of self-advocacy, the role of 'reluctant educator' is essential. These results emphasise the necessity of primary care providers taking an active role in enhancing their own understanding of transition-related healthcare.

Attempts to navigate the often confusing and complex pathway of transition-related healthcare in the NHS and the resulting long waiting times was described by participants as like being "stuck in a nightmare" (Parker, H). Many participants felt that the process of being referred to a GIS and attempts to receive the necessary transition-related medical interventions was challenging and convoluted, echoing the analogy of defeating 'bosses' in a videogame as described by Greg in the pilot study. Difficulties navigating access to transition-related healthcare and negotiating authenticity mentioned by participants in TESUP-1 were further mirrored in the pilot study focus group, where participants described feeling pressure to "perform" and "jump through hoops" for healthcare professionals. As discussed in chapter four, the feelings of restriction and pressure to appropriately perform a (typically binary) gender identity, fed into the lack of trust in healthcare providers and nondisclosure which was reported by participants in both the pilot study and the TESUP-1 interviews. Studies into trans experiences of medical transition in the UK have also reported these same concerns from participants. Harrison et al. (2020) found that participants in their study were unsatisfied with the support they received from healthcare professionals throughout their medical transition, and many felt the need to 'prove' their trans identity in order to receive timely and necessary medical interventions.

**Social Substance Use and Peer Group Influences.** Almost all of the participants in this study conceptualised substance use as a normal aspect of socialising. Many participants reported that drinking while socialising enhanced their enjoyment of the experience, increased their confidence, and enabled them to meet new people. LGBTQ individuals face the same risk factors for developing high-risk substance use habits as do the general population, such as enhancing one's mood (Sinha, 2008), fitting in with one's peer group (Keyzers et al., 2020), and wishing to feel more confident (Marsh et al., 2019). These studies largely align with the experiences of participants in the current study.

However, the qualitative interviews revealed that these risk factors involve multiple layers of complexity and were inherently informed by the participant's status as transmasculine. For example, participants in the high-risk group often reported that peer pressure was a prominent aspect of why they might use more substances. This was exemplified by Simon (H) who described how alcohol and cocaine use was normal within his peer group and so he typically engaged in it too because he felt that it was crucial that he "wasn't singled out for attention like that". The passive pressure that Simon experienced was multi-faceted; he desired acceptance and belonging within his group of friends and to achieve this he modelled his behaviour on that of his peers. This involved a) engaging in heavy drinking and simultaneous cocaine use, and b) non-disclosure of his transgender identity. The first behaviour was critical for preservation of the second, and in his view, not engaging in either would mean social rejection or ridicule. This example demonstrates how even general risk factors are informed by the individual's status as transmasculine.

Previous studies have explored the relationship between peer pressure and substance use; Demant et al. (2014) interviewed a sample of 45 LGBT youth and community stakeholders who reported that passive peer pressure is higher within LGBTQ communities. Furthermore, an examination of the genetic moderation of peer pressure on alcohol use (Griffin et al., 2015) revealed that for individuals with at least one copy of the D4 dopamine receptor gene (D4DR), peer pressure was associated with increased lifetime alcohol use. This is a particularly interesting finding in the context of the current study. As was stated in the literature review, the temperament dimension of Novelty Seeking (NS) has been shown to be related to the D4DR gene (Ebstein et al., 1996). Furthermore, in the present study, statistical analysis revealed that high scores in the NS domain were significantly correlated with alcohol and drug use, and that a one-point increase in the NS domain accounted for a .121-point increase in AUDIT score and a .206-point increase in DUDIT score. These results demonstrate strong support for previous studies which showed that NS is associated with high-risk alcohol and drug use and enhances our understanding of the role of dopaminergic activity in relation to substance use.

The negative relationship between tobacco consumption and NS was an unexpected finding; it was hypothesised that higher levels of NS would be associated with current tobacco consumption, but the opposite effect was found. For each point increase in NS, participants were 1.142 times less likely to be a current smoker. Since regression analysis can only investigate whether a variable (here, NS) has an impact on the topic of study (here, smoking), I cannot make any causative claims about the relationship of the two variables. These findings are contradictory to previous studies which have repeatedly found that high scores in the domain of NS are related to increased tobacco consumption (Laucht et al., 2005; Etter, 2010; Masiero et al., 2021). One possible reason for this discrepancy is that almost all previous research that focuses on personality traits and smoking habits was conducted on cisgender individuals. Therefore, it is possible that previous findings suggesting high NS is associated with increased tobacco consumption cannot be extended to transmasculine populations. Based on the qualitative interviews, it is possible that other factors such as the pursuit of masculinity or peer pressure may increase tobacco consumption.

Another risk factor which contributes to increased substance use, for both general populations and LGBTQ populations, is using substances to increase confidence and reduce social anxiety. However, for the participants in this study, the motivations behind the desire for more confidence were often connected to their experiences as a transmasculine individual. For example, Owen (H) reported that most of the times that he had disclosed his transgender status to other people he was under the influence of alcohol. Similarly, Cody (L) reported that they would purposefully get drunk in order to have difficult conversations with their friends about their mental health and gender dysphoria. For Cody, this was directly related to being transgender, not only because they were discussing trans experiences but also because they were discussing it with other trans and queer people. These results support previous research which indicates that the process of 'coming out' as LGBTQ is a risk factor for increased substance use (Shelton, 2017, pp.44). The process of 'coming out' as LGBTQ is conceptualised by many in the community as a lifelong experience, and something that inherently carries an element of risk (Felner et al., 2020). This complicates the relationship between coming out and substance use; it is not a one-time event with a high level of risk but a continuous or recurring situation which, for some people, might exist alongside chronic substance use issues.

Furthermore, the process of initially discovering and disclosing one's transgender identity may also coincide with increased exposure to, and involvement with, LGBTQ

communities. For many transgender individuals this is a positive experience, as noted by Cody (L), building community connections can involve a deep-rooted feeling of trust and belonging. However, it is well documented that many traditionally safe spaces for the LGBTQ community are bars and clubs, which places individuals in situations where there is a high exposure to, and a normalisation of, substance use (Hunt, 2012). This has important implications for GMS, in particular, the resilience factor of Community Connectedness (CC). In the present study CC was not associated with substance use; however, some studies have questioned the hypothesised moderating effect of CC between GMS stressors and negative outcomes (Jaggi et al., 2018; Puckett at al., 2019).

The qualitative results provide further challenges to the hypothesised protective role of CC. While some participants reported positive experiences of socialising with other trans people, others described finding it difficult to connect with their trans peers. There were two important distinctions here. First, the only participants who had positive experiences were those in the high-risk group, with the exception of Cody (L). This could imply a relationship between frequent socialising in LGBTQ peer groups and substance use, although the lack of significant relationship between CC and substance use may contradict this. It is important to note that the CC factor measures feelings of comfort and belonging with the transgender community, not the frequency or quality of time spent with other transgender people, which may obscure the link between quantitative and qualitative results.

Secondly, all of the participants who had positive experiences of transgender CC reported that this occurred in person, whereas those who had negative experiences all reported that they had sought out transgender communities online. Recent research into LGBTQ community engagement online and offline has indicated that online communities can provide a safe and validating space for LGBTQ youth, especially where they might not

have offline support or do not have access to LGBTQ spaces in their local area (Higa et al., 2014). The idea of a *safe space*, commonly found in trans and queer literature, refers to a space which provides the LGBTQ community freedom from discrimination, abuse, and violence, with the ability to discuss issues freely (Kenney, 2001). The narratives of those from the low-risk group about seeking connection in online transgender communities was often framed in a negative manner. The language used by some participants, such as "they went to all girl's schools" (Bailey, L) and "[they] identify as 30 genders and as a fox or whatever and use weird pronouns" (Darren, L), reveal their discomfort with unconventional trans identities and a subtle rejection of the variance in transmasculine experiences of gender.

Darren (L), in particular, described how his attempts at seeking online community connections were fraught with difficulties, often getting into arguments with other trans people about the necessity of gender dysphoria to identify as trans. As a result of this *insider* harm (as Scheuerman et al., 2018, term it), Darren felt rejected from the mainstream transgender community online and saw himself as belonging to a community of "rejected trans people". To understand this, it is important to understand the socio-political context of the UK at the time of interviews. From 2017, a significant political and social pushback on transgender rights has been taking place, often centred around the concept of selfdetermination of gender identity (Pearce et al., 2020a). This has most often been led by "trans-exclusionary radical feminists" (TERFs) who argue for the elimination of trans rights in the name of biological essentialism (Vincent et al, 2020). Where transfeminine people are often cast as predators or as a danger to society, TERF ideology renders transmasculine individuals as confused or deluded women, infantilising them and ignoring their lived experiences as male and/or masculine people. A common thread between transfeminine and transmasculine experiences of TERF ideology is the denial of the existence of happy, healthy transgender people, and a push toward eliminating all transition-related healthcare services.

This "war on trans existence" has only increased during the COVID-19 pandemic (Pearce et al., 2020b), and with lockdown measures taking place throughout the majority of 2020 in the UK (the year in which the TESUP-1 interviews took place), transgender people were essentially limited to accessing support or community connections *only* in an online context.

Throughout the interview with Darren (L), it was clear that he was struggling to consolidate his personal beliefs about gender with his own identity as a trans man. His intense gender dysphoria and internalised transphobia may well have been influencing his online communications with other transgender people. There is limited research into the schisms within transgender communities; however, one study (Pinter et al., 2021) which explored online visibility and disclosure of transgender identities discussed the pitfalls of some online transgender support groups. The authors conceptualised online support groups as 'doorways' and 'traps', whereby an individual might gain a 'doorway' to more support and a feeling of empowerment or might get 'trapped' by a community which privileges some experiences, identities, and beliefs over others. In the present study, Darren can be conceptualised as being 'trapped' in both sides of the community. He feels like he does not belong among others who prioritise self-determination over more traditional, pathology-based transgender identification. However, when he associates with other trans people who share his beliefs, this also has a negative effect on his mental health, because he sees himself as surrounded by peers who are in the midst of "alcohol and drugs and self-harm and suicides".

While Darren (L) discussed his experiences with intra-community rejection as having a substantial negative impact on his substance use, Zack (H) on the other hand, discussed his experiences with extra-community rejection as having a similar negative impact on his substance use. Zack conceptualised his connections with other trans people as increasing his substance use and framed this in the context of being rejected from mainstream (cis/het) society. This is an interesting perspective when one considers Zack's experiences in the context of GMS. Within the initial minority stress framework (1995), Meyer draws on societal reaction theory to explain how minority groups can be conceptualised as "deviant" because they depart from society's conventional understandings of appropriate ways of being. According to a critical review of GMS (Tan et al., 2019), cisnormativity (the assumption that it is 'normal' to be cisgender) contributes to why transgender identities may be seen as "deviant" by many Western societies. To take this concept of a "deviant" identity one step further here, some researchers have explored how identification with a "deviant" peer group can influence behaviour, particularly in relation to substance use. For example, one review of literature regarding peer group identification (Sussman et al., 2007) reported that within 37 studies which categorised individuals into 'types' of peer group, the "deviant" peer group were mostly likely to smoke cigarettes, drink alcohol, and take drugs. Throughout all of these studies, individuals who were included in the "deviant" peer group were LGBTQ individuals, people who were rejected from other peer groups, and those who identified themselves as punk/alternative/skater. This mirrors Zack's experiences of being "the queer group [...] the smokers and the people who did drugs [...] the alternative kinda punk crowd". These results suggest that experiences of rejection for transmasculine people, whether intra- or extracommunity rejection, have a substantial impact on substance use, and in Zack's words "if you're already a social outcast [...] it's very easy to do other things that people think are bad as well". Therefore, it is of critical importance to the health and wellbeing of transmasculine individuals, particularly teenagers and young adults, to foster an atmosphere of acceptance and support, both within trans-focused spaces and within society at large.

## 5.4.3. Personal and Psychological Risk Factors in Substance Use

**The Symbiotic Relationship of Mental Health and Substance Use.** Throughout TESUP-1, psychopathological symptoms and substance use were consistently observed as being highly interrelated. Both the global measures of psychopathological symptoms and individual symptom dimensions (as measured by the SCL-R-90) were correlated with high-risk alcohol and drug use, though this relationship was more frequently found in relation to high-risk drug use (See Table 5.6 and 5.7). To add further support to these findings, participants in both the low- and high-risk substance use groups in the interviews indicated that they would use alcohol or drugs to cope with mental health symptoms or gender dysphoria, and that such substance use in turn often worsened their mental and physical health (See Box 5.8).

In this study, psychopathological symptoms were measured using the SCL-90-R, which measures nine symptom dimensions and three global measures of symptoms and distress. The Global Severity Index (GSI; the mean value of the nine symptom dimensions divided by total number of items) is the most commonly used score to assess overall symptomatology and is viewed as the best single indicator of the current level of symptoms (Derogatis et al., 1994). In this study, GSI was significantly related to the use of drugs, but not alcohol or tobacco. Additionally, in the multiple regression analysis it was found that DUDIT score increased by 3.233 points for each point increase in GSI. These results indicate that for transmasculine individuals, GSI is an important indicator of high-risk drug use and would be a useful assessment tool to consider when planning interventions for individuals when multimorbidity is suspected.

There are only a few studies which utilise the SCL-90-R when exploring psychopathological symptoms in transgender individuals. Where it has been used, to the

author's knowledge, it has only been in non-UK populations of transgender people, using validated alternative language versions of the SCL-90-R such as Spanish (Castelo-Branco et al., 2021) and German (Auer et al., 2013). One study that established community sample norms in the UK (Francis et al., 1990) did not explicitly include transgender people, nor did another study which explored the nine symptom dimensions in a US sample with comorbid psychiatric diagnoses and SUD (Zack et al., 1998). The mean SCL-90-R scores from the present study will be compared to these studies to explore the psychopathology of transmasculine individuals in the UK (see Appendix F for means and standard deviations for the SCL-90-R in each of these studies).

When comparing the SCL-90-R results to the four studies in Appendix F it is immediately apparent that these results are substantially higher than both the Spanish FTM, German FTM, and British male community samples, and are much more similar to the US outpatient comorbid SUD sample. Initially it was hypothesised that the results would be similar to other samples of transmasculine individuals because not all of the sample in this study would report a SUD. While 49.5% and 43.8% of the sample met the cut-off score indicating high-risk alcohol use and drug use respectively, the SCL-90-R scores for the entire sample are comparable to scores in Zack et al.'s (1998) study in which all participants had a diagnosis of SUD and were attending an outpatient mental health unit.

Zack et al.'s (1998) exploratory factor analysis of the SCL-90-R found that the measure primarily represents general psychiatric distress and provides little information regarding specific symptom patterns in non-normative samples. Furthermore, the factor analysis identified three higher-order factors, one of which was Hostility and Suspiciousness; this factor comprised questions from the HOS, PAR, and PSY dimensions. The present study lends support to these findings, particularly because only three dimensions were positively correlated with both high-risk alcohol and drug use, and these were HOS, PAR, and PSY.

This may indicate that participants were coping with distressing symptoms of excessive anger, paranoid ideations, and psychosis by self-medicating using alcohol and drugs (Broman et al., 2019). On the other hand, it is possible that the increased symptoms in HOS, PAR, and PSY were as a result of high-risk alcohol and drug use, for example, as a result of substanceinduced psychosis (Tandon & Shariff, 2019). Longitudinal analyses which focus on the transgender community are necessary to begin to delineate between these possibilities

Some participants identified substance use as being intertwined with their mental health, and two participants in particular, Darren (L) and Zack (H), went into detail about how substance use is connected to self-harm (or non-suicidal self-injury; NSSI). Some common functions of NSSI are to regulate negative emotions or inflict self-punishment (Cipriano et al., 2019). In extracts from Darren and Zack their motivation to engage in substance use is comparable to the motivations to engage in NSSI, as mentioned above. For example, for Darren (L) drinking alcohol was a less harmful method of regulating negative emotions when his alternative was NSSI; both substance use and NSSI were means of coping with distress and symptoms of poor mental health, but he perceived the former to be less damaging. On the other hand, for Zack (H) it was specifically the damaging nature of substances that he sought out as a method of self-punishment. Zack's motivations, however, were more complex. He reminisced about the first time he smoked a cigarette, age 12, highlighting the pursuit of masculinity which provided him with gender euphoria, while simultaneously stating that it was "another form of self-harm". These extracts illustrate that while substance use and NSSI functioned as a method of regulating negative emotions, substance use (and perhaps NSSI by extension) was sometimes complicated by the paradoxical motivations of self-punishment and pursuing masculinity and gender euphoria. This suggests that while many participants were using substances to regulate negative emotions, the reasons behind substance use in transmasculine people are more complicated

and are entangled in social expectations of masculinity, gender dysphoria, and euphoria. It is recommended that culturally competent and gender-affirming healthcare and substance use interventions are provided to this population, with a particular focus on the psychosocial context of their lived experiences related to substance use and coping with negative events and emotions.

When interpreting the symptom dimensions, it is important to be aware of the differences between transgender and cisgender experiences of both psychopathological symptoms and experiences of healthcare. The PAR dimension was designed to be representative of symptoms of paranoid ideation which commonly occur in psychiatric conditions such as BPD, PTSD, and psychotic spectrum disorders (Marwaha et al., 2014). It has been reported that transgender individuals have a greater than 20-fold incidence of personality disorder diagnosis and a greater than 3-fold incidence of psychotic spectrum disorder diagnosis, compared to a cisgender control group (Hanna et al., 2019). However, it is important to note that perceived and actual discrimination are associated with both paranoid and psychotic symptoms (Pearce et al., 2019). It is crucial to take this into consideration in the present study, as the GMSR subscales of gender-related rejection and victimisation were positively correlated with high-risk drug use. This provides support for Pearce et al. (2019) and suggests that previous experiences of gender-related rejection and victimisation have a negative impact on mental health and substance use.

A recent literature review on psychosis in transgender individuals (Barr et al., 2021) cites diagnostic biases as a potential reason for the increased rate of psychosis diagnoses in this community. Referencing the historical conflation of transness to psychopathology, and the pathologisation of normal reactions to violence and oppression, the authors highlight the need for culturally sensitive and gender-affirming interventions. Barr et al. also note that symptoms of psychotic spectrum disorders are not always subjectively distressing which may

be relevant for the participants in this study. The Likert scale which measures symptom distress intensity ranged from 0 (not at all) to 4 (extremely) and according to Derogatis (1994) a score under 1 indicates the absence of substantial psychiatric distress. The HOS, PAR, and PSY means in this study ranged from 1.17 to 1.21, indicating a low level of distress that is only slightly past the clinically significant cut-off, which suggests that although these symptoms were associated with high-risk alcohol and drug use, participants did not necessarily perceive these symptoms as distressing. This may be because their efforts to use substances to lower their distress did effectively reduce the intensity of symptoms.

Distress (In)tolerance and Emotional Regulation. Many of the participants in the qualitative interviews across both low- and high-risk groups said that they did not know how to cope with stressful situations and negative emotions in a healthy or productive manner. As mentioned in the previous section (5.4.3.1), a common reason for engaging in substance use is to regulate negative emotions. Numerous studies have found that intolerance of emotional and physical distress is a primary motivation underlying the development and maintenance of SUD, and that the ability to tolerate distress is changeable depending on the personal and social context (Otto et al., 2005; Leyro et al., 2010; Veilleux et al., 2022). Distress (in)tolerance has been conceptualised as hierarchical in nature, consisting of the higher-order construct of Global Experiential Distress (In)tolerance and five related lower-order dimensions: a) tolerance of uncertainty, b) tolerance of ambiguity, c) tolerance of frustration, d) tolerance of negative emotion, and e) tolerance of physical discomfort (Leyro et al., 2010). The quantitative and qualitative results of TESUP-1 will now be discussed in the context of this hierarchical concept of distress (in)tolerance.

Participants frequently referenced struggles to tolerate uncertainty, ambiguity, and frustration, often in the context of access to transition-related healthcare. The difficulties in accessing appropriate healthcare were compounded by the lack of healthcare provider

knowledge and uncertainty about legal aspects of transition, such as changing name and gender marker. For some participants, the inability to tolerate these confusing and stressful situations prompted an urge to cope by using substances. Previous studies have demonstrated the relationship between low distress tolerance and substance use, for example, Özdel and Ekinci (2014) reported that substance-dependent participants scored significantly lower on a measure of distress tolerance, and significantly higher in a measure of depression and anxiety, compared to a healthy control group. Although the present study did not include a measure of distress (in)tolerance, the correlation analyses revealed that depression and anxiety had a moderate positive correlation with high-risk drug use, but not alcohol use. It is interesting that the participants who reported using alcohol to cope with distress intolerance were all in the low-risk group; this may indicate that though these participants may turn to alcohol as a maladaptive coping mechanism, they are not at high-risk of developing hazardous use or dependence on alcohol. However, this may not be the case for those who use drugs to cope with distress.

The difference between whether the participants chose to cope with distress using alcohol or drugs may be related to the type of distress they were struggling with. The participants mentioned in the previous paragraph were all low-risk and used alcohol to cope with frustration, ambiguity, and uncertainty, whereas the participants who turned to drugs were typically (but not always) in the high-risk group and more often reported previous trauma related to substance use, physical discomfort, and an inability to regulate negative emotions. For example, Simon (H) reflected on how witnessing his father overdose on heroin was the event that triggered his current struggles with alcohol and cocaine use. This agrees with previous literature which has indicated that exposure to parental SUDs is predictive of the development of SUD in adolescence and early adulthood (Biederman et al., 2000). An inability to tolerate physical discomfort was also noted among some of the participants, for

152

example, Bailey (L) suffered with migraines and would routinely combine opiate painkillers with alcohol "just to pass out for a few hours". In Bailey's description, it is clear that it was not just struggles to tolerate the physical discomfort but also the resulting negative emotions that prompted seeking relief in alcohol and drugs.

In the high-risk group, gender dysphoria was a frequently mentioned negative experience which combined intolerance of negative emotions and physical discomfort, and consequently resulted in substance use. Notably, gender dysphoria typically arose in the context of intimate and sexual contact. Logan (H) recalled that before he was on testosterone, he would often drink alone to cope with intense feelings of gender dysphoria and would go clubbing to "hook up with someone" and achieve validation of his gender identity. Similarly, Owen (H) routinely drank alcohol in order to cope with dysphoria, again in the context of intimacy and sexual relationships. In Connolly and Gilchrist's (2020) systematic review they note that gender dysphoria is a risk factor for the development and maintenance of SUDs. Very few studies focus on the relationship between substance use and intimacy for transgender people. Those that have explored this topic focus solely on the association between transgender people who participate in sex work and high-risk substance use (Keuroghlian et al., 2015; Scheim et al., 2017). The results from the present study indicate a critical gap in the research about the intimate and sexual lives of transgender people and how dysphoria may be related to increased substance use. Future research is necessary in order to explore lived experiences of sex and substance use, in order to tailor appropriate interventions.

The results indicate that distress intolerance is a common difficulty facing the transmasculine community, but evidence-based interventions for distress intolerance in the context of SUD are limited. A recent systematic review (Warner & Murphy, 2021) offers promising preliminary support for dialectical behavioural therapy skills training (DBT-ST) in

the treatment of SUD. However, there have been few studies which involve randomised controlled trials comparing this intervention to alternatives; additionally, there is no literature which indicates that transgender people would benefit from this intervention. Another study (Muhomba et al., 2017) which focused on DBT-ST for university students presenting with serious psychological concerns (such as substance use and NSSI), found that emotional regulation and dysfunctional coping skills decreased, and adaptive coping skills increased, after 7-10 weeks of DBT-ST. Again, this research was limited by the lack of a control group, and the inability to measure treatment efficacy over time, however the results indicate the potential benefits of this intervention. Given the difficulties that the participants in the current study were facing (such as NSSI, high-risk substance use, emotional dysregulation, and dysfunctional coping skills) it is recommended that future research should focus on identifying whether DBT-ST would be efficacious for the transgender community. Group skills training is not only a financially beneficial method of healthcare provision, but access to community-level psychological interventions can decrease GMS through strengthening individual- and community-level resilience, which in turn, improves mental health and quality of life (Meyer & Frost, 2013).

**Personality Traits.** The findings of this study indicated that various personality traits were significantly associated with high-risk substance use. Most notably, novelty-seeking (NS) significantly predicted current tobacco use and high-risk drug and alcohol use. As discussed previously statistical analyses revealed that NS contributed to an increase in high-risk alcohol and drug use, which confirmed initial hypotheses; however, the finding that high levels of NS was associated with a decreased likelihood of current tobacco use was unexpected (for a full discussion of this result please see section *5.4.3.2*).

The overall personality profile of the participants in TESUP-1 was comparable to the limited number of published studies which focus on transgender individuals. Miyajima et al.

(2014) examined differences in personality traits between trans men and trans women, and their findings indicated that trans men were higher in cooperativeness (CO) and lower in selftranscendence (ST). The present study had comparable results to that of Miyajima et al., though their sample had higher results in self-directedness (SD) and CO. This could indicate cultural differences between the UK and Japan; however, it may also be representative of the effect of substance use in the current sample. Though there was no association between SD and substance use, there was a significant negative correlation between CO and both highrisk alcohol and drug use. Furthermore, a multiple regression analysis revealed that CO significantly predicted alcohol use status, and the AUDIT score decreased by .155 for each point increase in CO. This result provides support for studies which have focused on personality profiles of people who use substances which indicate that low CO is associated with increased alcohol use (Amau et al., 2008; Foulds et al., 2016). It is also possible that the association between CO and substance use was simultaneously related to GMS. The CO scale represents how well an individual relates to and accepts other people; in this study, genderrelated rejection and victimisation were positively associated with substance use. Additionally, some participants in the interviews reported feeling rejected from society and having experienced discrimination in healthcare settings which may contribute to a lower propensity for cooperativeness and an increase in substance use. Further investigation is required to explore whether GMS might mediate the association of personality traits, such as CO, with high-risk substance use.

In order to determine whether there was a distinct personality profile for those engaged in high-risk substance use compared to low-risk substance use, independent sample *t*-tests were conducted. The results indicated that high-risk substance users had statistically significantly higher NS and ST, and lower HA and CO, compared to the low-risk group (see Table 5.14). These results are comparable to previous studies, for example, Le Bon et al. (2004) explored the differences in personality profiles between people with heroin dependence, alcohol dependence, and a control group. Le Bon et al.'s results indicated that NS and ST were significantly higher in both of the substance dependent groups. However, the results of the current study differ from that of Le Bon et al. in the dimensions of HA and CO; in the present study these dimensions were both significantly lower in the high-risk group, whereas in Le Bon et al.'s study HA was significantly higher in substance dependent groups compared to the control group, and there was no significant difference in CO. More recent studies have also indicated that high levels of NS and ST are associated with SUD (Kotov et al., 2010; Foulds et al., 2017; Hashemi et al., 2019) which aligns with the findings of the present study. However, Hashemi et al. (2004). When Cloninger developed the TCI measure, he theorised that a pattern of high NS and low HA, RD, and PS would predict SUD; the present study provides partial support for this theory in that high NS and low HA was indicative of high-risk substance use.

Table 5.14 – Independent samples <i>t</i> -test exploring the	difference between low- and high-risk SUD on the TCI-140
---	--

	Low-Risk SUD		High-Ri	isk SUD	Independent Samples t-test					
	М	SD	М	SD	df	t	р	Cohen's d		
NS	56.30	10.141	63.13	8.146	103	-3.391	.001**	.742		
HA	75.86	13.843	66.84	14.617	103	2.968	.004**	.633		
RD	66.18	11.858	62.51	9.396	103	1.615	.110	.343		
$\mathbf{PS}^{\mathrm{a}}$	60.09	15.017	62.37	9.971	95.55	837	.405	.178		
$SD^{a}$	60.02	14.333	55.91	8.717	91.73	1.617	.110	.346		
CO	75.62	10.734	65.95	10.768	103	4.218	.000**	.899		
ST	40.46	9.696	47.31	9.441	103	-3.309	.001**	.715		

<sup>a</sup> Welch test statistic is reported where Levene's test indicated that the homogeneity of variances assumption was violated for this variable. \*\* Correlation is significant at the 0.01 level.

> The interviews with participants in this study illuminate some of the differences between the quantitative findings of this study compared to earlier research. For example, Zack (H) used alcohol and drugs as a method of NSSI which may suggest a lack of concern about the consequences of potential harm involved in substance use. Throughout Zack's

interview it is clear that he is sociable, comfortable with taking risks, and sensation seeking, all of which are indicators of high NS and low HA. Other participants such as Parker (H) and Rowan (H) reported impulsively using substances to manage feelings of distress or symptoms of ADHD, demonstrating interactions between impulsive substance use (indicative of NS) and struggling with the anticipation of negative emotional or psychological states (indicative of HA). Additionally, it was *only* participants in the high-risk group who reported conceptualising themselves as being impulsive, and participants such as Gale (H) and Simon (H) discussed how their impulse control was worse when they were under the influence of substances. Despite some participants taking measures to minimise harm when engaging in substance use, the draw of new experiences or heightened enjoyment of events when using alcohol or drugs was often enough for them to impulsively engage in risky activities related to increased substance use.

## 5.4.4. Protective Factors in Substance Use

**Personal and Community-Level Resilience**. The previous section (*5.4.4.3*) discussed how the personality trait of CO significantly contributed to lower AUDIT scores and highlighted a possible connection between this dimension and GMS. It is important to note that CO is a character dimension; the character dimensions of the TCI (SD, CO, and ST) are held to be rooted in socialisation, culture, and personal experiences, and are therefore changeable across the lifespan (Cloninger et al., 1993). This is in contrast to the temperament dimensions which are thought to be more stable over time (Cloninger et al., 2019). Since the domain of CO is thought to be moderately changeable, there is the possibility of developing psychosocial interventions in order to increase CO and, consequently, reduce substance use. There are some preliminary studies which have attempted to enhance character dimensions in order to achieve better outcomes. For example, Cloninger (2004) reports that therapeutic interventions which focus on enhancing CO can increase happiness and wellbeing.

Additionally, intervention studies have demonstrated the potential of mindfulness-based meditation (Crescentini et al., 2018) and person-centred psychotherapy (Cloninger & Cloninger, 2016). However, it is crucial that any intervention offered to transgender clients is gender affirmative and considers the role of gender minority stress.

It was initially hypothesised that the GMS subscale of identity pride (P) would act as a protective factor against substance use, based on the proposed mechanism of resilience in the GMS theory (Testa et al., 2015). However, quantitative analyses revealed that P was significantly correlated to high-risk drug use. Bockting et al. (2019) reported that individuals who have high levels of identity pride may be reacting to stigma and discrimination by increasing their connections to the transgender community. It is notable that individuals with high identity pride may be more likely to disclose their trans status to others, and therefore may be at higher risk of experiencing discrimination, rejection, and victimisation (Breslow et al., 2015). Additionally, Nuttbrock et al. (2015) indicated that transgender individuals may re-conceptualise non-conformity or "social deviancy" as a positive aspect of their identity. Some studies have suggested that identity pride (Bockting et al., 2013) or conceptualising non-conforming identities (Barr et al., 2016) as a positive can buffer the negative effects of discrimination and GMS. However, the current study indicates that individuals who associate with a "deviant" peer group, have high levels of identity pride, or frequently participant in LGBTQ events, have a higher risk of engaging in substance use. The findings of this study contradict previous research, therefore particular emphasis was placed on investigating the relationship between identity pride and substance use in TESUP-2 and will be discussed in more detail in chapter six.

Although the GMS subscale of CC was not found to be related to substance use, the qualitative results indicated some participants did believe that attending "queer events" or associating with other transgender people tended to increase their substance use. In contrast

to this, some participants conceptualised their relationship to family and friends who do not drink or take drugs to be a protective factor against high-risk substance use. It is notable that five of the six interview participants in the low-risk group reported that their friends, partners, and family didn't often drink alcohol or use drugs. These findings provide support for previous studies into the positive effect of peers on substance use. Coyle et al. (2015) demonstrated that positive peer influence (such as abstinence and supportive friendships) was a significant predictor of lower substance use and better mental health. Additionally, Beard and Wolff (2020) explored the role of positive influence of peers on substance use in university-age adults and found that positive peer affiliation was related to lower nicotine and drug use, and that it moderated the relationship between sensation-seeking and substance use. As sensation-seeking is highly related to novelty seeking (NS), Beard and Wolff's (2020) findings, together with the results of the present study which indicated that NS significantly contributed to tobacco, alcohol, and drug use, have important implications for potential interventions into high-risk substance use.

Harm Avoidance. In the interviews, many participants described their experiences with harm avoidance in the context of the subdomain HA in the TCI, in addition to the more general measures they take to reduce the risk of harm when they are engaged in substance use. Typically, participants in the high-risk group described the methods they used to mitigate potential physical or emotional harm as a result of substance use. Participants in the low-risk group tended to discuss health concerns or previous negative experiences which led to them stopping using substances. These findings align with the quantitative results which indicated that participants in the low-risk group had higher levels of HA compared to the high-risk group. A high score in the domain of HA relates to fear of uncertainty, shyness, excessive worry, and being easily fatigued; all of these aspects can be conceptualised as features of anxiety (Markett et al., 2016).

Two participants discussed the impact that parental substance use had on their own beliefs and behaviours regarding substance use. Both participants (Aaron, L; Simon, H) disclosed that their fathers' problematic substance use had made them afraid of using substances, because they wanted to avoid becoming like their fathers. It is notable that although both discussed wanting to avoid substance use due to their family history, it was only Aaron who limited his use. Simon reported that he "used to be proper against drugs" until he witnessed his father overdose on heroin, which led to his current high-risk use of alcohol and cocaine. There is a growing body of evidence that has reported that adverse childhood experiences (such as abuse, parental substance use, and domestic violence) are associated with negative outcomes in adult life, including poor mental and physical health, and substance use (Choi et al., 2017; Hughes et al., 2017).

Participants often discussed how negative side-effects of alcohol and drug use, such as feeling unwell, embarrassment, and risks to health, all contributed to their decision to reduce or quit using substances. For Simon (H), it was after getting an infection in his nose from frequent cocaine use that he decided to seek help and began attending an alcohol and drug recovery service. The reported reasons for reducing or quitting substance use largely aligns with previous research which has indicated that common motivations to stop using substances include health concerns (Smith et al., 2010) and recent negative experiences, often described as "hitting rock bottom" (Stokes et al., 2018). Though no participants reported explicitly that transition-related motivations prompted abstinence from substance use, some did mention that when they began medically transitioning their motivation to use substances changed from being a coping method to being a way to enjoy or enhance socialising. This provides some support for previous research which has indicated that, for those who desire it, medical transition improves quality of life and mental health (Baker et al., 2021). Participants in the high-risk group discussed their methods of reducing harm whilst using substances. Being aware of potential negative effects on physical health and taking measures to mitigate this risk was commonly reported. Gale (H) in particular went into detail about staying alert to health risks when using alcohol and drugs (stimulants) at the same time. They expressed a high level of concern about their own wellbeing and the wellbeing of their friends and discussed methods of risk management such as buying drugs for the group from the same person so that they all experienced the same effects. Throughout Gale's interview it was clear that they put a lot of thought into their substance use, such as planning when to take specific substances to achieve the desired effects. Although Gale was in the high-risk group, they were the only participant who reported that they did not use substances in order to cope with difficult situations or manage mental health symptoms, and overall, they displayed multiple distress tolerance skills. These experiences, though only reported by one participant, suggests that a potential relationship between distress intolerance and harm avoidance may exist in the context of substance use and this warrants further investigation.

#### **5.5.** Conclusions

The results from TESUP-1 indicate that high-risk substance use in the transmasculine community is a prevalent issue; over half of the sample engaged in hazardous or harmful alcohol use, and just under half engaged in risky drug use. Additionally, interviews with a nested sample of participants indicated that for some individuals who did not currently engage in substance use, they had previously experienced substance use-related issues, and continued to feel the urge to use alcohol or drugs.

No specific stage of transition was related to increased substance use, however, difficulties in the process of transition might trigger an increase in substance use. It is important to note that improvements are necessary in the operationalisation of the concept of 'stage of transition'. In order to accurately investigate the issues affecting transmasculine individuals at different stages of transition, and to begin to formulate appropriate support, it is recommended that participatory research is conducted with the transmasculine community. It is noted that this is an ethical method of engaging marginalised or oppressed populations in research (Strang et al., 2019).

The role of masculinities, and how they intersected with a transgender identity, was frequently mentioned by interview participants. The concept of compensatory masculinity was a particularly important factor in understanding high rates of substance use. Participants felt acute pressure to always appear masculine and to 'pass' as male to others. This was not only due to personal preference but also an avoidance of potential danger that they might face due to their transgender identity. This study raised some questions about the validity of using the GRCS with the transmasculine population. First, I contend that transmasculine individuals have a unique experience of gender socialisation which is inherently disparate from cisgender socialisation experiences. Second, participant feedback indicated that many of the questions did not apply to non-binary individuals, and that their responses would change depending on the social context, particularly regarding if they were disclosing their trans identity or not. These issues call into question the appropriateness of the GRCS and suggest that a separate version of the measure should be developed and validated for use with transmasculine individuals.

Difficulties in accessing and navigating transition-related healthcare were cited as having a considerable influence on urges to engage in substance use. Quantitative results revealed that 65.7% of participants had experienced difficulties in accessing medical or mental health treatment due to being trans. It was frequently reported that healthcare professionals lacked trans-specific knowledge and experience and many participants were unsatisfied with the care provided by NHS GIS. Long waiting times and a feeling of having to 'prove' their trans identity or 'jump through hoops' to access healthcare were reported by most of the participants who were trying to access NHS transition-related services. Some participants expressed a lack of trust in GIS and voiced concerns about the current diagnostic system. The resulting stress from poor access to GIS and a lack of communication from healthcare professionals are likely to have contributed to increased substance use.

Peer pressure was often described as being heightened in importance due to their transgender status; a desire to be seen as cisgender or be seen as 'normal' meant that some participants would engage in high-risk substance use. Additionally, substance use was often connected to being transgender due to some participants getting drunk to 'come out' as trans to others, attending LGBTQ events which are often centred around alcohol, and pride in one's transgender identity. The hypothesised protective factor of community connectedness was not supported, and the qualitative results indicated that both socialising with other transgender people and being rejected by other transgender people contributed to an increase in substance use. Further research is necessary to explore the roles of identity pride and community connectedness in substance use in transgender communities, with particular focus on online/offline communities and how schisms in various trans communities contribute to isolation, seeking support, and substance use.

Psychopathological symptoms were consistently associated with high-risk substance use. Though it is worth noting that despite participants indicating a high number of symptoms, their distress regarding these symptoms was relatively low. It is important to take into consideration the impact of GMS because some symptoms may be a normal reaction to traumatic experiences. Some participants reported that they used substances as a distress tolerance method, though there were paradoxical accounts of conceptualising substance use both as a form of self-harm and a method of achieving validation of a masculine gender identity. These results indicate that the relationship between substance use and psychopathological symptoms is complex and entangled with gender identity and social expectations. For this reason, it is critical that any interventions gender-affirming and consider the psychosocial context of their lived experiences. Furthermore, due to the high occurrence of distress intolerance in this study, it is suggested that DBT-ST is likely a cost-effective and efficacious method of helping this community regulate emotions, reduce distress, and build community connections and individual-level resilience.

Personality traits were revealed to be a significant factor which contributed to highrisk substance use. There are only a few studies which investigate the personality profiles of transgender individuals, and to my knowledge, none which focus on substance use in the transgender population. This novel finding of the significance of personality traits in transmasculine substance use warrants further investigation. In particular, it would be useful to explore whether therapeutic interventions which focus on changing character traits, such as mindfulness-based meditation and person-centred psychotherapy, could be efficacious in reducing substance use.

Various protective factors were demonstrated to be related to a reduction or cessation of substance use; a high HA score, previous negative experiences of substance use, and health concerns were the most common factors. Participants who felt supported by their partners, family, or friends also reported low-risk substance use. Some participants who were categorised as engaging in high-risk substance use took measures to reduce risks to themselves and others when they were intoxicated. Although it was not indicated that transition-related motivations prompted a reduction in substance use, some participants described how beginning to medically transition led to an increase in self-esteem and overall wellbeing and led to more positive motivations for using substances.

### **6. TESUP-2**

#### **6.1. Descriptive Statistics**

# 6.1.1. Demographics

The age of the 82 participants ranged between 18–54 years (M = 26.17, SD = 6.21). 34.1% (N = 28) of the sample identified their gender identity to be non-binary which included identities such as transmasculine, genderqueer, agender, non-binary, and genderless. The remaining 65.9% (N = 54) identified as male. The largest proportion of participants considered themselves to be mid transition (39%, N = 32), although 2.4% (N = 2) declined to answer this question. The most common sexual orientation was bisexual (N = 27), and the majority of the sample identified as White British (N = 75, 91.5%).

Completed level of education was assessed using the established levels of educational attainment in the UK (Gov.uk, 2022) which range from 0 to 8. Levels in this sample ranged from 0 (no completed education) to level 7 (postgraduate master's level). Over half of the sample (62.2%, N = 51) reported completing either an undergraduate or postgraduate degree or certificate. The majority of participants were in paid employment (65.9%, N = 54). Please see Table 3.3 for quantitative demographics variables.

An additional question was added to this second questionnaire regarding disability. 74.4% (N = 61) reported that they did not have any disabilities or health conditions. The remaining 25.6% (N = 21) of the sample reported having a disability or health condition that impacts their daily life. Of these 21 participants, 57.1% (N = 12) had multiple health conditions or disabilities. The most frequently mentioned condition was autism spectrum disorder (N = 10), followed by psychiatric disorders (N = 8), followed by long-term health conditions (N = 10). **Tobacco Use.** Use of tobacco was determined through questions relating to currentand past-tobacco use. The majority (73.2%, N = 60) of the sample indicated that they had used tobacco in the past. Just over half of the sample (52.4%, N = 43) reported current use of tobacco. Only 12 participants (N = 14%) reported never having used tobacco. Please see Table 3.3 in Chapter 3 for quantitative tobacco use data.

Of those who reported current tobacco use (N = 43), the most frequently reported method of tobacco use was smoking manufactured cigarettes (N = 25), followed by using a tobacco pipe (N = 10), and smoking hand-rolled cigarettes (N = 6). The frequency of use ranged from 1 - 15 times per day, the most common frequency of use was between 1-10times per day (N = 41) with only two participants reporting a frequency of use greater than 10 times per day, both of which were fifteen times per day.

Alcohol Use. Use of alcohol was determined through administering the AUDIT (Babor et al., 1992). A cut-off score of 8 or over is suggestive of harmful and/or hazardous drinking habits and a cut-off score of 20 or over indicates that the individual may be at risk of, or are experiencing, alcohol dependence (Babor et al., 1992). For this study's sample, the mean score on the AUDIT was 11.96 (SD = 8.81), with scores ranging from 0–27 points. Over half of the sample met or exceeded the cut-off score of 8 points (56.01%, N = 46), and over a quarter of the sample met or exceeded the cut-off score of 20 points (26.8%, N = 22). Please see Table 3.4 for means and standard deviations of AUDIT.

**Drug Use.** Use of drugs was determined through administering the DUDIT (Berman et al., 2005). A cut-off score of 6 or more points is recommended to identify those with drug-related problems (Berman et al., 2005). For this study's sample, the mean score on the DUDIT was 10.98 (SD = 11.55), with a range of 0 - 30 points. Over half of the sample

(51.2%, N = 42) met or exceeded the cut-off score of 6 points. Please see Table 3.4 for means and standard deviations of DUDIT.

Participants were asked to indicate which drugs they usually take and had the option to select multiple drugs. A total of 42 participants indicated at least one drug that they typically used. The most commonly used drug was cannabis (46.3%, N = 38). Drugs also indicated in free-text comments but not included in the questionnaire were Valium and prescription painkillers taken recreationally. It is interesting to note that where heroin was the only drug not selected in TESUP-1, whereas in TESUP-2, four participants indicated that they use heroin. Please see Table 3.5 for frequencies of drug type used.

# 6.1.2. Correlations

Please refer back to each scales' description in the Methods chapter (*section 3.4.2. TESUP Materials*) for definitions of the abbreviations in each scale. Please see Appendix K for additional information and scattergrams of all correlations.

**Demographics.** Table 6.1 shows the correlations between demographic information (age, gender, education level), transition stage, and substance use (tobacco, alcohol, drugs) on data collected from 82 transmasculine individuals.

There were no correlations between any of these variables and the use of tobacco, alcohol, or drugs. Unsurprisingly, there was a significant correlation between level of education and age ( $\rho(82) = .299$ , p = .007, 95% CI [.064, .518]).

Interestingly, the use of tobacco was negatively correlated with both the use of alcohol (r(82) = -.320, p = .003, 95% CI [-.507, -.139]) and the use of drugs (r(82) = -.419, p

<.001, 95% CI [-.551, -.275]). Additionally, the use of alcohol had a very strong positive

correlation to the use of drugs (r(82) = .830, p < .001, 95% CI [.726, .905]).

correlations with confidence intervals										
Variable	М	SD	1	2	3	4	5	6	7	
1. Age ( <i>r</i> )	26.17	6.218								
2. Gender binary $(r_{pb})$			.042							
3. Transition Scale ( <i>r</i> )	3.36	.903	[142, .288] .189	.017						
4. Education (ρ)	5.04	1.666	[094, .416] .299*	[202, .240] 008	.108					
5. Tobacco score (r)	4.35	1.201	[.064, .518] 062	[237, .219] .027	[144, .345] .096	.152				
6. AUDIT score ( <i>r</i> )	11.96	8.814	[259, .144] 112	[178, .231] .059	[154, .325] .154	[106, .379] .106	320**			
7. DUDIT score ( <i>r</i> )	10.98	11.554	[314, .127] 034	[167, .283] .055	[061, .359] .136	[096, .307] .088	[507, 139] 419**	.830**		
			[233, .204]	[171, .284]	[050, .307]	[112, .295]	[551, 275]	[.726, .905]		

Table 6.1. Demographics, stage of transition, and substance use correlations. Means, standard deviations, and

*Note.* M and SD represent the mean and standard deviation, respectively. R and  $\rho$  indicate Pearson's correlation and Spearman's Rank correlation, respectively.  $r_{pb}$  indicates point-biserial correlation.

Values in square brackets are the 95% confidence interval for each correlation.

\* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

Quality of Life. Table 6.2 shows correlations between the four domains of the WHOQOL-BREF and substance use (tobacco, alcohol, drugs).

Results demonstrated a weak negative relationship between physical health QoL and

alcohol use (r(82) = -.226, p = .041, 95% CI [-.413, -.033]) as well as drug use (r(82) = -.226, p = .041, 95% CI [-.413, -.033])

.221, p = .046, 95% CI [-.402, -.054]). There was an unexpected finding of a moderate

positive relationship between tobacco use and environmental QoL (r(82) = .385, p < .001,

95% CI [.187, .545]).

Variable	М	SD	1	2	3	4	5	6	7
1. Physical Health QoL ( <i>r</i> )	13.26	2.622							
2. Psychological Health QoL ( <i>r</i> )	11.68	2.738	.562** [.355,						
3. Social and relationships QoL $(r)$	13.60	3.121	.706] .281*	.458**					
4. Environment QoL ( <i>r</i> )	13.54	2.474	[.037, .528] .631**	[.251, .635] .571**	.303**				
5. Tobacco score ( <i>r</i> )	4.35	1.201	[.454, .768] .150	[.353, .729] .174	[.098, .502] .154	.385**			
6. AUDIT score ( <i>r</i> )	11.96	8.814	[075, .366] 226*	[047, .369] .193	[122, .417] 054	[.187, .545] 040	320**		
7. DUDIT score ( <i>r</i> )	10.98	11.554	[413, 033] 221*	[.023, .371] .207	[243, .130] 030	[255, .149] 098	[507, 139] 419**	.830**	
			[402, 054]	[.038, .388]	[212, .161]	[276, .073]	[551, 275]	[.726, .905]	

Table 6.2. Quality of Life and substance use correlations. Means, standard deviations, and correlations with confidence intervals

*Note. M* and *SD* represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation. \* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

**Gender Role Conflict.** Table 6.3 shows correlations between the four factors of the GRCS and substance use (tobacco, alcohol, drugs). Tobacco was not significantly correlated to any GRC factors.

Results showed that both alcohol and drug use were correlated with three of the four factors: SPC, RE, RABBM. Concerning the SPC factor, alcohol use had a moderate positive correlation (r(82) = .371, p = .001, 95% CI [.200, .541]) and drug use had a weak positive correlation (r(82) = .280, p = .013, 95% CI [.098, .460]). Concerning the RE factor, there was a weak positive correlation with both alcohol use (r(82) = .238, p = .036, 95% CI [.031, .434]) and drug use (r(82) = .235, p = .039, 95% CI [.051, .433]). Finally, the RABBM factor had a moderate correlation with both alcohol use (r(82) = .451, p < .001, 95% CI [.282, .618]) and drug use (r(82) = .492, p < .001, 95% CI [.320, .652]).

Variable	М	SD	1	2	3	4	5	6	7
1. SPC ( <i>r</i> )	44.46	11.597							
2. RE ( <i>r</i> )	37.14	10.679	.372**						
3. RABBM ( <i>r</i> )	25.28	8.626	[.106, .587] .570**	.701**					
4. CBWFR ( <i>r</i> )	20.47	5.242	[.365, .735] .496**	[.587, .800] .235*	.289*				
5. Tobacco score ( <i>r</i> )	4.35	1.201	[.289, .672] 043	[041, .509] 209	[.066, .515] 179	031			
6. AUDIT score ( <i>r</i> )	11.96	8.814	[301, .192] .371**	[469, .062] .238*	[416, .069] .451**	[279, .198] .214	320**		
7. DUDIT score ( <i>r</i> )	10.98	11.554	[.200, .541] .280*	[.031, .434] .235*	[.282, .618] .492**	[.029, .400] .132	[507, - .139] 419**	.830**	
			[.098, .460]	[.051, .433]	[.320, .652]	[077, .355]	[551, - .275]	[.726, .905]	

Table 6.3. Gender Role Conflict Scale and substance use correlations. Means, standard deviations, and correlations with confidence intervals

*Note. M* and *SD* represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation. \* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

**Gender Minority Stress and Resilience.** For reasons of available space Table 6.4 shows correlations between the nine constructs of the GMSR and substance use (tobacco, alcohol, drugs) and Table 6.5 shows the respective confidence intervals. The constructs of ND and CC were found to have no significant associations with the use of any substance.

Tobacco use had weak negative correlations with three constructs: the construct V (r(82) = -.280, p = .011, 95% CI [-.477, -.038]), the construct R (r(82) = -.256, p = .019, 95% CI [-.499, -.006]), and the construct D (r(82) = -.223, p = .044, 95% CI [-.418, -.025]).

Alcohol use had a weak positive correlation with two constructs: the construct D (r(82) = .244, p = .044, 95% CI [.074, .464]), and the construct R (r(82) = .258, p = .019, 95% CI [.093, .445]). The use of alcohol also had a moderate positive correlation with four constructs: the construct NE (r(82) = .311, p = .004, 95% CI [.115, .508]), the construct P
(r(82) = .359, p = .001, 95% CI [.190, .528]), the construct IT (r(82) = .363, p = .001, 95% CI [.209, .512]), and the construct V (r(82) = .382, p < .001, 95% CI [.189, 598]).

Drug use had a weak positive correlation with five constructs: the construct NA (r(82) = .220, p = .047, 95% CI [.038, .404]), the construct IT (r(82) = .260, p = .018, 95% CI [.090, .428]), the construct R (r(82) = .260, p = .018, 95% CI [.080, 451]), the construct NE (r(82) = .285, p = .009, 95% CI [.098, .478]), and the construct D (r(82) = .292, p = .008, 95% CI [.103, .519]). The use of drugs also had a moderate positive correlation with two constructs: the construct P (r(82) = .390, p < .001, 95% CI [.224, .544]), and the construct V (r(82) = .452, p < .001, 95% CI [.281, .652]).

Variable	М	SD	1	2	3	4	5	6	7	8	9	10	11	12
1. Discrimination ( <i>r</i> )	4.29	2.052												
2. Rejection $(r)$	5.32	2.863	.692**											
3. Victimisation ( <i>r</i> )	4.23	2.856	.557**	.704**										
4. Non-affirmation $(r)$	12.77	6.632	.216	.288**	.311*									
5. Internalised Transphobia (r)	16.43	8.579	.246*	.207*	.320**	.394**								
6. Pride ( <i>r</i> )	13.95	6.450	.128	011	.078	.201	228*							
7. Negative Expectations ( <i>r</i> )	19.93	7.478	.273*	.283*	.362**	.523**	.430**	.273*						
8. Non-disclosure $(r)$	13.18	4.536	.072	.046	.057	.137	.333**	392**	.135					
9. Community Connectedness (r)	10.68	4.234	050	166	197	.200	314**	.273*	104	033				
10. Tobacco score $(r)$	4.35	1.201	223*	256*	280*	118	166	023	135	.013	.073			
11. AUDIT score $(r)$	11.96	8.814	.244*	.258*	.382**	.202	.363**	.359**	.311**	070	156	320**		
12. DUDIT score $(r)$	10.98	11.554	.292*	.260*	.452**	.220*	.260*	.390**	.285**	071	107	419**	.830**	

Table 6.4. Gender Minority Stress and Resilience measure and substance use correlations including means and standard deviations

*Note. M* and *SD* represent the mean and standard deviation, respectively. See table 6.5 for confidence intervals for each correlation value. \* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

Variable	М	SD	1	2	3	4	5	6	7	8	9	10	11	12
1. Discrimination ( <i>r</i> )	4.29	2.052												
2. Rejection $(r)$	5.32	2.863	[.540, .803]											
3. Victimisation ( <i>r</i> )	4.23	2.856	[.375, .720]	[.537, .828]										
4. Non-Affirmation ( <i>r</i> )	12.77	6.632	[.013, .447]	[.054, .514]	[.077, .526]									
5. Internalised Transphobia (r)	16.43	8.579	[.020, .489]	[.020, .505]	[.126, .513]	[.142, .598]								
6. Pride ( <i>r</i> )	13.95	6.450	[069, .361]	[268, .254]	[138, .305]	[036, .422]	[455, .025]							
7. Negative Expectations ( <i>r</i> )	19.93	7.478	[.017, .510]	[.061, .498]	[.158, .547]	[.351, .661]	[.210, .610]	[.057, .478]						
8. Non-Disclosure ( <i>r</i> )	13.18	4.536	[161, .274]	[150, .248]	[138, .238]	[061, .333]	[.093, .530]	[569, 183]	[084, .332]					
9. Community Connectedness (r)	10.68	4.234	[127, .158]	[383, .062]	[404, .005]	[041, .418]	[541, 096]	[.020,	[379, .139]	[270, .221]				
10. Tobacco score ( <i>r</i> )	4.35	1.201	[418,	[499, -	[477,	[356,	[421,	[273, .220]	[364,	[208,	[170, .307]			
11. AUDIT score $(r)$	11.96	8.814	[.074,	[.093,	[.189,	[.029,	[.209,	[.190,	[.115,	[272,	[.046, - 107]	[507, - 139]		
12. DUDIT score $(r)$	10.98	11.554	[.103, .519]	[.080, .451]	[.281, .652]	[.038, .404]	[.090, .428]	[.224, .544]	[.098, .478]	[261, .111]	[303, .074]	[551, 275]	[.726, .905]	

Table 6.5. Gender Minority Stress and Resilience measure and substance use correlation confidence intervals, with means and standard deviations

*Note. M* and *SD* represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation (see table 6 for correlation values.

**Psychopathological Symptoms.** For reasons of available space Table 6.6 shows correlations between all 12 dimensions of the SCL-90-R and substance use (tobacco, alcohol, drugs) and Table 6.7 shows the respective confidence intervals. Typically, GSI (the average score of all 90 items of the measure) is thought to be the best indicator for the current level of the symptoms (Derogatis, 1994).

Tobacco use had a negative correlation with all three global indices, indicating that the use of tobacco was associated with fewer psychopathological symptoms. There was a weak negative correlation with PST (r(82) = -.260, p = .018, 95% CI [-.467, -.055]) and with GSI (r(82) = -.290, p = .008, 95% CI [-.505, -.085]), and there was a moderate negative correlation with PDSI (r(82) = -.316, p = .004, 95% CI [-.527, -.068]).

Tobacco use was also negatively correlated with eight of the nine symptoms dimensions; there was no correlation between tobacco use and depression. There was a weak negative correlation with PAR (r(82) = -.218, p = .049, 95% CI [-.405, -.028]), INS (r(82) = -.227, p = .040, 95% CI [-.448, -.005]), SOM (r(82) = -.259, p = .019, 95% CI [-.453, -.096]), PHO (r(82) = -.285, p = .009, 95% CI [-.496, -.071]), and PSY (r(82) = -.291, p = .008, 95% CI [-.473, -.107]). There was also a moderate negative correlation with OBC (r(82) = -.300, p = .006, 95% CI [-.496, -.096]), HOS (r(82) = -.301, p = .006, 95% CI [-.488,-.109]), and ANX (r(82) = -.314, p = .004, 95% CI [-.527, -.095]).

Alcohol use had a positive correlation with all three global indices. There was a moderate correlation with PSDI (r(82) = .389, p < .001, 95% CI [.193, .566]), and there was a strong correlation with GSI (r(82) = .667, p < .001, 95% CI [.526, .768]) and with PST (r(82) = .673, p < .001, 95% CI [.549, .773]).

Alcohol use was also positively correlated with all nine symptom dimensions. There was a moderate correlation with OBC (r(82) = .416, p < .001, 95% CI [.237, .577]) and DEP

(r(82) = .424, p < .001, 95% CI [.239, .594]). There was a strong correlation with PHO (r(82) = .567, p < .001, 95% CI [.417, .691]), INS (r(82) = .602, p < .001, 95% CI [.450, .723]), ANX (r(82) = .636, p < .001, 95% CI [.512, .742]), SOM (r(82) = .676, p < .001, 95%CI [.536, .778]), PAR (r(82) = .724, p < .001, 95% CI [.589, .824]), HOS (r(82) = .736, p < .001, 95% CI [.627, .821]), and PSY (r(82) = .746, p < .001, 95% CI [.633, .833]).

Drug use had a positive correlation with all three global indices. There was a moderate correlation with PSDI (r(82) = .360, p < .001, 95% CI [.183, .523]), and there was a strong correlation with GSI (r(82) = .660, p < .001, 95% CI [.524, .763]) and PST (r(82) = .697, p < .001, 95% CI [.588, .788])

Drug use was also positively correlated with all nine symptom dimensions. There was a moderate correlation with DEP (r(82) = .373, p = .001, 95% CI [.196, .541]) and OBC (r(82) = .424, p < .001, 95% CI [.259, .582]). There was a strong correlation with INS (r(82)= .543, p < .001, 95% CI [.381, .682]), PHO (r(82) = .609, p < .001, 95% CI [.463, .726]), ANX (r(82) = .627, p < .001, 95% CI [.495, .732]), SOM (r(82) = .702, p < .001, 95% CI [.570, .805]), PAR (r(82) = .719, p < .001, 95% CI [.576, .819]), HOS (r(82) = .740, p < .001, 95% CI [.595, .845]), and PSY (r(82) = .773, p < .001, 95% CI [.684, .844]).

Variable	М	SD	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
1. SOM ( <i>r</i> )	1.3994	.89252															
2. OBC ( <i>r</i> )	2.0000	.93843	.764**														
3. INS ( <i>r</i> )	1.7344	.96071	.810**	.864**													
4. DEP ( <i>r</i> )	1.8039	.91488	.687**	.840**	.884**												
5. ANX ( <i>r</i> )	1.5866	.93092	.863**	.856**	.894**	.848**											
6. HOS ( <i>r</i> )	1.3313	1.00051	.867**	.732**	.797**	.672**	.846**										
7. PHO ( <i>r</i> )	1.5244	.99154	.830**	.828**	.860**	.794**	.898**	.860**									
8. PAR ( <i>r</i> )	1.2683	.95033	.804**	.619**	.726**	.622**	.800**	.766**	.713**								
9. PSY ( <i>r</i> )	1.3085	.98623	.880**	.723**	.823**	.731**	.875**	.886**	.844**	.859**							
10. GSI ( <i>r</i> )	1.5640	.84725	.914**	.889**	.940**	.879**	.962**	.900**	.928**	.831**	.927**						
11. PST ( <i>r</i> )	60.41	26.613	.902**	.810**	.877**	.878**	.907**	.876**	.884**	.851**	.912**	.951**					
12. PSDI ( <i>r</i> )	2.1650	.52059	.641**	.824**	.795**	.834**	.794**	.637**	.743**	.510**	.643**	.794**	.613**				
13. Tobacco score ( <i>r</i> )	4.35	1.201	259*	300**	227*	166	314**	301**	285**	218*	291**	290**	260*	316**			
14. AUDIT score (r)	11.96	8.814	.676**	.416**	.602**	.424**	.636**	.736**	.567**	.724**	.746**	.667**	.673**	.389**	320**		
15. DUDIT score (r)	10.98	11.554	.702**	.424**	.543**	.373**	.627**	.740**	.609**	.719**	.773**	.660**	.697**	.360**	419**	.830**	

Table 6.6. Symptom Checklist - 90 - Revised and substance use correlations with means and standard deviations.

*Note. M* and *SD* represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation. \* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

Variable	М	SD	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
1. SOM ( <i>r</i> )	1.3994	.89252															
2. OBC ( <i>r</i> )	2.0000	.93843	[.667, .843]														
3. INS ( <i>r</i> )	1.7344	.96071	[.718, .883]	[.784, .925]													
4. DEP ( <i>r</i> )	1.8039	.91488	[.540, .806]	[.755, .905]	[.822, .928]												
5. ANX ( <i>r</i> )	1.5866	.93092	[.788, .919]	[.794, .906]	[.848, .930]	[.779, .902]											
6. HOS ( <i>r</i> )	1.3313	1.00051	[.817, .911]	[.637, .821]	[.710, .867]	[.550, .778]	[.786, .894]										
7. PHO ( <i>r</i> )	1.5244	.99154	[.757, .896]	[.751, .893]	[.776, .925]	[.700, .865]	[.843, .938]	[.803, .903]									
8. PAR ( <i>r</i> )	1.2683	.95033	[.719, .866]	[.467, .747]	[.598, .833]	[.464, .754]	[.702, .876]	[.645, .852]	[.577, .828]								
9. PSY ( <i>r</i> )	1.3085	.98623	[.829, .918]	[.615, .818]	[.741, .892]	[.620, .821]	[.828, .914]	[.828, .927]	[.776, .900]	[.791, .907]							
10. GSI ( <i>r</i> )	1.5640	.84725	[.874, .948]	[.838, .930]	[.911, .961]	[.821, .922]	[.944, .976]	[.855, .934]	[.889, .959]	[.748, .896]	[.893, .953]						
11. PST ( <i>r</i> )	60.41	26.613	[.869, .931]	[.725, .874]	[.813, .921]	[.681, .862]	[.871, .937]	[.840, .910]	[.842, .919]	[.797, .893]	[.886, .935]	[.932, .965]					
12. PSDI ( <i>r</i> )	2.1650	.52059	[.488, .771]	[.700, .910]	[.661, .888]	[.676, .928]	[.672, .885]	[.497, .757]	[.598, .842]	[.308, .689]	[.482, .771]	[.657, .895]	[.424, .760]				
13. Tobacco score (r)	4.35	1.201	[453, 057]	[496, 096]	[448, 005]	[400, .077]	[527, 095]	[488, 109]	[496, 071]	[405, 028]	[473, 107]	[505, 085]	[467, 055]	[527, 068]			
14. AUDIT score (r)	11.96	8.814	[.536, .778]	[.237, .577]	[.450, .723]	[.239, .594]	[.512, .742]	[.627, .821]	[.417, .691]	[.589, .824]	[.633, .833]	[.526, .768]	[.549, .773]	[.193, .566]	[507, 139]		
15. DUDIT score (r)	10.98	11.554	[.570, .805]	[.381, .682]	[.381, .682]	[.196, .541]	[.495, .732]	[.595, .845]	[.463, .726]	[.576, .819]	[.648, .844]	[.524, .763]	[.588, .788]	[.183, .523]	[551, 275]	[.726, .905]	

Table 6.7. Symptom Checklist - 90 - Revised and substance use correlation confidence intervals with means and standard deviations.

Note. M and SD represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation (see table 6 for correlation values.

**Personality.** Table 6.8 shows the correlations between the seven dimensions of the TCI-140-R and substance use (tobacco, alcohol, drugs).

Tobacco use had a weak negative correlation with ST (r(82) = -.281, p = .011, 95% CI [-.471, -.090]) and a moderate negative correlation with NS (r(82) = -.470, p < .001, 95% CI [-.628, -.266]). Tobacco use also had a weak positive correlation with SD (r(82) = .228, p = .040, 95% CI [-.001, .421]).

Alcohol use had a weak negative correlation with RD (r(82) = -.248, p = .025, 95% CI [-.425, -.060]), a moderate negative correlation with HA (r(82) = -.419, p < .001, 95% CI [-.573, -.251]), and a strong negative correlation with CO (r(82) = -.594, p < .001, 95% CI [-.732, -.444]). Alcohol use also had a strong positive correlation with ST (r(82) = .571, p < .001, 95% CI [.417, .690]).

Drug use had a moderate negative correlation with HA (r(82) = -.436, p < .001, 95% CI [-.607, -.278]), and a strong negative correlation with CO (r(82) = -.524, p < .001, 95% CI [-.719, -.318]). Drug use also had a weak positive correlation with NS (r(82) = .240, p = .030, 95% CI [.081, .389]) and a strong positive correlation with ST (r(82) = .608, p < .001, 95% CI [.455, .727]).

Table 6.8. Temperament and Character Inventory - 140 - Revised and substance use correlations. Means, standard deviations, and correlations with confidence intervals

Variable	М	SD	1	2	3	4	5	6	7	8	9	10
1. Novelty Seeking (r)	59.87	9.701										
2. Harm Avoidance ( <i>r</i> )	69.92	13.941	189									
3. Reward Dependence (r)	63.50	10.837	[440, .036] .103	.337**								
4. Persistence ( <i>r</i> )	62.12	12.795	[165, .356] .013	[.122, .536] 195	032							
5. Self-directedness (r)	56.66	10.502	[277, .298] 254*	[407, .050] 262*	[329, .285] .094	.367**						
6. Cooperativeness (r)	68.84	11.375	[486, .052] 114	[438, - .052] .450**	[173, .352] .522**	[.099, .571] 085	.108					
7. Self-transcendence (r)	44.97	10.292	[414, .169] .383**	[.266, .641] 301**	[.321, .669] .113	[358, .134] .131	[191, .362] 264*	334**				
8. Tobacco score ( <i>r</i> )	4.35	1.201	[.217, .540] 470**	[561, - .067] .043	[110, .304] .014	[080, .361] 096	[439, - .064] .228*	[556, - .118] .113	281*			
9. AUDIT score ( <i>r</i> )	11.96	8.814	[628, - .266] .175	[148, .250] 419**	[218, .281] 248*	[362, .181] .050	[001, .421] 161	[102, .338] 594**	[471, - .090] .571**	320**		
10. DUDIT score (r)	10.98	11.554	[011, .377] .240*	[573, - .251] 436**	[425, - .060] 197	[151, .242] .077	[342, .035] 212	[-732, - .444] 524**	[.417, .690] .608**	[507, - .139] 419**	.830**	
.,			[.081, .389]	[607, - .278]	[383, - .022]	[109, .269]	[383, - .032]	[719, - .318]	[.455, .727]	[551, - .275]	[.726, .905]	

Note. M and SD represent the mean and standard deviation, respectively. Values in square brackets are the 95% confidence interval for each correlation. \* Correlation is significant at the 0.05 level. \*\* Correlation is significant at the 0.01 level.

## **6.2. Inferential Statistics**

In the same manner as phase one, separate multiple regression analyses using the forward method were carried out for the outcome variables of alcohol and drug use to investigate whether the predictor variables which most highly correlated with those outcomes could significantly predict participant's AUDIT and DUDIT scores, respectively. A binary logistic regression was conducted between the variables which most highly correlated with smoking to investigate if they could predict whether an individual was a current smoker or non-smoker.

For a predictor variable to be included in the regression analyses the inclusion criteria was a correlation coefficient of  $r \ge .35$  and a p-value of  $p \le .01$ .

## 6.2.1. Binary Logistic Regression – Smoking Status

Based on the inclusion criteria stated above, two predictor variables were included which were Environment QOL and Novelty Seeking (NS). A binary logistic analysis was conducted to explore if Environment and NS could significantly predict whether an individual was a non-smoker smoker or a current smoker.

82 cases were included in the analysis and the full model significantly predicted smoking status (omnibus  $\chi^2 = 23.27$ , df = 2, p < .001). The model could account for between 24.7% and 33% of variance in smoking status, with 69.8% of current smoking status successfully predicted and 61.5% of current non-smoking status successfully predicted. The overall prediction rate was 65.9%. Please see Table 6.9 for coefficients, Wald statistics, probability values, and standardised beta (with 95% confidence intervals). Overall, 65.9% of predictions were accurate. NS significantly contributed to predicting smoking status (p < .001); for each point increase in NS, participants were 0.86 times less likely to be a current smoker.

Table 6.9. Binary	/ Logistic	Regression	predicting c	current smoking s	status (smoke	r/non-smoker).
			r · · · · · · · · · · · · · · · · · · ·		· · · · · · · · · · · · · · · · · · ·	

Variable	df	Sig.	В	SE	Exp(β) [95%CI]
Constant	1	.000	8.904	2.446	7363.179
Novelty Seeking (NS)	1	.000**	151	.041	.860 [.794, .931]

Note. Omnibus  $\chi^2$  = 23.274, df = 1, p <.001. R<sup>2</sup> = .330 (Nagelkerke). R<sup>2</sup> = .247 (Cox & Snell). \*\*\*Significant at the .001 level.

## 6.2.2. Multiple Regression – Alcohol Use

Based on the inclusion criteria ( $r \ge .35$  and a p-value of  $p \le .01$ ), ten predictor variables were included. Please see Appendix L for the correlations and p-values. These were Success, Power, and Control (SPC), Restricted Affectionate Behaviour Between Men (RABBM), Pride (P), Internalised Transphobia (IT), Victimisation (V), Positive Symptom Total (PST), Positive Symptom Distress Index (PSDI), Harm Avoidance (HA), Cooperativeness (CO), and Self-Transcendence (ST). A multiple regression analysis using the forward method was conducted to explore if the ten predictor variables could significantly predict alcohol use.

82 cases were included in analysis, with three cases excluded for missing data in any of the ten predictor variables. In total 79 cases were analysed, and the results indicated that the model which was the most significant predictor of AUDIT scores was model 2 (see Table L.1), which included two predictors: PST (p < .001) and HA (p < .001). This model could account for 60.4% of variance and was a significant predictor of alcohol use status, F(2, 78) = 58.038, p < .001, R<sup>2</sup> = .604 (see Table 6.10). The final predictive model was:

AUDIT score = 16.60 + (.212\*PST) + (-.250\*HA)

AUDIT score increased by .212 for each point increase in PST and decreased by .25 for each point increase in HA.

Table 6.10. Multiple regression of AUDIT using forward method, Model 2: PST and HA

Variable	В	SE B	β	t	р
(Constant)	16.601	3.472		4.782	.000**
PST	.212	.023	.654	9.068	.000**
HA	250	.044	409	-5.673	.000**

*Note*: B = unstandardized beta, SEB = standard error for unstandardized beta,

\*\* Significant at the 0.001 level.

 $<sup>\</sup>beta$  = standardized beta.

## 6.2.3. Multiple Regression–Drug Use

Based on the inclusion criteria previously stated ( $r \ge .35$  and a p-value of  $p \le .01$ ), eight predictor variables were included. Please see Appendix M for the correlations and pvalues. These were Restricted Affectionate Behaviour Between Men (RABBM), Pride (P), Victimisation (V), Positive Symptom Total (PST), Positive Symptom Distress Index (PSDI), Harm Avoidance (HA), Cooperativeness (CO), and Self-Transcendence (ST). A multiple regression analysis using the forward method was conducted to explore if the eight predictor variables could significantly predict drug use.

82 cases were included in analysis, with one case excluded for missing data in any of the eight predictor variables. In total 81 cases were analysed, and the results indicated that the model which was the most significant predictor of DUDIT scores was model 3 (see Table M.1), which included the predictor variables PST (p < .001), HA (p < .001), and V (p = .022). This model could account for 68% of variance and was a significant predictor of drug use status, F(3, 80) = 54.428, p < .001,  $R^2 = .68$  9 (see Table 6.11). The final predictive model was:

DUDIT score = 
$$13.752 + (.272*PST) + (.653*V) + (-.316*HA)$$

DUDIT score increased by .272 for each point increase in PST and increased by .653 for each point increase in V. DUDIT score decreased by .316 for each point increase in HA.

Variable	В	SE B	β	t	р
(Constant)	13.752	4.409		3.119	.003**
PST	.272	.029	.632	9.291	.000**
HA	316	.054	386	-5.830	.000**
V	.653	.279	.163	2.341	.022*

Table 6.11. Multiple regression of DUDIT using forward method, Model 3: PST, HA, and V

*Note:* B = unstandardized beta, SEB = standard error for unstandardized beta,  $\beta$  = standardized beta.

\* Significant at the 0.05 level. \*\* Significant at the 0.001 level.

## **6.3.** Longitudinal Statistics

Twelve months separated TESUP-1 (November 2019) and TESUP-2 (November 2020). A one-way analysis of variance (ANOVA) was conducted between all independent variables to investigate if there were any statistically significant differences between the variables. To account for multiple comparisons, p-values were adjusted using the Bonferroni method. It is important to note that because 23 participants who completed TESUP-1 did not participate in TESUP-2, the mean for each variable given in the ANOVA results is based on the 82 participants who completed all the measures in both phase one and two and is therefore different to the means stated in TESUP-1 (see Table 3.3 in Chapter 3).

## 6.3.1. Substance use

Tobacco Use. A one-way repeated measures ANOVA (See Table 6.12) determined that there was no significant difference in mean rates of tobacco use between TESUP-1 and TESUP-2 (F(1, 81) = .391, p = .534,  $\eta_p^2 = .005$ ).

Alcohol Use. In TESUP-1 the mean AUDIT score was 7.26 (SD = 6.21), with 49.5% (N = 52) of the sample meeting the cut-off of 8 points (indicating harmful alcohol use), and

6% (N = 7) of these meeting the cut-off of 20 points (indicating alcohol dependence). In TESUP-2 the mean AUDIT score increased to 11.96 (SD = 8.81) which exceeds the cut-off of 8 points; 56.1% (N = 46) of the sample met the cut-off of 8 points and 26.8% (N = 22) met the cut-off of 20 points.

A one-way repeated measures ANOVA (See Table 6.12) determined that the mean AUDIT scores significantly differed between TESUP-1 and TESUP-2 (F(1, 81) = 15.609, p < .001,  $\eta_p^2 = .162$ ). Mean AUDIT scores increased by 4.707 points, 95% CI [2.337, 7.078].

**Drug use.** In TESUP-1 the mean DUDIT score was 5.11 (SD = 6.83); with 43.8% (N = 46) of the sample meeting the cut-off score of 6 points. In TESUP-2 the mean DUDIT score increased to 10.98 (SD = 11.55) which exceeds the cut-off of 6 points; 51.2% (N = 42) of the sample met the cut-off of 6 points.

A one-way repeated measures ANOVA (See Table 6.12) determined that mean DUDIT scores were statistically significantly different between TESUP-1 and TESUP-2  $(F(1, 81) = 20.501, p < .001, \eta_p^2 = .202)$ . Mean DUDIT scores increased by 5.866 points, 95% CI [3.288, 8.444].

stage of transition, an	a education	between Phas	se I and Phase	e 2. Means a	nd standard dev	lations.	
Measure	Pha	se 1	Pha	.se 2	<i>F</i> (1, 81)	${\eta_p}^2$	
	M	SD	M	SD			
Tobacco	4.48	1.434	4.35	1.201	.391	.005	
AUDIT	7.25	6.210	11.96	8.814	15.609**	.162	
DUDIT	5.11	6.831	10.98	11.554	20.501**	.202	
Age	25.26	6.39	26.09	6.202	.232	.003	
Stage of Transition	3.09	1.028	3.37	.908	3.189	.039	
Educational level	4.68	1.729	5.05	1.679	1.912	.024	

Table 6.12. Analysis of Variance (ANOVA) in Tobacco Use, AUDIT score, DUDIT score, age, stage of transition, and education between Phase 1 and Phase 2. Means and standard deviations.

Note. M and SD represent the mean and standard deviation, respectively.

\*\* Significant at the 0.001 level.

## 6.3.2. Demographics

A one-way repeated measures ANOVA explored significant differences in three demographics variables (age, stage of transition, education) between TESUP-1 and TESUP-2 (See Table 6.12). It was determined that there were no significant differences in the mean scores of any of the three demographic variables.

# 6.3.3. Quality of life

A one-way repeated measures ANOVA explored significant differences in four QOL domains between TESUP-1 and TESUP-2 (See Table 6.13). It was determined that there were no significant differences in the mean scores of any of the four QOL domains.

Fliase 2. Means and standard	ueviation	5.				
Measure	Pha	se 1	Pha	se 2	<i>F</i> (1, 81)	${\eta_p}^2$
	М	SD	М	SD		
Physical Health QOL	13.96	3.106	13.26	2.622	2.895	.035
Psychological Health QOL	11.46	3.173	11.68	2.738	.226	.003
Social and Relationships	13.34	3.703	13.60	3.121	.192	.002
QOL						
Environment QOL	13.32	2.848	13.54	2.474	.353	.004

Table 6.13. Analysis of Variance (ANOVA) in the 4 domains of QOL between Phase 1 and Phase 2. Means and standard deviations.

*Note. M* and *SD* represent the mean and standard deviation, respectively. \*\* Significant at the 0.01 level.

## 6.3.4. Gender Role Conflict

A one-way repeated measures ANOVA explored significant differences in four GRC

domains between TESUP-1 and TESUP-2 (See Table 6.14). It was determined that there

were no significant differences in the mean scores of any of the four GRC domains.

Measure	Pha	ise 1	Pha	ise 2	F(1, 81)	$\eta_p^2$
	М	SD	М	SD	_ 、	II.
SPC	42.06	11.898	44.46	11.597	1.771	.022
RE	36.44	10.929	37.14	10.679	.175	.002
RABBM	23.86	8.830	25.58	8.626	.987	.013
CBWFR	19.17	5.785	20.47	5.242	1.863	.024

Table 6.14. Analysis of Variance (ANOVA) in the 4 domains of GRCS between Phase 1 and Phase 2. Means and standard deviations.

Note. M and SD represent the mean and standard deviation, respectively.

\*\* Significant at the 0.01 level.

## 6.3.5. Gender Minority Stress and Resilience

A one-way repeated measures ANOVA explored significant differences in nine GMSR sub-domains between TESUP-1 and TESUP-2 (See Table 6.15). It was determined that there were no statistically significant differences in the mean scores of the GMSR subdomains of Gender-related Discrimination ( $F(1, 81) = .201, p = .201, \eta_p^2 = .020$ ), Genderrelated Victimisation ( $F(1, 81) = 2.769, p = .100, \eta_p^2 = .033$ ), Non-affirmation (F(1, 81) =.062,  $p = .062, \eta_p^2 = .001$ ), Internalised Transphobia ( $F(1, 81) = 1.8, p = .183, \eta_p^2 = .022$ ), Non-disclosure ( $F(1, 81) = 0, p = .988, \eta_p^2 = 0$ ), and Community Connectedness (F(1, 81) =.063,  $p = .063, \eta_p^2 = .042$ ).

The mean scores that significantly differed between TESUP-1 and TESUP-2 were Gender-related Rejection (R) (*F* (1, 81) = 6.246, *p* = .014,  $\eta_p^2$  = .072) which increased by 1.024, 95% CI [.209, 1.840]; Pride (P) (*F* (1, 81) = 4.775, *p* = .032,  $\eta_p^2$  = .056) which increased by 2, 95% CI [.179, 3.821]; and Negative Expectations for the Future (NE) (*F* (1, 81) = 4.767, *p* = .032,  $\eta_p^2$  = .056) which increased by 2.451, 95% CI [.217, 4.685].

Thuse T und Thuse 2. Weaths and standard de Viations.							
Measure	Phase 1		Phase 2		<i>F</i> (1, 81)	${\eta_p}^2$	
	М	SD	М	SD			
Discrimination	3.88	2.213	4.29	2.052	1.665	.020	
Rejection	4.29	2.706	5.32	2.863	6.246*	.072	
Victimisation	3.48	3.171	4.23	2.856	2.769	.033	
Non-affirmation	12.50	7.354	12.77	6.632	.062	.001	
Internalised Transphobia	14.74	8.550	16.43	8.579	1.800	.022	
Pride	11.95	6.543	13.95	6.450	4.775*	.056	
Negative Expectations	17.48	8.291	19.93	7.478	4.767*	.056	
Non-disclosure	13.17	5.298	13.18	4.536	0	0	
Community	12.00	5.038	10.68	4.234	3.554	.042	
Connectedness							

Table 6.15. Analysis of Variance (ANOVA) in the 9 domains of GMSR between Phase 1 and Phase 2. Means and standard deviations.

Note. M and SD represent the mean and standard deviation, respectively.

\* Significant at the 0.05 level.

## 6.3.6. Psychopathological Symptoms

A one-way repeated measures ANOVA explored significant differences in nine symptoms dimensions and three global indices (See Table 6.16). It was determined that there was no statistically significant difference in the mean scores of the symptoms dimensions of Obsessive-Compulsive (OBC) (F(1, 81) = 3.71, p = .058,  $\eta_p^2 = .046$ ), Interpersonal Sensitivity (INS) (F(1, 81) = 3.197, p = .078,  $\eta_p^2 = .040$ ), Depression (DEP) (F(1, 81) =1.263, p = .264,  $\eta_p^2 = .016$ ), Paranoid Ideation (PAR) (F(1, 81) = 3.692, p = .058,  $\eta_p^2 = .046$ ), or the global index of Positive Symptom Distress Index (PSDI) (F(1, 81) = 2.136, p = .148,  $\eta_p^2 = .027$ ).

The mean symptom dimension scores that significantly differed between TESUP-1 and TESUP-2 were: Somatisation (SOM) (F(1, 81) = 7.933, p = .006,  $\eta_p^2 = .093$ ) which increased by .342, 95% CI [.100, .584]; Anxiety (ANX) (F(1, 81) = 5.267, p = .024,  $\eta_p^2 =$ .064) which increased by .329, 95% CI [.044, .615]; Hostility (HOS) (F(1, 81) = 3.590, p =.044,  $\eta_p^2 = .051$ ). which increased by .303, 95% CI [.008, .599; Phobic Anxiety (PHO) (F(1, 81) = 13.758, p < .001,  $\eta_p^2 = .152$ ) which increased by .548, 95% CI [.254, .842; and Psychosis (PSY) (*F* (1, 81) = 8.051, p = .006,  $\eta_p^2$  = .095) which increased by .362, 95% CI [.108, .615]

The mean global index scores that were statistically significantly different between TESUP-1 and TESUP-2 were: Global Severity Index (GSI) (F(1, 81) = 5.292, p = .024,  $\eta_p^2 = .064$ ) which increased by .287, 95% CI [.039, .536], and Positive Symptom Total (PST) (F(1, 81) = 4.092, p = .047,  $\eta_p^2 = .050$ ) which increased by 7.397, 95% CI [.115, 14.680].

Table 6.16. Analysis of Variance (ANOVA) in the 9 symptom dimensions and 3 global indices of SCL-90-R between Phase 1 and Phase 2. Means and standard deviations.

Measure	Pha	Phase 1		Phase 2		$\eta_p^2$
	М	SD	М	SD		
SOM	1.0278	.73299	1.3994	0.89252	7.933**	.093
OBC	1.6782	.94850	2.0000	0.93843	3.709	.046
INS	1.4530	.88722	1.7344	0.96071	3.197	.040
DEP	1.6218	.91215	1.8039	0.91488	1.263	.016
ANX	1.2321	.90302	1.5866	0.93092	5.267*	.064
HOS	.9915	.88473	1.3313	1.00051	4.174*	.051
РНО	.9542	.94438	1.5244	0.99154	13.758**	.152
PAR	1.000	.77478	1.2683	0.95033	3.692	.046
PSY	.9256	.75152	1.3085	0.98623	8.051**	.095
GSI	1.2550	.73981	1.5640	0.84725	5.292*	.064
PST	52.60	23.623	60.41	26.613	4.092*	.050
PSDI	2.0061	.56514	2.1650	0.52059	2.136	.027

Note. M and SD represent the mean and standard deviation, respectively.

\* Significant at the 0.05 level. \*\* Significant at the 0.01 level.

## 6.3.7. Personality

A one-way repeated measures ANOVA explored significant differences in seven TCI

domains between TESUP-1 and TESUP-2 (See Table 6.17). It was determined that there

were no statistically significant differences in the mean scores of any of the seven TCI

domains.

Measure	Phase 1		Phase 2		<i>F</i> (1, 81)	${\eta_p}^2$
	M	SD	М	SD		
Novelty Seeking	60.58	9.086	59.87	9.701	.106	.001
Harm Avoidance	71.90	15.078	69.92	13.941	.396	.005
Reward Dependence	64.66	11.498	63.50	10.837	.279	.004
Persistence	60.85	13.014	62.12	12.795	.669	.009
Self-directedness	58.21	12.330	56.66	10.502	.456	.006
Cooperativeness	71.64	11.710	68.84	11.375	1.682	.036
Self-transcendence	43.01	10.520	44.97	10.292	.993	.014

Table 6.17. Analysis of Variance (ANOVA) in the 7 domains of TCI-140 between Phase 1 and Phase 2. Means and standard deviations.

Note. M and SD represent the mean and standard deviation, respectively.

\* Significant at the 0.05 level.

## 6.4. TESUP-2 TA Results

## 6.4.1. Impact of COVID-19 on Substance Use

**Tobacco Use.** Participants' reports of the impact of COVID-19 on tobacco use were varied and most of the participants who reported currently smoking in TESUP-1 reported no changes to this in TESUP-2. Those who had reported increased tobacco use had experienced redundancy or changes to their employment (See Box 6.1: 1.1.1), increased anxiety due to longer waits for gender identity services (GIS) appointments (1.1.2) or had experienced an increase in alcohol use which was the primary context for their tobacco use (1.1.3). Participants who had decreased their use of tobacco had often done so as a consequence of COVID-19 lockdown measures reducing social smoking opportunities (1.1.4).

Box 6.1: Theme 1: Impact on Substance Use - participant extracts

#### 1.1 Tobacco Use

- 1.1.1. "I've been smoking more tobacco because I was made redundant in the middle of a pandemic!" R\_004, aged 40.
- **1.1.2.** "I smoke more when I'm anxious such as waiting so long for a GIC appointment because of COVID, it has increased my anxiety and depression and therefore I am smoking more" -*R\_083, aged 35.*
- 1.1.3. "I smoke more now, it's when I've had a drink which I've been doing more recently" -R\_035, aged 24.
- **1.1.4.** "I have smoked less tobacco in the last year due to coronavirus. I am mostly a social smoker and therefore have not been many times to socialise, or smoke"  $R_050$ , aged 32.

#### 1.2. Alcohol Use

- **1.2.1.** "My alcohol intake has increased a very small amount, although this is because I'm more likely to have a casual drink at dinner rather than get drunk"  $-R_0.034$ , aged 19.
- **1.2.2.** "I've drank more alcohol during lockdown periods, including drinking during the day and not just evenings"  $R_039$ , aged 30.
- **1.2.3.** "I drink more often than I did last year and when I do drink, I drink a higher volume of alcohol"  $-R_038$ , aged 19.
- **1.2.4.** "I feel guilty for spending money on alcohol, or remorseful if I get hungover"  $-R_053$ , aged 29.
- 1.2.5. "I have drunk less alcohol as I haven't been going out due to COVID. R\_009, aged 24

#### 1.3. Drug Use

**1.3.1.** "My cannabis intake is up quite a bit more due to being at home more"  $-R_004$ , aged 40.

- **1.3.2.** "The pandemic has put much more pressure on my life and work, I increased the use of marijuana to relax"  $R_073$ , aged 29.
- **1.3.3.** "I have started to take stronger drugs and more frequently to deal with stress"  $R_032$ , aged 20.
- **1.3.4.** "I've used MDMA a lot since lockdown but I've being doing it at home alone rather than at gigs like I used to, I think this is bad and I don't like myself for doing it"  $-R_0.082$ , aged 35.

**Alcohol Use.** Participant reports of increased alcohol use during COVID-19 were frequent. Some participants described how, due to COVID-19, they were more often 'casually' consuming alcohol in their home (1.2.1). In these contexts, participants reported that drinking alcohol now occurred throughout the day rather than solely on weekends or evenings (1.2.2). Other participants reported both an increased frequency and quantity of alcohol consumption (1.2.3), these participants (who could be described as engaging in binge-drinking) also reported feelings of guilt about the financial impact or impact on their health (1.2.4). Some participants described a reduction in alcohol use; most of these participants reported that they only drank alcohol in social situations and had not been able to do so due to COVID-19 lockdown measures (1.2.5).

**Drug Use.** Most of the participants who indicated they used drugs or had used drugs in the last year reported that their drug use had increased during COVID-19. Many participants who detailed their use of cannabis explained that as they had more time at home due to lockdown their intake of cannabis had increased (1.3.1). Due to increased stress in the past year, some participants had begun to use cannabis more frequently to help them relax (1.3.2) or had started to take stronger drugs (1.3.3). Participants who used drugs other than cannabis did not typically respond in the optional text entry boxes; of those that did, one individual indicated that they had started using MDMA in their house, alone, and reported feeling ashamed of this (1.3.4).

## 6.4.2. Impact of COVID-19 on Transition

**Disruption of Transition-Related Interventions.** Due to reduced access to nonurgent healthcare in the COVID-19 pandemic, many participants reported disruptions to ongoing treatment. Most commonly, limitations to appointments with a GP or nurse resulted in delays of testosterone injections (Box 6.2: 2.1.1). Furthermore, multiple participants had transition-related surgeries postponed, which caused significant personal distress (2.1.2). Moreover, participants who were non-UK nationals experienced additional difficulties due to differences in responses to COVID-19 between countries (2.1.3). Box 6.2: Theme 2: Impact on Transition – participant extracts

### 2.1. Disruption of Transition-Related Interventions

- **2.1.1.** "Earlier in the year I did have issues booking an appointment with the nurse at my local GP to administer my injection"  $R_013$ , aged 27.
- **2.1.2.** "My top surgery has postponed due to COVID-19 which caused a large amount of stress as I have been waiting for years for this surgery"  $R_005$ , aged 20.
- **2.1.3.** "The pandemic has significantly slowed down my legal transition as the courts in my country have closed during lockdown"  $R_104$ , aged 21.

### 2.2. Difficulties accessing GIS

- **2.2.1.** "[COVID-19] has delayed medical appointments at the GIC, making an already long waiting list even longer" *R\_040, aged 35.*
- **2.2.2.** "Everything about my transition has been put on pause this year, I feel as if I am stuck and unable to progress in a lot of ways"  $R_050$ , aged 32.
- **2.2.3.** "I've had no support from the GIC"  $-R_044$ , aged 24.
- **2.2.4.** "Before lockdown the GIC referred me to various services to make sure I was healthy for medical transition. That all stopped. I've been left without any contact at all and now I feel like shit"  $R_027$ , aged 24.

Difficulties accessing GIS. Service provision from GIS was disrupted due to

COVID-19, resulting in cancelled appointments, thus increasing already long waiting times

(2.2.1). The psychological impact of this was frequently reported by participants, who had

experienced increased distress and uncertainty about their medical transition (2.2.2).

Furthermore, participants reported a lack of support and communication from the GIS (2.2.3)

which left them feeling 'stuck', both in their transition and in life. Difficulties communicating

with the GIS often resulted in an impact on the mental health and self-esteem of participants

(2.2.4).

# 6.4.3. Impact of COVID-19 on Gender Minority Stress

**Transphobia in Society.** Many participants reported that a major source of genderrelated stress was facing discrimination and transphobia in UK society, which had been compounded by the distress and isolation they were experiencing due to COVID-19. Negative depictions of transgender people in the news and in media which were perceived to be increasing in the past year was a significant source of distress (Box 6.3: 3.1.1), and participants described that with transgender discussions so frequent in the media, this past year had felt like a constant debate for transgender rights (3.1.2). Participants reported experiencing transphobia in public, such as being harassed in male toilets (3.1.3), as well as experiencing transphobia online (3.1.4). These direct and indirect experiences of transphobia and discrimination were highly detrimental to the mental health and self-esteem of all those who reported these situations, and sometimes resulted in internalised transphobia (i.e., shame and hatred about one's transgender identity) (3.1.5). Participants also reported experiencing harassment and discrimination from employers and fellow employees alike (3.1.6). To avoid discrimination, some participants described how they felt they had to 'go stealth' (i.e., not disclosure their gender identity). However, as a result of non-disclosure they also experienced anxiety over being 'outed' at work which might risk their job (3.1.7).

Box 6.3: Impact on Gender Minority Stress - participant extracts

3.1. Transphobia in Society

- **3.1.1.** "Recent raises in transphobia in the UK have really messed with my mental health and my comfort with my gender. There's been a lot of awful transphobic news and rhetoric. I receive a lot more direct hatred and discrimination from other people"  $-R_{-}027$ , aged 24.
- **3.1.2.** "With transgender-focused discussions being so central in the media this year I've had to argue my rights more this year than ever before. With the pandemic it was already a hard year and the constant media debates about my rights is unpleasant to say the least. There has been an increase in trans related attacks and murders too this year, so I have to pay a lot more attention to my surroundings"  $-R_023$ , aged 25.
- **3.1.3.** "I was harassed in a bar, while using a stall in the men's room a security guard forced open the door on me and kicked me out"  $R_045$ , aged 20.
- **3.1.4.** "This year I experienced for the first time transphobic abuse online from people quizzing me on my right to have my pronouns respected. I felt ashamed to correct people misgendering me"  $R_040$ , aged 35.
- **3.1.5.** "I just really hate being trans ... I often see negativity and hatred of trans people online, and it makes me angry that I have to be like this. I feel like I have some kind of incurable disease which people are allowed to make horrible comments about. I just wish I wasn't like this as I feel like I will be depressed and unhappy forever"  $-R_029$ , aged 22.
- **3.1.6.** "I had to leave 2 jobs this year. In one my boss wouldn't recognise my identity, the other is that my colleagues made fun of me. This brought tremendous pressure to my life and work.  $-R_0073$ , aged 29.
- **3.1.7.** "If I tell more people about being trans it increases the risk of others I don't want to know finding out, which increases my anxiety. It would be great to be open to more people but it risks people at work finding out and it could risk losing my job"  $-R_039$ , aged 30.

## 6.4.4. Impact of COVID-19 on Mental Health

# Worsening of Pre-Existing Mental Health Conditions. Many participants indicated

that they had experienced a decline in their mental health conditions. Individuals who indicated that they had a mental health diagnosis described experiencing a worsening of symptoms during lockdown (Box 6.4: 4.1.1). The effect of COVID-19 on one individual's mental health had caused worsening of symptoms to the extent that it warranted admittance as an inpatient (4.1.2). Participants who indicated that they had a diagnosis of a neurodevelopmental condition such as attention deficit hyperactivity disorder (ADHD) or autism spectrum disorder (ASD) also experienced additional difficulties which they attributed to COVID-19. Participants with ASD had found lockdown measures to be distressing and described having trouble regulating their emotions (4.1.3), while participants with ADHD reported struggles with executive functioning and memory (4.1.4).

Box 6.4: Theme 4: Impact of COVID-19 on Mental Health - participant extracts

### 4.1. Worsening of Pre-Existing Mental Health Conditions

- **4.1.1.** "This year I have come to understand that I suffer from obsessive compulsive disorder and anxiety on a daily basis, I've had persistent and difficult symptoms most recently in the last few months" *R\_050, aged 32.*
- **4.1.2.** "It's had a massive effect, so much that recently I had to have a 3-month inpatient stay"  $R_003$ , aged 25.
- **4.1.3.** "I feel anxious and angry when I see people wearing masks because of my autism I can't read people's eyes, so now strangers look faceless and threatening. I find myself feeling increasingly angry and hopeless."  $R_029$ , aged 22.
- **4.1.4.** "My symptoms have been affecting my day-to-day life even more [during COVID-19], especially the symptoms of trouble focusing and poor short-term memory which have begun to seriously impede my personal and work life"  $-R_038$ , aged 19.

#### 4.2. Anxiety, Isolation, and Loneliness

- **4.2.1.** "My anxiety has severely increased. My future feels extremely uncertain, I feel disconnected from the world and any opportunities"  $-R_{-009}$ , aged 24.
- **4.2.2.** "I suffered from panic attacks quite frequently this year, mostly around the growth of the virus. I felt very anxious mostly about giving the virus to someone who was vulnerable. This spiralled into thoughts about mortality, that I will inevitably lose people that I love in my lifetime, and that I could die at any point, and that I'm not ready to die"  $R_0 12$ , aged 24.
- **4.2.4.** "I've struggled with anxiety which feels overpowering most days. I have going outside like I have trouble eating or drinking in public and I feel like I am being watched by others while I'm out"  $R_016$ , aged 28.

#### 4.3. Mitigating the Impact of COVID-19 on Mental Health

- **4.3.1.** "I developed anxiety, and started taking medication for it, which has considerably improved my mental health"  $R_0097$ , aged 19.
- **4.3.2.** "I found the initial lockdown hard as I didn't have anything to occupy myself with, however getting a job and volunteering at a hospital helped my mental health greatly  $-R_{-005}$ , aged 20.

**Anxiety, Isolation, and Loneliness.** An often-reported concern that the participants had was anxiety about the future; feeling overwhelmed and uncertain about themselves and the world was a consequence of this anxiety (4.2.1). Additionally, worries about transmission of COVID-19 to loved ones was reported, which resulted in frequent rumination about illness and death (4.2.2). For some participants, increased anxiety also led to feelings of agoraphobia and paranoia (4.2.3).

**Mitigating the Impact of COVID-19 on Mental Health.** Due to an increase in psychopathological symptoms as a consequence of COVID-19 there was a need to find methods of managing these symptoms. Some participants found relief in medications to manage their mental health (4.3.1), while others were able to mitigate the impact of COVID-19 on their mental health by gaining employment or a volunteer position (4.3.2).

## 6.4.5. Impact of COVID-19 on Quality of Life

**Psychological and Physical Health.** Due to the pandemic, many participants expressed feelings of anxiety about the potential impact of their deteriorating mental health (See Box 6.5; 5.1.1). Participants who had experienced a worsening of pre-existing mental health conditions reported that their quality of life had been particularly affected (5.1.2). Some participants also mentioned that during the pandemic they engaged in more trans rights activism, but that this had resulted in 'activism burnout' (5.1.3). Where participants did indicate an improvement in their QOL related to psychological or physical health, this was typically due to reaching a milestone in their transition, such as starting testosterone (5.1.4).

#### 5.1. Psychological and Physical Health

- **5.1.1.** "Lately it's been harder to ignore my anxiety and depression. I've been taking antidepressants but I do worry about the long term impact of the pandemic on my mental health in the future"  $-R_016$ , aged 28.
- **5.1.2.** "I already had severe mental health issues and the pandemic has been terrible for me ... I've lost so much and it has completely ruined my already not-great quality of life"  $R_025$ , aged 33.
- **5.1.3.** "The situation in the UK is getting worse for trans people so since I had more time I was doing more activism. But this resulted in exhaustion from activism burnout and had a bad effect on my mental health"  $-R_050$ , aged 32.
- **5.1.4.** "My quality of life has increased over the last year. I started testosterone which really positively impacted my mental health"  $R_097$ , aged 20.

### 5.2. Social and Environmental QOL

- **5.2.1.** "It's been harder to look forward to the future and plan things. I'm unable to see friends or anyone I care about, and I feel stuck in a cycle of waking up, going to work, and coming home to sleep"  $-R_0.016$ , aged 28.
- **5.2.2.** "I feel like my quality of life has decreased because I've lost so much freedom and been so isolated. I haven't seen anyone or gone anywhere in months and I just feel like this isn't much of a life"  $-R_028$ , aged 22.
- **5.2.3.** "My quality of life initially wasn't great because I had an unpleasant living situation but now I've moved and I feel much better"  $R_011$ , aged 27.
- **5.2.4.** "My relationship with my fiancé has gotten so much better because we've had lots of time to spend together"  $R_053$ , aged 29.

**Social and Environmental QOL.** Some participants reported that due to disruptions in their social lives as a result of COVID-19, they had been feeling isolated, and found it difficult to make plans or look forward to the future (5.2.1). Many of the participants who described these social difficulties also reported that isolation due to COVID-19 had a negative impact on their mental health (5.2.2). However, some participants described how moving house had a positive impact on their quality of life (5.2.3) and others had been able to improve their relationship with their partner(s), due to having more free time (5.2.4).

# 6.4.6. Impact of COVID-19 on GRC and Masculinities

**Trans Masculinities.** The majority of participants (N = 41) who left a comment about GRC and masculinities indicated that there had been no changes in this area. Some participants who reported being 'stealth' in their daily lives discussed restricted expression of emotions (See Box 6.6; 6.1.1). Additionally, some participants made a distinction between

how they express masculinity around cis men versus transmasculine people (6.1.2). For some, the extra time alone at home due to lockdown meant that they could explore their gender expression further without fear of judgement from others (6.1.3). Additionally, one participant discussed gender performativity and questioned whether they change their gender expression depending on if others are around (6.1.4). This brings up interesting questions regarding the purpose of gender roles and why people *do* gender. The previous quote (6.1.3) implies a difference between doing gender for oneself and doing gender for others, and this theme is continued in the participants' distinction between performativity of gender around cisgender men vs. other transmasculine people. This implies multiple functions of gender performativity, both personal and social.

Box 6.6: Theme 6: Impact of COVID-19 on GRC and Masculinities - participant extracts

- **6.1.1.** "Being stealth I do feel under pressure to conform to a masculine role, I try to do more physical jobs at work and to be more emotionally stoic, I generally avoid talking about emotions"  $R_012$ , aged 24.
- **6.1.2.** "I can't always find the words to express my feelings, but when I do it's usually because I'm more comfortable around other trans men than I am with cis men"  $-R_034$ , aged 19.
- **6.1.3.** "Since I'm home alone I am trying to be more comfortable with femininity, since I am the only person who sees it"  $-R_032$ , aged 20
- **6.1.4.** "I'm not sure if I'm more feminine or masculine when I work from home. If someone doesn't see performativity of gender, did it happen? Is gender in the expression or the perception?" *R\_015, aged 35*

### 6.4.7. Impact of COVID-19 on Personality Traits

Positive Impacts. The responses regarding changes to personality traits were varied.

Many participants did not think there had been any impact on their personality, and where

they did it was typically a negative impact. However, some participants who had started

testosterone reported that it had made them more confident and decisive (See Box 6.7; 7.1.1).

Others said that due to being unable to socialise with others they had become more confident

about doing activities alone which had increased their sense of independence (7.1.2).

**Negative Impacts.** It was more common for participants to report that COVID-19 had a negative effect on their personality traits and attitudes to others and the world in general. Some participants reported that they had become cynical and suspicious, particularly of the government, due to their attitudes toward the trans community and the response to COVID-19 (7.2.1). Some people thought they had become anti-social and lazy, due to lockdown restrictions (7.2.2), whereas others had lost self-confidence after struggling with employment and mental health during the pandemic (7.2.3).

Box 6.7: Theme 7: Impact of COVID-19 on Personality Traits - participant extracts

### 7.1. Positive Impact

- **7.1.1.** "I started taking testosterone and it's made it easier to make decisions without second-guessing myself and I now feel more confident"  $R_004$ , aged 40.
- **7.1.2.** "I've become more independent and confident, I've been doing more things without needing another person to be with me which feels really good"  $R_012$ , aged 24.

#### 7.2. Negative Impact

- **7.2.1.** "I feel like I've become quite cynical in general. I've become a lot more suspicious and mistrusting of the government lately. The general transphobia and all the vague or confusing rules in COVID have meant I've been pushed away from other people and a resentment of being told what to do"  $R_028$ , aged 22.
- **7.2.2.** "Now that I've been able to be more anti-social without it being an issue I think I've also become quite lazy"  $R_019$ , aged 30.
- **7.2.3.** "I think I've lost a lot of confidence since losing my job, plus my mental health is worse which hasn't helped"  $R_030$ , aged 32.

# 6.5. TESUP-2 IPA Results

The qualitative TESUP-2 study involved re-interviewing the same participants from

TESUP-1, one year after the original interview. Nine of the original thirteen participants

agreed to take part in the second interview. It is important to note that the AUDIT and

DUDIT scores of two participants in the low-risk group (Darren and Cody) increased to such

a degree that they could be reclassified as high-risk. However, to keep comparisons

consistent, the responses from participants have been organised according to their original grouping of low- or high-risk. The following results were identified across both the low- (L) and high-risk (H) groups and are discussed in a longitudinal context. Overall, four themes (see Table 6.18) were identified across both groups and each theme included up to three sub-themes. The themes are: (1) Navigating Healthcare during COVID-19; (2) Coping with Gender Minority Stress; (3) Psychosocial Influences on Substance Use: Life in Lockdown; and (4) Novelty Seeking and Harm Avoidance in Substance Use. Participant extracts from each group are provided to illustrate each theme.

Themes	Sub-Themes
1. Navigating Health and Healthcare during COVID-19	<ul><li>1.1. Barriers to Transition-Related Healthcare</li><li>1.2. Maintaining Health and Wellbeing during COVID-19</li></ul>
2. Coping with Gender Minority Stress	<ul><li>2.1. The Personal Impact of Pervasive Transphobia in the UK</li><li>2.2. Negotiating Identity: Authenticity, Pride, and Disclosure</li></ul>
<ol> <li>Psychosocial Influences on Substance Use: Life in Lockdown</li> </ol>	<ul> <li>3.1. COVID-19: Isolation, Mental Health, and Substance Use</li> <li>3.2. Distress Intolerance and Maladaptive Coping Mechanisms</li> <li>3.3. Strategies to Reduce Substance Use</li> </ul>
4. Novelty Seeking and Harm Avoidance in Substance Use	4.1. Novelty Seeking and Impulsivity 4.2. Harm Avoidance

Table 6.18. TESUP-2 IPA results: themes and related sub-themes

## 6.5.1. Navigating Health and Healthcare during COVID-19

**Barriers to Transition-Related Healthcare.** Many participants across both the lowand high-risk groups reported that transition-related healthcare had been substantially disrupted due to COVID-19 which resulted in feelings of frustration and hopelessness. For example, Rowan (H; Box 6.8; 8.1), who had received a referral to GIS in 2019, described how he had been told that it would be another two-year wait until he received his first appointment. This news understandably had a negative effect on his mental health, and he reported that he had been considering "DIY" hormones (self-administering hormones not prescribed to him) but was afraid of potential negative consequences from doing this (8.2). Some participants reported that due to COVID-19 their planned surgical interventions had been cancelled or delayed. Owen (H; 8.3) had been attempting to schedule top surgery with a private surgeon but due to the pandemic it had been cancelled three times, which was a significant cause of stress. Similarly, Zack (H; 8.4) had to cancel his top surgery due to testing positive for COVID-19; much to the concern of his friends, this event prompted an increase in alcohol use, and he began smoking cigarettes after previously quitting.

In contrast to Zack's experience of surgical delays negatively impacting his substance use, Jeremy (L; 8.5) reported that, despite his initial concerns expressed in the first interview about surgical cancellations affecting his ability to manage his alcohol use, this was not the case. Citing his improved ability to recognise the risk factors which could trigger an increase in alcohol use, Jeremy reported that the delay in the next stage of lower surgery was "not as big of a thing as I had thought or I had worried it would be".

Participants voiced their concern about the lack of communication between GIS and themselves, and in a similar manner to the first interviews, this lack of communication led some participants, such as Owen (H), to question the motivations and decision-making processes of the GIS (8.6). Furthermore, participants again mentioned feeling like both private and GIS doctors expected a linear, one-size-fits-all process of transition, and that they felt like they were expected to "jump through all these hoops in order" (Ezra, L; 8.7). This led participants feeling "incredibly stressed" and feeling as though they "constantly have to perform" in order to have their gender identity taken seriously by doctors and gain access to transition-related healthcare (Bailey, L; 8.8). **Box 6.8.** Representative extracts for "Barriers to Transition-related Healthcare" in "Navigating Health and Healthcare during COVID-19". Quotes are from both low substance use group (L) and high substance use group (H).

- **8.1** "[my transition] is basically stalled at the moment just because of the NHS wait ... I'm two years behind currently of where they are so they're just now seeing people from October 2017 [...] I got referred in 2019 so it'll be a few years yet" (**Rowan, H**)
- **8.2** "I'm becoming more and more tempted to go DIY with hormones ... just to feel normal ... but I'm so scared of the consequences ... the implications of it ... that's the thing I'm very worried about and ... I dunno I just want to feel ... normal ... but if I do that will it knock me out of healthcare further down the line" (**Rowan, H**)
- **8.3** "I've been having private consultations with multiple surgeons across the globe and there was three surgery dates that I was offered ... but they were all cancelled because of COVID ... so the pandemic has had a big negative impact on being able to access anything like that and it's so frustrating ... and you look at the waiting lists of all the GICs and frankly it's horrendous" (Owen, H)
- **8.4** "It was a few weeks after I'd had COVID and I already had the date confirmed ... but then she said I couldn't get it for like three months after I'd tested negative after that I immediately kind of reached out to my friends in my area because I still couldn't leave the house and I got my friend to buy me some beer some vodka and a packet of cigarettes ... and I got atrociously drunk that night ... and then the next night I got even more drunk ... I drank a disgusting amount of it ... it was kind of my way of dealing with the bad news ... and some of my friends kind of expressed concern like hey you were quite drunk that night and you're getting drunk this night ... and I was like yeah but this is my way of dealing with it I'm really upset about this I'm just gonna get really drunk" (Zack, H)
- **8.5** "[delayed lower surgery] was not as big of a thing as I had thought or had worried it would be ... even though at the moment it's at the point of potentially requiring two more operations ... and is likely to be three years plus ... because I think now I've learned that it's more when a series of stressors build up that tends to push me to the point of ... I will look for ways to harm myself to cope ... but it's been easier to manage those stressors earlier since recognising that" (Jeremy, L)
- **8.6** "I tried to set up an appointment and [the GIS] cancelled it ... then I tried to set up another one but they never got back to me ... and a lot of people have this experience ... so it leaves you wondering what they're doing ... like they aren't accepting new patients but they're not servicing the existing patients and ... like where's the time and resources going at the moment ... like what are they really doing ... the lack of communication just drives you crazy" (**Owen, H**)
- **8.7** "[GIS] have this set way that you're meant to do it ... like you're meant to go through stage A then B then C and you're meant to jump through all these hoops in order" (Ezra, L)
- **8.8** "I just hit roadblock after roadblock and I get incredibly stressed ... I feel like I constantly have to perform ... and if I don't then they will criticise you and ... they keep acting like I'm not taking it seriously enough because I'm not jumping through the hoops quickly enough" (Bailey, L)

# Maintaining Health and Wellbeing During COVID-19. Each participant in both

the low- and high-risk groups described the challenges of maintaining good physical health

and mental wellbeing during the COVID-19 pandemic and the related lockdown measures in

the UK<sup>3</sup>. One participant was briefly homeless and had to sleep on friends' sofas due to

<sup>&</sup>lt;sup>3</sup> The interviews took place in July and August 2021. At this time the participants were beginning to adjust to the removal of local and national lockdown measures after three national lockdowns since March 2020, and varying numbers of local lockdowns, depending on where the participants were located in the UK.

unstable housing in the pandemic (Cody, L; See Box 6.9; 9.1). Other participants experienced unemployment (Jeremy, L; 9.2) which had a substantial negative effect on their mental health. Another participant was struggling to adjust to working from home (Rowan; H; 9.3) which was causing him to feel isolated and resulted in difficulties maintaining his physical health. Additionally, Darren (L) compared being in lockdown to "being in jail" because his "already small world just got even smaller". For Darren (9.4), the consequences of the pandemic such as losing potential friendships and establishments or activities which were shut down, caused him to frequently ruminate about how his life could have been different or better if it were not for COVID-19. Each of these participants reported that the stress of their situations sometimes gave them the urge to drink alcohol to cope with the stress, though they did not always give in to these urges.

However, not all participants reported negative experiences as a result of the pandemic. Parker (H; 9.5) described how initially he was struggling with lockdown measures due to where he was living, but after moving house his mental health improved and he now had a more positive outlook. Another participant, Ezra (L; 9.6) reported that because both himself and his partner had to shield very strictly for much of the pandemic — due to his fiancé being immunocompromised — their relationship has become stronger, and they were able to enjoy their time at home without financial worries because they received COVID grants. It is notable that the participants who did not perceive the pandemic to be an entirely negative experience had a secure financial situation and were in a stable, committed relationship. This is in comparison to participants such as Rowan, Jeremy, Darren, and Cody who reported difficulties in their interpersonal relationships, isolation from living alone or working from home, or unstable housing.

A commonly reported concern in both the low- and high-risk groups was the worry of themselves or their loved ones contracting COVID-19. Rowan (H; 9.7) described how their

worry about getting COVID-19 and accidentally spreading it to people gave them a lot of

anxiety. Similarly, Bailey (L; 9.8) described feeling "paranoid" about leaving the house due

to worries about contracting COVID-19, and so they tended to just stay inside with their

family. This led to increased alcohol use to cope with the stress of caring for their child with

ASD, who was struggling with the changes due to lockdown measures.

**Box 6.9.** Representative extracts for "Maintaining Health and Wellbeing During COVID-19" in "Navigating Health and Healthcare during COVID-19". Quotes are from both low substance use group (L) and high substance use group (H).

- **9.1.** "I was briefly homeless right in the centre of the pandemic ... it was a crossover of tenancies and I couldn't find anywhere ... because of the pandemic ... so I had a few weeks of sleeping on friends' sofas" (**Cody**, **L**)
- **9.2.** "About a year ago I was made redundant during the pandemic ... it was for about three months and I was really struggling ... I'm the type that if I'm not doing something productive then it's a really big stressor for me ... it damaged my mental health quite a lot" (Jeremy, L)
- **9.3.** "I got this job and it's working from home full time and it's a good job but ... I get so tired and when I started this job it became much harder to keep myself exercising regularly ... and it's very isolating because I work from home so some days I don't see another person ... you're just online all day and alone" (**Rowan, H**)
- **9.4.** "I was beginning to make friendships and then suddenly we don't talk anymore ... and my previous [sport team] is totally gone now ... I quite often think about how things could have worked out if everything was normal ... I wonder if I'd have been better off if things had worked out normally ... I just want things back to how they were" (**Darren, L**)
- **9.5.** "It was last winter I was really struggling with my mental health with everything from the lockdown ... and where I was living wasn't good ... there was mould and water damage and kind of a dodgy street ... but since moving I swear to god I've been so happy it's really helped my mental health" (**Parker, H**)
- **9.6.** "my fiancé is immunocompromised ... so every lockdown we have to shield **hard** ...it sucked for a while but we got COVID grants and we didn't struggle financially ... we got some DIY stuff done in the house that we wouldn't have done if COVID hadn't happened ... and I think our relationship has gotten stronger ... we supported each other quite a lot" (**Ezra**, **L**)
- **9.7.** "I have a fear of spreading [COVID-19] to people ... not even the fear of catching it but it's the fear of giving it to other people ... just a feeling of a lot of anxiety when I do go out and it's really off-putting and I just worry the whole time ... I think it's best to just stay in" (**Rowan, H**)
- **9.8.** "I don't really want to go out anywhere because I'm so paranoid now ... even though I'm double vaccinated I don't want to risk it ... and so my youngest ... who's quite autistic ... is having epic meltdowns all the time then my partner gets quite grumpy and doesn't want to deal with anything because he's reached his limit and I've reached mine ... so when you finally get [the child] to bed it's like 'Ah let's have a drink' ... so you have a drink ... and sometimes you have a second drink and it's quite a generous drink (!) ... just to deal with the stress of it all" (**Bailey, L**)

## 6.5.2. Coping with Gender Minority Stress

The Personal Impact of Pervasive Transphobia in the UK. All nine participants in this study had experienced transphobia, the majority of these events had occurred in the past year, since the first interview. For many participants these experiences had an impact on how they perceived themselves and their choice to disclose or not disclose their trans status to others (see section 6.5.2.2.). Some participants described a nationwide sense of antipathy and normalised victimisation toward transgender people, describing the UK as "TERF Island". Ezra (L; see Box 6.10; 10.1) stated that although the increasing levels of transphobia in the UK had not directly affected his transition, it consistently affected him "on an emotional level". Rowan (H; 10.2) also referenced the hostile climate of the UK toward trans people, and the suicide of trans activist, Sophie Gwen Williams<sup>4</sup>, as affecting them and their worries about accessing transition-related healthcare.

Many participants described how a lot of their recent experiences with transphobia occurred in online spaces; through interactions on social media and the increasing focus of UK media and news outlets on debating trans rights. For example, when Parker (H; 10.3) would post videos of themself online they would "rack up thousands of negative comments ... really transphobic stuff", which they found extremely distressing. They also reported that they noticed transphobic comments online had increased in the past year. Bailey (L; 10.4) described how logging onto social media sites such as Facebook and Twitter felt like "wading into a battle" because they felt like they consistently had to defend the existence and rights of transgender people. This led to them feeling "exhausted" and "depressed" and occasionally gave them the urge to cope by drinking alcohol.

<sup>&</sup>lt;sup>4</sup> Sophie Gwen Williams, artist, activist, and co-founder of the trans health charity *We Exist*, committed suicide after being on the waiting list in Northern Ireland for her first appointment for four years, moving to London, and discovering she would face a further wait of five years to access transition-related healthcare.

Although much of the participants' experiences of transphobia occurred online, it also happened in their daily lives too. Jeremy (L; 10.5) detailed his experience of being fired from his job due to being transgender and reported that "these experiences have continued in other jobs as well". Another participant had experienced transphobia at work, but from customers, not their colleagues. Cody (L; 10.6) described how they received transphobic abuse from customers when they were at work in an LGBT café. They stated that it was not until the police were called and the situation was over that it sunk in how scared they felt. Despite not explicitly connecting their substance use to these events, many of these participants reported that they would want to drink or take drugs in response to stress or if they had a bad day.

**Box 6.10.** Representative extracts for "The Personal Impact of Pervasive Transphobia in the UK" in "Coping with Gender Minority Stress". Quotes are from both low substance use group (L) and high substance use group (H).

- **10.1.** "I feel a global sense of discrimination ... not personally directed at me but just from living in the UK ... a lot of people have dubbed it 'TERF Island' ... like with all the court cases and the Kira Bell thing ... on a personal level it hasn't directly affected my transition but it really does affect me on an emotional level" (**Ezra**, **L**)
- **10.2.** "I've been feeling pretty shit lately ... I saw earlier this year on the news about [Sophie Gwen Williams] and I was like this is ridi- I can't ... I'm just so frustrated and tired all the time with this ... and I worry about how long the waiting takes ... what it takes to get to that point that she was at" (**Rowan, H**)
- **10.3.** "I definitely get a lot of abuse online ... especially if I post videos online ... they always rack up thousands of negative comments ... really transphobic stuff ... sometimes even private messages and things *Interviewer: when you experience this online abuse how do you deal with that in the moment*? Parker: sometimes it's okay ... like it's the same shit over and over no one says anything new ... but time where I'm having a bad day or my dysphoria is bad it can be very upsetting ... sometimes I literally have to go to bed to just hide under the covers for an hour to ride out the bad feelings" (**Parker, H**)
- **10.4.** "Facebook is really bad ... and Twitter is just a cesspool ... [transphobia] even comes up in mainstream news and it's quite shocking how mainstream this crap is ... it's exhausting because you log on and you're constantly having to defend your own existence and the existence of people like you ... every day you have to do this ... it's like wading into a battle ... and it just gets to a point where it feels insurmountable and it's exhausting ... it can be a bit depressing for me ... and yeah sometimes I do get to the point where all I want is to get off Facebook and drink a glass of wine" (**Bailey, L**)
- **10.5.** "when I changed my name and started presenting in a masculine way ... [the boss] basically said yeah I'm getting rid of you ... one of the bullshit reasons was that they thought I was going to have a mental heath breakdown and they didn't want to deal with it ... all because I'm trans ... so that was a big stressor and these experiences have continued in other jobs as well" (Jeremy, L)
- **10.6.** "I work at a café that is LGBT focused and recently we had some people yelling and trying to get in ... it was like abuse and slurs aimed at being gay and trans ... but it wasn't until afterwards when my manager phoned the police that I was like 'oh shit ... that was scary' ... it's not happened again but now I feel jumpy when I'm in the shopping centre at night ... it makes you very hyper-aware" (Cody, L)

## Negotiating Identity: Authenticity, Disclosure, and Pride. The negative

experiences of transphobia, discrimination, and victimisation described in the previous subtheme, had a substantial influence on the participants. They disclosed the effects these experiences had on how they perceived their own identity, their feelings about being transgender, and on-going decision-making about to whom they should disclose their trans identity and when it is safe to do so. For example, Bailey (L; see Box 6.11; 11.1) described how an increase in transphobic encounters online has the potential to expand offline, causing anxiety and fear of future discrimination.

Some of the participants described a complicated relationship with their transgender identity, much of it being informed by a fear of receiving negative reactions from other people when they disclose their trans status. For example, Rowan (H; 11.2) experienced transphobia from a colleague who told him that they would "never see [him] as a man", which led to Rowan's decision to not-disclose their trans status in the future and, if questioned, he would tell people that he has a hormone disorder in order to "protect [himself] from assumptions". He also reported that while he has feelings of pride about the history of the trans community, he does not feel pride in his own identity and lamented that "if [he] was born cis then life would be a lot easier" (11.3).

Other participants, who chose not to disclose their trans status to other people, did so in order to be perceived as their authentic self, or, similarly to Rowan, to avoid being 'othered' when people found out that they are trans. However, they indicated that, although they found many benefits in not disclosing to others that they are trans, this also came with risk. For example, Owen (H; 11.4) valued being "stealth" because it gave him "control over [his] own story". However, he also went into detail about some of the risk involved in not disclosing his trans status, such as worries about how people might react if they found out
and concern about personal safety, citing "horror stories of people taking someone home from a club, they find out they're trans and they beat them or try to kill them" (11.5). The stress and anxiety that arose from concealing his trans status sometimes led to an increase in alcohol use for Owen (11.6).

There was one participant whose experiences over the past year had a substantial impact on his identity, beliefs, and feelings about being transgender. In the TESUP-1 interviews, Darren was the only participant who felt anger at being transgender and perceived no positives from his identity or his transition (See Box 5.1; 1.10, 1.11). He reported feeling intense gender dysphoria and described how his rejection from online trans communities and his own internalised transphobia led to an increase in alcohol consumption (See Box 5.3; 3.8). In Darren's second interview his perspective had changed profoundly. He reported that since the TESUP-1 interview he had "distanced [himself] from any kind of LGBT related discussions or groups" and because of that he realised "I don't have to be anything ... I don't have to look or act a certain way to fit in to my community or my culture" (See Box 6.11; 11.7). As a result, he reported that his dysphoria had lessened, and his self-image had improved. Where previously he had implied that people who identified as non-binary were not "actually trans" (See Box 5.3; 3.7), he now disagreed with that idea. He described how he now rejects the label of 'trans' and uses 'he/him' or 'they/them' pronouns, perceives gender roles and gender identity as "too restrictive" and does not "really think of [himself] in terms of gender anymore".

**Box 6.11.** Representative extracts for "Negotiating Identity: Authenticity, Disclosure, and Pride" in "Coping with Gender Minority Stress". Quotes are from both low substance use group (L) and high substance use group (H).

- **11.1.** "My local community has a Facebook group [...] one guy called me a 'lib-tard tranny' ... and I was like 'shit' ... like they live around here somewhere ... like that's a worry ... because I do find that there is a fear of discrimination ... and it makes me anxious ... like that you might come out to someone and they react badly" (**Bailey, L**)
- **11.2.** "In my last job someone said to me "I'll use whatever pronouns but I'll never see you as a man" ... people can just say shit and question you when you're transgender ... and I've decided that from now on if someone asks I'm just gonna say that I have a hormone disorder ... just to protect me from assumptions ... I hate that when people find out you're trans you're not yourself ... you're just a man with an asterisk" (**Rowan, H**)
- **11.3.** "*Interviewer: How do you feel about being transgender?* Rowan: I feel pride in the history ... like all the people who came before me and the community history ... but other than that I feel frustrated ... because if I was born cis then life would be a lot easier" (**Rowan, H**)
- **11.4.** "The appeal of being able to be stealth is the fact that I have control over my story and how I tell my story ... rather than my story being told for me by the way I look for example and being stealth I find I can relate to others easier because we have something in common ... and that being the identity of being a man [...] because I feel like the label of trans is so arbitrary and ... like in the same way my nationality is not a core part of my identity ... it just doesn't convey much information about who I am as a person ... whereas identities which relate to my interests and what I do give a lot more relevant information" (**Owen, H**)
- **11.5.** "Obviously there are downsides to being stealth … like a worry of what will they do if they find out … will they feel like I lied to them … will they start to see me as a medical curiosity or treat me as a man asterisk essentially … and you worry because there's these horror stories of people taking someone home from a club, they find out they're trans and they beat them or try to kill them … so it does make you worry … and then the longer it goes on the more stressful it becomes because you know that if they were to find out they will be mad at you … so you're risking your relationship with that person" (**Owen, H**)
- **11.6.** "I am an anxious person so having a few beers will help me relax in social situations ... like when I joined a sports team ... I felt quite intimidated like ... will they clock me ... like I wanted to socialise with these guys but I was so guarded ... so having a few beers helps me relax about that stuff ... and then you know ... if anything bad happened which it didn't ... but previously when I have experienced transphobia then having a drink is a way to chill out ... so I don't have to deal with it" (**Owen, H**)
- **11.7.** "Over the past year I've distanced myself from any kind of LGBT related discussions or groups ... I'm just doing stuff my own way now ... and since then I've realised that I don't **have** to be anything ... I don't have to look or act a certain way to fit in to my community or my culture ... I used to feel like there was something wrong with me because I didn't have things in common with other trans people ... but since I stopped being involved in all that I feel a lot better about myself ... I don't get as much dysphoria and I've gotten a better self-image" (**Darren, L**)

## 6.5.3. Psychosocial Influences on Substance Use: Life in Lockdown

# COVID-19: Isolation, Mental Health, and Substance Use. All but two participants

(Ezra, L and Bailey, L) reported that over the course of the pandemic and throughout national

and local lockdown measures, their substance use had increased. For many of these

participants an increase in substance use coincided with struggles with their mental health,

and those with pre-existing mental health diagnoses reported a worsening in symptoms which they thought had been exacerbated by the pandemic.

Darren (L; see Box 6.12; 12.1), who has a diagnosis of ASD, found it difficult to adjust to the frequent changes during the pandemic (such as changes to social distancing, mask wearing etc.), and reported high levels of anxiety because of his "intolerance of uncertainty". Although he denied that he would turn to alcohol to cope with negative emotions (12.2), he did recall having "some drinks" while watching football to relax and "to do something that [he] thinks other people do" (12.3). Additionally, he disclosed that when he is "struggling with [his] mental health" he is more likely to turn to alcohol to "slow down [his] thoughts and reset [his] brain a bit", although he admitted that this sometimes makes him feel worse (12.4). Rowan (H; 12.5) reported a similar experience of turning to alcohol to cope with the frustration and confusion he was feeling about the pandemic but finding that his "bad feelings would come back anyway".

Some participants reported that their mental health and substance use were interconnected; when they struggled with one, they also struggled with the other. They also reported that being in lockdown would prompt a worsening of both mental health and substance use. For example, Zack (H; 12.6) described how his mental health and level of substance use would often fluctuate. He identified the UK lockdown measures as an event that triggered struggles with his mental health and an increase in using alcohol and drugs to cope with isolation and boredom. Similarly, Owen (H; 12.7) disclosed that "since the pandemic hit [he is] drinking and consuming drugs a lot more than [he] used to" and that he frequently has a desire to consume alcohol and drugs; he believed this was "a symptom of [his] mental health not being as good as it used to". Additionally, Owen (12.8) described that due to being unable to socialise and feeling isolated, he would drink or smoke cannabis alone in his room in order to "emulate a social experience without actually having anyone to

209

socialise with". He described his increased substance use as "a really easy brute force way to turn [his] brain off" and reported that he turned to alcohol and drug use to cope with negative emotions because his previous coping mechanisms, such as socialising and swimming, were not available due to the pandemic (12.9). Likewise, Jeremy (L; 12.10) reported that he also felt like his typical coping mechanisms were not available to him during the pandemic, and that increased stress and interpersonal issues as a result of the pandemic made him more inclined to use substances and engage in binge-eating as a way to cope.

Some participants who had experienced an increase in substance use whilst in lockdown reported that they consequently found it difficult to reduce their use when they realised it was having a negative effect on their lives. Parker (H; 12.11) described how at the beginning of the pandemic they found that drinking regularly, during the day, or alone had become normalised through social media, and this behaviour continued throughout the national lockdowns. They reported that they then struggled to stop drinking when they realised it had become a problem for them, and they found it difficult to break out of their daily routine of drinking alcohol (12.12). Cody (L; 12.13) reported that they had begun drinking more in the first national lockdown. When those measures were eased, and they were able to go to pubs with their friends their substance use increased even more, and they found it difficult to stop. They stated that this increase in alcohol use didn't bother them "until [they] get the withdrawal" when they could not access it.

# **Box 6.12.** Representative extracts for "COVID-19: Isolation, Mental Health, and Substance Use" in "Psychosocial Influences on Substance Use: Life in Lockdown". Quotes are from both low substance use group (L) and high substance use group (H)

- **12.1.** "I have a lot of anxiety lately ... I have an intolerance of uncertainty ... and that's a big thing I like knowing what's going to happen and how things work ... and all this stuff of the random rules that keep changing ... I'm just not keen on all that like suddenly one thing is normal then the next you're told that you can't do that ... you can't plan anything or look toward the future because it all might suddenly disappear" (**Darren, L**)
- 12.2. "I only ever drink because I like that taste and that's all it is ... I don't use it as a coping mechanism ever" (Darren, L)
- 12.3. "A few weeks ago I had some drinks and watched the football ... it was nice to chill out a bit and have some time to ... I don't really want to say it was to feel like a normal person ... but I'm not really into football ... but it was kind of just to do something that I think other people do and ... try relax or something" (Darren, L)
- **12.4.** "When I'm struggling with my mental health and my thoughts are going too fast then ... I know with alcohol it can kind of ... slow down my thoughts and reset my brain a bit ... it's like a diversion to see if it helps but I know it might make things worse sometimes so I'm hesitant about doing that sometimes" (**Darren, L**)
- 12.5. "At the beginning of the pandemic especially ... I felt so frustrated and didn't get what was going on I just wanted to feel better so I would just get up and drink ... as a way to like regulate my bad moods ... I know it's a bad thing to wake up at 10 o'clock in the morning and have a glass of wine or whatever ... but I felt so lost ... but it didn't even make me feel better ... my bad feelings would come back anyway (**Rowan, H**)
- **12.6.** "I've periodically gone through times of not really drinking or doing drugs back to doing excessive amounts of both ... because of lockdown ... I mean it's not great for your mental health and mine would get worse being stuck in the house ... then you're stuck inside and bored and can't do anything so you may as well have a drink or ... do a line and watch cartoons ... or smoke some weed just to do something and feel better (**Zack, H**)
- **12.7.** "Since the pandemic hit I'm drinking and consuming drugs a lot more than I used to ... from what I've seen from others I think that's a common experience during the pandemic ... it's a bit worrying because it feels like it's a symptom of my mental health not being as good as it used to be because now it's becoming more of a constant wish to drink alcohol or have drugs essentially" (**Owen, H**)
- **12.8.** "Before the pandemic I would regularly drink on the weekend with my girlfriend ... but over the pandemic I can't do that and ... like if I'm in the mood to drink and socialise I will just do that on my own in a desperate attempt to emulate a social experience without actually having anyone to socialise with ... or I will smoke weed alone in my room and watch something that I would usually watch with friends ... just to simulate that experience" (**Owen, H**)
- **12.9.** "I'm a naturally anxious person and I overthink a lot ... so I think that having a substance like alcohol and weed is just ... a really easy brute force way to turn my brain off ... because usually the coping mechanisms and distractions I would have outside of the pandemic aren't there anymore ... like I would meet up with a friend or go for a swim but I can't do that ... so I've had to find a new way to distract myself essentially" (**Owen, H**)
- 12.10. "lockdown was very much a stressor ... everything about how I lived had changed and with the difficulties in my relationship with my partner ... he described it as he can't cope with a long-distance relationship and he kind of thought 'why bother' with it all ... and so I was feeling abandoned on that front ... it was a trigger into ... a slip into bad behaviour with alcohol ... to get to a level where I don't have to think about these things ... and even with things like binge-eating worsened for a time" (Jeremy, L)
- 12.11. "I started drinking more at the beginning of the first lockdown ... I think the boredom and the stress ... and I think seeing other people doing it too ... I felt like everyone was drinking more and I saw it on Facebook like people ordering cocktails to their house in the morning ... or being on a Zoom call you would just always get a drink ... like people would drink alone on a Monday morning and you had that exposure to it ... like this is now a normal thing and this drinking culture had been intensified" (**Parker, H**)
- 12.12. "This drinking continued throughout lockdown ... and it was only a week before I started work again I tried to stop ... I was drinking every single day and as soon as I started I couldn't stop ... day to day I find it hard to go without a drink and I was increasing the amount too because my tolerance was increasing" (Parker, H)
- 12.13. "In the first lockdown it was like if I had a hard day then I would sit in the house and drink ... then when the pubs reopened it was like finally you can go out and you can see your friends ... so now that's what I want to do all the time ... like we started going every night and then it's like even if we don't go to the pub we still drink every night ... but it's 100% social and it doesn't bother me until I'm getting the withdrawal from it ... then it's shit" (Cody, L)

**Distress Intolerance and Maladaptive Coping Mechanisms**. As stated in the previous sub-theme, many of the participants reported distressing and stressful experiences related to the pandemic, and because some of their usual methods of coping with stress were not available to them, sometimes they would turn to substance use to cope. Additionally, the same participants mentioned situations from TESUP-1 that would prompt substance use, showing that they were facing similar difficulties and responded to them in a consistent manner.

Rowan (H) was still struggling to manage his symptoms of ADHD which had been mentioned in the TESUP-1 interview (see Box 5.8; 8.3). His difficulties with executive functioning affected his mood during the day, and his sleeping problems at night sometimes led him to combine sleeping tablets and alcohol to get some restful sleep (See Box 6.13; 13.1). Similarly, Parker (H) described in the TESUP-1 interview (See Box 5.6; 6.2) how interpersonal difficulties with their partner would trigger the urge to drink alcohol to cope. In the TESUP-2 interview they expanded on this, explaining that their difficulties in managing stress and anxiety arising from their relationship would lead them to use alcohol or Valium in order to try to sleep, "manage overwhelming emotions", or to "feel more in control" (see Box 6.13; 13.2).

For some participants, substance use was directly related to their experiences as a transmasculine person. For example, Owen (H; 13.3) reported that drinking alcohol would give him the confidence to come out as trans to people because he would not focus on any potential negative outcomes. Additionally, Owen (13.4) described how smoking cannabis would reduce his dysphoria, because it would "create an artificially strange experience in [his] body" which would distract him from focusing on bodily dysphoria. Zack (H) also related his substance use issues with being transgender. In the TEP1 interview (see Box 5.8; 8.5) he reported that substance use was sometimes a form of self-harm, but also that it was

connected to the pursuit of masculinity. In the TEP2 interview, Zack (see Box 6.13; 13.5)

echoed his previous statements about substance use being a form of self-harm; he felt that

because he is transgender it is "easier to get depressed". He said that when he is in a

"depressive episode" he will engage in excessive substance use as "a mixture of self-

punishment and escapism". When he would be experiencing a depressive episode, he

described feeling like "the world doesn't care about me and I don't care about the world ...

so I'm just gonna destroy myself". He stated that this feeling had been so consistent

throughout his life it had become "comfortable" (13.6).

**Box 6.13.** Representative extracts for "Distress Intolerance and Maladaptive coping Mechanisms" in "Psychosocial Influences on Substance Use: Life in Lockdown". Quotes are from both low substance use group (L) and high substance use group (H).

- **13.1.** "At the minute I've been struggling with executive dysfunction during the day which affects my mental health ... it gets really fatiguing ... but then it gets to night and I can't sleep ... I know you're not supposed to take sleeping pill while you're drinking ... but the only time I can ever get good sleep is when I have the pills ... and I'll have that with some alcohol to help it work better" (**Rowan, H**)
- **13.2.** "My wife went out for lunch and didn't come back until like three in the morning ... and she does this sometimes and so I was getting super panicky by midnight and was just in a general panic ... kind of a mix of concern and being a bit annoyed ... so I ended up taking a Valium that I was keeping and had a little drink ... I just wanted to sleep really and just to manage overwhelming emotions and ... just feel more in control" (**Parker, H**)
- **13.3.** "At the thought of coming out ... my brain going into a billion directions of what horrible events could go down ... but once I've had a few drinks then that quiets down and ... I can speak freely without thinking so much of the consequences" (**Owen, H**)
- **13.4.** "I've noticed when I have a joint it will create an artificially strange experience in my body and it makes it much easier to forget about the ... like it will stop me from lingering on the less fun bits ... it's a very easy way to trick my brain into not thinking about the bits that cause dysphoria essentially" (**Owen, H**)
- **13.5.** "I think that being trans it's much easier to get quite depressed ... and when I receive bad news in my transition it makes me go 'fuck it' with the drinking again ... I just kinda say 'fuck it' and I just become destructive ... I guess because I'm sad and annoyed and I just want to take it out on myself when I get into a depressive episode ... I think it's a mixture of self-punishment and escapism" (**Zack**, **H**)
- **13.6.** "When I'm in a depressive episode it's like ... a whirlwind of pain ... and it gets quite comfortable when you think like ... the world doesn't care about me and I don't care about the world ... so I'm just gonna destroy myself ... *Interviewer: Why would you say that it feels comfortable?* Zack: I think because for so much of my life ... it's just been the status quo for so long it's just familiar ... spending so much of my life depressed or facing so much alcohol and substance abuse ... it's just comfortable like that" (Zack, H)

**Strategies to Reduce Substance Use.** Although substance use had increased in the year separating the two interviews for most of the participants in this study, many of them also recognised that it had become problematic for them in their daily lives and described the steps that they took to reduce their substance use.

As stated in the sub-theme "COVID-19: Isolation, Mental Health, and Substance Use", Parker (H; see Box 6.12; 12.11) found it difficult to stop drinking once they had started and described how their alcohol consumption had increased during lockdown. However, they had successfully reduced their use over time by employing various tactics such as avoiding situations where they might purchase alcohol (see Box 6.14; 14.1) and finding a productive distraction such as gardening or gaming when they felt the urge to drink (14.2). Ezra (L; 14.3), who did not experience an increase in substance use, attributed this to starting testosterone and not wanting to drink on his own or alone with his partner.

Some participants who had been struggling with both their mental health and substance use found that learning about what might trigger an episode of poor mental health, or seeking help for their symptoms, would have a positive impact on their ability to reduce substance use. For example, Jeremy (L; 14.4) had recently been diagnosed with borderline personality disorder (BPD) and reported that he had been able to better manage his substance use by identifying situations that typically caused a "flare-up of symptoms". Additionally, Zack (H; 14.5) had begun to try to reach out to friends in times of crisis and found it to be a constructive coping mechanism. Zack (14.6) also emphasised that despite his belief that being transgender had a negative impact on his mental health and substance use, he perceived his mental health to be continuously improving throughout his transition journey and the gender euphoria that he feels through this is a substantial positive influence on his life. He indicated that barriers in accessing transition-related healthcare had a negative effect on his substance use, but that he was proud of being transgender and that transitioning has had a "beyond

phenomenal" effect on his quality of life (14.7).

**Box 6.14.** Representative extracts for "Strategies to Reduce Substance Use" in "Psychosocial Influences on Substance Use: Life in Lockdown". Quotes are from both low substance use group (L) and high substance use group (H).

- **14.1.** "Sometimes I won't go to the shop for food because I know if I go then I'll buy alcohol ... so I avoid putting myself in that situation ... so I'll just get a takeaway instead or something and like if I was at a party with people doing lines or something ... then I know I'd have to leave" (**Parker, H**)
- **14.2.** I've got an allotment so sometimes I would go there ... plant something and check up on things ... just a distraction like ... quite often I will play PlayStation instead" (**Parker, H**)
- **14.3.** "I would say my alcohol intake has lessened since we last spoke ... largely because I've started testosterone and ... they have to monitor your liver and kidneys so that's always in the back of my mind ... as well as like ... the idea of myself and [David; fiancé] sitting in the house drinking by ourselves [laughs] it's not really appealing to me" (Ezra, L)
- **14.4.** "It was suggested to me that I might have borderline personality disorder ... and looking into it ... it almost felt like a relief now that there's a kind of known entity and it has made things like substance use a lot easier to manage ... by becoming more aware of what triggers a flare up of symptoms I can look at it more logically ... although it's still a challenge at the moment ... it has been easier in terms of alcohol to avoid overdoing it (**Jeremy, L**)
- 14.5. "I've been trying to reach out to friends a bit more ... I mean I don't want to worry them ... but I realised recently that isolating myself isn't the best ... and talking with them and reaching out a bit more is more constructive" (Zack, H)
- **14.6.** "I just wanted to say as difficult and as bad as my mental health has been as well as the substance abuse issues ... that transitioning has impacted my life beyond positively ... like every time I reach a transition milestone I feel so much euphoria and it does improve my mental health" (**Zack**, **H**)
- **14.7.** "My quality of life has gotten so much better since I've been able to transition ... and it's the barriers to transitioning ... especially in the past year with COVID that do have a negative effect ... but I'm very very proud of being trans and I feel so incredibly happy ... as painful as it is to not get the healthcare you need ... it's just very beautiful to be your authentic self and the effect of transitioning on my quality of life is beyond phenomenal" (**Zack, H**)

# 6.5.4. Novelty Seeking and Harm Avoidance in Substance Use

Novelty Seeking and Impulsivity. Following on from findings in TESUP-1 which

indicated that novelty-seeking and impulsivity were related to substance use, all participants

in the TESUP-2 interviews were asked whether they considered themselves to be an

impulsive person. It is noteworthy that all four participants in the high-risk group said that

they did think they act impulsively and that this impacts their substance use. Additionally, the

majority of participants in the low-risk group said that they did not consider themselves to be

impulsive, although it is interesting that Cody (whose AUDIT and DUDIT scores increased from TESUP-1 to TESUP-2, moving them from low- to high-risk) did consider themself to be impulsive, particularly in relation to substance use.

All participants in the high-risk group considered themselves to be impulsive with substance use. Zack (H; see Box 6.15; 15.1) reported that he is "incredibly" impulsive and that he will often take substances to the point of "blackout drunk" which has resulted in consequences such as "nice shaky tattoos from far too much alcohol". Owen (H; 15.2) also discussed consequences of substance use-related impulsivity, such as saying or doing things that he would not have done if he was sober. Rowan (H; 15.3) believed his impulsivity stemmed from ADHD and being unable to "see negative consequences in the long term". Rowan's response was similar to Cody's (L; 15.4) who also reported that they "don't actually think about the consequences of what [they're] drinking or doing" which led to increased substance use.

Jeremy (L; 15.5) reported that he could sometimes be impulsive with alcohol but that he believed that this was related to his BPD symptoms in that he was only impulsive when he was feeling "heightened emotions". However, his concerns about personal safety and concerns about becoming the victim of transphobic violence (on a night out while drinking, for example) typically prevented impulsive behaviours to do with alcohol. **Box 6.15.** Representative extracts for "Novelty Seeking and Impulsivity" in "Novelty Seeking and Harm Avoidance in Substance Use". Quotes are from both low substance use group (L) and high substance use group (H).

- **15.1.** "I think I'm incredibly impulsive ... I find it very difficult to manage my impulses and so I like completely just get blackout drunk quite a lot ... and because I have my own tattoo gun I now have some very nice shaky tattoos from far too much alcohol" (**Zack, H**)
- **15.2.** "There are times that I just drank too much on an impulse ... and those are the times that I might say stuff that is embarrassing or do stuff that I wouldn't do while sober" (**Owen, H**)
- **15.3.** "I think I am impulsive ... because I have ADHD I can't see negative consequences in the long term ... so it's very difficult for me to not do something ... like in the way I think and the things I say ... if I think about something then I really have to have it so in that way I'm impulsive with alcohol too" (**Rowan, H**)
- **15.4.** "I'm definitely impulsive when it comes to like alcohol and drugs ... I don't actually think about the consequences of what I'm drinking or doing ... I don't think like 'will this impact me later' ... I don't think about what it means or how I feel or anything I just do it" (**Cody, L**)
- **15.5.** "For me I think that when I **am** impulsive ... typically it's related to symptoms in that I have to feel very heightened emotions ... and then I might indulge in alcohol in an unhealthy way ... but I am quite analytical to the point of being immobilised and this typically overrides the impulsivity ... for example if I were to be on a night out and there's an opportunity for a no-strings attached hook-up ... well safety is a big thing and being quite a small person who is physically vulnerable ... and the UK laws about fraud for example it would prevent the impulsivity" (**Jeremy, L**)

**Harm Avoidance**. In a similar manner to Jeremy, the remaining participants in the low-risk group (with the exception of Cody), considered themselves to be quite cautious people, and often stated that their safety and wellbeing was a major factor in deciding whether or not to engage in substance use. For example, Ezra (L; see Box 6.16; 16.1) described himself as being "anxious or paranoid" about the potential negative effect of drugs, and so would "err on the side of caution" and not engage in drug use.

Physical and mental health was a prominent consideration for the participants in the low-risk group; this was not mentioned as a concern for participants in the high-risk group. For example, Darren (L; 16.2) was conscious of the possible negative influence that alcohol can have on his mental health and would "be mindful of how [he was] feeling that day and whether it is appropriate to drink". Similarly, Bailey (L; 16.3) was concerned about getting headaches from drinking alcohol (they previously had struggled with migraines in the past) and would not drink if they thought "it isn't worth the headache". However, they did state that they believed themselves to be susceptible to peer pressure and would "just go along with it" if their family or friends were drinking, regardless of their personal concerns about their health (16.4). Bailey's attitude of "just [going] along with it" may be reflective of a different type of harm avoidance, rather than avoidance of personal harm, they may be avoiding social harm; not wanting to be seen as different or become othered for refusing to drink. They reported previous experiences of this when they had to abstain from alcohol for a few years due to medication and disclosed that it affected their social life and that they "lost their entire social circle" (16.5).

**Box 6.16.** Representative extracts for "Harm Avoidance" in "Novelty Seeking and Harm Avoidance in Substance Use". Quotes are from both low substance use group (L) and high substance use group (H).

- **16.2.** "I think I'm quite cautious really ... a big thing is ... like how my mental health is doing ... so typically I will be mindful of how I'm feeling that day and whether it is appropriate to drink ... I try to do that with everything like whether I'm eating healthy food or too much sugar ... it's good to keep an eye on it" (**Darren, L**)
- **16.3.** "Interviewer: Do you think you can be an impulsive person or not? Bailey: no not really ... I do sit and think about things I'm doing Interviewer: Do you think this influences the way you consume alcohol? Bailey: yeah I think so ... I do sit and think like ... is it worth the headache ... I don't want to feel unwell you know ... so like I will question 'do I actually want to drink" (**Bailey, L**)
- **16.4.** "I think I'm fairly susceptible to peer pressure to be honest ... if people around me are drinking I'll just go along with it ... it becomes very passive and ... my family will drink a lot of wine and it's a very passive thing of having my glass refilled and ... yeah I'll just go along with it really" (**Bailey**, **L**)

**16.5.** "When I went teetotal because of my medical needs I lost my entire social circle ... they maybe thought I wasn't fun anymore or ... I don't know ... but they stopped inviting me out and I got quite bitter about it at the time" (**Bailey**, **L**)

**<sup>16.1.</sup>** "I think I typically err on the side of caution ... with alcohol I know where my lines are and I tend not to cross them ... and with drugs I'm always like ... anxious or paranoid that I'll have a bad trip and ruin the mood" (**Ezra, L**)

#### 6.6. TESUP-2 Discussion

This study investigated correlates and experiences of substance use in transmasculine individuals. TESUP-1 highlighted areas of concern regarding gender minority stress, psychopathological symptoms, access to healthcare, masculinities and social expectations, and the influence of personality traits on substance use. The aim of TESUP-2 was to investigate changes to substance use over time and explore how these areas of concern contributed to substance use in transmasculine individuals. The COVID-19 pandemic took place in between TESUP-1 and TESUP-2, therefore the concerns of participants and the effect of the variables which were investigated were inherently influenced by this event. This changed the original aim of the study from exploring naturally occurring change to substance use over time, and any potential influence of stage of transition, to exploring the impact of the COVID-19 pandemic on substance use.

The central research question for this study was: What are the factors that contribute to the development and maintenance of substance use in transmasculine individuals? In TESUP-2 the main research question was: How did the COVID-19 pandemic affect substance use in transmasculine individuals?

For the quantitative stage, 82 participants out of 105 from the TESUP-1 completed the same questionnaire as they had one year earlier. The TESUP-2 questionnaire included free-text boxes to enable further investigation in change over time, this data was analysed using thematic analysis (TA). For the following qualitative stage nine participants out of the 13 who were also interviewed in TESUP-1 took part in a follow-up interview, this data was analysed using interpretative phenomenological analysis (IPA). The TESUP-2 interviews focused on exploring the participants' perspective on changes in their life since the first interview, in addition to discussing topics specific to each individual which were based on the issues that they had emphasised in TESUP-1.

This chapter presents an integrated discussion of quantitative and qualitative findings in a longitudinal context. First the TESUP-2 findings are presented and discussed in a longitudinal context. To compare results from TESUP-1 to TESUP-2, all findings are presented in the context of the same four core themes from TESUP-1 (The core themes are: Transition, Masculinities, and the Function of Substance Use; Contexts of Substance Use and Meaning-Making; Risk Factors in Substance Use; and Protective Factors in Substance Use). Second, to consider new findings which emerged from TESUP-2, and incorporate findings from TESUP-1, each core theme in TESUP-2 contains different sub-themes from TESUP-1 which are discussed. Finally, the chapter concludes with a discussion of practical implications, recommendations for future research, and limitations.

# 6.6.1. A General Overview of Changes to Substance Use

Before beginning to discuss the correlates and experiences of change in substance use, I will first outline how the quantity and frequency of tobacco, alcohol, and drug use changed between TESUP-1 and TESUP-2.

**Tobacco Use.** A one-way repeated measures ANOVA indicated that the rate of current tobacco use had not significantly changed. In TESUP-1 (N = 82) 41 participants reported current use of tobacco, compared with 43 participants in TESUP-2 (N = 82). In both phases smoking manufactured or hand-rolled cigarettes were the most common methods of consumption. The thematic analysis of text-box data in the questionnaire demonstrated support for these results; most participants did not report experiencing any change in their tobacco use. Some participants reported an increase in frequency of tobacco use due to anxiety related to long waits for GIC appointments, or because they typically smoked when using alcohol and their alcohol use had increased. Studies which have focused on smoking

behaviours during the COVID-19 pandemic have indicated that individuals who smoke daily experienced an increase in the frequency of tobacco use, whereas those who do not smoke every day did not experience any change (Busse et al., 2021; Gendall et al., 2021). This is similar to the findings of the current study; the majority of participants were not daily smokers, which may be why there was no significant change in tobacco use. Additionally, in the TA results, those who reported an increase in tobacco use frequency had typically indicated that they smoked tobacco daily and had increased their use in response to increased stress.

Alcohol Use. The TESUP-2 questionnaire results indicated that both the mean AUDIT score had increased (indicating more harmful or dependent alcohol use) in addition to the number of participants who met these criteria increasing, which indicates that participants who previously did not meet the criteria for high-risk alcohol use now did one year later. A one-way repeated measures ANOVA indicated a statistically significant increase of 4.7 points in AUDIT scores between TESUP-1 (M = 7.26, SD = 6.21) and TESUP-2 (M =11.96, SD = 8.81). These results support a recent study which indicated that COVID-19 lockdown measures were a risk factor for increased alcohol consumption in people with alcohol use disorders, as well as alcohol use in people who were previously abstinent (Kim et al., 2020). The findings from Kim et al. support why some participants who previously did not have an alcohol use disorder did in TESUP-2 (an additional 13 participants in TESUP-2 reached the cut-off of 8 points, indicating high-risk alcohol use). TA results indicated that 'casual' daily drinking was commonplace during lockdown, and some participants engaged in frequent binge-drinking. Many of these participants indicated that this increase in alcohol use had a negative effect on their mental health. For the few participants who reported a decrease in alcohol use, this was typically because their ability to socialise had been limited due to lockdown, and they only drank alcohol in social situations. These experiences and the factors

which contributed to a change in alcohol use will be discussed in more detail in the following sections.

**Drug Use.** The TESUP-2 questionnaire results indicated that the mean DUDIT score had significantly increased by 5.87 points, from 5.70 to 10.98 points. Similar to the AUDIT results, an additional 13 participants met the cut-off score of 6 points, indicating high-risk drug use. The most common drug of choice in both TESUP-1 and TESUP-2 was cannabis, though the number of participants who reported cannabis use increased (TESUP-1 N = 25, TESUP-2 N = 38). This result indicates that individuals who did use drugs, but not cannabis, had begun using cannabis in the year between TESUP-1 and TESUP-2. Additionally, the frequencies of the use of different drugs indicates that while the use of cannabis increased, the use of stimulants (such as MDMA, ecstasy etc.) had decreased which could indicate that participants had changed the type of drug they typically use due to COVID-19 lockdown measures. This is supported by initial investigations into drug use patterns which indicate a shift from drugs such as MDMA to more cannabis and psychoactive substances (Jemberie et al., 2020). This finding is further supported by the TA results; participants who reported drug use in TESUP-1 indicated that they had increased their use of cannabis, largely due to lockdown and related increases in stress. This finding is mirrored in recent studies which have reported that cannabis use was particularly increased for individuals who were selfisolating during COVID-19 and were using cannabis as a coping mechanism (Bartel et al., 2020). It is notable that in TESUP-1 no participants indicated the use of heroin, but in TESUP-2, four participants reported using heroin. While it is possible that they had begun using heroin during the pandemic, it may also be the case that they elected not to disclose this information during TESUP-1. Participant experiences of drug use, and the factors which contributed to a change in drug use, will be discussed in more detail in the following sections.

# 6.6.2. Transition, Masculinities, and the Function of Substance Use

Stage of Transition and Access to Transition-Related Healthcare. Previous studies looking at treatment-seeking transgender people have reported that when transgender people receive the medical interventions they require (such as HRT or surgery), their mental health and quality of life improves (Nobili et al., 2018; Baker et al., 2021). Based on these findings, it was initially hypothesised that individuals at a later 'stage' of transition, who have potentially had more transition-related medical interventions, would have lower levels of alcohol and drug use. It was also hypothesised that this effect would be seen in the longitudinal analysis between TESUP-1 and TESUP-2; the prediction was that participants who had accessed transition-related healthcare between the two phases would show a reduction in substance use. However, neither of these hypotheses were confirmed through quantitative analysis of TESUP-1 and TESUP-2 data.

As was mentioned in the TESUP-1 discussion (see *5.4.2.1. Stage of Transition*), the operationalisation of the concept of 'stage of transition' was not as robust as originally thought and therefore the results are limited by the subjective nature of 'stage of transition', in addition to the self-defined category options (ranging from pre- to post-transition). In a similar manner to TESUP-1, the correlation analysis of 'stage of transition' found no associations with substance use. However, a comparison of the descriptive statistics regarding stage of transition (Table 3.3) reveals that very few participants had categorised themselves as being in a different stage in TESUP-2 than they were in TESUP-1. Taking into account that 23 participants did not complete the TESUP-2 questionnaire, these results reveal that only two participants re-categorised themselves from pre-transition to early transition (both indicating that they had started testosterone). Additionally, four participants re-categorised themselves from mid-transition to late transition, and one participant changed category from mid-transition. Overall, only seven participants out of 82 reported any

changes to their self-defined stage of transition in the year between questionnaires, which is much lower than was expected.

The TA and IPA results indicate that most of the participants who had experienced delays in their transition said that this was due to the COVID-19 pandemic. Participants reported that access to healthcare had been severely reduced as a consequence of the pandemic, and for some, on-going transition-related healthcare (such as HRT or surgical interventions) had been delayed or halted. Access to GIS had been limited and the resulting longer waiting times left participants feeling hopeless about the progress of their transition; this can be seen as reflected in the significant increase in Negative Expectations for the Future (NE) from the GMS which includes expectations of being denied appropriate medical care. Furthermore, the resultant anxiety and uncertainty of their situation can be seen in the significant increase in anxiety symptoms and GSI (an overall measure of severity of distress). Due to increased waiting times for initial GIS appointments, some participants, such as Rowan (H = High-risk group), considered self-medicating (or "DIY") HRT. van der Miesen et al.'s (2020) research about access to transition-related healthcare during the pandemic highlights that reduced access and longer waiting times may have resulted in an increase in self-medication of transition-related medication, and higher levels of stress, and anxiety; all of which are seen in the present study.

The barriers to transition-related healthcare which were reported as being largely due to COVID-19 restrictions compounded the existing difficulties which were expressed by participants in TESUP-1. In a similar manner to the first interviews, multiple participants raised the issue of poor communication between GIS and themselves, which led to feelings of suspicion and a general lack of trust. The language used by participants in TESUP-2 to describe their experiences was strikingly similar to TESUP-1, the same phrases were used by participants, such as "you're meant to jump through all these hoops in order" (Ezra, L = Low-

risk group) and "I feel like I constantly have to perform" (Bailey, L). These extracts not only reveal the disconnect between service-user and GIS perceptions of what it means to be trans, but also the consistency between phases, and how often it was mentioned, indicates that this is an on-going problem for many transmasculine individuals seeking transition-related medical interventions. Due to these barriers to healthcare increasing the potential for harm to transgender individuals, it is recommended that when reintroducing healthcare as normal post-COVID, gender-affirming care and access to transition-related treatment should be prioritised, and increased communication with service-users should be considered crucial to mitigate anxiety and stress arising from longer waiting times.

Additionally, for some participants these delays and cancellations resulted in increased substance use. For example, when Zack's (H) top surgery was cancelled because he tested positive for COVID-19 his alcohol use increased and he began smoking cigarettes again as a way to cope with not getting the surgery he needed. Interestingly, in contrast to Zack's experience, when Jeremy (L) experienced delays in getting the next stage of phalloplasty, he reported that it was "not as big of a thing as I had thought". In TESUP-1 Jeremy had concerns that any delay in surgery would cause an increase in alcohol use, but for him this was not the case. It is notable that Zack was in the high-risk group and Jeremy was in the low-risk group; this might partially explain their differing reactions to a similar event. Furthermore, the concept of distress (in)tolerance may be a contributing factor to why Zack experienced an increase in substance use, but Jeremy did not. Throughout his interview, Zack reported multiple instances of struggling to cope with stress and would typically use alcohol to or drugs to regulate his emotions. However, while Jeremy did report experiencing some difficulties in tolerating distress, he described how he had recently been able to better manage his substance use through introspection and regular sessions with his therapist. These results build on the findings of TESUP-1, which indicated that distress intolerance was a common

difficulty for transmasculine individuals and provides further support to the suggestion that DBT-ST (Warner & Murphy, 2021) may be a beneficial psychological intervention for this community. It may be particularly helpful because numerous participants indicated that they felt they had no healthy coping methods, or their usual ones were unavailable due to lockdown measures. DBT-ST would give individuals a variety of adaptive coping skills and may increase their ability to tolerate negative emotions and distressing events (Muhomba et al., 2017).

Gender Minority Stress, Resilience, and Substance Use. As previously mentioned, there was a statistically significant difference between the GMS sub-scales of gender-related rejection and negative expectations for the future between TESUP-1 and TESUP-2. This difference may be partially explained by the difficulties in accessing healthcare that many participants discussed; however, the TA and IPA results reveal that there were additional areas of difficulty that may have contributed to this difference in scores.

Participants frequently described an increase in their experiences of rejection at work, in public, and online. As a result of this, many individuals experienced an increase in anxiety, lower self-esteem, and feelings of alienation. They also cited that discriminatory media coverage debating transgender rights compounded feelings of isolation and mental health struggles that arose as a result of COVID-19. These qualitative findings are supported by recent studies into experiences of rejection and discrimination of LGBTQ+ people during the COVID-19 pandemic (Kneale & Becares, 2020), which demonstrated that transgender individuals experienced a higher increase in discriminatory experiences during the pandemic compared to cisgender LGB individuals, although reports of discrimination were raised for all LGBTQ groups. Previous research has demonstrated the harmful effects of rejection (R), and the expectation for rejection or other negative events (NE) on the mental health of transgender individuals (Rood, et al., 2016), as well as contributing to increased substance use (Connolly & Gilchrist, 2020). These studies indicate that the increase in GMS for participants in this study may be linked to the significant increase in alcohol and drug use from pre-pandemic to during the pandemic that was demonstrated, in addition to the significant increase in the overall measures of psychological distress: GSI and PST.

The role of resilience at the individual-level as conceptualised as Pride in one's identity (P) significantly increased from pre- to during pandemic levels. This was an unexpected finding, particularly when considering the increases in R and NE, in addition to GSI and PST – indicating increased psychological and gender-related distress. Singh, Hays, and Watson (2011) reported that a crucial aspect in building identity pride is to embrace one's own self-worth, which enables the individual to more strongly advocate for themselves in a cisnormative world. While the present study's significant increase in P may be at odds with some of the participant excerpts regarding the COVID-19 impact on gender minority stress, Breslow et al.'s (2015) research into the moderating role of resilience measures (Pride and Community Connections) between GMS and psychological distress indicates that the moderating role of resilience was not supported. This result suggests that although pride in one's identity is an important facet of one's personality in the face of distress and trauma, it is perhaps not as effective as external factors such as peer community connectedness (Bockting, et al., 2013), as is indicated by similar research into resilience to minority stress.

The results of this study do not support the hypothesised protective role of identity pride; this construct was moderately correlated with the use of alcohol and drugs. Although the correlational analysis did not reveal any significant association between non-disclosure and substance use, the interview participants who described themselves as 'stealth' reported having a complex relationship between non-disclosure, identity pride, and substance use. For example, Owen (H) valued having the choice whether to disclose his trans identity to others because it gave him "control over [his] own story". However, he also reported that there was an element of risk in not disclosing that he is trans to others, and his concern about his personal safety regarding being 'outed' as trans would sometimes result in an increase in alcohol use. Other participants voiced similar concerns in both TA and IPA results; hiding one's transgender status requires constant monitoring of one's appearance, speech, and behaviours (Beauregard et al., 2021) which leads to an increase in the allostatic load (physiological consequences of chronic stress) of the individual. This can then result in worse mental and physical health outcomes (Juster et al., 2019), and for the participants in the present study, sometimes led to an increase in substance use to manage their stress and anxiety.

Given the potentially harmful effect of lockdown measures on people's ability to socialise and connect with peer groups and communities, measures to promote community connections for transgender people would be a potential method to mitigate the increases to GMS and psychopathological symptoms which the results of this study indicate. However, it is important to note that for some participants, attempts to connect with the transgender community could have a negative impact as well. In TESUP-1, Darren (L) went into detail about his experiences of rejection from trans communities online, and how this had a negative impact on his substance use and resulted in poor self-esteem and internalised transphobia. In TESUP-2, Darren had "distanced [himself] from any kind of LGBT related discussion or groups", and this had led to a decrease in dysphoria and a better self-image. This indicates that there may be differences in the positive influence of building community connections online vs. offline, and future research should focus on comparisons between the wellbeing of transgender individuals who mainly socialise online or offline.

Finally, one element of GMS which was associated with tobacco, alcohol and drug use also significantly contributed to an increase in DUDIT score – this was experiences of gender-related victimisation. Although there was no significant increase in this domain between TESUP-1 and TESUP-2, across both phases there was a significant correlation between it and all types of substance use. The qualitative findings also revealed that an increase in gender-related victimisation, particularly receiving verbal harassment, was something that was noted by multiple participants. Participants such as Bailey (L) and Parker (H) shared that their experiences of receiving verbal harassment online led to feelings of emotional exhaustion and depression. Additionally, Cody (L) recalled how an incident of transphobic verbal abuse at work had increased their anxiety while in public and had caused them to become hypervigilant about their surroundings. Although Cody (L) and Parker (H) did not explicitly link these specific events to an increase in substance use, they both reported that "a stressful day at work" (Cody) or if "something big and bad happened" (Parker) they would feel the urge to use alcohol or drugs to cope. This finding supports the limited extant literature regarding the impact of gender minority stress. Dermondy et al. (2016) found that experiences of gender-related victimisation were related to an increase in binge-drinking, and Bouman et al. (2016) reported that verbal transphobic victimisation was correlated with an increase in four SCL subscales (somatisation, anxiety, phobic anxiety, and paranoid ideation). Bouman et al.'s results are particularly relevant to the present study, which found that scores in somatisation, anxiety, and phobic anxiety all significantly increased between TESUP-1 and TESUP-2. Overall, this suggests that for transmasculine individuals, experiences of genderrelated victimisation are damaging to their mental health and are a risk factor for drug use disorders. Therefore, interventions aimed at reducing substance use, particularly in the case of drug use disorders, should take a trauma-informed approach within a gender minority stress framework.

Gender Role Conflict and Masculinities in Substance Use. Although the longitudinal analyses revealed that no domain of the GRCS had significantly increased between TESUP-1 and TESUP-2, the correlational analysis found that in addition to RABBM being associated with alcohol and drug use as in TESUP-1, the domains of RE and SPC were also associated with both alcohol and drug use. Similar results emerged from the thematic analysis of questionnaire responses. Numerous participants found it increasingly difficult to recognise and express their emotions and reported that this was due to social isolation and lockdown measures over the past year. It is noteworthy that, in a similar manner to those who were interviewed in TESUP-1, some participants differentiated between how they embody masculinity among other transmasculine people and among cisgender men. Generally, participants felt more comfortable and less self-conscious around other transmasculine people, whereas among cis men, they sometimes felt pressure to act more masculine and were more aware of how they were being perceived by others. This finding is reflected in previous literature that has indicated that GRC is dependent on context (Vandello & Bosson, 2013). Multiple participants reported employing compensatory masculinity around cisgender men, and in TESUP-1, this was connected to an increase in substance use. This indicates that socialising with cis men is a salient situational cue that prompts increased GRC and substance use. Further investigation into the role of compensatory masculinity for transmasculine individuals is necessary in order to understand its possible contributions to increased GRC.

It is interesting to note that in the TESUP-1 interviews almost all of the participants mentioned the relationship between masculinity and substance use, with members of both the low- and high-risk groups voicing the opinion that if they did not consume substances in a masculine manner then they would not be perceived as male. However, in the TESUP-2 interviews, none of the participants discussed masculinity in this context. It is possible that this was due to lockdown measures limiting opportunities for socialising in the time between TESUP-1 and TESUP-2 interviews. This brings up an important question posed by one participant in the questionnaire: "If someone doesn't see performativity of gender, did it happen? Is gender in the expression or the perception?". That participants in TESUP-2 were

less concerned about appearing masculine through substance use, yet their use of alcohol and drugs had increased, implies that the function of masculinity in the context of substance use is often to communicate gender to others. This finding aligns with Butler's (1990) theory of gender performativity which states that one is discursively compelled to perform gender within the regulatory norms of their culture so as to assert their gender identity to others. Studies which have focused on the relationship between GRC and social support have reported that men who perceived themselves to have a strong social support network have lower GRC and less psychological distress. Given the findings from this study that suggested socialising with other men prompted engaging in compensatory masculinity, which included increased substance use, further research into the role of socialising and social support as it relates to GRC in transmasculine individuals is recommended.

## 6.6.3. Contexts of Substance Use and Meaning-Making

Navigating Health and Healthcare during COVID-19. Throughout the interviews with both low- and high-risk participants, the challenge of maintaining good physical and mental health during the COVID-19 pandemic and related lockdown measures was discussed frequently. The difficulties that people faced were varied; from unstable housing and unemployment to anxiety about themselves or their loved ones contracting COVID-19. What was consistent throughout the IPA interviews was that the majority of participants had experienced increased stress, anxiety, and isolation, which sometimes triggered the urge to drink alcohol to cope. For participants who did not report that the pandemic had had negative effects on their health and wellbeing, such as Parker (H) and Ezra (L), it was observed that they tended to have secure housing and finances, as well as stable romantic relationships. Recent literature regarding inequalities during the COVID-19 pandemic has revealed that the gap between high and low SES groups is widening (Binns & Low, 2021). Low SES groups have been disproportionately affected by COVID-19, in part due to crowded or unstable housing and having to work to maintain an income, making it difficult to effectively socially distance or self-isolate (Basu et al., 2021). Additionally, some studies have revealed that those people in a relationship during the pandemic reported less stress than did single people (Kowal et al., 2020), as well as lower levels of depression and anxiety (Odriozola-Gonzalez et al., 2020). However, neither of these studies assessed relationship quality or stability. Therefore, it is possible that it is not simply being in a romantic relationship that mitigated the effect of the pandemic, but that the support and security that the relationship offers is an important factor, as was observed in the interviews in this study.

An interesting finding is that despite almost all the interview participants suggesting that their quality of life had been negatively affected by COVID-19, the repeated measures ANOVA did not indicate any statistically significant change to any of the four QOL domains (physical health, psychological health, social relationships, and environmental health). This was an unexpected result, particularly because many participants in the questionnaire and interviews reported that their social life and interpersonal relationships had been negatively affected by the pandemic and by lockdown measures.

When comparing the QOL scores of participants in this study to the results of QOL in four different population samples (see Appendix G), it is demonstrated that the transmasculine sample in the present study had noticeably lower scores than the general population sample (Hawthorne et al., 2006), and are much more similar to the Spanish transgender sample (Gomez-Gil et al., 2013) and the UK psychiatric conditions sample (Skevington & McCrate, 2012). Like the psychiatric conditions and general population samples, the results of the present study indicate that psychological health was the lowestrated QOL domain; however, the scores from the present study were lower in each of the four domains of the QOL. This trend makes sense when considering the TA and IPA results. Struggling to maintain good mental health was the most frequently mentioned factor in the TA results which had a negative effect on participant's QOL. This was also mirrored in the IPA results; participants with pre-existing diagnoses reported a worsening of symptoms during the pandemic and almost all participants mentioned feelings of anxiety. These results indicate that, although poor QOL was not associated with an increase in substance use, struggles to attain a good QOL were common for participants in this study. This finding supports previous studies which indicate that transgender people have worse psychological health QOL than cisgender people (Nobili et al., 2018). It is important to note that the psychological QOL domain includes the question "are you able to accept your bodily appearance?". This question likely takes on a different meaning for transgender people compared to cisgender people. For example, in the present study only 23.5% (N = 19) of participants indicated that they were able to accept their bodily appearance, whereas the remaining 76.5% (N = 63) ranged from entirely unable to somewhat able to accept their bodily appearance. Previous studies have demonstrated that, when transgender people can access the transition-related medical interventions they require, their QOL improves (Newfield et al., 2006; Baker et al., 2021).

There was an unexpected finding that environmental QOL was positively correlated with tobacco use, indicating that feeling secure and safe in one's environment was associated with being a current smoker. Başar et al. (2016) found that high levels of perceived discrimination were associated with worse environmental QOL. In the present study tobacco use was associated with better environmental QOL and fewer experiences of gender-related discrimination, rejection, and victimisation, which lends support to Başar et al.'s findings. It also raises questions regarding the specific relationship between tobacco use, gender minority stress, and QOL. Were current smokers exposed to less gender minority stress because they had a safer environment? Or does tobacco use have a mediating role in the effect of GMS on QOL? Further research is necessary to explore tobacco use in transmasculine individuals, particularly because of its negative association with alcohol and drug use, distal gender minority stressors, in addition to psychopathological symptoms in this study.

Social Substance Use and Peer group Influences. One of the biggest differences between the interviews between TESUP-1 and TESUP-2 was the change in the nature of social and peer influences on substance use. In the TESUP-1 interviews, peer pressure and complying with social expectations were frequently mentioned as factors which increased substance use. However, in the TESUP-2 interviews, the participants had been in lockdown, taking social distancing measures, or self-isolating due to COVID-19, and therefore had not been socialising in the same way as they had in TESUP-1. Unlike the TESUP-1 interviews, where participants identified socialising as increasing their substance use, many individuals in TESUP-2 reported that feeling isolated and being unable to socialise as normal had resulted in an increase in substance use. It is notable that most of the participants who connected their increased substance use to isolation as a result of the pandemic were in the high-risk group. This provides support for recent research which has indicated that individuals who struggled with substance use prior to COVID-19 have been particularly affected by lockdown measures and increases to substance use (MacMillan et al., 2021).

One participant (Owen, H) described how he would drink alcohol or smoke cannabis alone in order to "emulate a social experience". For Owen, prior to the pandemic, substance use mainly occurred in social situations, therefore using substances alone was an attempt to recreate the positive emotions he would normally feel while socialising. However, it is also apparent from Owen's extracts that his substance use was not only to replicate socialising, but also to manage anxiety which stemmed from the pandemic. He described his alcohol and cannabis use as an "easy brute force way to turn [his] brain off" – from these extracts it is apparent that he was struggling to manage the effects of lockdown on his mental health. Similar findings have been reported in recent qualitative studies into the effect of UK lockdowns on health and wellbeing. Dedryver and Knai (2021) describe how people turned to online platforms to recreate a feeling of normalcy in their social lives, though it should be noted that the participants in their study reported that doing so was an unsatisfactory substitute for offline socialising. Additionally, participants in Smith et al.'s (2021) investigation into the effects of COVID-19 on individuals in recovery from SUD described how loneliness, boredom, and the loss of their usual social support prompted a relapse event. These studies demonstrate that Owen's experiences of increased substance use in response to isolation and worsening mental health have been noted in many young adults during the pandemic.

A desire to "feel like a normal person" (Darren, L), or efforts to establish a 'new normal', was frequently spoken of throughout the interviews. Difficulties with managing frustration, confusion, and loneliness were often met by an increase in substance use. Some participants (like Darren and Owen (H)) found comfort and escape from distress by simulating social situations through substance use. Other participants, such as Parker (H) and Cody (L), embraced the 'new normal' of increased alcohol use that they observed online at the beginning of the first lockdown. They both reported that they perceived drinking alone and drinking during the day to have been normalised through social media and cited that frequent exposure to alcohol consumption on social media platforms, such as Facebook, prompted their own increased alcohol use. Emerging research has indicated that exposure to alcohol references on social media is related to increased alcohol consumption (Geusens & Beullens, 2021). Interestingly, Geusens and Beullens reported that although the number of references on social media and traditional media (e.g., television and films) was comparable, traditional media was not associated with alcohol consumption. Furthermore, an analysis of

alcohol marketing campaigns during the first UK national lockdown found that 22% of online alcohol advertisements encouraged drinking at home by framing alcohol use as a method of alleviating boredom (Atkinson et al., 2020). As the COVID-19 pandemic is an on-going situation at the time of writing (late 2021), the long-term effects of increased social media use and increased substance use is not fully known. However, the participants in this study reported symptoms of alcohol dependence, including escalating alcohol use in response to tolerance, and subsequent withdrawal symptoms when unable to consume alcohol. This is an immediate concern; alcohol withdrawal syndrome is characterised by hallucinations, seizures, and in severe cases can result in coma (Mirijello et al., 2015). Therefore, as the UK attempts to recover normality or adjust to the 'new normal' it is of the utmost importance that healthcare services attend to the increase in SUD and the potential for harmful side-effects if individuals attempt to reduce substance use.

#### 6.6.4. Personal and Psychological Risk Factors in Substance Use

The Symbiotic Relationship of Mental Health and Substance Use. A common finding throughout the TA and IPA extracts was the participants in this study experienced worsening mental health which they perceived to be a result of COVID-19 and related lockdown measures. The quantitative analysis of SCL-90-R symptom dimensions and global index scores also reflects this finding; somatisation (SOM), anxiety (ANX), hostility (HOS), phobic anxiety (PHO), and psychosis (PSY) all significantly increased between TESUP-1 and TESUP-2. Additionally, global severity index (GSI) and positive symptom total (PST) both significantly increased.

It is striking that of the five symptom dimensions that had a statistically significant increase, four (SOM, ANX, HOS, PHO) are related to anxiety and thus may represent

variation in symptoms of anxiety or reactions to anxiety (Holi, 2003). The TA results found that increased anxiety was reported by many participants; worries about the future, rumination about illness and death, and feelings of agoraphobia characterised the struggles that participants were experiencing. Participants described the nature of their anxiety as "overpowering and constant", "persistent and difficult", and "severely increased". These extracts demonstrate the intense impact that increased anxiety was having on their daily lives. Furthermore, these findings were mirrored in the IPA results: Bailey (L) described feeling "paranoid" about catching COVID-19 which led to reluctance to leave the house and a consequential increase in alcohol use as a method of coping with stress. Literature regarding the effect of COVID-19 on the mental health of UK adults has indicated an 11.8% increase (from 25% to 36.8%) in the prevalence of poor mental health since the beginning of the pandemic, primarily affecting young adults (Smith et al., 2020). Furthermore, results from an investigation into COVID-19-related anxiety across the UK population has demonstrated that anxiety was significantly associated with an increase in somatic symptoms (Shevlin et al., 2020). Anxiety and somatic symptoms have been documented as commonly co-occurring alongside SUD though the specific nature of the relationship between them is not well understood (Hassan & Ali, 2011). These studies have particular relevance for the results of the present study and provides some insight into the significant increase of ANX, PHO, and SOM that was demonstrated by quantitative analyses. However, further investigation is required to clarify the association between anxiety, somatic symptoms, and substance use, and evaluate the potential impact of COVID-19 in these co-occurring disorders.

The negative effects of COVID-19 and lockdown measures on mental health were particularly pronounced in participants who had previously reported a psychiatric or neurodevelopmental diagnosis in TESUP-1. Darren (L) had disclosed his ASD diagnosis in TESUP-1, and in TESUP-2 it was apparent that the uncertainty and ambiguity of rule changes during COVID-19 lockdown measures had a negative impact on his mental health. The difficulties he was facing in coping with the lockdown were complicated by his ASD. Difficulties in coping with change and struggling to interact with others are core symptoms of ASD (Simonoff et al., 2020), and studies conducted prior to the current pandemic have demonstrated that failure to cope with change and poor interpersonal interaction have a negative effect on symptoms of depression and anxiety (Rodgers et al., 2019). One mixedmethods longitudinal analysis of the effects of COVID-19 on the mental health of autistic adults (Bundy et al., 2021) demonstrated that there was a substantial variation on how COVID-19 affected participants. Although Bundy et al.'s quantitative analysis showed that engaging in social activities and maintaining a routine contributed to lower levels of depression and anxiety, the themes derived from the qualitative analysis indicated that worries about the future and the ambiguity in government COVID-19 guidance left many participants feeling distressed. The findings of Bundy et al. are echoed in the experiences reported by Darren, in addition to numerous individuals in the TA results. Considering that research now indicates that ASD is up to 6.36 times more common among transgender individuals (Warrier et al., 2020), particularly among transgender men (Murphy et al., 2020), it is important that future research should focus on the long-term impact of COVID-19 on transgender individuals with ASD. People with a diagnosis of ASD (Calleja et al., 2019) and transgender people alike (Valentine & Shipherd, 2018) are minority groups who experience barriers to healthcare, stigma, and discrimination. Therefore, interventions aimed at mitigating the impact of COVID-19 in this population should pay particular attention to considerations of intersectionality and the unique stressors that autistic transgender people might face.

In order to investigate whether the variables which were most highly correlated with substance use could significantly predict current smoking, AUDIT, and DUDIT scores, regression analyses were conducted. The multiple regression analyses for alcohol use and for drug use both indicated that positive symptom total (PST) was a statistically significant predictor. For each point increase in PST, the AUDIT score increased by .212 and the DUDIT score increased by .272. This was a difference from the multiple regression analyses in TESUP-1 which indicated that global severity index (GSI), not PST, significantly contributed to the DUDIT score. It should be noted that GSI was excluded from the multiple regressions in TESUP-2 due to a high variance inflation factor (VIF) value which indicated multicollinearity between GSI and PST. GSI was excluded because it had the highest VIF value, indicating that its influence on the outcome variable was captured by other predictor variables (O'brien, 2007). This indicates that although both GSI and PST were highly correlated with alcohol and drug use, and both significantly increased between TESUP-1and TESUP-2, it was PST that better captured the influence on substance use. Some previous studies have indicated that PST is higher for SUD samples than the general population (Mercier et al., 1992; Miovsky et al., 2021); however, the relationship between PST and SUD in transgender populations is not known. Although previous studies have used the SCL-90 for transgender samples, none have reported PST scores in their analysis (Haraldson & Dahl, 2000; Auer et al., 2013; Castelo-Branco et al., 2021). This is possibly because GSI is typically thought to be the best global indicator of psychopathology (Derogatis, 1994); however, the results of the present study indicate that for transmasculine individuals PST may represent an important and overlooked factor which contributes to SUD. Future research should focus on exploring this relationship in more detail.

**Distress** (**In**)**tolerance and Emotional Regulation.** In a similar manner to the TESUP-1 interviews, participants who were re-interviewed in TESUP-2, described their struggles with tolerating distress and regulating their emotions, and how many would use alcohol or drugs to manage these difficulties. It was observed that in TESUP-2, participants

described how the same situations would result in an increase in substance use as in TESUP-1. For example, in both TESUP-1 and TESUP-2 interviews Rowan (H) reported struggling with symptoms of ADHD such as executive dysfunction and insomnia, which would lead him to combine sleeping pills and alcohol to go to sleep. Additionally, in both interviews Parker (H) described how it was typically interpersonal difficulties with their partner that would lead them to use alcohol or Valium, often to try to sleep. These results indicate that across both time points, participants were struggling with the same situations that would consistently trigger an increase in substance use. Furthermore, the similarities between Rowan (H) and Parker's (H) narratives are striking. Although their difficulties with emotional regulation and achieving restful sleep seemed to stem from different sources, it is clear from their interviews that a low mood during the day would have a negative impact on their ability to sleep later that night. Butler et al. (2020) demonstrated that transgender adults have short sleep duration and worse quality sleep compared to both cisgender and LGB groups, which they suggest is due to GMS and the increased prevalence of mood disorders in the transgender adult population. Furthermore, Auer et al. (2017) investigated sleeping problems in transgender individuals and reported that 81.2% of their trans male sample reported high sleep disturbances which had a significant negative impact on their OOL. Taking this research into account and considering studies which have demonstrated that insomnia causes alcohol dependence and cannabis initiation (Pasman et al., 2020), further research into this area is recommended. Research that focuses on the relationship between distress intolerance, sleep disorders, and related substance use would be particularly valuable, based on the findings from the longitudinal IPA results.

Some participants made a direct connection between their experiences as a transmasculine person and their substance use. The relationship between these two phenomena was exemplified in Owen's (H) narrative. He reiterated in TESUP-2, as in

TESUP-1, how his experiences of coming out to others as transgender typically happened under the influence of alcohol. He also went into detail about how smoking cannabis would "create an artificially strange experience in [his] body" which would then distract him from bodily dysphoria. For Owen, the use of cannabis was not only to mitigate feelings of loneliness and anxiety (as was discussed in section 6.6.4.2. Social Substance Use and Peer group Influence), but it was also a method of positively relating to his own body and reducing distress that arose from dysphoria. This is a particularly interesting finding which builds on the theory that individuals choose different substances to cope with different types of distress that was proposed in the TESUP-1 discussion (see section 5.4.4.2. Distress (In)Tolerance and *Emotional Regulation*). The increase in drug use that Owen experienced in response to both anxiety and dysphoria is clear in his DUDIT score which rose from 3 points in TESUP-1 (low-risk drug use) to 27 points in TESUP-2 (high-risk drug use). This further supports the theory that different types of distress might prompt seeking relief in different substances; specifically, coping with dysphoria using drugs that have a dissociative or psychedelic effect (as was the case for participants in the present study). Previous studies have explored the influence of personality traits on drug of choice with inconclusive results (Conway et al., 2002; Le Bon et al., 2004; Ruiz-Olivares et al., 2019), but recent research has revealed that emotional regulation is a mediator for SUDs (Okasha et al., 2021). However, to the author's knowledge no such investigations have been conducted that focus on transgender participants. It is important to note that IPA does not seek to make generalisations, therefore it cannot be said that this phenomenon will occur in different samples or populations. However, this was a consistent finding between TESUP-1 and TESUP-2 that was observed in multiple participants, therefore the potential importance of this result warrants further exploration in future research.

Personality Traits. A repeated measures ANOVA indicated that there was no significant change in any personality trait between TESUP-1 and TESUP-2. This finding aligns with extant literature which suggests that character traits are somewhat changeable, but this happens slowly over time, typically aligning with age (Clonginger, 2006). The temperament traits are thought to be relatively stable over time and have a neurobiological foundation (Cloninger, Pryzbeck & Svrakic, 1999). Quantitative analyses revealed that the personality profile of individuals engaged in high-risk substance use remained stable and was characterised by high levels of NS and ST, and low levels of HA and CO. A notable change was that HA contributed more to the outcome variables of alcohol or drug use in TESUP-2, whereas it was NS in TESUP-1 which had more of a contribution. In TESUP-2, NS was excluded from inclusion in the multiple regression analyses of alcohol and drug use, following the inclusion criteria of correlation coefficient of  $r \ge .35$  and p-value of  $p \le .01$ . In TESUP-2 NS was not correlated with alcohol use, and only had a weak correlation with drug use (r(82) = .240, p = .03). In order to investigate whether this may have occurred due to the attrition of 23 participants between phases and two a one-way between groups ANOVA was conducted. This showed that there was no statistically significant difference in tobacco, alcohol, or drug use between the 82 participants who completed both questionnaires and the 23 who only completed the first questionnaire (see Table 6.19).

between mose who completed both of just the first questionnane. Means and standard deviations.						
Measure	Group 1 (only completed 1 <sup>st</sup>		Group 2 (completed 1 <sup>st</sup>		<i>F</i> (1, 104)	Sig.
	questionnaire)		and 2 <sup>nd</sup> questionnaires)			
	M	SD	M	SD		
Tobacco	4.91	.996	4.48	1.434	1.88	.173
AUDIT	8.30	4.656	7.26	6.211	.565	.454
DUDIT	7.78	5.134	5.11	6.831	3.03	.085
<i>Note. M</i> and <i>SD</i> represent the mean and standard deviation, respectively.						

Table 6.19. Analysis of Variance (ANOVA) in Tobacco Use, AUDIT score, DUDIT score, between those who completed both or just the first questionnaire. Means and standard deviations
Following up with TESUP-1 results which indicated that perceived impulsivity was an indicator of high NS and also substance use, the TESUP-2 interviews focused on exploring the participants self-perception in the context of impulsivity and caution. Analysis of interview data revealed that all participants in the high-risk group conceptualised themselves as being impulsive, particularly in relation to substance use. Many of the participants who considered themselves impulsive described how they did not consider consequences in the moment, and some felt they were unable to "see negative consequences in the long term". Despite all high-risk participants describing negative consequences of substance use, they did not take this into consideration later when making the decision whether to engage in substance use or not. This finding is reflected in the literature regarding decision-making and substance use. Chen et al. (2020) conducted a meta-analysis and metaregression review into this topic and found that participants who used most substances (except cannabis) had significantly higher levels of risky-decision making compared to control groups. Furthermore, some studies have shown that impulsive decision-making is related to difficulties in achieving and maintaining abstinence and recovery from SUD (Stevens et al., 2014; Sliedrecht et al., 2021). Taken together, this evidence indicates that impaired decision-making is related to traits of impulsivity, and likely confers an increase in high-risk substance use and difficulty reducing substance use.

The finding that individuals categorised as low-risk did not perceive themselves to be impulsive, whereas the high-risk group did, is a potentially important result which could have implications for clinical assessment of risk. This finding is supported by further exploration of the participants' narratives of substance use and risk. For example, one participant whose AUDIT and DUDIT scores increased to such a degree that he was recategorized from lowrisk to high-risk, reported that he did consider himself to be impulsive. It would be interesting to explore whether self-perception of impulsivity, particularly in the context of substance use, is associated with the development or maintenance of a SUD. It may also be promising to investigate whether self-reported impulsivity may predict future risk of escalating substance use. Further research is necessary because if self-reported impulsivity is related to high-risk substance use, as it was in this study, clinicians could use this information to make faster assessments of risk which would be advantageous in primary care or emergency settings. Furthermore, it would enable healthcare professionals to implement tailored interventions to reduce substance use and related harm.

#### 6.6.5. Protective Factors in Substance Use

**Personal and Community-Level Resilience.** Although many of the participants in this study reported an increase in substance use, psychopathological symptoms, and overall distress in the year between TESUP-1 and TESUP-2, they also demonstrated substantial resilience in response to their struggles. The repeated measures ANOVA which explored differences in psychopathological symptoms revealed that although the mean PST (number of symptoms reported) had increased by 7.81 points between TESUP-1 (52.6) and TESUP-2 (60.41), there was no statistically significant change to PSDI (intensity of distress). This finding indicates that, despite experiencing more psychopathological symptoms, the participants did not report an increase in related distress. In the questionnaire and interviews, numerous participants described taking an active role in trying to seek help or improve their mental health. In the TESUP-2 questionnaire 28 participants chose to provide additional information about their mental health. Eight of these participants had started taking medication in the past year and all reported experiencing considerable improvements to their mental health as a result. Similarly, Jeremy (L) described how being diagnosed with

borderline personality disorder (BPD) had given him a better understanding of his mental health and what might trigger "a flare-up of symptoms". This enhanced understanding of himself and how he relates to others resulted in improvements to emotional regulation and a decrease in using substances to cope with distress.

The results of TESUP-1 indicated that the data did not support the hypothesised roles of identity pride (P) and community connectedness (CC) as protective factors. Quantitative analysis of TESUP-2 data confirms these findings: CC had no associations with the use of any substance and P was moderately correlated to high-risk alcohol and drug use. High-risk alcohol and drug use, in addition to P, all significantly increased between TESUP-1 and TESUP-2 and displayed moderate to high correlations with each other. To further investigate this unexpected finding, all interview participants were asked to go into detail about whether they felt pride in their transgender identity and what 'identity pride' meant to them.

Participants in both the low- and high-risk groups made a distinction between feeling pride in their personal transgender identity vs. feeling pride for the transgender community as a whole. Every participant who made this distinction reported feelings of pride toward the transgender community and the shared history, particularly regarding advocacy for trans rights. However, feeling pride in their own trans identity was not reported as frequently as pride for the community. Participants such as Cody (L) and Rowan (H) described how despite having admiration for the trans community, they did not personally feel proud to be trans because of the difficulties they had experienced. Similarly, Bailey (L) and Ezra (L) reported that although they did feel pride in their identity, it was difficult to express or appreciate this due to widespread transphobia in the general population.

Personal identity pride (a positive view of one's own transgender identity) and community pride (a positive view of other people who share that identity) are both components of the concept of identity pride within Testa et al.'s (2015) model of gender minority stress. Recent research which has utilised the gender minority stress framework to develop psychological interventions for transgender people has highlighted the positive influence of community-level resilience on individual resilience. The Transgender Resilience Intervention Model (TRIM) proposed by Matsuno and Israel (2018) identifies community, group, and individual interventions which can increase both individual-and community-level resilience for transgender people. Pilot studies which have used TRIM as a guide for providing interventions to increase positive appraisals of identity and wellbeing have demonstrated promising results. Clements et al., (2021) reported that participating in a strengths-based group intervention increased positive feelings about one's trans identity, in addition to increased levels of happiness and life satisfaction. Furthermore, Smith et al. (2021) propose a revised framework of adaptive coping methods that foster resilience in transgender people; these are psychological affirmations (such as self-belief and selfdefinition of identity) and social affirmations (such as advocacy and trans community connections). The authors propose that coping responses that enable gender affirmation provide resilience against gender minority stress.

Notably, the participants who did report feeling pride in their transgender identity also reported a feeling of belonging within the transgender community, had strong connections with other transgender people, and many of them engaged in advocacy and awareness work. This finding supports the frameworks proposed by Matsuno and Israel (2018) and Smith et al. (2021); not only did these participants report both individual and community pride, but they also provided positive and gender-affirmative reasoning for their pride. Although, similarly to TESUP-1, these same participants also reported that associating with other transgender people and participating in trans-focused events tended to increase their substance use. The difference here may be due to attitudes of peers toward substance use; as stated in the

TESUP-1 discussion, positive peer influences have been shown to significantly predict decreased substance use and improvements to mental health (Coyle et al., 2015). In addition to this, Matsuno and Israel (2018) suggest that group-level interventions that encourage finding, and becoming, a role model in the trans community can buffer negative effects of GMS. Therefore, future research should focus on developing interventions in order to build resilience and community connections within the transgender community, and which encourage positive peer influence and teach community leadership skills.

Harm Avoidance. Quantitative analysis into the influence of harm avoidance on substance use indicated that the personality trait of harm avoidance (HA) was moderately correlated with lower alcohol and drug use. Furthermore, multiple regression analyses revealed that AUDIT scores decreased by .25 and DUDIT scores decreased by .316 for each point increase in HA. These results suggest that high HA is a protective factor against developing an alcohol or drug use disorder. Previous research into the role of HA in SUD has had mixed results. Some studies indicate that SUD populations have higher scores in HA compared to the general population (Le Bon et al., 2004; Hashemi et al., 2019). However, Cloninger et al., (1993) suggest that low HA is associated with SUD. Kim et al. (2007) compared the personality traits of patients with SUD, depression, and schizophrenia, and found that SUD patients had the lowest scores in HA compared to the other two groups, which offers support for Cloninger et al.'s (1993) theory.

In a similar manner to TESUP-1, the interview participants went into detail about their experiences of HA in the context of personality traits, as well as the measures they take to reduce potential harm when using alcohol or drugs. As was noted in the previous section (6.6.5.3. Personality traits), participants were asked whether they considered themselves to be impulsive and there was a distinct split between the low- and high-risk groups. Most of the participants in the low-risk group characterised themselves as cautious people, particularly in

the context of substance use. Jeremy (L) and Ezra (L) were concerned about personal safety when using substances and neither of them engaged in drug use. Additionally, Darren (L) was very concerned about the possible negative effects that alcohol use could have on his mental health and was careful to "be mindful of how [he was] feeling that day" when deciding whether to drink. The personality trait of HA includes feeling of anticipatory worry and fear of uncertainty (Cloninger et al., 1993), both of which are demonstrated in the participant extracts about their caution regarding using substances. These results provide support for the quantitative finding that high HA is associated with lower substance use and offer an insight into attitudes towards substance use from the perspective of individuals at low risk of SUD.

A point to consider in relation to HA and substance use is the different types of perceived harm and whether that might influence engagement in substance use. For instance, Bailey (L) reported that they thought of themselves as a cautious person, and the potential for physical harm from alcohol use (in their case, migraines) would stop them from drinking. However, they also pointed out that if they felt pressure from family or friends to drink some alcohol they would "just go along with it" regardless of the effect on their health. This can be seen as avoidance of social harm, rather than the previously cited avoidance of physical or emotional harm that was reported by the rest of the low-risk group. It is recommended that future research should focus on how HA operates in different contexts, and the decisionmaking process of whether or not to engage in substance use in the context of perceived harm and HA.

#### 6.6.6. Conclusions

The results from TESUP-2 confirm the conclusions of TESUP-1 which indicated that high-risk substance use is a common issue for transmasculine individuals, and that COVID-19 had a substantial negative impact on substance use, mental health, and overall wellbeing. Rates of tobacco use stayed relatively stable between the two time points; however, the mean AUDIT and DUDIT scores significantly increased. This indicates that people were consuming more alcohol and drugs between TESUP-1 and TESUP-2. Additionally, many participants had changed the typical drugs they use; the use of stimulants such as MDMA had decreased, whereas the use of cannabis had increased.

A commonly reported concern for the participants was the closure of non-essential healthcare during COVID-19 which delayed access to transition-related healthcare. The increased waiting times, delayed or cancelled surgeries, and barriers to access on-going treatment, were all cited as events which would increase anxiety, distress, and substance use. Related to this, quantitative analyses revealed a significant increase in experiences of R and NE, and it was discovered that experiences of victimisation significantly contributed to an increase in drug use. These increases in GMS were felt to be due to higher levels of discriminatory media coverage of transgender people which had had a harmful effect when compounded by feelings of isolation and struggles with mental health that occurred as a result of COVID-19 lockdown measures. Therefore, as healthcare services are reintroduced, it is recommended that transition-related services and substance use recovery services should be prioritised and should take a trauma-informed approach within a gender minority stress framework.

It was found that although almost all participants reported that COVID-19 had a negative effect on their mental health or quality of life, it was participants who had secure

housing, finances, and a stable romantic relationship who were less affected. Although there was no statistically significant decrease to any QOL domain, many participants indicated that they believed their QOL had worsened as a result of COVID-19. Additionally, it was found that good environmental QOL was associated with current tobacco use. This was an unexpected finding that raises questions about the specific relationship between QOL, GMS, and tobacco use, and warrants further investigation.

The IPA results demonstrated that many participants were striving to restore a sense of normality in their lives during the pandemic, and for some, this culminated in an increase to alcohol and drug use. Participants in the high-risk group tended to increase substance use as a method of coping with isolation and loneliness. Some participants attempted to recreate social situations through the use of alcohol and cannabis. Whereas other participants tried to embrace online methods of social connection but reported that they were frequently exposed to a normalisation of frequent alcohol use that increased how often they drank and resulted in alcohol dependence and escalating use in response to increased alcohol tolerance.

Many participants reported worsening mental health during the pandemic, and this was confirmed by a significant increase in five symptom dimensions and two global distress dimensions. It is particularly notable that although PST significantly increased between the two phases, and it was a statistically significant predictor of increased alcohol and drug use, measures of overall distress did not increase as much. This indicates that although the participants experienced more symptoms their distress was not significantly increased. Because this was an unexpected finding and taking into consideration the lack of literature which explores PST in relation to substance use and transmasculine individuals, further research is recommended. Additionally, a particularly striking finding was the disproportionate impact that COVID-19 and related lockdown measures had on individuals with ASD. Due to this finding, and the barriers to healthcare that already exist for both

autistic and transgender individuals, interventions aimed at mitigating the negative effects of COVID-19 in this population should have a comprehensive understanding of intersectionality and incorporate this into assessment and treatment plans.

TESUP-2 narratives of distress intolerance were largely unchanged from TESUP-1, indicating that the same individuals were experiencing the same struggles, and continued to respond to these events in a dysfunctional manner, typically by using alcohol and drugs to cope. It was noted that low mood during the day had a particularly disruptive effect on sleep and symptoms of insomnia. Considering the high prevalence of sleep disorders among transmasculine individuals, further research into the relationship between distress intolerance and sleep disorders, and the potential impact on substance use, is recommended. Additionally, there was a novel finding of participants using dissociative and psychedelic drugs to distract themselves from gender dysphoria and create a more positive relationship with their own body. This indicates that specific types of distress may dictate which drug an individual may choose as a coping mechanism; to the author's knowledge there are no investigations related to this and the potential importance of this result warrants further research.

It was established that personality traits have a substantial influence on substance use – with some traits, such as high levels of NS being associated with increased substance use – and other traits, such as high levels of CO and HA, being associated with decreased substance use. All interview participants in TESUP-2 who categorised themselves as cautious were in the low-risk group and all of those who categorised themselves as impulsive were in the high-risk group. It was interesting that participants in the low-risk group in TESUP-1 whose substance use increased to such a level that they were recategorized as high-risk in TESUP-2 categorised themselves as impulsive, similarly to the other high-risk participants. These findings suggest that an individual's assessment of themselves as cautious or impulsive may

dictate their risk level as it relates to substance use and further experimental investigations are required with a larger sample size, as this finding was based only on the nine participants who were interviewed twice. Various protective factors were identified, most of these related to harm avoidance, which implies that personality traits are a major factor in relative risk of engaging in hazardous substance use or developing a SUD.

#### 7. Conclusions

#### 7.1. Summary of Findings

This thesis aimed to investigate factors associated with high-risk substance use in transmasculine individuals and to understand their lived experiences related to substance use. This was achieved by using a longitudinal mixed-methods design that enabled a wide-ranging exploration of the relationship between substance use and multiple contributing factors, in addition to an in-depth analysis of the lives of transmasculine individuals and what substance use meant to them.

This chapter will summarise the overall findings of this study and directly address the research questions that led to this investigation. The novel contributions that this thesis has made will be related to substance use disorders and trans studies. I will relate these key findings to theory and practice and will provide suggestions for future research. This chapter will conclude with a discussion of the limitations of this thesis.

The results of this thesis indicate that high-risk substance use is a prevalent issue affecting the transmasculine population; over half of the sample in this study engaged in high-risk substance use, and individuals who did not currently use substances had often experienced substance use-related issues in the past. Quantitative analysis revealed that gender minority stress, psychopathological symptoms, and personality traits were all major contributors to the development and maintenance of substance use for transmasculine individuals. Difficulties accessing transition-related healthcare, widespread transphobia in UK society, and navigating masculinities and social expectations were commonly reported experiences in this study which were associated with significant distress and often resulted in high-risk substance use. The impact of COVID-19 was substantial in this community. Alcohol and drug use significantly increased during this time, and the pandemic had a damaging effect on mental health and gender minority stress, which further contributed to high-risk substance use.

#### 7.2. Addressing the Research Questions

# 7.2.1. Is Stage of Transition Related to Substance Use?

In this study stage of transition was not related to substance use; however, specific difficulties throughout the process of transition were observed to trigger an increase in substance use. Typically, long waiting times and difficulties accessing transition-related healthcare, lack of communication with GICs, and experiences of discrimination or stigma in healthcare settings were events that were associated with an increase in substance use. The closure of non-essential healthcare during COVID-19 had a disproportionate effect on transgender individuals, and the resulting barriers to healthcare and delays in receiving treatment were all observed to increase distress, anxiety, and substance use.

This project has revealed that there are many possible avenues for mitigating the difficulties that this population experiences. Crucially, given the difficulties that the majority of participants in this study had experienced in accessing transition-related healthcare, and in light of worsening accessing during the COVID-19 pandemic, it is recommended that a shift be made to the Informed Consent Model of transgender healthcare delivery in the UK. Not only would this model empower transgender service-users to make informed choices in collaboration with healthcare professionals, but by de-pathologising trans identities (specifically, removing the requirement for a DSM-5 diagnosis of Gender Dysphoria), there is the possibility for decreasing waiting times for gender identity services, and decreasing the frequency of discrimination and transphobia in healthcare settings, which in the present study were related to increased substance use and poorer mental health.

It is important to note that the findings related to 'stage of transition' may be limited by the operationalisation of the concept. Stage of transition was measured through participant self-identification with a specific 'stage' with no beginning or end point suggested by the author, which meant the experiences of participants in each 'stage' varied considerably. Although the questions regarding 'stage of transition' were co-developed with transmasculine individuals in a focus group setting, Vincent (2018) notes that there is significant linguistic nuance within the trans community, and highlights that generational, classed, and socioeconomic differences influence the accessibility of language. Therefore, future efforts to adjust the operationalisation of 'stage of transition' should prioritise participatory research methods within the transgender community, in order to more accurately explore issues that may affect individuals at different 'stages' of transition.

# 7.2.2. Is Low QOL Associated with High-Risk Substance Use?

This thesis indicates that transmasculine individuals have worse QOL compared to previous studies on cisgender samples, particularly in the domain of psychological health. However, there was no association between psychological health QOL and substance use within the transgender group studied. In TESUP-1 drug use had a weak correlation between low physical health QOL and low environmental QOL, though this finding was not replicated in TESUP-2. Additionally, in TESUP-2 although there was no statistically significant change in any QOL domain, there was an unexpected relationship between being a smoker and better environmental QOL. Considering the similar relationship between being a smoker and having fewer experiences of gender-related discrimination, rejection, and victimisation, it is possible that current smokers were exposed to less gender minority stress because they had a safer environment. However, it is also possible that tobacco use mediates the relationship between

GMS and QOL, or alternatively it is possible that this unusual finding is a type 1 error, in which a statistically significant effect has been found that does not truly exist. Further research to clarify the specific role of tobacco use as it relates to QOL is necessary.

### 7.2.3. Does Increased GRC Correlate with Higher Substance Use?

High levels of GRC in the domain of RABBM were consistently associated with high-risk alcohol and drug use, and in TESUP-2, RE and SPC were also associated with highrisk alcohol and drug use, though to a lesser degree. This is an important finding as this is the first research to explore GRC in transmasculine individuals. However, the validity of the GRCS when used for transmasculine samples was called into question. Feedback from participants indicated that many of the questions were not applicable to non-binary individuals, and many participants indicated that their answers were dependent on social context, particularly regarding non-disclosure and perceived safety. Furthermore, in interviews, the concept of compensatory masculinity was often connected to both an increase in GRC and in substance use, which potentially indicates unique experiences of GRC that affect transmasculine individuals, but not cisgender individuals. I argue that transgender people have an inherently disparate experience of gender socialisation compared to cisgender people, which raises further questions regarding the application of the GRCS to transmasculine populations in its current form. Given these issues, I suggest that a separate version of the measure should be developed and validated for use with transmasculine individuals, with particular attention given to the inclusion of masculine non-binary individuals.

#### 7.2.4. Does Increased GMS Correlate with Higher Substance Use?

The results reveal that experiencing gender minority stress was common for the participants of this study, and the distal GMS factors in particular were associated with highrisk substance use. The relationship between GMS and drug use was more pronounced than with alcohol use, and in TESUP-2, gender-related victimisation was a significant predictor of high-risk drug use. Numerous participants felt as though negative interactions with others due to their gender identity had increased during the pandemic and described how discriminatory media coverage debating trans rights increased their feelings of anxiety and isolation during lockdown. Furthermore, difficulties accessing transition-related treatment increased during the pandemic, and likely contributed to the GMS domains of rejection and negative expectations for the future; both of which significantly increased between TESUP-1 and TESUP-2. These findings align with previous research in this area and suggest that interventions aimed at improving mental health or reducing substance use should take a trauma-informed approach within a gender minority stress framework. Utilising the GMS framework for clinical assessment and intervention is crucial, due to the historical pathologisation of trans experiences and reasonable responses to systematic oppression and gender-related victimisation. The results of this study suggest that this is particularly relevant for individuals seeking support to reduce drug use, therefore the importance of interventions which acknowledge the historical (and current) harm that the medico-legal system has inflicted on transgender people in the UK cannot be overstated.

### 7.2.5. Is Resilience a Protective Factor Against Issues with Substance Use?

Throughout both phases of this study, the hypothesised protective role of the resilience measures of identity pride and community connectedness were not supported.

Community connectedness was not associated with any type of substance use. Identity pride was associated with an *increase* in high-risk alcohol and drug use, which was the opposite of the hypothesised effect. Further investigation into identity pride in TESUP-2 revealed that participants distinguished between personal identity pride and community pride, typically only feeling pride for the latter. Some participants reported difficulties in expressing or feeling pride of their identity due to multiple negative experiences of GMS. Participants who did report feeling personal identity pride tended to also report a strong sense of belonging within the trans community, in addition to engaging in advocacy and awareness work. This provides support for the Transgender Resilience Intervention Model (TRIM; Matsuno & Israel, 2018), which suggests that group-based interventions that encourage positive appraisals of identity and finding or becoming a role model can buffer negative effects of GMS.

Taking into consideration the protective effect of positive peer influences that was discussed in TESUP-1, in addition to the promising initial findings from trials using TRIM, it is recommended that to build resilience and community connections within the trans community, researchers and clinicians should prioritise developing group-based interventions which encourage positive peer influence, positive self-appraisal, and community leadership skills. It is particularly important that such interventions should be trans-led and communitybased, given the frequency of discrimination and transphobia in healthcare settings.

Finally, the interviews with participants indicated that it tended to be participants in the high-risk substance use group who reported positive experiences of building connections in the trans community. Additionally, it was observed that multiple participants had negative experiences of online interactions with other transgender people, and a feeling of rejection from the community sometimes contributed to increased substance use. This is a particularly important finding given the current COVID-19 pandemic; many support or social groups moved online, and for individuals who struggle with online communication, this may have contributed to further isolation. Therefore, future research should focus on exploring the wellbeing of transgender individuals who socialise online versus offline and investigate how to enable building community connections in times of COVID-19.

#### 7.2.6. Are Psychopathological Symptoms Associated with Substance Use?

Psychopathological symptoms were consistently associated with high-risk substance use in both stages of TESUP. Many of these symptoms significantly increased during COVID-19, most notable, the mean number of symptoms (PST) endorsed increased by 7.4 points between TESUP-1 and TESUP-2. Interestingly, this increase was not related to an increase in distress. This may be because participants commonly reported coping with distress through substance use, which lowered their subjective rating of distress in the SCL-90-R. There is a lack of research which uses this measure of distress with transgender populations. Given the unexpected influence of PST on substance use, and its variation from previous literature which suggests that GSI is a better indicator of psychological distress, further investigations are necessary to clarify this relationship.

A striking result was the disproportionate affect that COVID-19 had on individuals with neurodevelopmental disorders, such as ASD and ADHD. These participants reported increased distress related to existing symptoms that were exacerbated by the ambiguity of COVID-19 restrictions and isolation during lockdown. Warrier et al. (2020) has indicated that ASD is more common in the transmasculine community. Therefore, it is crucial that any interventions that are aimed at mitigating the impact of COVID-19 in this population should be sensitive to issues around intersectionality and the unique stressors that transmasculine people with neurodevelopmental disorders might face.

# 7.2.7. Is There a Specific Personality Profile for Individuals who Engage in High-Risk Substance Use?

Throughout this project, personality traits were revealed to be a significant factor which contributed to high-risk substance use. To my knowledge this is the first study which explores the personality profiles of transmasculine individuals in the context of substance use. The results indicated that high-risk substance users had statistically significantly higher scores in NS and ST and lower HA and CO, compared to the low-risk group. In both TESUP-1 and TESUP-2, NS significantly predicted current smoking status; low NS was associated with current tobacco use. Based on interviews with participants, it was theorised that smoking behaviours were associated with a pursuit of masculinity or experiences of peer pressure. This finding contradicts previous research which indicates that high NS is associated with current tobacco use, however, all of these studies were based on cisgender samples. Further investigation is necessary; it would be particularly valuable to compare the personality traits and smoking behaviours of transgender individuals to that of a control group consisting of cisgender participants. It would also be interesting to explore the tobacco use of transfeminine individuals to transmasculine individuals, to further clarify correlates of smoking in the transgender population.

Given the significant influence of HA and CO on predicting low-risk substance use, it is worth exploring whether therapeutic interventions which focus on changing character traits, such as through mindfulness-based meditation, could be useful in reducing high-risk substance use. Additionally, the results suggest that the participant's own self-assessment of whether they are an impulsive or cautious person is related to whether they engage in highrisk substance use. This is a valuable finding, as it could provide clinicians with the ability to make faster preliminary assessment of risk, which would be useful in primary care and emergency settings. However, this finding was based only on the observations of interviews with nine participants at two time points, therefore further experimental investigations with a larger sample size are necessary to determine how and when to utilise this knowledge in clinical settings.

#### 7.2.8. What is the Impact of Transitioning and Masculinities on Substance Use?

Although no stage of transition was associated with lower levels of substance use, participants consistently reported that transitioning reduced dysphoria and increased feelings of confidence and happiness. Participants conceptualised moving along the transition path as changing their relationship with substance use from a maladaptive coping method to an enjoyable, social pastime. These results suggest that rather than a specific stage of transition being associated with changes to substance use, it is instead a sense of security and confidence in one's gender identity that may be related to a reduction in substance use and misuse.

Participants reported regularly enacting compensatory masculinity through increased substance use, so as to communicate their masculine identity and be gendered correctly by others. It is particularly important to transgender individuals to ensure that they are correctly gendered – not only for external validation of their gender identity – but also to ensure their own safety. Multiple participants described how non-disclosure of their trans identity was important to them, and how they would sometimes increase their substance use in response to peer pressure or social expectations, to preserve their 'stealth' status.

# 7.2.9. What Contexts and Situations are Associated with Substance Use and What Do These Mean to the Participants?

Typically, the participants in this study conceptualised substance use as either a coping mechanism or as a social activity. Participants described how in social settings they would use alcohol or drugs to increase confidence and reduce social anxiety. In this context, some participants would use substances to help ease coming out as trans to others or to have difficult conversations about dysphoria or mental health. It was also observed that gender dysphoria typically arose in the context of intimacy and sexual relationships, with some participants purposefully getting drunk to feel comfortable in these situations. A review of relevant literature reveals a critical gap in the research about the relationship between dysphoria, sex and intimacy, and substance use. Given the findings of this study and the lack of research on this topic, it is recommended that further research is conducted to tailor appropriate substance use reduction interventions to this community.

Due to psychopathological symptoms and experiences of GMS, participants frequently reported experiences of distress that they did not know how to tolerate. An inability to tolerate distress and to respond to distress with maladaptive coping mechanisms, such as substance use, was commonly reported. Furthermore, these narratives were largely unchanged between TESUP-1 and TESUP-2, indicating that participants were encountering the same struggles in their lives, and responding to them in a consistent manner. Low mood during the day had a particularly negative effect on ability to sleep and symptoms of insomnia; due to the high prevalence of sleep disorders in this population, further investigation into this is recommended.

Additionally, there was a novel finding that participants who found it difficult to tolerate ambiguity and frustration typically responded by an increase in alcohol use, whereas

participants who struggled to cope with mental health or dysphoria-related distress tended to cope by using drugs (namely cannabis and LSD). That individuals may respond to gender dysphoria using dissociative or psychedelic drugs, in order to distract themselves and build a more positive relationship with their body is an important finding. To my knowledge there have been no investigations into distress (in)tolerance and coping methods in the context of SUD in transgender individuals. It is recommended that future research should focus on this area to fully explore coping responses to different types of distress.

The high occurrence of distress intolerance in this study suggests that this may be a prevalent issue in the transgender community. Studies have demonstrated that DBT-ST is a cost-effective and efficacious method of helping people reduce distress and regulate their emotions. It should be explored whether this intervention may be useful for transgender individuals who are struggling to tolerate distress, particularly because the group-based format of DBT-ST would contribute to building community connections. To this end, it would be worth exploring whether DBT-ST delivered within a TRIM framework would be beneficial for transgender individuals, as a method of increasing resilience, providing peer support, and building healthy coping strategies.

#### 7.2.10. What Factors Contribute to a Cessation or Reduction of Substance Use?

Participants who felt supported by their partners, family, and friends were typically in the low-risk group and stated that because the people around them did not engage in substance use, neither did they. Some participants in the high-risk group described how building relationships with people who don't drink alcohol or use drugs was a positive influence on their own substance use. The effects of positive peer influence can also increase resilience, as discussed in TESUP-1. Therefore, community-based support and interventions, that include the opportunity to become a role model in the community, can contribute to a reduction or cessation of substance use.

Many of the protective factors that were identified related to harm avoidance. Concerns about the consequences of substance use on one's physical or mental health was an often-cited reason for not engaging in substance use, as well as worries about risks to personal safety while intoxicated.

Risk to personal safety while intoxicated was also a concern for some participants in the high-risk group and they described how staying alert to side-effects of mixing substances, and looking out for their friends, were their usual methods of risk management.

# 7.2.11. What Was the Impact of the COVID-19 Pandemic on Substance Use Within the Transmasculine Community?

The COVID-19 pandemic has had, and is continuing to have, a substantial impact on the transgender community. Both the frequency and quantity of alcohol and drug use increased significantly, in addition to the number of participants engaged in high-risk alcohol use, between TESUP-1 and TESUP-2. Some participants communicated their belief that alcohol use had been normalised and encouraged during the pandemic (particularly the first national lockdown), and that their alcohol use increased due to this. Other participants reported that their increase in substance use was an attempt to cope with anxiety and isolation that resulted from the pandemic and related lockdown measures. In addition to this, some participants described how they turned to alcohol and drugs to manage their stress when their usual coping methods (such as socialising and swimming) were unavailable to them due to the pandemic. Increased difficulties in access to transition-related healthcare were also associated with high-risk substance use. Participants reported concerns about longer waiting times and delays to surgical interventions which prompted an increase in substance use. Furthermore, participants also reported an increase in discrimination and victimisation during the pandemic, both of these factors were associated with high-risk substance use. It is recommended that improving access to transition-related healthcare and promoting communication between healthcare providers and service-users should be prioritised when seeking to restore healthcare services post-COVID.

# 7.3. Limitations

Whilst this research makes some novel and valuable contributions to our understanding of substance use and SUD in the transmasculine community, there are some limitations to this work.

Firstly, the participants in this study were overwhelmingly white British, and skewed to a younger age, therefore the findings of this study may only be generalisable to a rather homogenous group within the transmasculine community. On reflection, more steps to engage the BIPOC transmasculine community in this research and promote recruitment should have been taken; a more diverse insight into experiences of substance use would enhance our understanding of this issue. It is possible that the language used during recruitment limited engagement from BIPOC groups; research has shown that language which is associated with stigma in certain communities is a limiting factor to research recruitment (Ellard-Gray et al., 2015). Additionally, because I am white, potential BIPOC participants may not have trusted that I would be accurate in the interpretation of their experiences or had mistrust about the research process. In order to address these issues in

future, a potential strategy would be to form a research participant advisory committee, who would be able to provide guidance on the issues of cultural and linguistic differences. Such a committee would be representative of the demographics of the trans community and would strive to represent marginalised voices, this committee would also have the opportunity to review the interpretative and analytical process in order to ensure transparency and validity of the findings for all participant demographic groups.

Secondly, while an in-depth exploration of substance use in transmasculine individuals contributes important knowledge to the field of psychology and trans studies, it is difficult to understand these results in a wider context, due to the focus on transmasculine identities. Limiting the scope of this research to only transmasculine participants was necessary, but it makes comparisons to other transgender identities and to cisgender populations difficult. Research that includes people of all gender identities could offer further insights into the convergence and variation of experiences between different trans identities and cisgender identities.

Finally, the operationalisation and the appropriateness of some measures used in this study were questioned during analysis of the results. As has previously been pointed out, the operationalisation of 'stage of transition' and the GRCS both received critical feedback from participants as being too limiting and not reflective of their lived experiences. Furthermore, in the case of the GRCS, participants perceived some questions as hetero/cisnormative and were uncomfortable with the portrayal of gender roles in the questionnaire. Therefore, it is acknowledged that there are some limitations to using these measures in the study which may have impacted the results of the research. Future research using similar measures should seek to improve how the concepts of 'stage of transition' and 'gender role conflict' are operationalised for use with transmasculine participant samples.

### 7.4. Concluding Thoughts

This study has demonstrated how transmasculine individuals are disproportionately affected by SUD, and how the effects of COVID-19 and lockdown measures have negatively impacted the mental health of this community. Being transgender in the UK can feel, for some, like they have to defend their existence daily, and the knock-on effect of this and other experiences of gender minority stress results in worse health outcomes for trans individuals. Many participants had experienced barriers to accessing healthcare, particularly transitionrelated healthcare, and due to the COVID-19 pandemic these barriers have increased.

Although the narrative of this thesis has been one of distress, negative outcomes, and barriers, it is crucial to point out that this is largely due to the treatment that transgender individuals currently receive in UK society. Furthermore, it is hoped that by focusing on the ways that transmasculine people currently struggle, the situation can be improved for future generations. Many participants in this study expressed joy, gender euphoria, and pride in their community, they communicated that there is much more to being transgender than dysphoria, and that there is beauty in living an authentic life. Therefore, this thesis will close with the words of Zack, who had this to say about being transgender:

"I am very, very, proud of being trans, and I am very vocal about it. I just want to say, despite how bad things have been this year, transitioning has impacted my life beyond positively ... I've felt so much euphoria recently from reaching my transition milestones. It's not all doom and gloom, I love this community and it's so inspiring to see people fighting for trans rights. I'm happy being trans ... I wouldn't change it"

# References

Allport, G.W. (1937). Personality: A psychological interpretation. Holt, Rinehart, & Winston.

- Alsuhaibani, R., Smith, D. C., Lowrie, R., Aljhani, S., & Paudyal, V. (2021). Scope, quality, and inclusivity of international clinical guidelines on mental health and substance abuse in relation to dual diagnosis, social and community outcomes: a systematic review. *BMC Psychiatry*, 21(1), 1-23. <u>https://doi.org/10.1186/s12888-021-03188-0</u>
- Aluja, A., & Blanch, A. (2011). The five and seven factors personality models: differences and similitude between the TCI-R, NEO-FFI-R and ZKPQ-50-CC. *The Spanish Journal of Psychology*, 14(2), 659-666. <u>https://doi.org/10.5209/rev\_sjop.2011.v14.n2.14</u>
- American Psychiatric Association. (2013). *Diagnostic and statistical manual of mental disorders (DSM-5)*. American Psychiatric Publications.
- Anzani, A., De Panfilis, C., Scandurra, C., & Prunas, A. (2020). Personality disorders and personality profiles in a sample of transgender individuals requesting gender-affirming treatments. *International Journal of Environmental Research and Public Health*, 17(5), 1521. <u>https://doi.org/10.3390/ijerph17051521</u>
- Arcelus, J., Bouman, W. P., Van Den Noortgate, W., Claes, L., Witcomb, G., & Fernandez-Aranda, F. (2015). Systematic review and meta-analysis of prevalence studies in transsexualism. *European Psychiatry*, 30(6), 807-815, <u>https://doi.org/10.1016/j.eurpsy.2015.04.005</u>
- Arnau, M. M., Mondon, S., & Santacreu, J. J. (2008). Using the temperament and character inventory (TCI) to predict outcome after inpatient detoxification during 100 days of outpatient treatment. *Alcohol & Alcoholism*, 43(5), 583-588. <u>https://doi.org/10.1093/alcalc/agn047</u>
- Ashley, F. (2019). Gatekeeping hormone replacement therapy for transgender patients is dehumanising. *Journal* of Medical Ethics, 45(7), 480-482. <u>http://dx.doi.org/10.1136/medethics-2018-105293</u>
- Atkinson, A. M., Sumnall, H., & Meadows, B. (2021). 'We're in this together': A content analysis of marketing by alcohol brands on Facebook and Instagram during the first UK Lockdown, 2020. *International Journal of Drug Policy*, 98, 103376. <u>https://doi.org/10.1016/j.drugpo.2021.103376</u>
- Auer, M. K., Höhne, N., Bazarra-Castro, M. Á., Pfister, H., Fuss, J., Stalla, G. K., ... & Ising, M. (2013). Psychopathological profiles in transsexuals and the challenge of their special status among the sexes. *PLoS One*, 8(10), e78469. https://doi.org/10.1371/journal.pone.0078469
- Auer, M.K., Fuss, J., Höhne, N., Stalla, G. K., Sievers, C. (2014). Transgender transitioning and change of selfreported sexual orientation. *PLoS ONE* 9(10), e110016. <u>https://doi.org/10.1371/journal.pone.0110016</u>
- Auer, M. K., Liedl, A., Fuss, J., Nieder, T., Briken, P., Stalla, G. K., ... & Sievers, C. (2017). High impact of sleeping problems on quality of life in transgender individuals: a cross-sectional multicenter study. *PLoS One*, 12(2), e0171640. <u>https://doi.org/10.1371/journal.pone.0171640</u>
- Austin, A., & Craig, S. L. (2015). Transgender affirmative cognitive behavioral therapy: Clinical considerations and applications. *Professional Psychology: Research and Practice*, 46(1), 21. <u>https://psycnet.apa.org/doi/10.1037/a0038642</u>
- Austin, A., & Goodman, R. (2017). The impact of social connectedness and internalized transphobic stigma on self-esteem among transgender and gender non-conforming adults. *Journal of Homosexuality*, 64(6), 825– 841. <u>https://doi.org/10.1080/00918369.2016.1236587</u>
- Austin, A., & Goodman, R. (2018). Perceptions of transition-related health and mental health services among transgender adults. *Journal of Gay & Lesbian Social Services*, 30(1), 17-32, <u>https://doi.org/10.1080/10538720.2017.1408515</u>

- Babor, T.F., de la Fuente, J.R., Saunders, J., & Grant, M. (1992). *The Alcohol Use Disorders Identification Test. Guidelines for use in primary health care*. World Health Organization.
- Baker, K. E., Wilson, L. M., Sharma, R., Dukhanin, V., McArthur, K., & Robinson, K. A. (2021). Hormone therapy, mental health, and quality of life among transgender people: a systematic review. *Journal of the Endocrine Society*, 5(4). <u>https://doi.org/10.1210/jendso/bvab011</u>
- Bandura, A., & Walters, R. H. (1977). Social learning theory (Vol. 1). Prentice Hall.
- Barber, M. (2020). Alfred Schutz. The Stanford Encyclopedia of Philosophy Edward N. Zalta (ed.). Retrieved on 28/10/21 from <a href="https://plato.stanford.edu/archives/sum2020/entries/schutz/">https://plato.stanford.edu/archives/sum2020/entries/schutz/</a>
- Barger, B. T., Obedin-Maliver, J., Capriotti, M. R., Lunn, M. R., & Flentje, A. (2021). Characterization of substance use among underrepresented sexual and gender minority participants in The Population Research in Identity and Disparities for Equality (PRIDE) Study. *Substance Abuse*, 42(1), 104-115. <u>https://doi.org/10.1080/08897077.2019.1702610</u>
- Baron-Cohen, S., & Hammer, J. (1997). Is autism an extreme form of the "male brain"? *Advances in Infancy research*, *11*, 193-218.
- Barr, S. M., Budge, S. L., & Adelson, J. L. (2016). Transgender community belongingness as a mediator between strength of transgender identity and well-being. *Journal of Counseling Psychology*, 63(1), 87. <u>https://doi.org/10.1037/cou0000127</u>
- Bartel, S. J., Sherry, S. B., & Stewart, S. H. (2020). Self-isolation: A significant contributor to cannabis use during the COVID-19 pandemic. *Substance Abuse*, 41(4), 409-412. <u>https://doi.org/10.1080/08897077.2020.1823550</u>
- Başar, K., Öz, G., & Karakaya, J. (2016). Perceived discrimination, social support, and quality of life in gender dysphoria. *The Journal of Sexual Medicine*, 13(7), 1133-1141. <u>https://doi.org/10.1016/j.jsxm.2016.04.071</u>
- Basu, P., Bell, C., & Edwards, T. H. (2021). Covid social distancing and the poor: An analysis of the evidence for England. *The BE Journal of Macroeconomics*. <u>https://doi.org/10.1515/bejm-2020-0250</u>
- BBC News (January 2020). 'Transgender people face NHS waiting list hell'. Retrieved 08/04/21 from https://www.bbc.co.uk/news/uk-england-51006264
- Beard, S. J., & Wolff, J. M. (2020). The moderating role of positive peers in reducing substance use in college students. *Journal of American College Health*, 1–12. https://doi.org/10.1080/07448481.2020.1784907
- Berman A, Bergman H, Palmstierna T, Schlyter F. (2005). Evaluation of the Drug Use Disorders Identification Test (DUDIT) in criminal justice and detoxification settings and in a Swedish population sample. *Eur* Addict Res, 11(1), 22-31. <u>https://doi.org/10.1159/000081413</u>
- Bertrand, M. (2019). The Gender Socialization of Children Growing Up in Non-traditional Families. *AEA Papers and Proceedings*, 109, 115-21. <u>https://www.jstor.org/stable/26723925</u>
- Bhalla, I. P., & Rosenheck, R. A. (2018). A change in perspective: from dual diagnosis to multimorbidity. *Psychiatric Services*, 69(1), 112-116. <u>https://doi.org/10.1176/appi.ps.201700194</u>
- Biederman, J., Faraone, S. V., Monuteaux, M. C., & Feighner, J. A. (2000). Patterns of alcohol and drug use in adolescents can be predicted by parental substance use disorders. *Pediatrics*, 106(4), 792-797. <u>https://doi.org/10.1542/peds.106.4.792</u>
- Bilodeau, B. L., & Renn, K. A. (2005). Analysis of LGBT identity development models and implications for practice. New Directions for Student Services, 2005(111), 25-39, <u>https://doi.org/10.1002/ss.171</u>
- Binns, C., & Low, W. Y. (2021). The rich get richer and the poor get poorer: The inequality of COVID-19. Asia Pacific Journal of Public Health, 33(2-3), 185-187. <u>https://doi.org/10.1177%2F10105395211001662</u>

- Blazina, C., & Watkins, C. E., Jr. (1996). Masculine gender role conflict: Effects on college men's psychological well-being, chemical substance usage, and attitudes towards help-seeking. *Journal of Counseling Psychology*, 43(4), 461–465. https://doi.org/10.1037/0022-0167.43.4.461
- Block, J. (2010). The five-factor framing of personality and beyond: Some ruminations. *Psychological Inquiry*, 21(1), 2-25. <u>https://doi.org/10.1080/10478401003596626</u>
- Bockting, W., Coleman, E., Deutsch, M. B., Guillamon, A., Meyer, I., Meyer III, W., ... & Ettner, R. (2016). Adult development and quality of life of transgender and gender nonconforming people. *Current Opinion in Endocrinology, Diabetes, and Obesity*, 23(2), 188. <u>https://doi.org/10.1097/med.0000000000232</u>
- Boroughs, M. S., Bedoya, C. A., O'Cleirigh, C., & Safren, S. A. (2015). Toward defining, measuring, and evaluating LGBT cultural competence for psychologists. *Clinical Psychology: Science and Practice*, 22(2), 151-171. <u>https://doi.org/10.1111/cpsp.12098</u>
- Brändström, S., Richter, J., & Przybeck, T. (2001). Distributions by age and sex of the dimensions of temperament and character inventory in a cross-cultural perspective among Sweden, Germany, and the USA. *Psychological Reports*, 89(3), 747-758. <u>https://doi.org/10.2466%2Fpr0.2001.89.3.747</u>
- Breslow, A. S., Brewster, M. E., Velez, B. L., Wong, S., Geiger, E., & Soderstrom, B. (2015). Resilience and collective action: Exploring buffers against minority stress for transgender individuals. *Psychology of Sexual Orientation and Gender Diversity*, 2(3), 253. <u>http://dx.doi.org/10.1037/sgd0000117</u>
- Broman, C. L., Wright, M. K., Broman, M. J., & Bista, S. (2019). Self-medication-and substance use: A test of the hypothesis. *Journal of Child & Adolescent Substance Abuse*, 28(6), 494-504. <u>https://doi.org/10.1080/1067828X.2020.1789526</u>
- Budge, S. L., Adelson, J. L., & Howard, K. A. (2013a). Anxiety and depression in transgender individuals: the roles of transition status, loss, social support, and coping. *Journal of Consulting and Clinical Psychology*, 81(3), 545. <u>https://doi.org/10.1037/a0031774</u>
- Budge, S. L., Katz-Wise, S. L., Tebbe, E. N., Howard, K. A., Schneider, C. L., & Rodriguez, A. (2013b). Transgender emotional and coping processes: Facilitative and avoidant coping throughout gender transitioning. *The Counseling Psychologist*, 41(4), 601-647. https://doi.org/10.1177%2F0011000011432753
- Bundy, R., Mandy, W., Crane, L., Belcher, H., Bourne, L., Brede, J., ... & Cook, J. (2021). The impact of early stages of COVID-19 on the mental health of autistic adults in the United Kingdom: A longitudinal mixed-methods Study. *Autism*.(Online ahead of print). <u>https://doi.org/10.1177/13623613211065543</u>
- Busse, H., Buck, C., Stock, C., Zeeb, H., Pischke, C. R., Fialho, P. M. M., ... & Helmer, S. M. (2021). Engagement in health risk behaviours before and during the COVID-19 pandemic in German university students: Results of a cross-sectional study. *International Journal of Environmental Research and Public Health*, 18(4), 1410. <u>https://doi.org/10.3390/ijerph18041410</u>
- Butler, E. S., McGlinchey, E., & Juster, R. P. (2020). Sexual and gender minority sleep: A narrative review and suggestions for future research. *Journal of Sleep Research*, 29(1), e12928. <u>https://doi.org/10.1111/jsr.12928</u>
- Calleja, S., Kingsley, J., Amirul Islam, F. M., & McDonald, R. (2021). Barriers to accessing healthcare: Perspectives from autistic adults and carers. *Qualitative Health Research*. 32(2), 267-278. <u>https://doi.org/10.1177%2F10497323211050362</u>
- Campbell, J. L., & Snow, B. M. (1992). Gender role conflict and family environment as predictors of men's marital satisfaction. *Journal of Family Psychology*, 6(1), 84.
- Carr, L. T. (1994). The strengths and weaknesses of quantitative and qualitative research: what method for nursing? *Journal of Advanced Nursing*, 20(4), 716-721. <u>https://doi.org/10.1046/j.1365-2648.1994.20040716.x</u>

- Castelo-Branco, C., Ribera-Torres, L., Gómez-Gil, E., Uribe, C., & Cañizares, S. (2021). Psychopathological symptoms in Spanish subjects with gender dysphoria. A cross-sectional study. *Gynaecological Endocrinology*, 37(6), 534-540. <u>https://doi.org/10.1080/09513590.2021.1913113</u>
- Cavanaugh, T., Hopwood, R., & Lambert, C. (2016). Informed consent in the medical care of transgender and gender-nonconforming patients. *AMA Journal of Ethics*, *18*(11), 1147-1155, https://doi.org/10.1001/journalofethics.2016.18.11.sect1-1611
- Chapman, R. (2016). A case study of gendered play in preschools: how early childhood educators' perceptions of gender influence children's play. *Early Child Development and Care*, *186*(8), 1271-1284. <u>https://doi.org/10.1080/03004430.2015.1089435</u>
- Chamberlin, W. (1993). Gender role conflict as a predictor of problem solving, leadership style, authoritarian attributes, and conflict management attitudes (Doctoral dissertation, Columbia University). *Dissertation Abstracts International*, 52, 844.
- Chen, S., Yang, P., Chen, T., Su, H., Jiang, H., & Zhao, M. (2020). Risky decision-making in individuals with substance use disorder: A meta-analysis and meta-regression review. *Psychopharmacology*, 237(7), 1893-1908. <u>https://doi.org/10.1007/s00213-020-05506-y</u>
- Childress, J. (2020). Paternalism in Healthcare and Health Policy. In *Public Bioethics: Principles and Problems*. Oxford University Press.
- Choi, N., Herdman, K., Fuqua, D. R., & Newman, J. L. (2011). Gender-role conflict and gender-role orientation in a sample of gay men. *The Journal of Psychology*, 145(5), 507-519. <u>https://psycnet.apa.org/doi/10.1080/00223980.2011.590163</u>
- Choi, N., DiNitto, D. M., Marti, C. N., & Choi, B. Y. (2017). Association of adverse childhood experiences with lifetime mental and substance use disorders among men and women aged 50+ years. *International Psychogeriatrics*, 29(3), 359-372. <u>https://doi.org/10.1017/S1041610216001800</u>
- Choy, L. T. (2014). The strengths and weaknesses of research methodology: Comparison and complimentary between qualitative and quantitative approaches. *IOSR Journal of Humanities and Social Science*, *19*(4), 99-104.
- Cialdini, R.B., Trost, M. R. (1998). Social influence: Social norms, conformity, and compliance. In *The handbook of social psychology*. (4<sup>th</sup> Ed. Vol. 2.). 151–192. McGraw-Hill.
- Cipriano, A., Cella, S., & Cotrufo, P. (2017). Nonsuicidal self-injury: A systematic review. *Frontiers in Psychology*, 8, 1946. <u>https://doi.org/10.3389/fpsyg.2017.01946</u>
- Clancy, M. (2013). Is reflexivity the key to minimising problems of interpretation in phenomenological research? *Nurse Researcher*, 20(6), 12-16. <u>https://doi.org/10.7748/nr2013.07.20.6.12.e1209</u>
- Cloninger, C. R. (1986). A unified biosocial theory of personality and its role in the development of anxiety states. *Psychiatric Developments*, *3*(2), 167-226.
- Cloninger, C. R. (2004). Feeling good: the science of well-being. Oxford University Press.
- Cloninger, C. R. (2006). The science of well-being: an integrated approach to mental health and its disorders. World Psychiatry, 5(2), 71. <u>http://www.ncbi.nlm.nih.gov/pmc/articles/pmc1525119/</u>
- Cloninger, C.R. (2015). Validation of psychiatric classifications: the psychobiological model of personality as an exemplar. In *Alternative perspectives on psychiatric validation*. (pp. 201–223). Oxford University Press.
- Cloninger, C. R., & Cloninger, K. M. (2016). Person-centered psychotherapy. In Person centered psychiatry (pp. 247-261). Springer.

- Cloninger, C. R., Przybeck, T. R., Svrakic, D. M. (1999). *The temperament and character inventory-revised*. Washington University.
- Cloninger, C. R., Svrakic, D. M., & Przybeck, T. R. (1993). A psychobiological model of temperament and character. *Archives of General Psychiatry*, *50*(12), 975-990. <u>https://doi.org/10.1001%2Farchpsyc.1993.01820240059008</u>
- Cohn, A., & Zeichner, A. (2006). Effects of masculine identity and gender role stress on aggression in men. *Psychology of Men & Masculinity*, 7(4), 179. <u>http://dx.doi.org/10.1037/1524-9220.7.4.179</u>
- Collazo, A., Austin, A., & Craig, S. L. (2013). Facilitating transition among transgender clients: Components of effective clinical practice. *Clinical Social Work Journal*, 41(3), 228-237. <u>https://doi.org/10.1007/s10615-013-0436-3</u>
- Connolly, D., & Gilchrist, G. (2020). Prevalence and correlates of substance use among transgender adults: A systematic review. *Addictive Behaviors*, 106544. <u>https://doi.org/10.1016/j.addbeh.2020.106544</u>
- Conway, K. P., Swendsen, J. D., Rounsaville, B. J., & Merikangas, K. R. (2002). Personality, drug of choice, and comorbid psychopathology among substance abusers. *Drug and Alcohol Dependence*, 65(3), 225-234. <u>https://doi.org/10.1016/S0376-8716(01)00168-5</u>
- Corr, P. J. & Matthews, G. (2009). *The Cambridge handbook of personality psychology*. Cambridge University Press
- Cortese, J. (2003). Gender role conflict, personality, and help seeking in adult men (Doctoral dissertation, University of Southern California). *Dissertation Abstracts International*, 64, 4609.
- Courtenay, W. (2009). Theorising masculinity and men's health. in Broom, A., & Tovey, P. (eds) *Men's health: Body, identity, and social context*, (pp. 9-32).Wiley-Blackwell.
- Coyle, C., Bramham, J., Dundon, N., Moynihan, M., & Carr, A. (2015). Exploring the positive impact of peers on adolescent substance misuse. *Journal of Child & Adolescent Substance Abuse*, 25(2), 134-143. <u>https://doi.org/10.1080/1067828X.2014.896761</u>
- Creswell, J. W., Klassen, A. C., Plano Clark, V. L., & Smith, K. C. (2011). Best practices for mixed methods research in the health sciences. (pp.541-545). National Institute of Health.
- Darcy, C. (2018). Making the invisible visible: Masculinities and men's illicit recreational drug use. Irish Journal of Sociology, 26(1), 5-24. <u>https://doi.org/10.1177%2F0791603517735772</u>
- Dedryver, C. C., & Knai, C. (2021). 'It's easily the lowest I've ever, ever got to': A qualitative study of young adults' social isolation during the COVID-19 lockdowns in the UK. *International Journal of Environmental Research and Public Health*, 18(22). <u>https://doi.org/10.3390/ijerph182211777</u>
- De Fruyt, F., Van de Wiele, L., & Van Heeringen, C. (2000). Cloninger's psychobiological model of temperament and character and the five-factor model of personality. *Personality and Individual Differences*, 29(3), 441-452. <u>https://doi.org/10.1016/S0191-8869(99)00204-4</u>
- del Pino-Gutiérrez, A., Jiménez-Murcia, S., Fernández-Aranda, F., Agüera, Z., Granero, R., Hakansson, A., ... & Menchón, J. M. (2017). The relevance of personality traits in impulsivity-related disorders: From substance use disorders and gambling disorder to bulimia nervosa. *Journal of Behavioral Addictions*, 6(3), 396-405. <u>https://doi.org/10.1556/2006.6.2017.051</u>
- Dell'Orco, S., Sperandeo, R., Moretto, E., & Maldonato, N. M. (2018). Revision on psychometric properties of the temperament and character inventory in a clinical sample. *Frontiers in Psychology*, 9, 1951. <u>https://doi.org/10.3389/fpsyg.2018.01951</u>

- Demant, D., Hides, L., White, K. M., & Kavanagh, D. J. (2018). LGBT communities and substance use in Queensland, Australia: Perceptions of young people and community stakeholders. *PloS one*, 13(9), e0204730. <u>https://doi.org/10.1371/journal.pone.0204730</u>
- Dermody, S. S., Marshal, M. P., Burton, C. M., & Chisolm, D. J. (2016). Risk of heavy drinking among sexual minority adolescents: Indirect pathways through sexual orientation-related victimization and affiliation with substance-using peers. *Addiction*, 111(9), 1599-1606. <u>https://doi.org/10.1111/add.13409</u>
- Derogatis, L. R. (1983). SCL-90-R: Administration, Scoring and Procedures: Manual II. Clinical Psychometric Research.
- Derogatis, L. R. (1994). SCL-90-R: Administration, scoring and procedures manual (3rd ed.). NCS Pearson.
- Deutsch, M. B. (2012). Use of the informed consent model in the provision of cross-sex hormone therapy: a survey of the practices of selected clinics. *International Journal of Transgenderism*, *13*(3), 140-146. https://doi.org/10.1080/15532739.2011.675233
- Devor, A. H. (2004). Witnessing and mirroring: A fourteen stage model of transsexual identity formation. *Journal of Gay & Lesbian Psychotherapy*, 8(1-2), 41-67. <u>https://doi.org/10.1300/J236v08n01\_05</u>
- Dhejne, C., Van Vlerken, R., Heylens, G., & Arcelus, J. (2016). Mental health and gender dysphoria: A review of the literature. *International Review of Psychiatry*, 28(1), 44-57. https://doi.org/10.3109/09540261.2015.1115753
- Dickey, L. M. & Budge, S. L. (2020). Suicide and the transgender experience: A public health crisis. American Psychologist, 75(3), 380. <u>https://doi.org/10.1037/amp0000619</u>
- Dietert, M., & Dentice, D. (2013). Growing up trans: Socialization and the gender binary. *Journal of GLBT Family Studies*, 9(1), 24-42. <u>https://doi.org/10.1080/1550428X.2013.746053</u>
- Downing, J. M., & Przedworski, J. M. (2018). Health of transgender adults in the US, 2014–2016. *American Journal of Preventive Medicine*, 55(3), 336-344. <u>https://doi.org/10.1016/j.amepre.2018.04.045</u>
- Dyar, C., Sarno, E. L., Newcomb, M. E., & Whitton, S. W. (2020). Longitudinal associations between minority stress, internalizing symptoms, and substance use among sexual and gender minority individuals assigned female at birth. *Journal of Consulting and Clinical Psychology*, 88(5), 389. <u>https://doi.org/10.1037/ccp0000487</u>
- Eatough, Virginia and Smith, Jonathan A. (2017). Interpretative phenomenological analysis. In: Willig, C. and Stainton-Rogers, W. (eds.) *Handbook of Qualitative Psychology 2nd Edition*. (pp. 193-211). Sage.
- Ebstein, R. P., Novick, O., Umansky, R., Priel, B., Osher, Y., Blaine, D., ... & Belmaker, R. H. (1996).
  Dopamine D4 receptor (D4DR) exon III polymorphism associated with the human personality trait of novelty seeking. *Nature Genetics*, 12(1), 78-80. <u>https://doi.org/10.1038/ng0196-78</u>
- Ellard-Gray, A., Jeffrey, N. K., Choubak, M., & Crann, S. E. (2015). Finding the hidden participant: Solutions for recruiting hidden, hard-to-reach, and vulnerable populations. *International Journal of Qualitative Methods*, 14(5), <u>https://doi.org/10.1177/1609406915621420</u>
- Ellis, S. J., Bailey, L., & McNeil, J. (2015). Trans people's experiences of mental health and gender identity services: A UK study. *Journal of Gay & Lesbian Mental Health*, 19(1), 4-20, <u>https://doi.org/10.1080/19359705.2014.960990</u>
- Etter, J. F. (2010). Smoking and Cloninger's temperament and character inventory. *Nicotine & Tobacco Research*, 12(9), 919–926. <u>https://doi.org/10.1093/ntr/ntq116</u>
- Ervin, A. M. (2005). Male gender role conflict and internalized homonegativity: The impact on gay men's psychological well-being. *Dissertation Abstracts International: Section B: The Sciences and Engineering*, 65(7-B), 3704.

- Eyssel, J., Koehler, A., Dekker, A., Sehner, S., & Nieder, T. O. (2017) Needs and concerns of transgender individuals regarding interdisciplinary transgender healthcare: A non-clinical online survey. *PLoS One*, 12(8), <u>https://doi.org/10.1371/journal.pone.0183014</u>
- Farmer, R. F., & Goldberg, L. R. (2008). A psychometric evaluation of the revised Temperament and Character Inventory (TCI-R) and the TCI-140. *Psychological Assessment*, 20(3), 281. https://doi.org/10.1037/a0012934
- Farré, J. M., Fernández-Aranda, F., Granero, R., Aragay, N., Mallorquí-Bague, N., Ferrer, V., ... & Jiménez-Murcia, S. (2015). Sex addiction and gambling disorder: similarities and differences. *Comprehensive Psychiatry*, 56, 59-68. <u>https://doi.org/10.1016/j.comppsych.2014.10.002</u>
- Fischer, A. R. (2007). Parental relationship quality and masculine gender-role strain in young men: Mediating effects of personality. *The Counseling Psychologist*, 35(2), 328-358. <u>https://doi.org/10.1177%2F0011000005283394</u>
- Felce, D., & Perry, J. (1995). Quality of life: Its definition and measurement. *Research in Developmental Disabilities*, 16(1), 51-74. <u>https://doi.org/10.1016/0891-4222(94)00028-8</u>
- Felner, J. K., Wisdom, J. P., Williams, T., Katuska, L., Haley, S. J., Jun, H. J., & Corliss, H. L. (2020). Stress, coping, and context: examining substance use among LGBTQ young adults with probable substance use disorders. *Psychiatric Services*, 71(2), 112-120. <u>https://doi.org/10.1176/appi.ps.201900029</u>
- Fetters, M. D., Curry, L. A., & Creswell, J. W. (2013). Achieving integration in mixed methods designs principles and practices. *Health Services Research*, 48(6.2), 2134-2156. <u>https://doi.org/10.1111/1475-6773.12117</u>
- Finley, C. (2020). Paternalism and Autonomy in Transgender Healthcare. Undergraduate Honors Thesis Collection. 518. <u>https://digitalcommons.butler.edu/ugtheses/518</u>
- Foulds, J. A., Mulder, R. T., Newton-Howes, G., Adamson, S. J., Boden, J. M., & Sellman, J. D. (2016). Personality predictors of drinking outcomes in depressed alcohol-dependent patients. *Alcohol and Alcoholism*, 51(3), 296-301. <u>https://doi.org/10.1093/alcalc/agv122</u>
- Foulds, J. A., Boden, J. M., Newton-Howes, G. M., Mulder, R. T., & Horwood, L. J. (2017). The role of novelty seeking as a predictor of substance use disorder outcomes in early adulthood. *Addiction*, 112(9), 1629-1637. <u>https://doi.org/10.1111/add.13838</u>
- Fox, N. J. (2008). Postpositivism. The SAGE encyclopaedia of qualitative research methods, 2, 659-664.
- Frost, D. M., Lehavot, K., & Meyer, I. H. (2015). Minority stress and physical health among sexual minority individuals. *Journal of Behavioral Medicine*, 38(1), 1-8. <u>https://doi.org/10.1007/s10865-013-9523-8</u>
- Fricker, M. (2006). Powerlessness and social interpretation. *Episteme*, *3*(1-2), 96-108. https://doi.org/10.3366/epi.2006.3.1-2.96
- Galupo, M. P., Davis, K. S., Grynkiewicz, A. L., & Mitchell, R. C. (2014). Conceptualization of sexual orientation identity among sexual minorities: Patterns across sexual and gender identity. *Journal of Bisexuality*, 14(3-4), 433–456. https://doi.org/10.1080/15299716.2014.933466
- Galupo, M. P., Henise, S. B., & Mercer, N. L. (2016). "The labels don't work very well": Transgender individuals' conceptualizations of sexual orientation and sexual identity. *International Journal of Transgenderism*, 17(2), 93-104. <u>https://doi.org/10.1080/15532739.2016.1189373</u>
- Galupo, M. P., Pulice-Farrow, L., & Pehl, E. (2021). "There is nothing to do about it": Nonbinary individuals' experience of gender dysphoria. *Transgender Health*, 6(2), 101-110. <u>https://doi.org/10.1089/trgh.2020.0041</u>

- Gendall, P., Hoek, J., Stanley, J., Jenkins, M., & Every-Palmer, S. (2021). Changes in tobacco use during the 2020 COVID-19 lockdown in New Zealand. *Nicotine and Tobacco Research*, 23(5), 866-871. <u>https://doi.org/10.1093/ntr/ntaa257</u>
- George, O., & Koob, G. F. (2017). Individual differences in the neuropsychopathology of addiction. *Dialogues* in Clinical Neuroscience, 19(3), 217. <u>https://dx.doi.org/10.31887%2FDCNS.2017.19.3%2Fgkoob</u>
- Geusens, F., & Beullens, K. (2021). I see, therefore I am: exposure to alcohol references on social media, but not on traditional media, is related to alcohol consumption via drinking and non-drinking identity. *Health Communication*, 1-9. <u>https://doi.org/10.1080/10410236.2021.1954301</u>
- Gillon, R. (1985). Paternalism and medical ethics. British medical journal (Clinical research ed.), 290(6486).
- GIRES (2015). Retrieved 08/04/21, from https://www.gires.org.uk/nhs-gender-identity-clinics-england/
- Glynn, T. R., & van den Berg, J. J. (2017). A systematic review of interventions to reduce problematic substance use among transgender individuals: A call to action. *Transgender Health*, 2(1), 45-59. <u>https://doi.org/10.1089/trgh.2016.0037</u>
- Goldhammer, H., Crall, C., & Keuroghlian, A. S. (2019). Distinguishing and addressing gender minority stress and borderline personality symptoms. *Harvard Review of Psychiatry*, 27(5), 317-325. <u>https://doi.org/10.1097/hrp.0000000000234</u>
- Gómez-Gil, E., Gutiérrez, F., Cañizares, S., Zubiaurre-Elorza, L., Monràs, M., Esteva de Antonio, I., & Guillamón, A. (2013). Temperament and character in transsexuals. *Psychiatry Research*, 210(3), 969–974. https://doi.org/10.1016/j.psychres.2013.07.040
- Good, G. E., & Mintz, L. B. (1990). Gender role conflict and depression in college men: Evidence for compounded risk. *Journal of Counseling & Development*, 69(1), 17-21. <u>https://doi.org/10.1002/j.1556-6676.1990.tb01447.x</u>
- Good, G. E., & Wood, P. K. (1995). Male gender role conflict, depression, and help seeking: Do college men face double jeopardy? *Journal of Counseling & Development*, 74(1), 70-75. <u>https://doi.org/10.1002/j.1556-6676.1995.tb01825.x</u>
- Gov.uk (2022). What Qualification Levels Mean. Accessed 30/01/22 on https://www.gov.uk/what-different-qualification-levels-mean/list-of-qualification-levels
- Grant, J. E., Flynn, M., Odlaug, B. L., & Schreiber, L. R. (2011). Personality disorders in gay, lesbian, bisexual, and transgender chemically dependent patients. *The American Journal on Addictions*, 20(5), 405-411. <u>https://doi.org/10.1111/j.1521-0391.2011.00155.x</u>
- Green, J. (2004). Becoming a visible man. Vanderbilt University Press.
- Gusnard, D. A., Ollinger, J. M., Shulman, G. L., Cloninger, C. R., Price, J. L., Van Essen, D. C., & Raichle, M. (2003). Persistence and brain circuitry. *Proceedings of the National Academy of Sciences*, 100(6), 3479-3484. <u>https://doi.org/10.1073/pnas.0538050100</u>
- Ham, B. J., Choi, M. J., Lee, H. J., Kang, R. H., & Lee, M. S. (2005). Reward dependence is related to norepinephrine transporter T-182C gene polymorphism in a Korean population. *Psychiatric Genetics*, 15(2), 145-147. <u>https://doi.org/10.1097/00041444-200506000-00012</u>
- Hänel, H. C. (2020). Hermeneutical injustice, (self-) recognition, and academia. *Hypatia*, 35(2), 336-354. https://doi.org/10.1017/hyp.2020.3
- Hanna, B., Desai, R., Parekh, T., Guirguis, E., Kumar, G., & Sachdeva, R. (2019). Psychiatric disorders in the US transgender population. *Annals of Epidemiology*, 39, 1-7. https://doi.org/10.1016/j.annepidem.2019.09.009

- Haraldsen, I. R., & Dahl, A. A. (2000). Symptom profiles of gender dysphoric patients of transsexual type compared to patients with personality disorders and healthy adults. *Acta Psychiatrica Scandinavica*, 102(4), 276-281. <u>https://doi.org/10.1034/j.1600-0447.2000.102004276.x</u>
- Harley, J. A., Wells, J. E., Frampton, C., & Joyce, P. R. (2011). Bipolar disorder and the TCI: higher selftranscendence in bipolar disorder compared to major depression. *Depression Research and Treatment*, 2011. https://doi.org/10.1155/2011/529638
- Harrison, N., Jacobs, L., & Parke, A. (2020). Understanding the lived experiences of transitioning adults with gender Dysphoria in the United Kingdom: An interpretative phenomenological analysis. *Journal of LGBT Issues in Counseling*, 14(1), 38-55. <u>https://doi.org/10.1080/15538605.2020.1711292</u>
- Hashemi, S. G. S., Khoei, E. M., Hosseinnezhad, S., Mousavi, M., Dadashzadeh, S., Mostafaloo, T., ... & Yousefi, H. (2019). Personality traits and substance use disorders: comparative study with drug user and non-drug user population. *Personality and Individual Differences*, 148, 50-56. <u>https://doi.org/10.1016/j.paid.2019.05.015</u>
- Hassan, I., & Ali, R. (2011). The association between somatic symptoms, anxiety disorders and substance use. A literature review. *Psychiatric quarterly*, 82(4), 315-328. <u>https://doi.org/10.1007/s11126-011-9174-2</u>
- Hawthorne, G., Herrman, H., & Murphy, B. (2006). Interpreting the WHOQOL-BREF: Preliminary population norms and effect sizes. *Social Indicators Research*, 77(1), 37-59. <u>https://doi.org/10.1007/s11205-005-5552-1</u>
- Hayes, N. (2013). Doing qualitative analysis in psychology. Psychology Press.
- Heng, A., Heal, C., Banks, J., & Preston, R. (2018). Transgender peoples' experiences and perspectives about general healthcare: A systematic review. *International Journal of Transgenderism*, 19(4), 359-378, <u>https://doi.org/10.1080/15532739.2018.1502711</u>
- Herdman, K. J., Fuqua, D. R., Choi, N., & Newman, J. L. (2012). Gender Role Conflict Scale: Validation for a sample of gay men and lesbian women. *Psychological Reports*, 110(1), 227-232. <u>https://doi.org/10.2466%2F02.07.PR0.110.1.227-232</u>
- Herpertz, S. C., Huprich, S. K., Bohus, M., Chanen, A., Goodman, M., Mehlum, L., ... & Sharp, C. (2017). The challenge of transforming the diagnostic system of personality disorders. *Journal of Personality Disorders*, 31(5), 577. <u>https://doi.org/10.1521%2Fpedi\_2017\_31\_338</u>
- Hiestand, K. R., & Levitt, H. M. (2005). Butch identity development: The formation of an authentic gender. *Feminism & Psychology*, 15(1), 61-85. <u>https://doi.org/10.1177%2F0959353505049709</u>
- Higa, D., Hoppe, M. J., Lindhorst, T., Mincer, S., Beadnell, B., Morrison, D. M., & Mountz, S. (2014). Negative and positive factors associated with the well-being of lesbian, gay, bisexual, transgender, queer, and questioning (LGBTQ) youth. *Youth & Society*, 46(5), 663–687. <u>https://doi.org/10.1177%2F0044118X12449630</u>
- Holi, M. (2003). Assessment of psychiatric symptoms using the SCL-90. [Doctoral dissertation, Helsinki University]. Accessed on 21/01/22 at <u>https://helda.helsinki.fi/bitstream/handle/10138/22453/assessme.pdf</u>
- Horowitz, L. M., Rosenbert, S. E., Baer, B. A., Ureno, G., & Villasenor, V. S. (1988). Inventory of interpersonal problems: Psychometric properties and clinical applications. *Journal of Consulting and Clinical Psychology*, 56(6), 885–892. https://doi.org/10.1037/0022-006X.56.6.885
- Hughes, K., Bellis, M. A., Hardcastle, K. A., Sethi, D., Butchart, A., Mikton, C., ... & Dunne, M. P. (2017). The effect of multiple adverse childhood experiences on health: a systematic review and meta-analysis. *The Lancet Public Health*, 2(8), e356-e366. <u>https://doi.org/10.1016/S2468-2667(17)30118-4</u>

- Hughto, J. M., Quinn, E. K., Dunbar, M. S., Rose, A. J., Shireman, T. I., & Jasuja, G. K. (2021). Prevalence and co-occurrence of alcohol, nicotine, and other substance use disorder diagnoses among US transgender and cisgender adults. *JAMA Network Open*, 4(2), e2036512-e2036512. https://doi.org/10.1001/jamanetworkopen.2020.36512
- Hunt, J. (2012). Why the Gay and Transgender Population Experience Higher Rates of Substance Abuse. Centre for American Progress, Washington D.C. Accessed on 07/05/21 at https://cdn.americanprogress.org/wpcontent/uploads/issues/2012/03/pdf/lgbt\_substance\_abuse.pdf?\_ga=2.21460859.175235344 4.1620399028-1403356992.1620399028
- Informed Consent for Access to Trans Health. (n.d.). *Informed Consent for Access to Trans Health*. Retrieved 08/04/21 from <u>http://www.icath.org</u>
- Institute of Medicine. (2011). *The Health of Lesbian, Gay, Bisexual, and Transgender People: Building a Foundation for Better Understanding*. National Academies Press.
- Jäggi, T., Jellestad, L., Corbisiero, S., Schaefer, D. J., Jenewein, J., Schneeberger, A., ... & Garcia Nuñez, D. (2018). Gender minority stress and depressive symptoms in transitioned Swiss transpersons. *BioMed Research International*, 2018. <u>https://doi.org/10.1155/2018/8639263</u>
- James, S. E., Herman, J. L., Rankin, S., Keisling, M., Mottet, L., & Anafi, M. (2016). The Report of the 2015 U.S. Transgender Survey. Washington, DC: National Center for Transgender Equality.
- Jemberie, W. B., Williams, J. S., Eriksson, M., Grönlund, A. S., Ng, N., Nilsson, M. B., & Lundgren, L. M. (2020). Substance use disorders and COVID-19: multi-faceted problems which require multi-pronged solutions. *Frontiers in psychiatry*, 11. <u>https://doi.org/10.3389/fpsyt.2020.00714</u>
- John, O. P., Donahue, E. M., & Kentle, R. L. (1991). Big five inventory. Journal of Personality and Social Psychology. <u>https://doi.org/10.1037/t07550-000</u>
- John, A. N., Stoebenau, K., Ritter, S., Edmeades, J., & Balvin, N. (2017). Gender socialization during adolescence in low and middle-income countries: conceptualization, influences, and outcomes. UNICEF Office of Research. Accessed on 11/01/21 from https://www.unicef-irc.org/publications/ pdf/IDP\_2017\_01.pdf
- Jome, L. M., & Tokar, D. M. (1997). Dimensions of masculinity and major choice traditionality. Journal of Vocational Behavior, 52, 120-134. <u>https://doi.org/10.1037/0022-0167.45.4.424</u>
- Joyce, P. R., Mulder, R. T., Luty, S. E., McKenzie, J. M., Sullivan, P. F., & Cloninger, R. C. (2003). Borderline personality disorder in major depression: Symptomatology, temperament, character, differential drug response, and 6-month outcome. *Comprehensive Psychiatry*, 44(1), 35–43. <u>https://doi.org/10.1053/comp.2003.50001</u>
- Karimi, M., & Brazier, J. (2016). Health, health-related quality of life, and quality of life: what is the difference?. *Pharmacoeconomics*, 34(7), 645-649. https://doi.org/10.1007/s40273-016-0389-9
- Kassing, L. R., Beesley, D., & Frey, L. L. (2005). Gender role conflict, homophobia, age, and education as predictors of male rape myth acceptance. *Journal of Mental Health Counseling*, 27(4), 311-328. <u>https://doi.org/10.17744/mehc.27.4.9wfm24f52kqgav37</u>
- Kenney, M. (2001). Mapping gay LA: The intersection of place and politics. Temple University Press.
- Keuroghlian, A. S., Reisner, S. L., White, J. M., & Weiss, R. D. (2015). Substance use and treatment of substance use disorders in a community sample of transgender adults. *Drug and Alcohol Dependence*, 152, 139-146. <u>https://doi.org/10.1016/j.drugalcdep.2015.04.008</u>
- Keyzers, A., Lee, S. K., & Dworkin, J. (2020). Peer pressure and substance use in emerging adulthood: a latent profile analysis. *Substance Use & Misuse*, 55(10), 1716-1723. <u>https://doi.org/10.1080/10826084.2020.1759642</u>

- Kidd, J. D., Jackman, K. B., Wolff, M., Veldhuis, C. B., & Hughes, T. L. (2018). Risk and protective factors for substance use among sexual and gender minority youth: A scoping review. *Current Addiction Reports*, 5(2), 158-173. <u>https://doi.org/10.1007/s40429-018-0196-9</u>
- Kim, J. U., Majid, A., Judge, R., Crook, P., Nathwani, R., Selvapatt, N., ... & Lemoine, M. (2020). Effect of COVID-19 lockdown on alcohol consumption in patients with pre-existing alcohol use disorder. *The Lancet Gastroenterology & Hepatology*, 5(10), 886-887. https://doi.org/10.1016/S2468-1253(20)30251-X
- Kimmel, S. B., & Mahalik, J. R. (2005). Body image concerns of gay men: the roles of minority stress and conformity to masculine norms. *Journal of Consulting and Clinical Psychology*, 73(6), 1185. <u>https://doi.org/10.1037/0022-006X.73.6.1185</u>
- Klein, A., & Golub, S. A. (2016). Family rejection as a predictor of suicide attempts and substance misuse among transgender and gender nonconforming adults. *LGBT Health*, 3(3), 193-199. <u>https://doi.org/10.1089/lgbt.2015.0111</u>
- Koller, G., Zill, P., Skoruppa, T., Bondy, B., Preuss, U. W., & Soyka, M. (2008). Low level of harm avoidance is associated with serotonin transporter functional haplotype in alcohol-dependent individuals. *Psychiatric Genetics*, 18(2), 59-63. <u>https://doi.org/10.1097/ypg.0b013e3282f60333</u>
- Kollmayer, M., Schultes, M. T., Schober, B., Hodosi, T., & Spiel, C. (2018). Parents' judgments about the desirability of toys for their children: Associations with gender role attitudes, gender-typing of toys, and demographics. Sex Roles, 79(5-6), 329-341. <u>https://doi.org/10.1007/s11199-017-0882-4</u>
- Korcuska, J. S., & Thombs, D. L. (2003). Gender role conflict and sex-specific drinking norms: Relationships to alcohol use in undergraduate women and men. *Journal of College Student Development*, 44(2), 204-216. <u>https://doi.org/10.1353/csd.2003.0017</u>
- Kotov, R., Gamez, W., Schmidt, F., & Watson, D. (2010). Linking "big" personality traits to anxiety, depressive, and substance use disorders: a meta-analysis. *Psychological Bulletin*, 136(5), 768. <u>https://doi.org/10.1037/a0020327</u>
- Kowal, M., Coll-Martín, T., Ikizer, G., Rasmussen, J., Eichel, K., Studzińska, A., ... & Ahmed, O. (2020). Who is the most stressed during the COVID-19 pandemic? Data from 26 countries and areas. *Applied Psychology: Health and Well-Being*, 12(4), 946-966. <u>https://dx.doi.org/10.1111%2Faphw.12234</u>
- Kratzner, R. E. (2003). Gender role conflict, instrumentality-expressiveness, personality, and psychological distress in college males. [Master's thesis, St. Louis University].
- Kuper, L. E., Wright, L., & Mustanski, B. (2018). Gender identity development among transgender and gender nonconforming emerging adults: An intersectional approach. *International Journal of Transgenderism*, 19(4), 436-455, https://doi.org/10.1080/15532739.2018.1443869
- Laucht, M., El-Faddagh, M., Hohm, E., & Schmidt, M. H. (2005). Association of the DRD4 exon III polymorphism with smoking in fifteen-year-olds: a mediating role for novelty seeking? *Journal of the American Academy of Child & Adolescent Psychiatry*, 44(5), 477-484. <u>https://doi.org/10.1097/01.chi.0000155980.01792.7f</u>
- Le Bon, O., Basiaux, P., Streel, E., Tecco, J., Hanak, C., Hansenne, M., ... & Dupont, S. (2004). Personality profile and drug of choice; a multivariate analysis using Cloninger's TCI on heroin addicts, alcoholics, and a random population group. *Drug and Alcohol Dependence*, 73(2), 175-182. <u>https://doi.org/10.1016/j.drugalcdep.2003.10.006</u>
- Lefevor, G. T., Boyd-Rogers, C. C., Sprague, B. M., & Janis, R. A. (2019). Health disparities between genderqueer, transgender, and cisgender individuals: An extension of minority stress theory. *Journal of Counseling Psychology*, 66(4), 385. <u>https://doi.org/10.1037/cou0000339</u>
- Lev, A. I. (2004). *Transgender emergence: Therapeutic guidelines for working with gender-variant people and their families.* Haworth Clinical Practice Press.
- Levant, R. F., & Richmond, K. (2008). A review of research on masculinity ideologies using the Male Role Norms Inventory. *The Journal of Men's Studies*, 15(2), 130-146. <u>https://doi.org/10.3149%2Fjms.1502.130</u>
- Leyro, T. M., Zvolensky, M. J., & Bernstein, A. (2010). Distress tolerance and psychopathological symptoms and disorders: a review of the empirical literature among adults. *Psychological Bulletin*, 136(4), 576. <u>https://doi.org/10.1037/a0019712</u>
- Linander, I., Alm, E., Hammarström, A., & Harryson, L. (2017). Negotiating the (bio) medical gaze– Experiences of trans-specific healthcare in Sweden. *Social Science & Medicine*, 174, 9-16. <u>https://doi.org/10.1016/j.socscimed.2016.11.030</u>
- Livingston, N. A., Flentje, A., Heck, N. C., Szalda-Petree, A., & Cochran, B. N. (2017). Ecological momentary assessment of daily discrimination experiences and nicotine, alcohol, and drug use among sexual and gender minority individuals. *Journal of Consulting and Clinical Psychology*, 85(12), 1131. https://psycnet.apa.org/doi/10.1037/ccp0000252
- Lombardi, E. (2007). Substance use treatment experiences of transgender/transsexual men and women. *Journal* of LGBT Health Research, 3, 37–47. <u>https://doi.org/10.1300/J463v03n02\_05</u>
- Lykens, J. E., LeBlanc, A. J., & Bockting, W. O. (2018). Healthcare experiences among young adults who identify as genderqueer or nonbinary. *LGBT Health*, 5(3), 191-196. <u>https://doi.org/10.1089/lgbt.2017.0215</u>
- Macapagal, K., Bhatia, R., & Greene, J. (2016). Differences in healthcare access, use and experiences within a community sample of racially diverse lesbian, gay, bisexual, transgender, and questioning emerging adults. *LGBT Health*, 3(6), 434-442. <u>https://dx.doi.org/10.1089%2Flgbt.2015.0124</u>
- MacMillan, T., Corrigan, M. J., Coffey, K., Tronnier, C. D., Wang, D., & Krase, K. (2021). Exploring factors associated with alcohol and/or substance use during the COVID-19 pandemic. *International Journal* of Mental Health and Addiction, 1-10. <u>https://doi.org/10.1007/s11469-020-00482-y</u>
- Mahalik, J. R., & Lagan, H. D. (2001). Examining masculine gender role conflict and stress in relation to religious orientation and spiritual well-being. *Psychology of Men & Masculinity*, 2(1), 24-33. <u>https://doi.org/10.1037/1524-9220.2.1.24</u>
- Mak, J., Shires, D. A., Zhang, Q., Prieto, L. R., Ahmedani, B. K., Kattari, L., ... & Goodman, M. (2020). Suicide attempts among a cohort of transgender and gender diverse people. *American Journal of Preventive Medicine*, 59(4), 570-577. <u>https://doi.org/10.1016/j.amepre.2020.03.026</u>
- Manieri, C., Castellano, E., Crespi, C., Di Bisceglie, C., Dell'Aquila, C., Gualerzi, A., & Molo, M. (2014). Medical treatment of subjects with gender identity disorder: the experience in an Italian public health center. *International Journal of Transgenderism*, 15(2), 53-65. <u>https://doi.org/10.1080/15532739.2014.899174</u>
- Markett, S., Montag, C., & Reuter, M. (2016). Anxiety and harm avoidance. In *Neuroimaging personality, social cognition, and character* (pp. 91-112). Academic Press. https://doi.org/10.1016/B978-0-12-800935-2.00005-1
- Marwaha, S., Broome, M. R., Bebbington, P. E., Kuipers, E., & Freeman, D. (2014). Mood instability and psychosis: analyses of British national survey data. *Schizophrenia Bulletin*, 40(2), 269-277. <u>https://doi.org/10.1093/schbul/sbt149</u>
- Masiero, M., Cutica, I., Mazzocco, K., Zunino, A., Cropley, M., & Pravettoni, G. (2021). A Comprehensive Model of Tobacco Cigarette Smoking in Adolescence: The Role of Attachment Style and Personality. *The Journal of Psychology*, 155(7), 589-605. https://doi.org/10.1080/00223980.2021.1934374

- Matsuno, E., & Israel, T. (2018). Psychological interventions promoting resilience among transgender individuals: Transgender resilience intervention model (TRIM). *The Counseling Psychologist*, 46(5), 632-655. <u>https://doi.org/10.1177%2F0011000018787261</u>
- McCann, E., & Sharek, D. (2016). Mental health needs of people who identify as transgender: A review of the literature. Archives of Psychiatric Nursing, 30(2), 280-285, <u>https://doi.org/10.1016/j.apnu.2015.07.003</u>
- McCann, E., & Brown, M. (2017). Discrimination and resilience and the needs of people who identify as transgender: a narrative review of quantitative research studies. *Journal of Clinical Nursing*, 26(23-24), 4080-4093. <u>https://doi.org/10.1111/jocn.13913</u>
- McDermott, R. C., Schwartz, J. P., Lindley, L. D., & Proietti, J. S. (2014). Exploring men's homophobia: Associations with religious fundamentalism and gender role conflict domains. *Psychology of Men & Masculinity*, 15(2), 191–200. <u>https://doi.org/10.1037/a0032788</u>
- McDuff, D. R., & Baron, D. (2005). Substance use in athletics: a sports psychiatry perspective. *Clinics in Sports Medicine*, 24(4), 885-897. <u>https://doi.org/10.1016/j.csm.2005.06.004</u>
- McGuire, J. K., Doty, J. L., Catalpa, J. M., & Ola, C. (2016). Body image in transgender young people: Findings from a qualitative, community-based study. *Body Image*, 18, 96-107. https://doi.org/10.1016/j.bodyim.2016.06.004
- McInroy, L. B., McCloskey, R. J., Craig, S. L., & Eaton, A. D. (2019). LGBTQ+ youths' community engagement and resource seeking online versus offline. *Journal of Technology in Human Services*, 37(4), 315-333. <u>https://doi.org/10.1080/15228835.2019.1617823</u>
- McLellan, A. T. (2017). Substance misuse and substance use disorders: why do they matter in healthcare? *Transactions of the American Clinical and Climatological Association, 128*, 112. http://www.ncbi.nlm.nih.gov/pmc/articles/pmc5525418/
- McNeil, J., Bailey, L., Ellis, S., Morton, J., & Regan, M. (2012). Trans mental health study 2012. Scottish Transgender Alliance. Accessed 05/04/21: http://www. scottishtrans. org/wp-content/uploads/2013/03/trans\_mh\_study. pdf.
- McQuillan, M. T., Kuhns, L. M., Miller, A. A., McDade, T., & Garofalo, R. (2021). Gender minority stress, support, and inflammation in transgender and gender-nonconforming youth. *Transgender Health*, 6(2), 91-100. <u>https://doi.org/10.1089/trgh.2020.0019</u>
- Mercier, C., Brochu, S., Girard, M., Gravel, J., Ouellet, R., & ParÉ, R. (1992). Profiles of alcoholics according to the SCL-90-R: a confirmative study. *International Journal of the Addictions*, 27(11), 1267-1282. <u>https://doi.org/10.3109/10826089209047349</u>
- Merleau-Ponty, M. (1962). Phenomenology of perception. Routledge & Kegan Paul.
- Metzger, N. Y., & Boettger, S. (2019). The effect of testosterone therapy on personality traits of trans men: a controlled prospective study in Germany and Switzerland. *Psychiatry Research*, 276, 31–38. <u>https://doi.org/10.1016/j.psychres.2019.03.053</u>
- Meyer, I. H. (1995). Minority stress and mental health in gay men. *Journal of Health and Social Behavior*, 38-56. <u>https://doi.org/10.2307/2137286</u>
- Meyer, I. H. (2003). Prejudice, social stress, and mental health in lesbian, gay, and bisexual populations: Conceptual issues and research evidence. *Psychological Bulletin*, 129, 674–697. <u>https://doi.org/10.1037/0033-2909.129.5.674</u>
- Meyer, I. H., & Frost, D. M. (2013). Minority stress and the health of sexual minorities. In C. J. Patterson & A. R. D'Augelli (Eds.), *Handbook of psychology and sexual orientation* (pp. 252–266). Oxford University Press.

- Miller, C. F., Trautner, H. M., & Ruble, D. N. (2006). The role of gender stereotypes in children's preferences and behavior. In L. Balter & C. S. Tamis-LeMonda (Eds.), *Child psychology: A handbook of contemporary issues* (pp. 293–323). Routledge.
- Miller, L. R., & Grollman, E. A. (2015). The social costs of gender nonconformity for transgender adults: Implications for discrimination and health. *Sociological Forum 30*(3), 809-831. <u>https://doi.org/10.1111/socf.12193</u>
- Miovský, M., Lukavská, K., Rubášová, E., Šťastná, L., Šefránek, M., & Gabrhelík, R. (2021). Attention Deficit Hyperactivity Disorder among clients diagnosed with a substance use disorder in the therapeutic communities: Prevalence and psychiatric comorbidity. *European Addiction Research*, 27(2), 87-96. <u>https://doi.org/10.1159/000508571</u>
- Mirijello, A., D'Angelo, C., Ferrulli, A., Vassallo, G., Antonelli, M., Caputo, F., ... & Addolorato, G. (2015). Identification and management of alcohol withdrawal syndrome. *Drugs*, 75(4), 353-365. <u>https://doi.org/10.1007/s40265-015-0358-1</u>
- Miyajima, E., Taira, N., Koda, M., & Kondo, T. (2014). Differences in personality traits between male-tofemale and female-to-male gender identity disorder subjects. *Psychiatry Research*, 220(1-2), 496-499. <u>https://doi.org/10.1016/j.psychres.2014.07.069</u>
- Monk, D., & Ricciardelli, L. A. (2003). Three dimensions of the male gender role as correlates of alcohol and cannabis involvement in young Australian men. *Psychology of Men & Masculinity*, 4(1), 57. <u>https://doi.org/10.1037/1524-9220.4.1.57</u>
- Muller, A. E., Skurtveit, S., & Clausen, T. (2016). Many correlates of poor quality of life among substance users entering treatment are not addiction-specific. *Health and Quality of Life Outcomes*, *14*(1), 1-10. <u>https://doi.org/10.1186/s12955-016-0439-1</u>
- Murphy, J., Prentice, F., Walsh, R., Catmur, C., & Bird, G. (2020). Autism and transgender identity: implications for depression and anxiety. *Research in Autism Spectrum Disorders*, 69, 101466. <u>https://doi.org/10.1016/j.rasd.2019.101466</u>
- Nadal, K. L., Skolnik, A., & Wong, Y. (2012). Interpersonal and systemic microaggressions toward transgender people: Implications for counseling. *Journal of LGBT Issues in Counseling*, 6(1), 55–82. <u>https://doi.org/10.1080/15538605.2012.648583</u>
- Narang, P., Sarai, S. K., Aldrin, S., & Lippmann, S. (2018). Suicide among transgender and gendernonconforming people. *The Primary Care Companion for CNS Disorders*, 20(3), 26899. <u>https://doi.org/10.4088/PCC.18nr02273</u>
- Newfield, E., Hart, S., Dibble, S., & Kohler, L. (2006). Female-to-male transgender quality of life. Quality of Life Research, 15(9), 1447-1457. <u>https://doi.org/10.1007/s11136-006-0002-3</u>
- Newman, B. M., & Newman, P. R. (2017). *Development through life: A psychosocial approach*. Cengage Learning
- NICE. (2011). Alcohol-use disorders: Diagnosis, assessment, and management of harmful drinking and alcohol dependence [Clinical guideline CG115]. Retrieved on 12/05/21 at https://www.nice.org.uk/guidance/cg115/chapter/1-Guidance#interventions-for-alcohol-misuse
- NICE. (2007). Drug misuse in over 16s: Psychosocial interventions [Clinical guideline CG51]. Retrieved on 12/05/21 at https://www.nice.org.uk/guidance/cg51
- NHS England (2019). Service specifications for gender dysphoria. Retrieved 08/04/21 at <u>https://www.england.nhs.uk/wp-content/uploads/2019/07/service-specification-gender-dysphoria-</u> services-non-surgical-june-2019.pdf
- NHS (2019). Guide to NHS Waiting times in England. Retrieved 08/04/21 at https://www.nhs.uk/using-the-nhs/nhs-services/hospitals/guide-to-nhs-waiting-times-in-england/

- Nieder, T. O., Eyssel, J., & Köhler, A. (2019). Being trans without medical transition: Exploring characteristics of trans individuals from Germany not seeking gender-affirmative medical interventions. Archives of Sexual Behavior, 1-12, <u>https://doi.org/10.1007/s10508-019-01559-z</u>
- Nobili, A., Glazebrook, C., & Arcelus, J. (2018). Quality of life of treatment-seeking transgender adults: a systematic review and meta-analysis. *Reviews in Endocrine and Metabolic Disorders*, *19*(3), 199-220. <u>https://doi.org/10.1007/s11154-018-9459-y</u>
- Nock, M. K., & Prinstein, M. J. (2004). A functional approach to the assessment of self-mutilative behavior. *Journal of Consulting and Clinical Psychology*, 72(5), 885. <u>https://doi.org/10.1037/0022-006X.72.5.885</u>
- Nuttbrock, L. A. (2012). Culturally competent substance abuse treatment with transgender persons. *Journal of Addictive Diseases*, 31(3), 236–241. <u>https://doi.org/10.1080/10550887.2012.694600</u>
- Nuttbrock, L., Bockting, W., Rosenblum, A., Hwahng, S., Mason, M., Macri, M., & Becker, J. (2015). Gender abuse and incident HIV/STI among transgender women in New York City: buffering effect of involvement in a transgender community. *AIDS and Behavior*, *19*(8), 1446-1453. <u>https://doi.org/10.1007/s10461-014-0977-7</u>
- O'brien, R. M. (2007). A caution regarding rules of thumb for variance inflation factors. *Quality & Quantity*, 41(5), 673-690. <u>https://doi.org/10.1007/s11135-006-9018-6</u>
- Odriozola-González, P., Planchuelo-Gómez, Á., Irurtia, M. J., & de Luis-García, R. (2020). Psychological symptoms of the outbreak of the COVID-19 confinement in Spain. *Journal of Health Psychology*, 1359105320967086. <u>https://doi.org/10.1177%2F1359105320967086</u>
- Okasha, T., Abd Elsamie, A., Azzam, H., Elserafi, D., Morsy, M., & Shorub, E. (2021). Emotional Regulation as a Mediating Factor in Substance Use Disorders. *Addictive Disorders & Their Treatment*, 20(3), 202-210. <u>https://doi.org/10.1097/ADT.00000000000241</u>
- O'Neil, J. M. (1981). Male sex-role conflict, sexism, and masculinity: Implications for men, women, and the counseling psychologist. *The Counseling Psychologist*, 9, 61-80. https://doi.org/10.1177/001100008100900213
- O'Neil, J.M. (2008). Summarizing 25 years of research on men's gender role conflict using the gender role conflict scale: New research paradigms and clinical implications. *The Counseling Psychologist*, 36(3), 358-445. <u>https://doi.org/10.1177%2F0011000008317057</u>
- O'Neil, J. M. (2013). Gender role conflict research 30 years later: An evidence-based diagnostic schema to assess boys and men in counseling. *Journal of Counseling & Development*, 91(4), 490-498. <u>https://doi.org/10.1002/j.1556-6676.2013.00122.x</u>
- O'Neil, J. M. (2015). *Men's gender role conflict: Psychological costs, consequences, and an agenda for change*. American Psychological Association.
- O'Neil, J. M., Helm, B., Gable, R., David, L., & Wrightsman, L. (1986). Gender Role Conflict Scale (GRCS): College men's fears of femininity. *Sex Roles*, *14*, 335-350. <u>https://doi.org/10.1007/BF00287583</u>
- O'Neil, J. M., Wester, S. R., Heesacker, M., & Snowden, S. J. (2017). Masculinity as a heuristic: Gender role conflict theory, superorganisms, and system-level thinking. In R. F. Levant & Y. J. Wong (Eds.), *The psychology of men and masculinities* (pp. 75–103). American Psychological Association. <u>https://doi.org/10.1037/0000023-004</u>
- ONS (Office for National Statistics). (July, 2019) *Adult Smoking Habits in Great Britain*. Accessed 05/04/21 at: https://www.ons.gov.uk/peoplepopulationandcommunity/healthandsocialcare/drugusealcoholandsmoking/datasets/adultsmokinghabitsingreatbritain

- Osborne, T. L. (2004). Male gender role conflict and perceived social support: Predicting help seeking in college men. *Dissertation Abstracts International: Section B: The Sciences and Engineering*, 65(6-B), 3175.
- Otto, M. W., Powers, M. B., & Fischmann, D. (2005). Emotional exposure in the treatment of substance use disorders: Conceptual model, evidence, and future directions. *Clinical Psychology Review*, 25(6), 824-839. <u>https://psycnet.apa.org/doi/10.1016/j.cpr.2005.05.002</u>
- Özdel, K., & Ekinci, S. (2014). Distress intolerance in substance dependent patients. *Comprehensive Psychiatry*, 55(4), 960-965. <u>https://doi.org/10.1016/j.comppsych.2013.12.012</u>
- Palmer, M., Larkin, M., de Visser, R., & Fadden, G. (2010). Developing an interpretative phenomenological approach to focus group data. *Qualitative Research in Psychology*, 7(2), 99-121. http://dx.doi.org/10.1080/14780880802513194
- Panhwar, A. H., Ansari, S., & Shah, A. A. (2017). Post-positivism: An effective paradigm for social and educational research. *International Research Journal of Arts & Humanities (IRJAH)*, 45(45).
- Parker, L. L., & Harriger, J. A. (2020). Eating disorders and disordered eating behaviors in the LGBT population: a review of the literature. *Journal of Eating Disorders*, 8(1), 1-20. https://doi.org/10.1186/s40337-020-00327-y
- Parolin, M., Simonelli, A., Mapelli, D., Sacco, M., & Cristofalo, P. (2016). Parental substance abuse as an early traumatic event. Preliminary findings on neuropsychological and personality functioning in young drug addicts exposed to drugs early. *Frontiers in Psychology*, 7, 887. <u>https://doi.org/10.3389/fpsyg.2016.00887</u>
- Pascoe, E. A., & Smart Richman, L. (2009). Perceived discrimination and health: a meta-analytic review. *Psychological Bulletin*, 135(4), 531. https://dx.doi.org/10.1037%2Fa0016059
- Pasman, J. A., Smit, D. J., Kingma, L., Vink, J. M., Treur, J. L., & Verweij, K. J. (2020). Causal relationships between substance use and insomnia. *Drug and Alcohol Dependence*, 214, 108151. <u>https://doi.org/10.1016/j.drugalcdep.2020.108151</u>
- Pearce, J., Rafiq, S., Simpson, J., & Varese, F. (2019). Perceived discrimination and psychosis: a systematic review of the literature. *Social Psychiatry and Psychiatric Epidemiology*, 54(9), 1023-1044. <u>https://doi.org/10.1007/s00127-019-01729-3</u>
- Pearce, R., Erikainen, S., & Vincent, B. (2020a). TERF wars: An introduction. *The Sociological Review*, 68(4), 677-698. <u>https://doi.org/10.1177%2F0038026120934713</u>
- Pearce, R., Erikainen, S., & Vincent, B. (2020b). Afterword: TERF wars in the time of COVID-19. *The Sociological Review*, 68(4), 882-888. <u>https://doi.org/10.1177%2F0038026120934712</u>
- Pelissolo, A., Moukheiber, A., & Mallet, L. (2015). Obsessive-compulsive disorders and anxiety disorders: A comparison of personality and emotionality patterns. *Psychiatry Research*, 229(3), 695-701. <u>https://doi.org/10.1016/j.psychres.2015.08.020</u>
- Perroud, N., Hasler, R., Golay, N., Zimmermann, J., Prada, P., Nicastro, R., ... & Baud, P. (2016). Personality profiles in adults with attention deficit hyperactivity disorder (ADHD). *BMC Psychiatry*, 16(1), 1-9. <u>https://doi.org/10.1186/s12888-016-0906-6</u>
- Perrykkad, K., & Hohwy, J. (2019). When big data aren't the answer. Proceedings of the National Academy of Sciences, 116(28), 13738-13739. https://doi.org/10.1073/pnas.1902050116
- Peterson, F. (1999, August). *Gender-role conflict as a risk factor among male smokers*. Paper presented at the meeting of the American Psychological Association, Boston
- Pietkiewicz, I., & Smith, J. A. (2014). A practical guide to using interpretative phenomenological analysis in qualitative research psychology. *Psychological Journal*, 20(1), 7-14. http://dx.doi.org/10.14691/CPPJ.20.1.7

- Pinter, A. T., Scheuerman, M. K., & Brubaker, J. R. (2021). Entering doors, evading traps: Benefits and risks of visibility during transgender coming outs. *Proceedings of the ACM on Human-Computer Interaction*, 4(CSCW3), 1-27. <u>https://doi.org/10.1145/3434181</u>
- Price, P. C., Jhangiani, R. S., & Chiang, I. C. A. (2015). Understanding Null Hypothesis Testing. *Research Methods in Psychology*. B.C. Open Textbook Collection, 2<sup>nd</sup> Canadian Edition.
- Puckett, J. A., Matsuno, E., Dyar, C., Mustanski, B., & Newcomb, M. E. (2019). Mental health and resilience in transgender individuals: What type of support makes a difference?. *Journal of Family Psychology*, 33(8), 954. <u>https://doi.org/10.1037/fam0000561</u>
- Ratnayake, S. (2022). 'I will never love anyone like that again': cognitive behavioural therapy and the pathologisation and medicalisation of ordinary experiences. *Medical Humanities*, 48(2), e7-e7. <u>https://doi.org/10.1136/medhum-2021-012210</u>
- Reichl, C., & Kaess, M. (2021). Self-harm in the context of borderline personality disorder. *Current Opinion in Psychology*, 37, 139-144. <u>https://doi.org/10.1016/j.copsyc.2020.12.007</u>
- Reisner, S. L., Pardo, S. T., Gamarel, K. E., Hughto, J. M. W., Pardee, D. J., & Keo-Meier, C. L. (2015a). Substance use to cope with stigma in healthcare among US female-to-male trans masculine adults. *LGBT Health*, 2(4), 324-332. <u>https://doi.org/10.1089/lgbt.2015.0001</u>
- Reisner, S. L., Bradford, J., Hopwood, R., Gonzalez, A., Makadon, H., Todisco, D., ... & Boswell, S. L. (2015b). Comprehensive transgender healthcare: the gender affirming clinical and public health model of Fenway Health. *Journal of Urban Health*, 92(3), 584-592. https://doi.org/10.1007/s11524-015-9947-2
- Rich, A. (2021). Epidemiology of chronic disease and multimorbidity for transgender people living with and without HIV. [Doctoral Thesis, University of British Columbia]. Retrieved from https://open.library.ubc.ca/collections/ubctheses/24/items/1.0396096
- Rodgers, J., Herrema, R., Garland, D., Osborne, M., Cooper, R., Heslop, P., & Freeston, M. (2019). Uncertain futures: Reporting the experiences and worries of autistic adults and possible implications for social work practice. *The British Journal of Social Work*, 49(7), 1817-1836. <u>https://doi.org/10.1093/bjsw/bcy117</u>
- Rood, B. A., Reisner, S. L., Surace, F. I., Puckett, J. A., Maroney, M. R., & Pantalone, D. W. (2016). Expecting rejection: Understanding the minority stress experiences of transgender and gender-nonconforming individuals. *Transgender Health*, 1(1), 151-164. <u>https://doi.org/10.1089/trgh.2016.0012</u>
- Rubio, V. G., Bermejo, V. J., Caballero, S. S. M., & Santo-Domingo, C. J. (1998). Validation of the alcohol use disorders identification test (AUDIT) in primary care. *Revista Clínica Española*, 198(1), 11.
- Ruiz-Olivares, R., Lucena, V., Raya, A. F., & Herruzo, J. (2019). Personality profiles and how they relate to drug consumption among young people in Spain. *Personality and Individual Differences*, 149, 291-295. <u>https://doi.org/10.1016/j.paid.2019.06.015</u>
- Salvatore, S., & Valsiner, J. (2010). Between the general and the unique: Overcoming the nomothetic versus idiographic opposition. *Theory & Psychology*, 20(6), 817-833. https://doi.org/10.1177%2F0959354310381156
- Sánchez, F. J., Westefeld, J. S., Liu, W. M., & Vilain, E. (2010). Masculine gender role conflict and negative feelings about being gay. *Professional Psychology: Research and Practice*, 41(2), 104. <u>https://dx.doi.org/10.1037%2Fa0015805</u>
- Scandurra, C., Amodeo, A. L., Valerio, P., Bochicchio, V., & Frost, D. M. (2017). Minority stress, resilience, and mental health: A study of Italian transgender people. *Journal of Social Issues*, 73(3), 563-585. https://doi.org/10.1111/josi.12232

- Scheim, A. I., Bauer, G. R., & Shokoohi, M. (2017). Drug use among transgender people in Ontario, Canada: Disparities and associations with social exclusion. *Addictive Behaviors*, 72, 151-158. <u>https://doi.org/10.1016/j.addbeh.2017.03.022</u>
- Schelle, K. J., Olthof, B. M., Reintjes, W., Bundt, C., Gusman-Vermeer, J., & Van Mil, A. C. (2015). A survey of substance use for cognitive enhancement by university students in the Netherlands. *Frontiers in Systems Neuroscience*, 9(10). <u>https://doi.org/10.3389/fnsys.2015.00010</u>
- Scheuerman, M. K., Branham, S. M., & Hamidi, F. (2018). Safe spaces and safe places: Unpacking technologymediated experiences of safety and harm with transgender people. *Proceedings of the ACM on Human-Computer Interaction*, 2(CSCW), 1-27. <u>http://dx.doi.org/10.1145/3274424</u>
- Schilt, K. (2006). Just one of the guys? How transmen make gender visible at work. *Gender & Society*, 20(4), 465-490. <u>https://doi.org/10.1177%2F0891243206288077</u>
- Schrock, D., & Schwalbe, M. (2009). Men, masculinity, and manhood acts. Annual review of Sociology, 35, 277-295. <u>https://doi.org/10.1146/annurev-soc-070308-115933</u>
- Schulte, S., Moring, J., Meier, P. S., & Barrowclough, C. (2007). User involvement and desired service developments in drug treatment: Service user and provider views. *Drugs: Education, Prevention, and Policy*, 14(3), 277-287. https://doi.org/10.1080/09687630701267317
- Schulz, S. L. (2018). The Informed Consent Model of transgender care: An alternative to the diagnosis of gender dysphoria. *Journal of Humanistic Psychology*, 58(1), 72–92. https://doi.org/10.1177%2F0022167817745217
- Schwartz, J. P., Waldo, M., & Daniel, D. (2005). Gender-Role Conflict and self-esteem: Factors associated with partner abuse in court-referred men. *Psychology of Men & Masculinity*, 6(2), 109. https://doi.org/10.1037/1524-9220.6.2.109
- Schweizer, V. J., & Mowen, T. J. (2020). Discrimination and risky sexual behavior, substance use, and suicidality among transgender individuals. *Deviant Behavior*, 115. <u>https://doi.org/10.1080/01639625.2020.1839817</u>
- Serna, G. S. (2004). The confounding role of personality in the relation to gender role conflict and substance abuse and sexual aggression against women (Doctoral dissertation, ProQuest Information & Learning).
- Sharpe, M. J., & Heppner, P. P. (1991). Gender role, gender-role conflict, and psychological well-being in men. Journal of Counseling Psychology, 38(3), 323. <u>https://doi.org/10.1037/0022-0167.38.3.323</u>
- Shaw, R.L. (2010). Embedding reflexivity within experiential qualitative psychology. *Qualitative Research in Psychology*, 7(3), 233- 243. https://doi.org/10.1080/14780880802699092
- Shelton, M. (2017). Fundamentals of LGBT substance use disorders: Multiple identities, multiple challenges. Harrington Park Press/The Haworth Press.
- Sher, K. J., Bartholow, B. D., & Wood, M. D. (2000). Personality and substance use disorders: a prospective study. *Journal of Consulting and Clinical Psychology*, 68(5), 818. <u>http://dx.doi.org/10.1037/0022-006X.68.5.818</u>
- Shevlin, M., Nolan, E., Owczarek, M., McBride, O., Murphy, J., Gibson Miller, J., ... & Bentall, R. P. (2020). COVID-19-related anxiety predicts somatic symptoms in the UK population. *British Journal of Health Psychology*, 25(4), 875-882. <u>https://doi.org/10.1111/bjhp.12430</u>
- Shinebourne, P. (2011). The Theoretical Underpinnings of Interpretative Phenomenological Analysis (IPA). *Existential Analysis: Journal of the Society for Existential Analysis*, 22(1), 16.

- Simonoff, E., Kent, R., Stringer, D., Lord, C., Briskman, J., Lukito, S., ... & Baird, G. (2020). Trajectories in symptoms of autism and cognitive ability in autism from childhood to adult life: findings from a longitudinal epidemiological cohort. *Journal of the American Academy of Child & Adolescent Psychiatry*, 59(12), 1342-1352. <u>https://doi.org/10.1016/j.jaac.2019.11.020</u>
- Sinha, R. (2008). Chronic stress, drug use, and vulnerability to addiction. Annals of the New York Academy of Sciences, 1141(105). https://dx.doi.org/10.1196%2Fannals.1441.030
- Sipes, M. L. (2005). A partial test of male gender role conflict theory: Current perceptions of gender role socialization, masculinity ideology, and gender role conflict; relations with personality and prediction of interpersonal problems. [Doctoral dissertation, University of Akron].
- Simonsen, G., Blazina, C., & Watkins Jr, C. E. (2000). Gender role conflict and psychological wellbeing among gay men. *Journal of Counseling Psychology*, 47(1), 85. <u>https://doi.org/10.1037/0022-0167.47.1.85</u>
- Skevington, S. M., & McCrate, F. M. (2012). Expecting a good quality of life in health: assessing people with diverse diseases and conditions using the WHOQOL-BREF. *Health Expectations*, 15(1), 49-62. <u>https://doi.org/10.1111/j.1369-7625.2010.00650.x</u>
- Sliedrecht, W., Roozen, H. G., Witkiewitz, K., de Waart, R., & Dom, G. (2021). The association between impulsivity and relapse in patients with alcohol use disorder: a literature review. *Alcohol and Alcoholism*, 56(6), 637-650. <u>https://doi.org/10.1093/alcalc/agaa132</u>
- Sloan, C., Conner, M., & Gough, B. (2015). How does masculinity impact on health? A quantitative study of masculinity and health behaviour in a sample of UK men and women. *Psychology of Men and Masculinity*, 16 (2). pp. 206-217. https://doi.org/10.1037/a0037261
- Smith, J. A. (2004). Reflecting on the development of interpretative phenomenological analysis and its contribution to qualitative research in psychology. *Qualitative Research in Psychology*, 1(1), 39-54, <u>https://doi.org/10.1191/1478088704qp004oa</u>
- Smith, J. A., & Osborn, M. (2003). Interpretative phenomenological analysis. In J. A. Smith (Ed.), *Qualitative psychology: A practical guide to research methods*. Sage Publications, Inc. (pp. 51–80).
- Smith, J., Flower, P., & Larkin, M. (2009). *Interpretative Phenomenological Analysis: Theory, Method, and Research*. London: Sage.
- Smith, D. C., Cleeland, L., & Dennis, M. L. (2010). Reasons for quitting among emerging adults and adolescents in substance-use-disorder treatment. *Journal of Studies on Alcohol and Drugs*, 71(3), 400-409. <u>https://dx.doi.org/10.15288%2Fjsad.2010.71.400</u>
- Smith, L., Jacob, L., Yakkundi, A., McDermott, D., Armstrong, N. C., Barnett, Y., ... & Tully, M. A. (2020). Correlates of symptoms of anxiety and depression and mental wellbeing associated with COVID-19: a cross-sectional study of UK-based respondents. *Psychiatry Research*, 291, 113138. <u>https://doi.org/10.1016/j.psychres.2020.113138</u>
- Smith, E., Carter, M., Walklet, E., & Hazell, P. (2021). Investigating the experiences of individuals in recovery from problem substance use and their perceptions of the COVID-19 pandemic. *Drugs and Alcohol Today*, 22(1), 17-27. <u>https://doi.org/10.1108/DAT-01-2021-0003</u>
- Spicer, S. S. (2010). Healthcare needs of the transgender homeless population. *Journal of Gay & Lesbian Mental Health*, 14(4), 320-339. https://doi.org/10.1080/19359705.2010.505844
- Stevens, L., Verdejo-García, A., Goudriaan, A. E., Roeyers, H., Dom, G., & Vanderplasschen, W. (2014). Impulsivity as a vulnerability factor for poor addiction treatment outcomes: a review of neurocognitive findings among individuals with substance use disorders. *Journal of Substance Abuse Treatment*, 47(1), 58-72. <u>https://doi.org/10.1016/j.jsat.2014.01.008</u>

- Stieglitz, K. A. (2010). Development, risk, and resilience of transgender youth. *Journal of the* Association of Nurses in AIDS Care, 21(3), 192-206. <u>https://doi.org/10.1016/j.jana.2009.08.004</u>
- Stokes, M., Schultz, P., & Alpaslan, A. (2018). Narrating the journey of sustained recovery from substance use disorder. Substance Abuse Treatment, Prevention, and Policy, 13(1), 1-12. <u>https://doi.org/10.1186/s13011-018-0167-0</u>
- Stonewall. (2018). *LGBT in Britain: Trans Report*. Retrieved from: https://www.stonewall.org.uk/system/files/lgbt\_in\_britain\_-\_trans\_report\_final.pdf
- Strang, J. F., Klomp, S. E., Caplan, R., Griffin, A. D., Anthony, L. G., Harris, M. C., ... & van der Miesen, A. I. (2019). Community-based participatory design for research that impacts the lives of transgender and/or gender-diverse autistic and/or neurodiverse people. *Clinical Practice in Pediatric Psychology*, 7(4), 396. <u>https://doi.org/10.1037/cpp0000310</u>
- Su, D., Irwin, J. A., Fisher, C., Ramos, A., Kelley, M., Mendoza, D. A. R., & Coleman, J. D. (2016). Mental health disparities within the LGBT population: A comparison between transgender and nontransgender individuals. *Transgender Health*, 1(1), 12-20. https://dx.doi.org/10.1089%2Ftrgh.2015.0001
- Sue, S., Zane, N., Nagayama Hall, G. C., & Berger, L. K. (2009). The case for cultural competency in psychotherapeutic interventions. *Annual Review of Psychology*, 60, 525-548. https://doi.org/10.1146/annurev.psych.60.110707.163651
- Sussman, S., Pokhrel, P., Ashmore, R. D., & Brown, B. B. (2007). Adolescent peer group identification and characteristics: A review of the literature. *Addictive Behaviors*, 32(8), 1602-1627. <u>https://doi.org/10.1016/j.addbeh.2006.11.018</u>
- Tan, K. K., Treharne, G. J., Ellis, S. J., Schmidt, J. M., & Veale, J. F. (2019). Gender minority stress: A critical review. *Journal of Homosexuality*. https://doi.org/10.1080/00918369.2019.1591789
- Tandon, R., & Shariff, S. M. (2019). Substance-induced psychotic disorders and schizophrenia: pathophysiological insights and clinical implications. *American Journal of Psychiatry*, 176(9), 683-684. <u>https://doi.org/10.1176/appi.ajp.2019.19070734</u>
- Tate, N.S. (1998). Social support and depression in male GP surgery attendees: An investigation of the link with gender role conflict. [Doctoral dissertation, University of Newcastle, UK].
- Tatum, A. K., Catalpa, J., Bradford, N. J., Kovic, A., & Berg, D. R. (2020). Examining identity development and transition differences among binary transgender and genderqueer nonbinary (GQNB) individuals. *Psychology of Sexual Orientation and Gender Diversity*. https://doi.apa.org/doi/10.1037/sgd0000377
- Taylor, E. T. (2013). Transmen's healthcare experiences: Ethical social work practice beyond the binary. Journal of Gay and Lesbian Social Services, 25(1), 102-120, https://doi.org/10.1080/10538720.2013.750575
- Taylor, J. (2018). Out of the darkness and into the shadows: The evolution of contemporary bisexuality. *The Canadian Journal of Human Sexuality*, 27(2), 103-109. <u>https://doi.org/10.3138/cjhs.2018-0014</u>
- Taylor, J., Zalewska, A., Gates, J. J., & Millon, G. (2019). An exploration of the lived experiences of non-binary individuals who have presented at a gender identity clinic in the United Kingdom. *International Journal of Transgenderism*, 20(2-3), 195-204. https://doi.org/10.1080/15532739.2018.1445056
- Testa, R. J., Habarth, J., Peta, J., Balsam, K., & Bockting, W. (2015). Development of the gender minority stress and resilience measure. *Psychology of Sexual Orientation and Gender Diversity*, 2(1), 65. <u>https://doi.org/10.1037/sgd0000081</u>

- Testa, R. J., Michaels, M. S., Bliss, W., Rogers, M. L., Balsam, K. F., & Joiner, T. (2017). Suicidal ideation in transgender people: Gender minority stress and interpersonal theory factors. *Journal of Abnormal Psychology*, 126(1), 125. <u>https://doi.org/10.1037/abn0000234</u>
- Testa, R. J., Sciacca, L. M., Wang, F., Hendricks, M. L., Goldblum, P., Bradford, J., & Bongar, B. (2012). Effects of violence on transgender people. *Professional Psychology, Research, and Practice, 43*, 452–459. http://dx.doi.org/10.1037/a0029604
- Tiffany, S. T., Friedman, L., Greenfield, S. F., Hasin, D. S., & Jackson, R. (2012). Beyond drug use: a systematic consideration of other outcomes in evaluations of treatments for substance use disorders. *Addiction*, 107(4), 709-718. https://doi.org/10.1111/j.1360-0443.2011.03581.x
- Timmins, L., Rimes, K. A., & Rahman, Q. (2017). Minority stressors and psychological distress in transgender individuals. *Psychology of Sexual Orientation and Gender Diversity*, 4(3). https://doi.org/10.1037/sgd0000237
- Tokar, D. M., Fischer, A. R., Schaub, M., & Moradi, B. (2000). Masculine gender roles and counseling-related variables: Links with and mediation by personality. *Journal of Counseling Psychology*, 47(3), 380. <u>https://doi.org/10.1037/0022-0167.47.3.380</u>
- Tomkin, L. & Eatough, V. (2010). Reflecting on the use of IPA with focus groups: Pitfalls and Potentials. *Qualitative Research in Psychology*, 7(3), 244-262, https://doi.org/10.1080/14780880903121491
- TransActual (2021). Trans Lives Survey 2021: Enduring the UK's Hostile Environment. Retrieved on 06/12/21 from: https://static1.squarespace.com/static/5e8a0a6bb02c73725b24dc9d/t/6152eac81e0b0109491dc518/
- Trofimova, I. (2014). Observer bias: an interaction of temperament traits with biases in the semantic perception of lexical material. *PloS One*, 9(1), e85677. https://dx.doi.org/10.1371%2Fjournal.pone.0085677
- Uher, J. (2013). Personality psychology: Lexical approaches, assessment methods, and trait concepts reveal only half of the story—Why it is time for a paradigm shift. *Integrative Psychological and Behavioral Science*, 47(1), 1-55. https://doi.org/10.1007/s12124-013-9230-6
- Valentine, S. E., & Shipherd, J. C. (2018). A systematic review of social stress and mental health among transgender and gender non-conforming people in the United States. *Clinical Psychology Review*, 66, 24-38. https://dx.doi.org/10.1016%2Fj.cpr.2018.03.003
- Vandello, J. A., & Bosson, J. K. (2013). Hard won and easily lost: A review and synthesis of theory and research on precarious manhood. *Psychology of Men & Masculinity*, 14(2), 101-113. <u>https://doi.org/10.1037/a0029826</u>
- van der Miesen, A. I., Raaijmakers, D., & van de Grift, T. C. (2020). "You have to wait a little longer": Transgender (mental) health at risk as a consequence of deferring gender-affirming treatments during COVID-19. *Archives of Sexual Behavior*, *49*, 1395-1399. https://doi.org/10.1007/s10508-020-01754-3
- Vassileva, J., & Conrod, P. J. (2019). Impulsivities and addictions: a multidimensional integrative framework informing assessment and interventions for substance use disorders. *Philosophical Transactions of the Royal Society B*, 374(1766), 20180137. <u>https://doi.org/10.1098/rstb.2018.0137</u>
- Vegter, V. (2013). Conceptualizing masculinity in female-to-male trans-identified individuals: A qualitative inquiry. *Canadian Journal of Counselling and Psychotherapy*, 47(1), 88-108.
- Velez, B. L., & Moradi, B. (2016). A moderated mediation test of minority stress: The role of collective action. *The Counseling Psychologist*, 44(8), 1132-1157. https://doi.org/10.1177%2F0011000016665467

- Vincent, B. (2018). Studying trans: recommendations for ethical recruitment and collaboration with transgender participants in academic research. *Psychology & Sexuality*, 9(2), 102-116. https://doi.org/10.1080/19419899.2018.1434558
- Vincent, B., Erikainen, S., & Pearce, R. (2020). *TERF Wars: Feminism and the fight for transgender futures*. Sage.
- Vogel, D. L., & Heath, P. J. (2016). Men, masculinities, and help-seeking patterns. In APA handbook of men and masculinities. (pp. 685-707). American Psychological Association.
- Vogel, D. L., Wester, S. R., Hammer, J. H., & Downing-Matibag, T. M. (2014). Referring men to seek help: The influence of gender role conflict and stigma. *Psychology of Men & Masculinity*, 15(1), 60–67. https://doi.org/10.1037/a0031761
- Wanta, J. W., Niforatos, J. D., Durbak, E., Viguera, A., & Altinay, M. (2019). Mental health diagnoses among transgender patients in the clinical setting: An all-payer electronic health record study. *Transgender health*, 4(1), 313-315. https://doi.org/10.1089/trgh.2019.0029
- Waszkiewicz, E. (2006). Getting by Gatekeepers: Transmen's Dialectical Negotiations within Psychomedical Institutions [Doctoral Thesis, Georgia State University]. https://scholarworks.gsu.edu/sociology\_theses/13
- Watt, S. O., Tskhay, K. O., & Rule, N. O. (2018). Masculine voices predict well-being in female-tomale transgender individuals. *Archives of sexual behavior*, 47(4), 963-972. https://doi.org/10.1007/s10508-017-1095-1
- Walker, P., et al. (1979). Standards of Care, the hormonal and surgical sex reassignment of gender dysphoric persons. Accessed 04/05/21 at: http://www.wpath.org
- Walker, D. F., Tokar, D. M., & Fischer, A. R. (2000). What are the eight popular masculinity- related instruments? Underlying dimensions and their relations to sociosexuality. *Psychology of Men and Masculinity*, 1, 98-108.
- Warner, N., & Murphy, M. (2021). Dialectical behaviour therapy skills training for individuals with substance use disorder: A systematic review. *Drug and Alcohol Review*. <u>https://doi.org/10.1111/dar.13362</u>
- Wester, S. R., McDonough, T. A., White, M., Vogel, D. L., & Taylor, L. (2010). Using gender role conflict theory in counseling male-to-female transgender individuals. *Journal of Counseling & Development*, 88(2), 214-219. <u>https://doi.org/10.1002/j.1556-6678.2010.tb00012.x</u>
- Wester, S. R., Christianson, H. F., Vogel, D. L., & Wei, M. (2007). Gender role conflict and psychological distress: The role of social support. *Psychology of Men & Masculinity*, 8(4), 215–224. https://doi.org/10.1037/1524-9220.8.4.215
- Whittle, S., Turner, L., Al-Alami, M., Rundall, E., & Thom, B. (2007). *Engendered penalties: Transgender and transsexual people's experiences of inequality and discrimination*. Wetherby: Communities and Local Government Publications.
- WHOQOL Group. (1997). Measuring quality of life. World Health Organization.
- WHOQOL Group. (1998). Development of the World Health Organization WHOQOL-BREF quality of life assessment. *Psychological medicine*, 28(3), 551-558. https://doi.org/10.1017/s0033291798006667
- Winter, S., Diamond, M., Green, J., Karasic, D., Reed, T., Whittle, S., & Wylie, K. (2016). Transgender people: health at the margins of society. *The Lancet*, 388(10042), 390-400. https://doi.org/10.1016/s0140-6736(16)00683-8
- Willis, P., Dobbs, C., Evans, E., Raithby, M., & Bishop, J. A. (2020). Reluctant educators and self-advocates: Older trans adults' experiences of health-care services and practitioners in seeking gender-affirming services. *Health Expectations*, 23(5), 1231-1240. <u>https://doi.org/10.1111/hex.13104</u>

- WPATH (2001). The Harry Benjamin International Gender Dysphoria Association's Standards of Care for Gender Identity Disorders, sixth version. Retrieved 08/04/21, from <u>https://www.cpath.ca/wpcontent/uploads/2009/12/WPATHsocv6.pdf</u>
- WPATH (2012). Standards of Care for the Health of Transsexual, Transgender, and Gender-Nonconforming People. Retrieved 08/04/21, from <u>https://www.wpath.org/media/cms/Documents/SOC%20v7/Standards%20of%20Care\_V7%20Full%20</u> Book\_English.pdf
- Wright, T., Nicholls, E. J., Rodger, A. J., Burns, F. M., Weatherburn, P., Pebody, R., ... & Witzel, T. C. (2021). Accessing and utilising gender-affirming healthcare in England and Wales: trans and non-binary people's accounts of navigating gender identity clinics. *BMC health services research*, 21(1), 1-11. <u>https://doi.org/10.1186/s12913-021-06661-4</u>
- Wolf, E. C., & Dew, B. J. (2012). Understanding risk factors contributing to substance use among MTF transgender persons. *Journal of LGBT Issues in Counseling*, 6(4), 237-256. https://doi.org/10.1080/15538605.2012.727743
- Wong (Eds.), *The psychology of men and masculinities* (pp. 75–103). American Psychological Association. <u>https://doi.org/10.1037/0000023-004</u>
- World Health Organization. (1996). WHOQOL-BREF: introduction, administration, scoring and generic version of the assessment: field trial version, December 1996 (WHOQOL-BREF). World Health Organization.
- Yang, L. (2015). *The relationship between male gender role and men's experience of depression*. Unpublished dissertation. Widener University.
- Yerke, A. F., & Mitchell, V. (2011). Am I man enough yet? A comparison of the body transition, self-labelling, and sexual orientation of two cohorts of female-to-male transsexuals. *International Journal of Transgenderism*, 13(2), 64-76, https://doi.org/10.1080/15532739.2011.622125
- Zack, M., Toneatto, T., & Streiner, D. L. (1998). The SCL-90 factor structure in comorbid substance abusers. *Journal of Substance abuse*, 10(1), 85-101. <u>https://doi.org/10.1016/s0899-3289(99)80143-6</u>
- Zeluf, G., Dhejne, C., Orre, C., Mannheimer, L. N., Deogan, C., Höijer, J., & Thorson, A. E. (2016). Health, disability, and quality of life among trans people in Sweden–a web-based survey. *BMC public health*, 16(1), 1-15. https://doi.org/10.1186/s12889-016-3560-5
- Zosuls, K. M., Ruble, D. N., Tamis-LeMonda, C. S., Shrout, P. E., Bornstein, M. H., & Greulich, F. K. (2009). The acquisition of gender labels in infancy: Implications for gender-typed play. *Developmental Psychology*, 45(3), 688. <u>https://doi.org/10.1037/a0014053</u>

## **Appendix A: Pilot Study SSIS for Interviews**

- Can you tell me about how you came to know that you're transgender? - How old were you?
  - How did you feel? How did you deal with that internally?
  - What actions did you take? If none, why?
- What path did you take to get to where you are today? -Successes and difficulties?
- Can you tell me about your coming out experience? - How did you feel?
  - How did your family and friends react?
  - -What action did you take? If none, why?
- After you came out what did you imagine your transition would be like? What did you hope for?
  - How does that compare to your experiences thus far?
- Do you celebrate milestones in your transition?
  -Which ones and why?
  -If not, why? Did you used to and have now stopped? If so, why?
- How would you describe your method of transition?
  -What were your aims and goals?
  -How did you achieve them?
  -Have your aims and goals/route of transition changed over time?
  -Current aims/goals?
- What role would you say the GIC has played in your transition? -Positive or negative? Why?
- Where do you see yourself in 1, 5, 10 years?
- Is there anything you discovered through transitioning that you wish you had known about sooner?
- What advice would you give to someone just beginning their transition? -What do you wish you had known?

## **Appendix B: Pilot Study SSIS for Focus Group**

- Does anyone here celebrate milestones in their transition?
   How?
  - Why is it significant/ what made you want to do that?
- Has transitioning affected how you view your own identity (your role in society, in your family, who you are as a person)
  - How? Positive/negative?
  - How has it affected everyday life and how you interact with other people?
- What role has medical care played in your transition?
  - NHS/Private
  - Was it different to how you expected? What did you expect? What did you find?
- In what ways could healthcare provisions for trans men be improved?
  - Fertility treatment was flagged up in interviews
  - Communication between GIC and individual was flagged up in interviews
  - Is there anything you wouldn't change about healthcare services?
  - What other services would be useful during transition?
- How would you describe the various stages of transitioning in general? - Would it be linear?
  - Can it be conceptualised in this way?
  - [specifically] How to ask participants about where they are in their transition?
- Do you have any ethical concerns regarding studies into trans men?
  - What would you like researchers to know?
  - How would you like to be involved in research?
  - Do you think focus groups like this are useful? Do you think they ought to be done every time?

# **Appendix C: TESUP-1 SSIS for Interviews**

- Can you tell me a bit about yourself?
- Can you tell me about your transition?
  How did you get to the point you're at today?
  Triumphs, struggles?
- What drugs/drinks do you usually have? (AND how much in 1 session & per week)
- In what situations do you use drugs/alcohol?
  Who do you use drugs/alcohol with?
  Do you ever do it alone? In what situations?
- Are there any events or situations that trigger the urge to have alcohol or drugs? - Celebrations, stressful life events, transphobia, socialising?
- How did you first start drinking/using drugs?
  Have your habits around drinking/drugs changed over time?
- How does drinking/using drugs make you feel?
  Does this differ from when you're sober? How?
- Why do you think you drink/use drugs?
  - What are the good aspects
  - bad aspects of your alcohol/drug use?
- Is there anything that makes you drink/use drugs more or less?
  - Why do you think this is?
  - How do you cope if you have the urge to drink/use drugs but can't?
- Did transitioning change how you use drugs/alcohol?
  Did it change your attitude to drugs/alcohol?
  Has your drug/alcohol use affected your transition in any way?
- Do you think masculinity is related to your attitudes/behaviours with drugs/alcohol?
- How do you think drinking/using drugs has an impact on your physical health -or mental health? positive and negative
- Do you think your alcohol/drug use is manageable?
  If you felt you needed to would you seek help for your alcohol/drug use?
  Do you know where to go if you want support?

# **Appendix D: TESUP-2 SSIS for Interviews**

- Can you tell me about what may have changed in your life since we last spoke one year ago?
- Has your transition journey changed in any way over the past year?
  - Identity (pride in identity? what makes you proud/why)
  - Goals & Progress (first appointment, hormones, surgery)
- Can you describe your health and wellbeing over the past year? - physical health
  - mental health
  - social life
  - transition progress and support in transition impact on wellbeing
  - if negative experiences discuss coping strategies and social support
- Has the amount or frequency of alcohol or drug use changed in the past year? - what elicited this change
  - can you describe your feelings or thoughts regarding this change
- Do you think your substance use was affected in anyway by the COVID-19 pandemic or lockdown measures?
  - Follow up on why (either way)
  - Focus on transition, mental health, socialising/isolation (if relevant)
- Can you describe to me a time in the past year where you have taken drugs or drank alcohol?
  - how did you feel about yourself
  - what were your thoughts during this period of substance use
  - what was happening in your life at the time
  - how did you deal with what was happening in your life at the time
- What do you think are your typical reasons for using drugs or drinking alcohol
  - can you describe what happens when this reason occurs
  - have there ever been other reasons for drinking or using drugs
  - have you ever used substances to enhance a good mood or regulate negative emotions? can you give me an example of this?
- Do you consider yourself to be an impulsive person?

- (yes) Can you describe a time where you acted impulsively – consequences, feelings about self – does this impact your substance use?

- (no) do you conder yourself to be a cautious person – what factors do you consider when making a decision – does this impact your substance use?

- Do you feel like you have ever experienced discrimination or victimisation due to your transgender identity?
  - can you describe a time when this happened?
  - how did you respond in the moment and afterwards
  - can you describe how you felt when this happened (generally and about yourself)
- Is there anything you'd like to add that we haven't discussed? Do you have any questions or concerns?

#### Individual topics to cover.

Aaron: frequent urges to drink (due to stress)

Bailey: use of painkillers (sometimes with alcohol) to manage pain or distress

**Cody:** substance use to lower inhibitions and have difficult conversations with friends (dysphoria/mental health)

**Darren:** struggles with trans identity/social isolation – harmful coping strategies (self-harm and alcohol use)

Ezra: process of decision-making in substance use (RISK vs. reward)

Gale: process of balancing risk and safety in substance use (impact of peers)

Jeremy: worries about progress of lower surgery (impact on substance use)

**Logan:** difficulties self-regulating alcohol consumption (binge-drinking)

**Owen:** substance use to lower inhibitions, have difficult conversations with people (if appropriate ask about alcohol use to reduce dysphoria and be intimate with girlfriend)

**Parker:** substance use to regulate emotions (triggers: arguing with girlfriend, boredom, poor mental health)

Rowan: coping mechanisms for distress (alcohol use, disordered eating, dysphoria)

Simon: Experience of substance use reduction services (progress on reducing cocaine use)

**Zack:** influence of mental health on substance use and coping mechanisms (self-harm, substance use as self-harm)

#### **Appendix E: TESUP-1 Stage of Transition Comments**

#### R\_001

I feel like I'm post-transition because I've discharged myself from the GIC, and I'm happy with my body in its current form. I haven't had bottom surgery so I guess some people would say I'm not post-transition, but I don't want it.

#### R\_004

I haven't had a GIC appointment yet. I've been waiting 2.5 years. I can't get a bridging prescription from my GP despite mental health issues. However, I feel that I am early in my trans journey as I am very self aware. I have at least changed my name (although not by deed poll yet). I know who I am, just I'm not far on the journey to be that guy.

#### R\_005

I have been out socially for 3 years and have started T, however as I have more changes to come and still am waiting for top surgery and hysto I'm only considering myself as mid transition

#### **R\_006**

I have completed all of the aspects of transition that I am certain I want to go through. I have socially transitioned, I am on testosterone medication, and have had chest masculinisation surgery. I am still considering other surgeries, but for now I am finished with my transition.

#### R\_007

I don't think I'll ever be post-transition, but I'm not planning on having any further surgeries or interventions (other than restarting testosterone as I've currently paused to try and conceive)

#### **R\_008**

I perceive transition to start from the beginning of social coming out, and the end is highly personal for each not cis people to decide

#### R\_011

I have been on hormones for 6 years, had top surgery 4 years ago, and will not be having bottom surgery. I pass 100% of the time, and do not often think of being trans any more unless it's prompted. I would consider this transitional period of my life to be over, and I am now in the process of acclimatising as a man in society.

#### R\_012

I have been on testosterone for 2 years and am looking into top surgery (which looks like it's going to take a while, thanks NHS..) and eventually probably bottom surgery. The longer my transition goes on, the more I want phalloplasty despite the risks it seems, but I guess we'll see.

#### R\_013

I identify as late transition as it has been a few years already since I began transitioning (5), I have been taking hormones for 4 years now and have been living comfortably (rarely misgendered if at all) for the last couple of years. I have also had top surgery and my body has been significantly masculinised by the hormones. I am not considering lower surgery for the time being. I don't think, personally, that there is such as stage as 'post transition' for me.

#### R\_015

No option to suit me, feel I am myself but not "doing" anything actively

#### **R\_016**

I feel like my transition is mostly over because socially it's something that isn't really relevant any more as people largely don't know I'm trans, despite the fact that I am still waiting to undergo chest surgery which for me is a necessary step in my transition. It almost feels like the large part of transition for me is more about coming out, mentally sorting through my perception and feelings about my gender and how I'm seen by others, and allowing me to be more authentic in how I present myself. However, HRT has been a large part of the process in my opinion, and being on testosterone for over 5 years kind of makes me feel that my transition is kind of over, despite ongoing changes.

#### R\_017

On hormones and blockers, no surgeries (on waiting list)

#### R\_018

I've had top surgery and hormones, but am holding off lower surgery until after I qualify as a nurse

#### R\_019

I disagree with having to put myself on the scale. I've never specifically worked towards transitioning as a thing so its kinda impossible to place where I am. All changes towards transitioning have been as/when seems appropriate rather than as a part of some big effort to transition

#### R\_021

Fully socially transitioned for 8 months, HRT for 2 weeks.

I don't really like this scale. Not everyone transitions in anyway so there's no space for them. Transition is multifaceted and non linear so such simple categorisation seems inept to capture reality of experiences.

#### R\_023

I have presented as masculine since puberty and before ... I have undergone transition to further confirm that presentation .. however having never felt comfortable identifying as female ... I am equally not comfortable as being seen as male .. and have little in common with either ... it's been a journey to get where I am ... for the ease of others mostly, I use male pronouns ... gender neutral would be a preference .. but I am tired of the battle of identity ... and for work reasons ... it's ok for now ...my identity is fluid and evolving ... and who knows where it will take me next

#### R\_024

I feel at the point in my transition where I am comfortable. I do not wish to go ahead with any further medical intervention, nor alter any social aspects of my presentation.

#### R\_025

I am completely socially transitioned, on testosterone and happy! I hope to have surgeries in the future so I do not consider myself post transition.

#### R\_026

I've put early transition, because although I began socially transitioning in 2015, I haven't started any medical transition and I only just had my deed poll for name change.

#### R\_027

Out to all my friends, they refer to me as he/him and by my name, out to my 2 sisters who are trying. Have not yet told Mother. Seen GP to be referred to a GIC

#### R\_028

I don't really know muh about trans stuff so it's just a guess I suppose. I've transitioned socially and am in the process of legally changed my name. I have not seeked any professional help with my issues (medical phobia) so I've been dealing with it on my own.

#### R\_030

Nearing the end of phalloplasty

I consider myself to be Very early in transition. I live full time as male, I have changed my name legally, and everyone in my life calls me the right pronouns. But Im not on testosterone yet. And I do not have a supportive family, as they are the only people to not respect my pronouns.

#### R\_033

i would say i am both late and post transition. i feel like post transition was a mindset I reached a few months after top surgery. i started to fully feel confident in myself and my gender. however, i am still awaiting phalloplasty (with a couple of months to go until stage 1) and i think i may feel that i am "more" post transition in a few years when that is out of the way - but as most trans men stop pursuing further medical interventions post top surgery, and i have now been fully socially transitioned for six years, including the majority of my adult life, and essentially in all aspects of my life i feel i am seen as indisputably male, i feel that post transition is a label i resonate with for now

#### R\_034

- I have socially transitioned (although family still use birth name & she/her pronouns) - I am currently in the process of legally changing my name and getting blood work done for hrt

#### R\_035

I consider myself mid-transition as I am on testosterone but have yet to have any surgery

#### R\_038

I've been socially transitioned for several years, have been on testosterone for  $\sim 10$  months, and have my top surgery consultation scheduled for later this year, hopefully to get it performed around Easter. Once this is performed, I'd consider myself late transition. I also plan to have bottom surgery with the NHS, but due to the waiting lists, I expect this won't be for another 4 years or so, and then another year or two for all the stages of the surgery + medical tattooing to be complete. At that point I would consider myself fully transitioned.

#### R\_039

Not sure if I will ever feel I have "finished" my transition, or if I do, it will take many years. 4 years in (from first coming out) and 3 years on from starting hormones I still like I'm slowing becoming more and more me.

#### **R\_40**

Come out to family and on waiting list for GIC but it's a very... VERY long wait.

I feel I am in mid transition although I have only been able to medically transition at a later age.

#### R\_042

This is an interesting and difficult to answer question; I don't think I will ever know how far along my transition I am because I am alsoway transitioning yet simultaneously not? I chose mid as I have started T but have not had any surgeries I desire.

#### R\_043

I completed transition (as much as I am able/willing to do) in 2014. I am on testosterone, I've had top surgery but declined bottom surgery due to my age and unrelated medical condition. I fully and 100% pass as male.

#### R\_044

Nearly 3 years on Testosterone but still to get top surgery, having top surgery will be my last step in my transition. Socially transition was my starting point (coming out and changing my name)

#### R\_045

I'm on my way to getting Testosterone but currently haven't started physically transitioning, but I've been out for 4 years

#### R\_046

Socially fully transitioned in terms of name change, informed folk of my pronouns, dress how I feel matches my gender etc. Medically part transitioned? 8 months on T. Considering top and bottom surgery in the future.

#### R\_047

I don't have any plans to transition right now

#### R\_048

Personally I'd say Im 100% socially transitioned, but not medically or legally transitioned at all. While I feel somewhat far along as I was non binary for years but in reality I've been out to myself and others as a trans man for less than a year.

#### R\_049

2 months on HRT, socially transitioned at 18/19

Socially transitioning not yet medically transitioning

#### R\_052

I would say I am mid transition as I have been on testosterone for over a year and am planning on having surgery in January. After surgery I would consider myself late transition, and after bottom; Post-transition.

#### R\_053

I put pre-transition as I am not on hormones and haven't been to see my GP for a referral yet. Though I do bind and pack and present as male socially.

#### R\_055

I'm having top surgery within the next year which is what o need in order to feel comfortable with my chest

#### R\_056

On hrt for 2 years, on a waiting list for top surgery

#### R\_058

Been on testo for about two years, getting top surgery soon. No plans for bottom surgery yet.

#### R\_059

I've had top surgery, I'm on testosterone. I do want bottom surgery but not right now

#### R\_060

I consider myself late transition bc I only have two more stages of lower surgery left and this is the end of my transition in my eyes

#### R\_061

I haven't begun my transition yet as I still need to discuss referral with my GP.

I define my transition as early based on the goals I personally have and how long it will take for those to be accomplished. I imagine that differs for everyone.

#### R\_063

I've fully socially transitioned, legally changed my name and gotten most of the id I have changed to reflect my gender. I haven't yet started my medical transition, but I think the rest puts me pretty far along the path of what I would consider "fully transitioned" for myself.

#### **R\_064**

Waiting for top surgery, prescribed testosterone but currently not taking. Transitions began at coming out, transition end after top surgery

#### R\_065

This one is difficult to quantify since what treatments I want and/or need has evolved over time. I am at a point now where I still have treatments yet to go however my dysphoria is diminished to a point of being barely an issue.

#### R\_066

There was not an option to suit me.

#### R\_067

I am currently taking testosterone under the guidance of a private GIC and am on the waiting list for an NHS GIC for top and lower surgery, and have been for two years so far. I would describe myself as mid transition as I have passed one of the major milestones I had knew I wanted to pass for my transition but have not reached a point where I no longer want to persue procedures, medications or treatments.

#### **R\_069**

I define the start of my transition as when I realised I was trans and started playing with it at 15, 16. I didn't come out to my family until I was 20 or so. I thought of myself pre-transition for years as I am not medically transitioned yet, but have come to think of it as how comfortable I am with myself and how I present.

#### **R\_070**

I have been on hormones replacement for just under a year and a half. I plan to eventually have various surgeries (mastectomy, phalloplasty) if I can afford them privately

I think of myself as mid transition because i have socially transitioned but not medically

#### **R\_080**

I am late in my transition, I am only waiting for one more surgery and then I will consider myself post-transition

#### R\_086

I consider myself post transition because people always see me as male and i do not want any more surgeries but i will always be on testosterone

#### **R\_088**

I wouldn't say I agree with a linear timeline of transition but if i have to put something i would say late transition because although i dont feel like it will ever be finsihed developing i have been trans for a long time now (6 years)

#### **R\_089**

I'm fucking done with everything!!!

#### R\_091

I would say that I have fully transitioned socially, but have not yet started medically transitioning due to long waiting lists at the gender clinic. I would also say that my transition started before I came out, when I was "coming out" to myself, and learning to accept myself as trans, even still in the closet. I was surprised that it took me a lot longer to be comfortable with being visibly and openly trans than it did for other people to accept me, as people around me were pretty supportive, whereas I was very nervous about being seen as trans by other people, and put a lot of energy into passing as cis male when I first came out.

#### R\_094

I wouldn't ever say that I would be post transition because I think my gender identity is always transitioning into something more comfortable to myself and something i gain understanding of as time passes

#### R\_098

I consider the start of my transition to be the day I started social transition (coming out in school + work, changing my name, and living as male). I have now had all the medical procedures I feel I need,

and changed (almost) all things legally. I don't call myself post-transition because a) I'm not quite yet sure if I will need more surgeries in the future - I don't know how my feelings on it will change - and b) there are still some lingering legal documents to change.

#### **R\_099**

I placed myself mid-transition since I have a fairly medical view of my own transition. I'd classify my coming out as the start of my transition, and now that I'm on HRT I feel I have made good progress towards living my own life as myself, and thus think mid-transition is an acceptable place. I'd classify getting top/bottom surgery or hysto as moving into my late transition, as well as being on HRT for a significant amount of time (3+ years) I am done with my transition when I have reduced my dysphoria to manageable background levels

#### R\_101

If you are thinking of this in the medicalised way that the gender clinic do then I would say late transition because I only have one more surgery, however I do not identify with this way of conceptualising transition, it is much more flexible and personal than any cis practicioner could understand

#### **R\_104**

I wouldn't normally think of transition on a scale but this fits the best

# Appendix F: SCL-90 Comparison Table

	Phase 1 (present study; Mills, 2022)	Phase 2 (present study; Mills, 2022)	Spanish FTM Sample (Castelo- Branco et al., 2021)	German FTM Sample (Auer et al., 2013)	US Outpatient Comorbid SUD sample (Zack et al., 1998)	British cis male community sample (Francis et al., 1990)
Age $(M)$	25	26	27.01	32.38	37.1	44
Total ( <i>N</i> )	105	82	76	32	740	150
	M (SD)	M (SD)	M (SD)	M(SD)	M (SD)	M (SD)
SOM	1.26 (.82)	1.39 (.89)	.69 (.66)	.56 (.50)	1.18 (.76)	.43 (.57)
OBC	1.83 (.91)	2 (.93)	.91 (.84)	.58 (.56)	1.67 (.86)	.59 (.63)
INS	1.65 (.91)	1.73 (.96)	.84 (.72)	.59 (.77)	1.57 (.89)	.58 (.72)
DEP	1.77 (.88)	1.80 (.91)	1.14 (.92)	.45 (.55)	2.13 (.89)	.42 (.65)
ANX	1.45 (.93)	1.58 (.93)	.75 (.75)	.36 (.51)	1.68 (.90)	.45 (.60)
HOS	1.21 (.98)	1.33 (1)	.65 (.78)	.46 (.66)	1.17 (.93)	.44 (60)
PHO	1.21 (1.03)	1.52 (.99)	.50 (.67)	.19 (.42)	1.00 (.94)	.24 (.50)
PAR	1.18 (.86)	1.26 (.95)	.89 (.85)	.49 (.66)	1.41 (.93)	.54 (.65)
PSY	1.17 (.85)	1.30 (.98)	.66 (.64)	.25 (.35)	1.19 (.78)	.27 (.48)
GSI	1.45 (.80)	1.56 (.84)	.83 (.69)	.45 (.48)		.44 (.47)
PST	59.1 (25.23)	60.41 (26.61)				14.46 (10.43)
PSDI	2.07 (.54)	2.16 (.52)				1.38 (.56)

**Table F.1.** SCL-90 Comparison table between TESUP-1 and -2 and four different samples from studies using the SCL-90 to measure psychopathological symptoms.

# Appendix G: WHOQOL-BREF Comparison Table

					I IIZ	1.117
					UK	UK
	Phase 1	Phase 2	AUS	Spanish	'psychiatric	<i>'lifestyle</i>
	(present	(present	general	transgender	conditions'	conditions'
	study:	study:	population	sample	sample	sample
	Mills	Mills	(Hawthorne	(Gomez-Gil	(Skevington	(Skevington
	2022)	2022)				
	2022)	2022)	et al., 2006)	et al., 2013)	& McCrate,	& McCrate,
					2012)	2012)
Age (M)	25	26	48.2	31.2	44.5	44.5
Total (N)	105	82	866	193	77	121
	M (SD)	M(SD)	M (SD)	M (SD)	M (SD)	M(SD)
Dhygiaal Haalth	60.13	57.88	73.5	63.51	54.57	54.06
Fliysical Health	(18.27)	(16.38)	(18.1)	(17.79)	(20.62)	(20.20)
Developical Health	47.65	48.01	70.6	56.09	45.93	54.15
Psychological Health	(18.70)	(17.11)	(14.0)	(16.27)	(25.99)	(16.76)
Social and	58.73	60.01	71.5	60.35	61.91	56.71
Relationships	(21.72)	(19.51)	(18.2)	(21.88)	(20.80)	(23.95)
Environment	58.45	59.64	75.1	55.44	61.00	60.45
Environment	(16.59)	(15.46)	(13.0)	(27.18)	(17.02)	(17.12)

**Table G.1.** WHQOL-BREF Comparison table between TESUP-1 and -2 and four different samples from studies using the WHOQOL-BREF to measure quality of life.

## **Appendix H: TESUP-1 Correlations Additional Information**

### **1. Demographics**

For correlations with demographic data Pearson (r) was used, or Spearman (p) for nonparametric data. Point bi-serial correlation was used for the gender variable which was measured as either binary or non-binary.

Figure H.1. Scatterplots of tobacco, AUDIT, DUDIT, age, stage of transition and level of education



# 2. Quality of Life

# **Figure H.2.** Scatterplots of tobacco, AUDIT, DUDIT, and the four domains of the WHOQOL-BREF measuring quality of life

	DOM1	DOM2	DOM3	DOM4	TobaccoTotal	AUDIT score	DUDIT score
DOM1		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0				° ∞&o 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
DOM2					0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	◦ ◦ ◦ ◦ ◦ ◦ ◦ ◦ ◦ ◦ ◦ ◦ ◦ ◦	
DOMB		0 0 0 0 0000 0 0 0000 0 0 0000 0 0 0 000000 0 0 0 00000 0 0 0 00 0 0 0 00 0		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		000 0000000 0000000 00000000 000000000	
DOM					0 00 0 0 000000 0 0 0 0 0 0 0 0 0 0 0		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
TobaccoTotal	0         000000000000000000000000000000000000	00000000000000000000000000000000000000		coccocce     cocccce     cocccce     cocccce     cocccce     cocccce     cocccce     cocccce     cocccce     coccce     coccce     coccce     coccce     coccce     coccce     cocce     coccee     cocee     cocee     cocee     cocee     cocee     cocee     cocee     cocee     coceee     coceee     coceee     coceee     coceee     coceee		0000000000 0 0 0 0 0 0 000 000000 0 0 000 000000	D 00 0 0 0 0 00 D000 00 D 00 0 00000 0 0 D 00 0 00000 0 0 D 0 0 0 0
AUDIT score		° °		۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵	0 000 000 00 00 00 00 00 00 00 00 00 00		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
DUDIT score	0 00 0 0 0 0 0 0000000000 0 000000000		○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○	0 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 0	

# 3. GRCS

<b>Figure H.3</b>	<b>3.</b> Scatterplots of	of tobacco, AUD	IT. DUDIT.	and the four	domains of th	ne GRCS
0	1	,	, , ,	/		

_	SPC	RE	RABBM	CBWFR	TobaccoTotal	AUDIT score	DUDIT score
SPC					0         0		SPC
æ		. hadha.			0 00000000 0 0 00000000 0 0 000000000 0 0 00000000		<b>RE</b> 0 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
RABBM		ိုင်ငံ ကို ကို ကို ကို ကို ကို ကို ကို ကို ကို		° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	0 00 00 00 00 0 0 00 00 000 0 00 0 000 0 00 0		ADBW
CBWFR							CDWFR
TobaccoTotal		000000 0 00 00000 0 00 0 0 0 0 0 0 0 0	00000000000000000000000000000000000000	00 0 000000000000000000000000000000000		0000000000 0 00 00 0 000 000 00 0 0 0000 000000	Tobucco 0 00 00 00 00000 00 0 00 0 0000 0 0 0 0 0 0 0000 0 0 0 0 0 0 0 0 0 0 0
AUDIT score	۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵	۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵	္ စိုင္လိမ္းစိုင္ရ စိုင္လိမ္းစိုင္ စိုင္လမ္းစိုင္ရ စိုင္လမ္းစိုင္ စိုင္လမ်းစိုင္ စိုင္လင္ စိုင္လ္က စိုင္လမ္းစိုင္ စိုင္လမ္းစိုင္ စိုင္လမ္းစိုင္ စိုင္လမ္ကေ စိုင္လင္ စိုင္လ္က စိုင္လမ္းစိုင္ စိုင္လ္က စိုင္လမ္းစိုင္ စိုင္ စိုင္လမ္းစိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိုင္ စိေ စိုင္ စိုင္ စိေ စိုင္ စိေ စိုင္ စိေ စိုင္ စိေ စိုင္ စိေ စိုင္ စိေ စိုင္ စိေ စိုင္ စိေ စိေ စိုင္ စိေ စိုင္ စိေ စိေ စိုင္ စိေ စိုင္ စိုင္ စိေ စိေ စိုင္ စိေ စိုင္ စိုင္ စိေ စိေ စိုင္ စိေ စိုင္ စိုင္ စိေ စိေ စိုင္ စိေ စိုင္ စိုင္ စိေ စိေ စိာက စိေ စိာ စက စိုင္ စိေ စိာ စိာ စက စိုင္ စက စိုက စိာ စိာ စက စိာ စက စိာ စကေ စိာ စက စိ	۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵	0 000 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		AUDIT score
DUDIT score			° ,00 ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °			00000000000000000000000000000000000000	

# 4. GMSR

Figure H.4. Scatterplots of tobacco, AUDIT, DUDIT, and the nine domains of the GMSR

Г	D_sum	R_sum	V_1um	NA_sum	17_sum	P_sum	NE_sum	ND_sum	00_sum	TebaccoTotal	AUDIT score	DUDIT score
0,um			0 03 0 0 03 0 0 03 0 0 0000 0 0 0000 0 0 0000 0 0 0000 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0	0 20 0 0 20 0 0 5 0 0 5 07 07 0 5 0000000 00 0 6 00 00 0 0 6 00 0 0 6 00 0 0 00 0	0         0         0           0         0         0           0         0         0           0         0         0           0         0         0           0         0         0           0         0         0           0         0         0           0         0         0           0         0         0           0         0         0           0         0         0           0         0         0           0         0         0           0         0         0	9 9 9000 0000 0000 00000 00000 0000 00	0 0
Rem			0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0				0 0 0 0 0 0 0 0 0 0 0 0 0 0		0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
V,um				0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 6 6 6 80 6 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7		۵ ۵۵ ۵ می ۵ می ۵ می ۵ می ۵ می ۵ می ۵ می		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
NL.un			0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0							0 00 0 00 00 00 0 0 0 0 0 0 0 0 0 0 0		
flow									3 0 0 0 3 0 0 0 3 0 0 5 0 0 0 0 0 0 0 0 0 0 0 0 0		င္ ္ ္ ္ ္ ္ ္ ္ ္ ေ ေ က ေနာ္က အေ က ေနာ္က အေ က ေနာ္က အ က ေ က ေ က ေ က ေ က ေ က ေ က ေ က ေ က ေ က	a con a contract of the contra
P_m						HI L						
NE,un												o contraction of the second se
m, 01										00         0000000         0000000         0000000           00000000         00         0         0           0000000         00         0         0		
00 <sup>°</sup> aun			0 0 0 0 0 0 0 0 0 0 0 0 0 0									
Tebacco Tetal			0000033 0 0000 50 0 0 0000 50 0 0 0000 50 0	0.910 0000000000000000000000000000000000		00 00 000000000 200 00 000 0 0 000 0000000 0 0 000000 0 0 00000 0 0 00000	00000 0 00 0000 0 000 0 0 0 0 0000 0 000 0 0 0 0 0	4 02 0 20 0 0 0 0 0000000000 0 0 0 0000000000	0 0 00 0 0 0 0 0 0 0 0 00000 00000 00 000000		0000000000 0 00 0 0 0 0 0 0 0 0 0 0 0	0         0         0         0         0         0           0         0         0         0         0         0         0           0         0         0         0         0         0         0         0
AUDIT score		0         0	မ မ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ	ိ မမိ ဂုဂ်င်္သာ မန်းမှ ဂုဂ်င်္သာ မန်းမှ စစ်သို့ မန်းမှ စစ်သို့ မန်းမှ စစ်သို့ မန်းမှ စစ်သို့		ت	မ မန္က ေျပာရာန္က ေ ေျပာရာန္က ေ ေ ေရာက္ ေ ေ ေ ေ ေ ေ ေ ေ ေ ေ ေ ေ		68 6 6 6 6 6 6 6 7 6 6 7 6 7 6 7 6 7 6 7 7 6 7 7 7 7	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		
DUDIT score			0.5 3 5 7.5 9 0.5 0 0.5 0 0.5 0 0.5 0 0.5 0 0.5 0 0.5 0 0.5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵	0 0			0 00 0	2 0 5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0         0         0         0         0           0	6 9 9 0 9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	

# 5. Psychopathological Symptoms

Figure H.5. Scatterplots of tobacco, AUDIT, DUDIT, and the nine symptom domains of t	the
SCL-90-R	

	SOM	OBC	INS	DEP	ANX	HOS	PHO	PAR	PSY	i elsacco i otal	AUDI I score	DUDIT sears
wos							2 2 2 2 2 2 2 2 2 2 2 2 2 2			0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		
OBIC			0 0 0 0 0 0 0 0 0 0 0 0 0 0									
91					2 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9					0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 0 0 0 0 0 0 0 0 0 0 0 0 0
8			0 0				а с с с с с с с с с с с с с с с с с с с					
MAC		ိ မ ရန္က နိုင္ငံ မ ရန္က နိုင္ရ မ ရန္က နိုင္ရ မ ရန္က က က ရန္က က ရန္က က က ရန္က က ရန္က က ရန္က က က ရန္က က ရန္က က က ရန္က က က က က က က က က က က က က က က က က က က										
80H										000 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		
OHE	0 0 8 2 2 2 2 2 2 2 2 0 2 2 0 2 2 2 2 2 0 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2			0 0 0 0 0 0 0 0 0 0 0 0 0 0					0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
bitd	6 6 6 6 6 6 6 6 6 6 7 6 6 7 6 7 6 7 6 7				6 6 8 9 6 6 6 6 6 6 6 6 6 6 8 8 8 8 8 8						0 0 0 0 0 0 0 0 0 0 0 0 0 0	
ASI						0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0				0 00 0 000 0 0000000000000000000000000		
TebacoTetal	00990000 0900 00000 00000 000 000 0000 00000 000	00 11 00         00000000         0         0         0         0           0 2 0         0 0         0         0         0         0         0         0           0         0 0         0         0         0         0         0         0         0         0           0         0 0         0 <th>00000000000000000000000000000000000000</th> <th>0000 00 00 000000 0 000000 0 0 0 0 00000 0 0 0 0 0 000000</th> <th>00000000000000000000000000000000000000</th> <th>0 000000 0 0000000 0 0 00000 00 000 0 0 00 0</th> <th>9 6 10 10 1 1 0 0000 0 50 000 000 0 3 0 90 000 0 0 0 0 0 500 0 0 0 0 0 500 0 0 0 500 0 0 500 0 0 500 0 500 0 50 0 50</th> <th>CO CO CO C</th> <th>0000000000 00 0 0 00000 0 0000 00 0 0 0 00000 0 0 00 0</th> <th></th> <th>00000000000 0 0 00 0 0     000 0 0     000 0 0 0     000 0 0 0     000 0 0     000 0     000 0     000 0     000 0     000 0     000 0     000 0     000 0     000 0     000 0</th> <th>0         0         0         0         0         0           0         0         5         0</th>	00000000000000000000000000000000000000	0000 00 00 000000 0 000000 0 0 0 0 00000 0 0 0 0 0 000000	00000000000000000000000000000000000000	0 000000 0 0000000 0 0 00000 00 000 0 0 00 0	9 6 10 10 1 1 0 0000 0 50 000 000 0 3 0 90 000 0 0 0 0 0 500 0 0 0 0 0 500 0 0 0 500 0 0 500 0 0 500 0 500 0 50	CO CO CO C	0000000000 00 0 0 00000 0 0000 00 0 0 0 00000 0 0 00 0		00000000000 0 0 00 0 0     000 0 0     000 0 0 0     000 0 0 0     000 0 0     000 0     000 0     000 0     000 0     000 0     000 0     000 0     000 0     000 0     000 0	0         0         0         0         0         0           0         0         5         0
AUDIT scene	ေ လာ ကို ရှိမ်း ကို ရှိမ်း ကို ရှိမ်း ကို ရှိမ်း ကို ရှိမ်း ကို ရှိမ်း ကို ရှိမ်း ကို ရှိမ်း ကို ရောက် ကို ကို ကို ကို ကို ကို ကို ကို ကို ကို ကို			ာ မိုးနှိုင် နိုင်ငံသည် နိုင်သည် နိ နိုင်သည် နိ နိ နိ နိ နိ နိ					ာ ေရွိသို့ သိုင် ရွှေရှိသို့ သိုင် ရွှေလို ကြီးလာရှိ ရွက်ကောက် ရွှေလို ကြီးလာရှိ ကြီးကောက် ရွှေကြီးကြီးလာရှိ	1000 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		
DUDIT score	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0				000 00 00 00 00 00 00 00 00 00 00 00 00	ου ο ο Ο ο ο ο ο			0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		

	GSI	PST	PSDI	TobaccoTotal	AUDIT score	DUDIT score
GSI		۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		° 8 0 8 0 8 0 8 0 9 0 9 0 9 0 9 0 9 0 9 0 9 0 9
PST	2000 000 000 000 000 000 000 000 000 00					8000 <sup>000</sup> 0 0 8000 <sup>000</sup> 0 8000 <sup>0000</sup> 0 8000 <sup>000</sup> 0 8000 <sup></sup>
IDSA						° ° ° ° ° ° ° ° ° ° ° ° ° °
TobaccoTotal	© ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞ ∞	∞∞∞∞ ∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞∞ ∞ ∞∞ ∞	000000 0000 00000 0000 000000 0000 000000		coccocccccc         cocc         cocccccccc         cocccccccc         coccccccccc         coccccccccc         cocccccccccc         cocccccccccc         cocccccccccc         cocccccccccccccccc         coccccccccccccccccccccccccccccccccccc	• • • • • • • • • • • • • • • • • • •
AUDIT score	٥ ٥ ٥ ٥ ٥ ٥ ٥	ိ ၀ိုလို လိုလို လိုက္ရ ကေလ့ ဂိုလိုလို ဂိုလို ကေလ့ ဂိုလိုလိုလို ဂိုက္ရ ကေလ့ ဂိုလိုက္ရပ္ပဲ ကေလ့ ဂိုလိုက္ရပ္ပဲ ကေလ့ ဂိုလ္ရွိလို ဂိုက္ရဲ ကေလ့ ဂိုလိုက္ရပ္ပဲ ကေလ့ ဂိုလို ဂိုက္ရဲ ကေလ့ ဂိုလို ဂိုက္ရဲ ကိုလို ဂိုလို ဂိုလို ကိုလို ကိုလိုက္ရဲ ကိုလို ကိုလိုက္ရဲ့ ကိုလိုက္ရဲ့		0 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵
DUDIT score	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	DUDIT score

# **Figure H.6.** Scatterplots of tobacco, AUDIT, DUDIT, and the three global measures of the SCL-90-R

# 6. Personality Traits

Figure H.7. Scatterplots of tobacco, AUDIT, DUDIT, and the seven dimensions of the TCI-
140

,	NSR	HAR	RDR	PSR	SDR	COR	STR	TebaccoTotal	AUDIT score	DUDIT score
NSR										9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9
NH.		<u> </u>						00000000000000000000000000000000000000		
ROR								0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		Abb Abb Abb Abb Abb Abb Abb Abb
PSR								0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	80 ° 342 S 342 S 345	M4 M4 M4 M4 M4 M4 M4 M4 M4 M4
SDR								0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		A conservation of the cons
COR								0 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 0 0 0 0 0 0 0 0 0 0 0 0 0
STR								0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
TobaccoTotal	0 9:0 00 0 0 0 0	0 80 8880 80 80 80 80 80 00 00 0 0 0000 80 60 0 80 0 0 0 0 00008880 80 00 80 80000 0 0 80 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	00 0 0000 000000 00 000 6 0000 0 00 0 000 00	00 0 00 000 000 00 0 0 0 0 0 0 0 0 0 0	0 0000000 000000 000000 000 0 0 0 00000 00000 0 0 000000	0 00 0000 00000000 00 00 0 0 000 0000 0 0 0 000 00		00000000000 0 00 0 0 000 000 0 0 0 000 00	Tagent
AUDIT scere		0 9 9 9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		ိန္လ 8.0 ိုင်ကို ရှိနိုင်ရ ၀.၀ ရှိနိုင်ရန် ၀.၀ ရှိနိုင်ရန် ၀.၀ ရှိနိုင်ရန်				0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		
DUDIT score			00000 00000 00000 00000 00000 000000 0000		00000 00000000000000000000000000000000		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	

#### **Appendix I: TESUP-1 AUDIT Multiple Regression**

#### **Additional Information**

Prior to conducting a multiple regression analysis, the assumptions of the test were checked. All variables had a linear relationship with the dependent variable (AUDIT). Data were checked for outliers or missing data, and 13 cases were excluded, bringing the total participants to 92. According to Tabachnick and Fidell (2013), a sample of 92 would be acceptable using their formula of N > 50 + 8m, where N is number of participants and m is number of independent variables.

A forward method was chosen for the multiple regression analysis. This method begins with a model that includes no variables, then it adds variables one at a time according to which is the best fit for the model and stops adding them when the model stops improving. Here, the model stopped adding variables when the *p* values of the remaining variables were all over .05, and therefore would not provide a significant improvement to the model.

Model	Variable	В	SE B	β	t	р	VIF
1	(Constant)	19.504	3.463		5.633	.000**	
	CO	172	.048	350	-3.544	.001**	1
2	(Constant)	11.082	5.292		2.094	.039*	
	CO	155	.048	316	-3.211	.002**	1.029
	NS	.121	.058	.204	2.077	.041*	1.029

**Table I.1.** AUDIT multiple regression using forward method; Model 1 and 2.

*Note*: B = unstandardized beta, SE B = standard error for unstandardized beta,  $\beta$  = standardized beta.

\* Significant at the 0.05 level. \*\* Significant at the 0.01 level.




Figure I.2. AUDIT Multiple Regression Residual Scatterplot



#### **Appendix J: TESUP-1 DUDIT Multiple Regression**

### **Additional Information**

Prior to conducting a multiple regression analysis, the assumptions of the test were checked. All variables had a linear relationship with the dependent variable (DUDIT). Data were checked for outliers or missing data, and 8 cases were excluded, bringing the total participants to 97. According to Tabachnick and Fidell (2013), a sample of 113 would be acceptable using their formula of N > 50 + 8m, where N is number of participants and m is number of independent variables. However, the forward stepwise multiple regression method can be used even when the number of predictors is very large, this is because it adds variables one at a time and thus does not have to consider the full model. After the regression analysis was conducted, the minimum sample size was checked (using the above formula) using only the variables which were added to the model. This indicated that the acceptable minimum sample size would be 66, which meets the assumptions of the method.

A forward method was chosen for the multiple regression analysis. This method begins with a model that includes no variables, then it adds variables one at a time according to which is the best fit for the model and stops adding them when the model stops improving. Here, the model stopped adding variables when the p values of the remaining variables were all over .05, and therefore would not provide a significant improvement to the model.

Model	Variable	В	SE B	β	t	р	VIF
1	(Constant)	.324	1.232		.263	.793	
	GSI	3.651	.739	.452	4.941	.000**	1
2	(Constant)	-11.382	3.573		-3.185	.002*	
	GSI	3.233	.710	.400	4.554	.000**	1.03
	NS	.206	.059	.305	3.466	.001**	1.03

**Table J.1.** DUDIT multiple regression using forward method; Model 1 and 2.

*Note:* B = unstandardized beta, SE B = standard error for unstandardized beta,

 $\beta$  = standardized beta.

\* Significant at the 0.05 level. \*\* Significant at the 0.01 level.

## Figure J.1. DUDIT Multiple Regression P-plot



## Normal P-P Plot of Regression Standardized Residual

317

Figure J.2. DUDIT multiple regression residual scatterplot



## **Appendix K: TESUP-2 Correlations Additional Information**

## 1. Demographics

For correlations with demographic data Pearson (r) was used, or Spearman (p) for nonparametric data. Point bi-serial correlation was used for the gender variable which was measured as either binary or non-binary.

Figure K.1. Scatterplots of tobacco, AUDIT, DUDIT, age, stage of transition and level of education



# 2. Quality of Life

Figure K.2. Scatterplots of tobacco, AUDIT, DUDIT, and the four domains of the	e
WHOQOL-BREF measuring quality of life	

_	TobaccoTotal	Alcohol_sum	Drug_sum	DOM1	DOM2	DOM3	DOM4
		000000000 0	<b>0</b> 00 0	00 0000000 000	00 0000000000	0 0 0 0 0 0 0 0 0	0 000 00000 0 0 0
coTotal		00 0 0 0 0 00 0 00 00 0 0 00000 00000	••••••••••••••••••••••••••••••••••••••	0 000000 00 000	• • • • • • • • • • • • • • • • • • •	0000000	
Tobac		00	0 0	o o	0 0	0 0	o o
		0000000	0 0 00 0	0 0000 00 0	00 0 000 0	•••••	0 0 00 00 0
Acohol_sum	0 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0						Alcohul_sum
	0 0 0 0	°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°		° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	° <b>ွ</b> ° <mark>ွ</mark> ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °		°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°
Taum	° ° °	° ° °		° °	°°°°	° °	° ° ° ° °
Drič	° °					× · · · · · · · · · · · · · · · · · · ·	
	0 0 0 0 0	ο ο ο ο <sup>ο</sup> ααο ο ααααάτασο ο ααο		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 00 000000 00 00 0
DOM1							0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
DOM2							00000000000000000000000000000000000000
DOM3		0 0 00 0 00 00 0 00 000 0 0 0000 0 000 0 0 0 0 000 0 0 0 0 0 0 0 0 0	0 0 00 00 00 00 00 00 00 00 00				0 0 0 0 0 0 0 0 0 0 00000 0 0 000000 0 0 0 00000 0
DOM4	• • • • • • • • • • • • • • • • • • •	ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο	° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	

## 3. GRCS

TobaccoTotal	Alcohol_sum	Drug_sum	SPC	RE	RABBM	CBWFR
	00000000 0	000 0 0000 0	0 0000000000000000000000000000000000000	00 00 00 00 0 00 0 0 00 0	0 00 00 0 0 0 0 0	00000 0 00 00000
Leco Total	00 CO 0 CO 000000 000000		0000 0000000000000000000000000000000000	00 0000000000000 0 0	00000 000000000000000000000000000000000	00 00 000000 0 00
	0 0	o o	0 0	o o	o o	0
	<u> </u>	o o ∞ ∞ o	00 0 0 00 0	0 0 00 0 0 0	0 00 00 00	
Acceled Junn 0 00 0 0 0 0 0 0 0 00000 00 0 0 0 00000 00 0 0 0						
00000 0	°°°°		C C C C C C C C C C C C C C C C C C C	00000 000000 0000000000000000000000000	°	°
0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	° ° ° ° °		° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	° ° ° ° ° °	° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	°°°°
6 ° °	00000000000000000000000000000000000000		° ° ° °		° ° ° °	° ° °
	ດວິດແມ່ນ ເພິ່ມ ເພ		00000000000000000000000000000000000000	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°	တ ထိတ် ဝဝ ဝဝ ဝဝ
SC 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	8 0 0 0 0 0 0 0 0 0 0 0 0 0					
AE	08 080 080 090 00 00 00 00 00 00 00 00 00 00 00 0		۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵ ۵			
RadeBM 0 00 00 0 0 00 00 0 0 00 00 0 0 0 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°					
	00000000000000000000000000000000000000					

# Figure K.3. Scatterplots of tobacco, AUDIT, DUDIT, and the four domains of the GRCS

## 4. GMSR

_	TobaccoTotal	Alcohol_sum	Drug_sum	DOM1	DOM2	DOM3	DOM4
TobaccoTotal		000000000 0 00 0 0 0 0 0 00 00 0 0 00 00	000 0 0000 0 000 0000000	0 000000 000 0 0000 0 00 0 000000 00 000	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 00 00000000 0 0 0 0 0 000000000 0
		0 00 0 0 0 0	0 0 00 0	0 0000 00 0	0 0 00 <b>0</b> 00 <b>0</b> 0	• • • • •	0 0 00 00 0
Acohol_sum			°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°				
	00 <b>000</b> 00 0	°°°°		° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	° 8 ° 8 ° 8 ° 8 ° 8 ° 8 ° 8 ° 8 ° 8 ° 8	° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°
uns Bru	° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	° ° ° °		° ° ° °	°°°°°°	° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	
	0 0 0 0 0 0 0 0 0 0 0 0	00 0, 0, 0 0, 0, 0 0, 0, 0 0, 0, 0 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0					
DOM1							
DOMZ							
DOM3			0 0 00 0 00 0 00 0				
DOM4	0 00 00 00 0 0 0 000000000000 000 0						
	TobaccoTotal	Alcohol_sum	Drug_sum	DOM1	DOM2	DOM3	DOM4

# Figure K.4. Scatterplots of tobacco, AUDIT, DUDIT, and the nine domains of the GMSR

# 5. Psychopathological Symptoms

Figure K.5. Scatterplots of tobacco,	AUDIT, DUDIT,	, and the nine symptom	domains of the
SCL-90-R			

	TobaccoTetal	Alcohal_sum	Drug_sum	SOM	OBC	INS	DEP	ANX	HOS	PHO	PAR	PSY
Teal		00000000 0	0 000 0 0 00 0 0		0 000 000 0 000 0 0	0 000000000000000000000000000000000000			0 200000000 0 0 0 0 0 0 0 0 0 0 0	8086 08 06060 8 08 8 60 0		88 000 0 00 0 0 86 0 0 0 0
Tehatce		00 00 0 0 00 00000 0 0 0 00 0 0 0 0	600 0 000 0 000000 p a 0 0 000 0	000 0000 00000000000000000000000000000	00 00 00 000000000 0 n 0 00 00 0 0 0 0	a a ann an ann an an ann an an an an an	0 000 000 000 000 000 000 000 000 000	00 00 0 000000000000000000000000000000	00 000 000 000000000 0 0 0000 00 0	0 0 0 0 0 0 0 0 0	0 0 0 0 0 0000000000000000000000000000	0 0 0 0 0 00 0000000000000000000000000
5	· · · · · ·	L I	۵۵ و هما ۵۵ و م	୍ କୁହିଲ୍ଲ ୧ କୁହିଲ୍ଲ ୧ କୁହିଲ୍ଲ		ေ ငူသိသိမ္တာ ေ ငု ငု ငု ငု ငု ငု ငု ငု	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	00000000000000000000000000000000000000	00 4 00800 % 00 00 00	م <sup>68</sup> مهم م م م <sup>0</sup> م		ະ ເສັ້ນເພື່ອ ຄຸດເວັ້
e_booke			° o o o ° ° ° 88°°88 88°°88 8°°88 8°°88 8°°88 8°°88 9°°88 9°°88 9°°88 9°°88						° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °		3 3 3 6 8 0	0 30° 0 0 ° ° 9 ° ° ° ° 9 ° ° ° ° ° 9 ° ° ° ° ° °
	0 0 0	°.				, .			0.00 <sup>00</sup> 80 <sup>8</sup> 0 0.0000000000000000000000000000000000	0.5380 0.5380 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. 0.	8 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
Pre Charle		ຊ ເບິ່ວ ຜູ້ ດີ ເດັ	h.l.	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	00, 00 0 0 0 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0			2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
	0	°	د د 8 <sup>°°</sup>		م م م م	6000	0 0 <sup>30</sup> 0,0	، میس <sup>و</sup> د	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	6 6 6 6 6 6 7 8	c B og	م <sup>ع 6</sup> شر
WOS		င္ ၀ ၀ ၀ ရန္က ရန္က်ိဳး င္ ၀ ၀ ၀ ရန္က ရန္က မိုင္ ၀ ၀ မိုင္ ၀ စိုင္ ၀ စိုင္ ၀ စိုင္ ၀ ၀ ၀ ၀ ၀ ရန္က ရန္က စိုင္ ၀ ၀ ၀ ၀ ၀ ရန္က ရန္က ၀ ၀ ၀ ၀ ၀ ၀ ၀ ရန္က ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀ ၀				ိုင္လိုင္လိုင္လ စစ္ေရွမိုင္ စစ္ေရွမိုင္ စစ္ေရ စစေေရ စစေရ စစေေရ စစေေရ စစေေရ စစေေရ စစေေရ စစေေရ စစေေရ စစေေရ စစေေရ စစေေရ စစေေရ စစေေရ စစစေရ စစေေရ စစေေရ စစစေရ စစစေရ စစစေရ စစစစစစစစ		မိုက္ကိုလ္လိုင္ရဲ ကို လိုင္ရမ္က ကို လိုက္လိုင္ရမ္းတဲ့ ကို လိုင္ရမ္းတဲ့ ကို လိုက္လိုင္ရမ္းတဲ့ ကို လိုက္ရမ္းတဲ့ ကို လိုင္ရမ္းတဲ့ ကို လိုက္ရမ္းတဲ့ ကို ကို ကို ကို ကို ကို ကို ကို ကို ကို		ေရးမိန့်နိုင်ငံ မိန့်မိန့်မိုင်ငံ မိန့်မိန့်မိန့်မိုင်ငံ မိန့်မိန့်မိန့်မိန့်မိန့်	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
8	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			د د د د د د د د د د د د د د د د د د د		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.000 9.000 8.0000 8.00000 8.0000 8.0000 8.00000 8.00000 8.0000 8.00000 8.00000 8.00000 8.000000 8.00000000	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ပ စိ စိုရိပ် ရှိနှင့် စိုင်ရှိ ရှိနှင့် စိုင် စိုင်ရှိ အမ်ကန်နိုင် စိုင် စိုင်ရှိ အမ်
		000 0 900 0 900 0 900 0 900 0 900 0 900 0 900 0		07 0 0 500 8 8 8 8 0 0 8 0 0 8 0 0 8 0 0 8 0	diale.	600 000 000 000 000 00 00 00 00 00 00 00			* 8 * * * 880 3 * 8 *	60 0 0 00 2 0 0 0 8 0 0 0 8 0 8 0 8 0 8 0	5 80 0 08 08 0 380 0 8 0 8 0 8 0	80 0 00 00 00 00 00 00 0 00 00 00 00 00 0
SN									0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			င်္နိ ရိန္တန္က ရန္က နိုင္ငံ က်က္က အားလို အားလို က်က္က အားလို
	8 0 8 0 8 0 8 0 8 0 8 0 8 0 8 0 8 0 8 0			ତିତ୍ତି ତୁର୍ଦ୍ଦୁର ତୁର୍ଦ୍ଦୁର ତୁର୍ଦ୍ଧ ତୁର୍ଦ୍ଧ ତୁର୍ଦ୍ଦୁ ତୁର୍ଦ୍ଦୁ ତୁର୍ଦ୍ଦୁ ତୁର୍ଦ୍ଦୁ ତୁର୍ଦ୍ଦୁର ତୁର୍ଦ୍ଦୁର ତୁର୍ଦ୍ଦୁର ତୁର୍ଦ୍ଦୁର ତୁର ତୁର ତୁର ତୁର ତୁର ତୁର ତୁର ତୁର ତୁର ତ			2 000 0 2 0 0 0 2 0 0 0 2 0 0 0 2 0 0 0 0	00000 010000 8000000 8000000	80°00 88°08 88°	2000 0 2000 0 200000000		
DEP DEP			600 0000 0000 0000 000000	00000000000000000000000000000000000000						ີ ວິ ເ ອີດອີດ ອີດອີດອີດ ອີດອີດອີດ ອີດອີດອີດ ອີດອີດອີດ		
		စိပ္သာမ်ိဳး ရောင် စစ်စိုင် စ စစ်စိုင် ရ စစ်စိုင် ရ စစ်စိုင် ရ စစ်စိုင် ရ စစ်စိုင် ရ စစ်စိုင် စ စစ်စိုင် စ စစ်စိုင် စ စစ်စိုင် စ စစ်စိုင် စစ်စိုက စစ်စိုင် စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စစိုက စစ်စစိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စိုက စစ်စစိုက စစစိုက စစိနာက စစိုက စစိုက စစိုက စစိုက စစိုက စစိုက စစိုက စစိ စစိုက စစိုက စစိုက စစိုက စစိုက စစိုက စစိုက စိနာစိ စစိုက စစိုက စစိ စစိုက စစစစိ စစိ စစစစိ စစစစ စစစစ		မီ ၀၀ မ ဖ စစ်စ ရမ္မာစ ရမ္မေစ ရမ္မေစ ရမေနာစ ရ ရ ရေနာစစ ရမရေနာစ ရ ရ ရေနာစ ရ ရေ ရမရရ ရေ ရေနာစ ရ ရ ရေ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ ရ	45°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°	ိင္လ်ိဳင္လိုင္ စစ္အာစို စစ္အစို စစ္အစို စစ္အစို စ္စစ္အစိ		000 98 900 80 900 99 900 99	0 00 00 0 000 0 2008 0 2008 0 80 0 80 0	90 0 90 0 0 8 0 0 0 0 0	0 08 0 08 00 86 000 8000 8000 8000 8000	ି ଦୁଦ୍ଦୁ କୁନ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁ କୁ କୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁନ୍ଦୁ କୁ କ କ କୁନ୍ଦୁ କୁନ କ କ କ କ କ କ କ କ କ କ ନ କ କ କ କ କ କ
AMK						200 200 200 200 200 200 200 200 200 200	ိုင် (၂) (၂) (၂) (၂) (၂) (၂) (၂) (၂) (၂) (၂)				၀ နှင့် နိုင်ငံ ၀ ရေး၀ စမ္ဘာ့စေရန် ၀ စ စန္တာ စန်နိုင်ငံ ၈ ရန်နိုင်ငံ ၈ ရနိုင် ၈ ရန်နိုင်ငံ ၈ ရန်နိုင်ငံ ၈ ရန်နိုင်ငံ ၈ ရန်နိုင်ငံ ၈ ရန်နိုင် ၈ ရနိုင် ၈ ရနိုင် ၈ ရနိုင် ၈ ရနိနိုင် ၈ ရနိနိုင် ၈ ရနိနိ	
		000 00 00 00 00 00 00 00 00 00 00 00 00		င်း ကို ကို ကို ကို ကို ကို ကို ကို ကို ကို ကို ကို ကို ကို ကို							00000 0000 0000 0000 0000 0000 0000 0000	
804		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		6 6 6 6 6 6 7 6 6 8 8 6 6 8 8 6 6 8 8 8 6 6 8 8 8 6 6 8 8 6 8 8 8 8 8 8 8 9 8 9	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0000 0000 00000 00000 00000 00000 0000	020 6200 00000 00000 00000 00000 00000			4 3 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 20 05 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0											
OHd	0 0 0 0 0 00000 0 0 0 0000 0 0	°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°	° ° & & & & & & & & & & & & & & & & & &	د د مورد د مورد مورد د مورد د مورد مورد مورد مورد مورد مورد مورد مورد	୍ଟ୍ର ୧୦୦୦ ୧୦୦୦ ୧୦୦୦ ୧୦୦୦	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	د میں میں میں میں میں میں میں میں میں میں				00 00 00 00 00 00 00 00 00 00	ိ ိ ေ စိ ေ ဝိ ့ စိ ေ က စက္လိုင္လို စိ စိ
		°G <sup>2</sup> ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	9 0 0 0 9 0 0 0 9 0 0 0 9 0 0 9 0 0 9 0 0 9 0 0 9 0 9	00300 000 000 000 000 00 00 00 00 00 00		၀ အ ပုဝိ န စြော စ စေ စစ္စ စစ္စ စစ္စ စစ္စ စစ္စ စစ္စ စစ္စ	000 <sup>0</sup> 0 <sub>00</sub> 0 0000000 00000000000000000000000					
PHR.						0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0				0000 000 0000 0000 0000 000 000 000 00		0 0
		00 0 0 0 0 0 0 0 0 0 0	00 0		000 00 00 00 0 0 0 0 0 0 0 0 0 0 0 0 0	0 3 00 0	003 D 0 0 0 D 0 00 00 D 0 0 0 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		50 0	0 0		0 0 0 000 0 000 0
754	00000000000000000000000000000000000000			0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		8 8 9 5 5 6 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9		2 c c c c c c c c c c c c c c c c c c c		5 ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °		
-								े २०२० २०२० २०२० २०२० २०२० २०२० २०२० २०		3 8 9 000 9 8 8 9 9 9 9 9 9 9		



**Figure K.6.** Scatterplots of tobacco, AUDIT, DUDIT, and the three global measures of the SCL-90-R

# 6. Personality Traits

# **Figure K.7.** Scatterplots of tobacco, AUDIT, DUDIT, and the seven dimensions of the TCI-140

TobacceTotal	Aleshel_sum	Drug_sum cco o	SMEAN (NSR)	SMEAN(HAR) BC C C CO BEBE	SMEAN(RDR)	SMEAN(PSR)	SMEAN(SD8)	SMEAN(SDR)	SMEAN(COR) 0 00:000:00:00:0	SMEAN(STR) 000000 00000 0
	000 0 0 0 0	0000 0	000000	a o a c co	0 0 0 0 00 0 0	0 <b>00 00</b> 0		0 00 0 00 00	000 00 000 0	00 00 00 0 0
100 01 001	0000000 000000 00000	0.00.0 00.0 0.0000000	20 60 0000 <b>0000</b> 0 0 0	0 00 000000 0 0 00 000 00	0 0 00 <b>00000000</b> 00000000000000000000000	00.0 00000000 0	0000 0000000 0 0	C.000.020200000000000000000000000000000	0 000000 80.00000 0 0	0 0 00 00 00000000000000
	00	0 0	0 0	0 0	0 0	0 0	00	0 0	00	0 0
	0 00 0 0 0 0	0 0 00 0	0000000	0 0 0 00 00	0 0 00 000	0 0 0 00 0	0 0 0 0 0	0 0000 000	0 0 = 00 0 0	0 00 0200
g		°.	<u>8</u>	×		0 0	°.8	° 2	<sup>0</sup>	0 0
		ംം	2000 C	• ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	0,000 0	* #8	0 0°88000	್ ಕ್ಷಮ್ಮೆಕ್ ಯಂ	ି <b>ଲ୍ଲି</b> କି ୦ ୦ <sup>୧</sup> ୦୦  ଦ	°eee°
	h.	800 0 0		- <del>6</del> - 00,00	· *** •*	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	0 0 <sup>0</sup> 000 8	0 0 0 0		
		00000000000000000000000000000000000000	°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°		• • • • • • • • • • • • • • • • • • •		° 8° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
° 8 8		80 m o	0.55 Date		~~ <u>~</u> ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	oo a boot oo	• • • • • • • • • • •	000 000 0 mm 0	§	°°, °°, °° m
00000	°°°°°°°°			and the second s	°	,	000000 000000		· 🚆	°°&#**</td></tr><tr><td>0 0 0 0</td><td>° •</td><td></td><td>° °</td><td>a 4</td><td>,</td><td>° = •9</td><td>° •</td><td>° •</td><td>• •</td><td></td></tr><tr><td></td><td>8 0000</td><td>-</td><td>0000 0</td><td>°°°°</td><td>° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °</td><td>· · · · · · · · · · · · · · · · · · ·</td><td>• 8 ° •</td><td>· · · · ·</td><td>° ° ° ° ° °</td><td>• • • • •</td></tr><tr><td>8 8 8</td><td>0, 00 00000</td><td></td><td>00000 00000000000000000000000000000000</td><td>00000000000000000000000000000000000000</td><td>0 0 0 00<sup>0</sup>0 0 00 0</td><td></td><td></td><td></td><td>° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °</td><td></td></tr><tr><td>0</td><td>0</td><td>0</td><td></td><td>0</td><td>0</td><td>0</td><td>0</td><td>¢</td><td>0</td><td>•</td></tr><tr><td>8 a 0 0 g</td><td>° ° °</td><td></td><td></td><td></td><td>\$ °</td><td>° 8 8 ° °</td><td>· · · · · · · · · · · · · · · · · · ·</td><td>° ° °</td><td>° 8</td><td>•<i>°°</i></td></tr><tr><td></td><td></td><td>1000 - 10000 - 10000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 -</td><td></td><td></td><td></td><td>· · · · ·</td><td>°°°8 °°8 °</td><td></td><td></td><td>·</td></tr><tr><td>8 ° 8 ° 8</td><td>86°° ***</td><td>· · ·</td><td></td><td>°°°°°°°</td><td>· · · · · ·</td><td>6,80 B</td><td></td><td>**************************************</td><td></td><td></td></tr><tr><td>0</td><td>0</td><td>0</td><td></td><td>•</td><td>0</td><td>0</td><td>0</td><td>0</td><td>° •</td><td>0</td></tr><tr><td>° e 8 8</td><td>9,0 ° ° °</td><td>800 00 0</td><td>000 00 00 00 00 00 00 00 00 00 00 00 00</td><td></td><td>° ° °</td><td></td><td>00 0<sup>0</sup> 0 0</td><td>96<sup>0</sup> 0 0</td><td>00,00 00,00</td><td>\$ ° ° ° .</td></tr><tr><td></td><td>890 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0</td><td>8° °° °</td><td>0 980 0 58 8 9</td><td></td><td>• • • • • • • • • • • • • • • • • • •</td><td>6600 6600 6600</td><td>2 '80 8<sup>5</sup>0'0 0 0</td><td>° • • • • • • • • • • • • • • • • • • •</td><td>° ° ° ° ° °</td><td></td></tr><tr><td></td><td></td><td></td><td></td><td><b>.</b></td><td></td><td></td><td></td><td></td><td></td><td></td></tr><tr><td>a</td><td>0</td><td>0</td><td>•</td><td></td><td>•</td><td>0</td><td>0</td><td>ō</td><td>0</td><td>0</td></tr><tr><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></tr><tr><td></td><td></td><td>88 80 0 20 0 90</td><td></td><td>• • • • • • • • • • • • • • • • • • •</td><td></td><td>· · · · · · · · · · · · · · · · · · ·</td><td>8.8.</td><td></td><td>· · · · · · · · · · · · · · · · · · ·</td><td>, , , , , , , , , , , , , , , , , , ,</td></tr><tr><td></td><td>**************************************</td><td></td><td>5 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9</td><td>°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°</td><td>h.</td><td>· · · · · · · · · · · · · · · · · · ·</td><td></td><td></td><td></td><td></td></tr><tr><td></td><td>000 0 000 <b>10</b>0</td><td></td><td>0 1890 0 U 0 18 0 0 0 10</td><td></td><td></td><td>8. <b>3</b> 5 6 6</td><td>0 808200 0 808200</td><td>0 00000000 0</td><td></td><td>000 0000000000000000000000000000000000</td></tr><tr><td>0 0</td><td>8</td><td>8</td><td>0 0</td><td>0 0</td><td></td><td>0 a</td><td>• •</td><td>° 0</td><td>° °</td><td>0 0</td></tr><tr><td>0 0</td><td>0 g</td><td>0 0</td><td>0.0</td><td>0 0</td><td>0 0</td><td></td><td>° °</td><td>0 0</td><td>o 0 0</td><td>0 0</td></tr><tr><td>- 8 8</td><td>°8 .</td><td>8</td><td>° ° °</td><td>8.00</td><td>, • • • • •</td><td></td><td>8000</td><td>° ° °</td><td>8.00</td><td>· · · · ·</td></tr><tr><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°</td><td></td><td></td><td></td></tr><tr><td></td><td>်မ လို စ ရွက်စစ</td><td></td><td></td><td>0 0 000 0 0 0 000</td><td>0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0</td><td></td><td>8000</td><td>0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0</td><td>· • • • • • •</td><td>000,8 0</td></tr><tr><td>0 0</td><td>a 0</td><td>000</td><td>0 C</td><td>• • •</td><td>0 0 0</td><td></td><td>°</td><td>c °</td><td>5 °</td><td></td></tr><tr><td>0 0 0 0 0</td><td>0 0 0 0 0</td><td>0 0</td><td>0 0 0 20 0</td><td></td><td>• •</td><td>0 0</td><td>_</td><td>0 0 0 0 0</td><td>• • •</td><td>0 00 000</td></tr><tr><td></td><td></td><td>00000</td><td></td><td>°°°°°°°°°</td><td></td><td>°°°° °</td><td></td><td></td><td>° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °</td><td>• • • • • • • • • • • • • • • • • • • •</td></tr><tr><td></td><td>000 0000000000000000000000000000000000</td><td></td><td>0000 00000 00000 00000 00000 00000</td><td>0 00 00 0 000 0 00 00 000 00 00 000 00</td><td>00000000000000000000000000000000000000</td><td>0 0 00 0 0000 0 0 0 0000 00 0 0 0000</td><td></td><td>0 000 00 0000 0 00 0000000 0000 00 0000 0</td><td>0 00 0 0 0000 0 0 0 000 0 0 00 0 0 000 0 0 0 0 000 0 0 0</td><td></td></tr><tr><td>0 0 0</td><td></td><td>0,0 ° °</td><td></td><td></td><td></td><td></td><td></td><td></td><td>00000000000000000000000000000000000000</td><td>00000 00000</td></tr><tr><td>G O</td><td></td><td></td><td></td><td>• •</td><td></td><td></td><td></td><td></td><td>0 0</td><td></td></tr><tr><td>0</td><td>0</td><td>0</td><td>0</td><td>a</td><td>0</td><td>0</td><td>0</td><td></td><td>0</td><td>0</td></tr><tr><td></td><td></td><td>000 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0</td><td>° ° 6 000 ° 9 000 ° 8</td><td>e en en</td><td>°°°°°°</td><td></td><td>° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °</td><td></td><td>° 6. 8</td><td>°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°°</td></tr><tr><td></td><td></td><td></td><td></td><td></td><td>° 2800 ° 2</td><td></td><td>ି <b>କ</b>ର୍ଦ୍ଦିକର କୁତ୍କର୍ବ୍ଦି</td><td></td><td></td><td></td></tr><tr><td>0 0</td><td>°°° °</td><td>• • •</td><td>*** **********************************</td><td>* °</td><td>0 0</td><td>• • • •</td><td>°°°°</td><td></td><td>° ° °</td><td></td></tr><tr><td>a a a</td><td>0<sub>0</sub> 0</td><td>00 0</td><td>0 0 0</td><td>° ° °</td><td>000</td><td>0000</td><td>o _ o</td><td>• • •</td><td></td><td>0 0 0</td></tr><tr><td></td><td>808 0 808 0 808 0 808 0 808 0 80 0 80 0</td><td></td><td></td><td>60000000000000000000000000000000000000</td><td>60000 00000 00000</td><td></td><td></td><td></td><td></td><td></td></tr><tr><td></td><td>0 0 00 000 0 0 0 0 0 00 00000000000000</td><td></td><td>,</td><td>ి తెంప్లర్ల సరక్షణ</td><td>1996</td><td></td><td>00000000000000000000000000000000000000</td><td>Son to align of the son</td><td></td><td>500 00 00 00 00 00 00 00 00 00 00 00 00</td></tr><tr><td>0 0</td><td>° °</td><td>0</td><td>00</td><td>0</td><td>0</td><td>0 0</td><td>° 0</td><td>° 0</td><td></td><td>۰ ۰</td></tr><tr><td></td><td></td><td></td><td></td><td>· · · · ·</td><td></td><td>-</td><td></td><td></td><td></td><td></td></tr><tr><td>a o</td><td>e _ 62</td><td>• • • •</td><td>0000</td><td>° , 89</td><td></td><td>°</td><td>° 08 80 0</td><td>° - °</td><td>°</td><td></td></tr><tr><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></tr><tr><td></td><td></td><td></td><td>° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °</td><td>08 00 00,0 00 00 00,0</td><td>0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0</td><td></td><td></td><td>000 000 0 000 000 0 000 00 0</td><td>0 9 90 0 2<sup>2</sup> 800</td><td></td></tr><tr><td>• • • • •</td><td>000 00 98 0</td><td>870 o o</td><td>00° 00° 00</td><td>° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °</td><td>° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °</td><td>069 00°</td><td>00 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0</td><td>0<sup>0</sup> 0 0 0 0 0 0</td><td>9 0<sup>00</sup>8000</td><td></td></tr></tbody></table>

### **Appendix L: TESUP-2 AUDIT Multiple Regression**

### **Additional Information**

Prior to conducting a multiple regression analysis, the assumptions of the test were checked. All variables had a linear relationship with the dependent variable (AUDIT). Data were checked for outliers or missing data, and 3 cases were excluded, bringing the total participants to 79. According to Tabachnick and Fidell (2013), a sample of 130 would be acceptable using their formula of N > 50 + 8m, where N is number of participants and m is number of independent variables. However, the forward stepwise multiple regression method can be used even when the number of predictors is very large, this is because it adds variables one at a time and thus does not have to consider the full model. After the regression analysis was conducted, the minimum sample size was checked (using the above formula) using only the variables which were added to the model. This indicated that the acceptable minimum sample size would be 66, which meets the assumptions of the method.

A forward method was chosen for the multiple regression analysis. This method begins with a model that includes no variables, then it adds variables one at a time according to which is the best fit for the model and stops adding them when the model stops improving. Here, the model stopped adding variables when the p values of the remaining variables were all over .05, and therefore would not provide a significant improvement to the model.

Model	Variable	В	SE B	β	t	р	VIF
1	(Constant)	-1.065	1.819		586		
	PST	.214	.028	.661	7.727	.000**	1
2	(Constant)	16.601	3.472		4.782	.000**	
	PST	.212	.023	.654	9.068	.000**	1
	HA	250	.044	409	-5.673	.000**	1

Table L.1. AUDIT multiple regression using forward method; Model 1 and 2.

*Note:* B = unstandardized beta, SE B = standard error for unstandardized beta,

 $\beta$  = standardized beta.

\* Significant at the 0.05 level. \*\* Significant at the 0.01 level.





Figure L.2. AUDIT multiple regression residual scatterplot



Scatterplot

#### **Appendix M: TESUP-2 DUDIT Multiple Regression**

### **Additional Information**

Prior to conducting a multiple regression analysis, the assumptions of the test were checked. All variables had a linear relationship with the dependent variable (DUDIT). Data were checked for outliers or missing data, and 1 case was excluded, bringing the total participants to 81. According to Tabachnick and Fidell (2013), a sample of 113 would be acceptable using their formula of N > 50 + 8m, where N is number of participants and m is number of independent variables. However, the forward stepwise multiple regression method can be used even when the number of predictors is very large, this is because it adds variables one at a time and thus does not have to consider the full model. After the regression analysis was conducted, the minimum sample size was checked (using the above formula) using only the variables which were added to the model. This indicated that the acceptable minimum sample size would be 74, which meets the assumptions of the method.

A forward method was chosen for the multiple regression analysis. This method begins with a model that includes no variables, then it adds variables one at a time according to which is the best fit for the model and stops adding them when the model stops improving. Here, the model stopped adding variables when the *p* values of the remaining variables were all over .05, and therefore would not provide a significant improvement to the model.

Model	Variable	В	SE B	β	t	р	VIF
1	(Constant)	-7.164	2.297		-3.118	.003**	
	PST	.298	.035	.692	8.526	.000**	1
2	(Constant)	17.209	4.272		4.028	.000**	
	PST	.294	.029	.682	10.285	.000**	1.001
	HA	345	.054	421	-6.351	.000**	1.001
3	(Constant)	13.752	4.409		3.119	.003**	
	PST	.272	.029	.632	9.291	.000**	1.112
	HA	316	.054	386	-5.830	.000**	1.055
	V	.653	.279	.163	2.341	.022*	1.169

Table L.1. DUDIT multiple regression using forward method; Model 1, 2 and 3.

*Note*: B = unstandardized beta, SEB = standard error for unstandardized beta,

 $\beta$  = standardized beta.

\* Significant at the 0.05 level. \*\* Significant at the 0.01 level.





Histogram

Figure M.2. DUDIT multiple regression residual scatterplot



Scatterplot

**Regression Standardized Predicted Value**